# An evaluation of a patient safety culture tool in Saudi Arabia



Thesis by

### Mamdooh Shrier Alonazi

Supervisors: Professor Allen Hutchinson, Dr. Rachel O'Hara,

Dr. Jenny Freeman

A thesis submitted in fulfilment of the requirements for the degree of Doctor of Philosophy Faculty of Medicine, Dentistry and Health School of Health and Related Research The University of Sheffield June 2011



### **IMAGING SERVICES NORTH**

Boston Spa, Wetherby West Yorkshire, LS23 7BQ www.bl.uk

# PAGE NUMBERING AS ORIGINAL



#### **IMAGING SERVICES NORTH**

Boston Spa, Wetherby West Yorkshire, LS23 7BQ www.bl.uk

## **BEST COPY AVAILABLE.**

## VARIABLE PRINT QUALITY

## Summary

#### Background

Safety culture is considered to be an essential element of patient safety. Several tools are available to assess patient safety culture in hospitals. One of the most common methods of assessing safety culture is the use of safety climate questionnaires.

#### **Research** question

Is there an existing patient safety culture measure that can be demonstrated to be a valid and reliable tool for use with the workforce in hospitals in Saudi Arabia?

#### Aim and objectives

This study aims to identify whether there is an existing English language tool that would be suitable for assessing patient safety culture in Saudi context. The objectives of the study are:

- 1. To select an appropriate questionnaire to assess hospital patient safety culture.
- 2. To evaluate the face validity of the selected patient safety climate questionnaire.
- 3. To assess the psychometric properties of the selected patient safety climate questionnaire in hospitals in Saudi Arabia.
- 4. To develop the most appropriate measure for assessing patient safety culture for use in hospitals in Saudi Arabia.

#### Methods

Qualitative methods were used to evaluate face validity (n=12 hospital staff). Quantitative methods were used to assess psychometric properties (n=862 doctors and nurses in three hospitals in Saudi Arabia).

#### Findings

Evaluation of face validity identified a need for minor changes to the Hospital Survey on Patient Safety Culture (HSOPSC) questionnaire wording before it was used to collect data for psychometric assessment. The results of Confirmatory Factor Analysis (CFA) and reliability analysis showed an unsatisfactory fit for the factor structure of the original HSOPSC questionnaire to the Saudi data. Exploratory Factor Analysis (EFA) was used on one half of the Saudi dataset to produce an optimal model(s). This was followed by CFA of the resulting measurement model on the second (validation) half of the data to test the fit of the resulting optimal factor structure. The result of EFA showed that eight factors (23 safety climate items) is the optimal model to the Saudi data.

I

All factors consisted of two to four items. The items loading were between 0.43 and 0.97. The result of CFA confirmed the eight factors solution (CFA=0.94, RMSEA=0.045, SRMR=0.040, TLI=0.97). The results of EFA, CFA, correlation and reliability analysis (Cronbach's alpha) showed that the optimal model for the Saudi data consists of eight patient safety culture dimensions (23 safety climate items).

#### Conclusion

This is one of very few studies to provide an assessment of an American patient safety culture tool using data from Saudi Arabia. The results indicate the importance of appropriate validation of patient safety climate questionnaires prior to applying them to populations outside contexts in which they were developed. The validated Saudi English language version of the HSOPSC questionnaire is an appropriate patient safety climate questionnaire to assess patient safety culture in Saudi hospitals.

## Acknowledgments

Firstly, all praise to Allah almighty who enabled me to accomplish this task. I am thankful for all his blessings showering me and his unlimited beneficence in my life.

I would like to express my sincere gratitude to my supervisors, Professor Allen Hutchinson, Dr. Rachel O'Hara and Dr. Jenny Freeman for their invaluable advice, guidance and encouragement throughout the duration of this project.

I want to express my great appreciation to my parents (to my late father's memory) who supported and encouraged me to continue my study. I would like to express my sincere gratefulness to my mother who unendingly prays, offers advice and encouragement, and her endless love and carrying. I must admit that without my brothers (Gaze, Saleh, Farhan, Naif and Mosab and their children), my wife (am Zaid) and my children (Zaid, Remas, Tareq and Nouf) for their continuous love, encouragement and support I would not have reached this point.

I would like to thank my uncle and his family (Abo Hadi) for their encouragement and support. I would like to thank my sisters and my aunt for their love and support.

I gratefully acknowledge the people who they helped me in data collection. I would like to thank Dr. Chris Stride for his help and advice in data analysis. I would like to thank the course instructor (Charlotte McClelland) for the statistical analysis training and support course.

I would like to thank my government for providing me financial support during this research work. A special thanks for Dr. Tariq Mashhour for his support in this work.

Finally, I would like to thank my relatives (uncles, aunts and their children) and friends back home (Mohamed, Mamdouh, Jaber, Ouied, Abduallah and Nassar) who have helped me in any way during this work. Special thanks for Abdul Rauf for his help and support.

Ш

## Table of Contents

Summary	Ι
AcknowledgmentsI	Π
List of TablesV	Π
List of Figures	Х
List of Appendices	(I
AbbreviationsXI	Π
GlossaryX	V
Chapter one: Introduction	. 1
1.1 Background	.1
1.2 Research problem	.3
1.3 Research question	.3
1.4 Research aim and objectives	.3
1.5 Significance of the study	.4
1.6 The Saudi health care system	.4
1.6.1 Patient safety in hospitals in Saudi Arabia	.>
1.7 Thesis structure	.6
1.8 Summary	.6
Chapter two: Literature review	.7
2.1 Introduction	.7
2.2 Organisational safety	.7
2.2.1 Safety in organisations	.7
2.2.2 Safety in healthcare organisations	11
2.2.3 Patient safety in healthcare	12
2.3 Safety culture	15
2.3.1 Overview of safety culture	15
2.3.2 Safety culture and safety climate	23
2.3.3 Patient safety culture	33
2.4 Measurement of patient safety climate	38
2.4.1 Existing patient safety climate questionnaires	39
2.4.2 Development and validation of patient safety climate questionnaires	51
2.4.3 Selecting a suitable questionnaire for measuring patient safety climate	57
2.5 Overall summary of chapter two	59
Chapter Three: Methodology	72
3.1 Introduction	72
3.2 Research aim and objectives	72
3.3 Study design	73
3.4 Study setting and sample	78
3.5 Research Ethics Review	78
3.6 The fieldwork and plan	79
3.7 Summary	30
Chapter 4: Choosing an appropriate patient safety climate questionnaire for use	in
Saudi hospitals	31
4.1 Introduction	81
4.2 Selection of appropriate patient safety climate questionnaires	31
4.2.1 Step1: Identifying existing patient safety culture questionnaires	32

4.2.2	Step 2: Assessment of psychometric properties (validity and relia	ability)
and di	mensionality of patient safety climate questionnaires	84
4.2.3	Step 3: Assessment of general characteristics of patient safety c	limate
questi	onnaires	85
4.2.4	Step 4: Select appropriate patient safety climate questionnaires	87
4.2.5	Comparison of the two questionnaires	96
4.3 Fa	ce validity testing of two patient safety climate questionnaires	98
4.3.1	Introduction	98
4.3.2	Objectives of interviews	99
4.3.3	Setting of interviews	99
4.3.4	Sample of interviews	99
4.3.5	Recruitment (interview process)	99
4.3.6	The structure of the interview	101
4.3.7	Analysis of interviews data	103
4.3.8	Characteristics of participants	104
4.3.9	Results of the face validity testing of the Hospital Survey on F	'atient
Safety	<sup>•</sup> Culture (HSOPSC)	105
4.3.10	Results of the face validity testing of the Teamwork and Safety C	limate
Survey	y 115	
4.3.11	Results of the comparison of the two patient safety c	limate
questio	onnaires	123
4.3.12	Decision and changes (making a choice of questionnaire and cha	anges
to que	stionnaire's language and structure)	128
4.4 Ov	erall summary	132
Chapter Fi	ve: Data collection: Using the Hospital Survey on Patient Safety C	ulture
questionna	ire	133
5.1 Int	roduction	133
5.2 Stu	Jdy setting	133
5.3 Po	pulation of study	134
5.4 Sa	mple size	134
5.5 Da	ta collection	135
5.5.1	King Fahad National Guard hospital at King Abdul-Aziz Medical Cit	y.137
5.5.2	King Fahad Medical City	139
5.5.3	King Faisal Specialist Hospital and Research Centre	141
5.6 Da	ta management	145
5.6.1	Introduction	145
5.6.2	Data Preparation	145
5.6.3	Data entry	146
5.6.4	Data editing (validation and checking)	147
5.6.5	Double entry:	149
5.6.6	Incomplete questionnaires	150
5.7 Su	mmary	151
Chapter 6:	Assessing the psychometric properties of the Hospital Survey on P	atient
Safety Cult	ure	152
6.1 Inti	roduction	152
6.2 De	scriptive Analysis	153
6.2.1	Demographic information	153
6.2.2	Items analysis	156
6.3 Ps	ycnometric Analysis of the Hospital Survey on Patient Safety Culture	168
6.3.1	Introduction	168

6.3.2 Assessment of the Suitability of the data for factor analysis	168
6.3.3 Psychometric Data Analysis Strategy	172
6.4 Testing the original USA HSOPSC model	172
6 4 1 Confirmatory Easter Analysis	172
6.4.2 Poliobility Apolygia of the ariginal UCA UCODCC model	1/3
6.5 Identify in a patienal model	1/0
	178
6.5.1 Exploratory Factor Analysis (EFA)	179
6.5.2 Testing the fit of the optimal model: CFA	212
6.5.3 Construct validity (factor correlations of the optimal model)	214
6.5.4 Reliability analysis (internal consistency)	215
6.6 Proposed optimal model	216
6.7 Summary of overall findings	217
Chapter Seven: Discussion	218
7.1 Introduction	
7.2 Face validity findings	218
7.3 Testing the original model findings	274
7.4 Developing the optimal model	<u>22</u> 4
7.5 Mothodological issues	250
7.5 Nethouological issues	250
7.0 Sludy Strengths and inflidence and impact of the study	
7.7 Summary of the main findings and impact of the study	254
Chapter eight: Conclusion	256
8.1 Introduction	256
8.2 Implications of the study	257
8.3 Conclusion	259
References	260

.

## List of Tables

Table 2-1 Patient safety culture dimensions       35         Table 2-3 Dimensions of patient safety climate questionnaires       44         Table 2-4 A comparison of the common patient safety climate dimensions that are mentioned in four review papers.       46         Table 2-5 Psychometric properties of Hutchinson et al's study (2006)       54         Table 2-7 Psychometric properties of Materson et al's study (2009)       59         Table 2-7 Psychometric properties of Bodur and Filiz's study (2010)       61         Table 2-9 Psychometric properties of Smits et al's (2008) study       63         Table 2-10 Psychometric properties of Pfeiffer and Masner's study (2010)       63         Table 2-10 Psychometric properties of Preiffer and Masner's study (2010)       63         Table 4-1 Steps of selecting appropriate patient safety climate questionnaires       82         Table 4-1 Steps of selecting appropriate patient safety climate questionnaires       82         Table 4-2 Excluded patient safety climate questionnaires in step 3       86         Table 4-3 Excluded patient safety climate questionnaires       82         Table 4-4 Characteristics of Safety Attitude Questionnaire       93         Table 4-5 Characteristics of Safety Attitude Questionnaire       93         Table 4-7 Characteristics of Patient Safety Culture Survey on Health Care       94         Table 4-8 Characteristics of Patient Safety Culture Survey on Health Care <th>Table 2-1 Differences and similarities between safety culture and safety climate 32</th>	Table 2-1 Differences and similarities between safety culture and safety climate 32
Table 2-3 Dimensions of patient safety climate questionnaires       44         Table 2-4 A comparison of the common patient safety climate dimensions that are mentioned in four review papers.       46         Table 2-5 Psychometric properties of Hutchinson et al's study (2006)       54         Table 2-5 Psychometric properties of Waterson et al's study (2009)       59         Table 2-7 Psychometric properties of Bodur and Filiz's study (2001)       61         Table 2-9 Psychometric properties of Materson et al's study (2010)       63         Table 2-10 Psychometric properties of Pfeiffer and Masner's study (2010)       63         Table 2-110 Psychometric properties of Pfeiffer and Masner's study (2010)       63         Table 2-12 Excluded patient safety climate questionnaires       80         Table 4-1 Steps of selecting appropriate patient safety climate questionnaires       82         Table 4-3 Excluded patient safety climate questionnaires in step2       84         Table 4-3 Excluded patient safety climate questionnaires in step3       86         Table 4-3 Characteristics of Safety Climate Survey       90         Table 4-4 Characteristics of Safety Attitude Questionnaire       91         Table 4-5 Characteristics of Safety Attitude Questionnaire for assessing patient safety culture       94         Table 4-10 Demographic information of participants (n=12)       104         Table 4-10 Demographic information of participants (n=12	Table 2-7 Differences and similarities between safety culture and safety climate
Table 2-4 A comparison of the common patient safety climate dimensions that are mentioned in four review papers	Table 2-2 Patient Safety culture unitensions
Table 2-4 A Comparison of the Common Patient Safety Climate Climate Climates Climate Climates Clim	Table 2-3 Dimensions of patient safety climate question rates dimensions that are
Table 2- 5 Psychometric properties of Hutchinson et al's study (2006)	Table 2-4 A comparison of the common patient safety climate of mensions that are
Table 2- 6 The original analysis of the HSOPSC (Sorra and Nieva, 2004)	Table 0.5 Development in a file of the table of the table (0000)
Table 2- 6 The original analysis of the HSOPSC (Sorra and Nieva, 2004)       57         Table 2- 7 Psychometric properties of Waterson et al's study (2009)       59         Table 2- 8 Psychometric properties of Smits et al's (2008) study       63         Table 2- 10 Psychometric properties of Pfeiffer and Masner's study (2010)       61         Table 2- 10 Psychometric properties of Pfeiffer and Masner's study (2010)       65         Table 3- 1 Fieldwork timetable       80         Table 4- 2 Excluded patient safety climate questionnaires in step2       84         Table 4- 3 Excluded patient safety climate questionnaires in step 3       86         Table 4- 5 Characteristics of the original Hospital Survey on Patient Safety Culture 88       86         Table 4- 6 Characteristics of Tearmwork and safety climate survey       91         Table 4- 7 Characteristics of Tearmwork and safety Culture Survey on Health Care       93         Table 4- 9 Steps of selection of an appropriate questionnaire       94         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient safety culture       96         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 10 Clarity and understanding of the HSOPSC       106         Table 4- 10 Clarity and understanding of the respondents about Teamwork and Safety Climate Survey       112         Table 4- 16 General comments of the respondents about Tea	Table 2-5 Psychometric properties of Hutchinson et al's study (2006)
Table 2- 7 Psychometric properties of Waterson et al's study (2009)       59         Table 2-8 Psychometric properties of Smits et al's (2008) study       63         Table 2-10 Psychometric properties of Pfeiffer and Masner's study (2010)       65         Table 2-11 Psychometric properties of Pfeiffer and Masner's study (2010)       65         Table 2-11 Psychometric properties of Pfeiffer and Masner's study (2010)       65         Table 4-1 Steps of selecting appropriate patient safety climate questionnaires       80         Table 4-2 Excluded patient safety climate questionnaires in step2       84         Table 4-3 Excluded patient safety climate questionnaires in step 3       86         Table 4-4 Characteristics of the original Hospital Survey on Patient Safety Culture 8       90         Table 4-5 Characteristics of Teamwork and safety climate survey       90         Table 4-6 Characteristics of Safety Attitude Questionnaire       93         Table 4-7 Characteristics of Patient Safety Culture Survey on Health Care       94         Organisations       94         Table 4-9 Steps of selection of an appropriate questionnaire for assessing patient       94         Table 4-10 Demographic information of participants (n=12)       104         Table 4-10 Larity and understanding of the HSOPSC       106         Table 4-13 General comments of the respondents about HSOPSC       116         Table 4-14 Clarit	Table 2-6 The original analysis of the HSOPSC (Sorra and Nieva, 2004)
Table 2- 8 Psychometric properties of Bodur and Filiz's study (2010)       .61         Table 2- 10 Psychometric properties of Smits et al's (2008) study       .63         Table 2- 10 Psychometric properties of Pfeiffer and Masner's study (2010)       .65         Table 3- 1 Fieldwork timetable       .80         Table 4- 2 Excluded patient safety climate questionnaires in step2       .84         Table 4- 3 Excluded patient safety climate questionnaires in step2       .84         Table 4- 4 Characteristics of the original Hospital Survey on Patient Safety Cultures       .90         Table 4- 5 Characteristics of Safety Climate Survey       .90         Table 4- 6 Characteristics of Safety Climate Survey       .91         Table 4- 7 Characteristics of Safety Attitude Questionnaire       .93         Table 4- 8 Characteristics of Patient Safety Culture Survey on Health Care       .94         Organisations       .94         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient safety culture       .96         Table 4- 10 Demographic information of participants (n=12)       .104         Table 4- 10 Clarity and understanding of the HSOPSC       .106         Table 4- 14 Clarity and understanding of Teamwork and Safety Climate Survey       .114         Table 4- 15 Number of participants identifying unclear items       .107         Table 4- 16 General comments of the respondents ab	Table 2- 7 Psychometric properties of Waterson et al's study (2009)
Table 2- 9 Psychometric properties of Smits et al's (2008) study	Table 2- 8 Psychometric properties of Bodur and Filiz's study (2010)         61
Table 2- 10 Psychometric properties of Pfeiffer and Masner's study (2010)	Table 2- 9 Psychometric properties of Smits et al's (2008) study
Table 3- 1 Fieldwork timetable       80         Table 4- 1 Steps of selecting appropriate patient safety climate questionnaires       82         Table 4- 2 Excluded patient safety climate questionnaires in step 3       86         Table 4- 3 Excluded patient safety climate questionnaires in step 3       86         Table 4- 4 Characteristics of the original Hospital Survey on Patient Safety Culture 88       84         Table 4- 5 Characteristics of Safety Climate Survey       90         Table 4- 6 Characteristics of Teamwork and safety climate survey       91         Table 4- 7 Characteristics of Patient Safety Culture Survey on Health Care       93         Organisations       94         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient       94         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient       96         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 13 General comments of the respondents about HSOPSC       106         Table 4- 14 Clarity and understanding of Teamwork and Safety Climate Survey       112         Table 4- 15 Number of participants identifying unclear items       107         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       116         Table 4- 17	Table 2-10 Psychometric properties of Pfeiffer and Masner's study (2010)65
Table 3- 1 Fieldwork timetable       80         Table 4- 1 Steps of selecting appropriate patient safety climate questionnaires       82         Table 4- 2 Excluded patient safety climate questionnaires in step2       84         Table 4- 3 Excluded patient safety climate questionnaires in step3       86         Table 4- 4 Characteristics of the original Hospital Survey on Patient Safety Culture 88       86         Table 4- 5 Characteristics of Safety Climate Survey       90         Table 4- 7 Characteristics of Safety Attitude Questionnaire       93         Table 4- 8 Characteristics of Patient Safety Culture Survey on Health Care       93         Organisations       94         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient       94         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 11 Clarity and understanding of the HSOPSC       106         Table 4- 13 General comments of the respondents about HSOPSC       117         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       122         Table 4- 17 Comparison of the questionnaires in terms of clarity (n=12)       124         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       122         Table 4- 17 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 17	
Table 3- 1 Fieldwork timetable       80         Table 4- 1 Steps of selecting appropriate patient safety climate questionnaires       82         Table 4- 2 Excluded patient safety climate questionnaires in step2       84         Table 4- 3 Excluded patient safety climate questionnaires in step3       86         Table 4- 4 Characteristics of the original Hospital Survey on Patient Safety Culture 88       86         Table 4- 5 Characteristics of Safety Climate Survey       90         Table 4- 6 Characteristics of Teamwork and safety climate survey       91         Table 4- 7 Characteristics of Safety Attitude Questionnaire       93         Table 4- 9 Characteristics of Patient Safety Culture Survey on Health Care       93         Organisations       94         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient         safety culture       96         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 10 Carity and understanding of the HSOPSC       106         Table 4- 12 Number of participants identifying unclear items       107         Table 4- 15 Number of participants identifying unclear items       117         Table 4- 16 General comments of the respondents about Teamwork and Safety       114         Table 4- 17 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 16 Gen	
Table 4- 1 Steps of selecting appropriate patient safety climate questionnaires.       82         Table 4- 2 Excluded patient safety climate questionnaires in step 2       84         Table 4- 3 Excluded patient safety climate questionnaires in step 3       86         Table 4- 4 Characteristics of the original Hospital Survey on Patient Safety Culture 88       86         Table 4- 5 Characteristics of Safety Climate Survey.       90         Table 4- 6 Characteristics of Teamwork and safety Climate survey.       91         Table 4- 7 Characteristics of Safety Attitude Questionnaire       93         Table 4- 8 Characteristics of Patient Safety Culture Survey on Health Care       93         Organisations       94         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient       94         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 10 Clarity and understanding of the HSOPSC       106         Table 4- 13 General comments of the respondents about HSOPSC       114         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       112         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       112         Table 4- 17 Comparison of the questionnaires in terms of clarity (n=12)       126      <	Table 3-1 Fieldwork timetable
Table 4- 1 Steps of selecting appropriate patient safety climate questionnaires.       82         Table 4- 2 Excluded patient safety climate questionnaires in step2       84         Table 4- 3 Excluded patient safety climate questionnaires in step 3       86         Table 4- 4 Characteristics of the original Hospital Survey on Patient Safety Culture 88       86         Table 4- 5 Characteristics of Safety Climate Survey       90         Table 4- 6 Characteristics of Teamwork and safety climate survey       91         Table 4- 7 Characteristics of Safety Attitude Questionnaire       93         Table 4- 8 Characteristics of Patient Safety Culture Survey on Health Care       93         Organisations       94         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient       94         Safety culture       96         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 13 General comments of the respondents about HSOPSC       114         Table 4- 14 Clarity and understanding of Teamwork and Safety Climate Survey       112         Table 4- 15 Number of participants identifying unclear items       117         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       122         Table 4- 17 Comparison of the questionnaires in ter	
Table 4- 2 Excluded patient safety climate questionnaires in step 2       84         Table 4- 3 Excluded patient safety climate questionnaires in step 3       86         Table 4- 4 Characteristics of the original Hospital Survey on Patient Safety Culture 88       76         Table 4- 5 Characteristics of Safety Climate Survey       90         Table 4- 6 Characteristics of Safety Climate Survey       90         Table 4- 7 Characteristics of Safety Attitude Questionnaire       91         Table 4- 8 Characteristics of Patient Safety Culture Survey on Health Care       93         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient safety culture       94         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 13 General comments of the respondents about HSOPSC       114         Table 4- 14 Clarity and understanding of Teamwork and Safety Climate Survey       112         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       112         Table 4- 18 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 18 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 19 Advantages and disadvantages of questionnaires	Table 4-1 Steps of selecting appropriate patient safety climate questionnaires     82
Table 4-3       Excluded patient safety climate questionnaires in step 3	Table 4-2 Excluded nations safety climate questionnaires in step2
Table 4: 4 Characteristics of the original Hospital Survey on Patient Safety Culture 88         Table 4: 5 Characteristics of Safety Climate Survey	Table 4-3 Excluded patient safety climate questionnaires in step 3
Table 4- 5 Characteristics of Safety Climate Survey       90         Table 4- 6 Characteristics of Safety Climate Survey       91         Table 4- 7 Characteristics of Safety Attitude Questionnaire       93         Table 4- 8 Characteristics of Patient Safety Culture Survey on Health Care       93         Organisations       94         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient       94         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient       96         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 11 Clarity and understanding of the HSOPSC       106         Table 4- 12 Number of participants identifying unclear items       107         Table 4- 15 Number of participants identifying unclear items       114         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       116         Table 4- 17 Comparison of the questionnaires (12 interviews)       122         Table 4- 18 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 19 Advantages and disadvantages of questionnaires       127         Table 4- 20 Changes made on the HSOPSC questionnaire	Table 4-4 Characteristics of the original Heapital Survey on Patient Safety Culture 88
Table 4- 0 Characteristics of Safety Climate Survey       90         Table 4- 7 Characteristics of Safety Attitude Questionnaire       93         Table 4- 8 Characteristics of Patient Safety Culture Survey on Health Care       94         Organisations       94         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient       96         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 11 Clarity and understanding of the HSOPSC       106         Table 4- 13 General comments of the respondents about HSOPSC       114         Table 4- 16 General comments of the respondents about HSOPSC       116         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       122         Table 4- 17 Comparison of the questionnaires (12 interviews)       124         Table 4- 18 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 20 Changes made on the HSOPSC questionnaires       127         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       129         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       131         Table 5- 1 data collection of KFNGH       139	Table 4-5 Characteristics of Cafety Olimete Survey on Patient Salety Culture 88
Table 4- 0 Characteristics of Teamwork and safety climate survey       91         Table 4- 7 Characteristics of Safety Attitude Questionnaire       93         Table 4- 8 Characteristics of Patient Safety Culture Survey on Health Care       94         Organisations       94         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient       96         Safety culture       96         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 10 Larity and understanding of the HSOPSC       106         Table 4- 13 General comments of the respondents about HSOPSC       114         Table 4- 16 General comments of the respondents about HSOPSC       116         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       112         Table 4- 17 Comparison of the questionnaires (12 interviews)       124         Table 4- 18 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 19 Advantages and disadvantages of questionnaires       127         Table 4- 20 Changes made on the HSOPSC questionnaire       129         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       131         Table 5- 1 data collection of KFNGH       139         Table 5- 2 Data col	Table 4-6 Characteristics of Safety Climate Survey
Table 4- 7 Characteristics of Safety Attitude Questionnaire       93         Table 4- 8 Characteristics of Patient Safety Culture Survey on Health Care       94         Organisations       94         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient       94         Safety culture       96         Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient       96         Safety culture       96         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 11 Clarity and understanding of the HSOPSC       106         Table 4- 12 Number of participants identifying unclear items       107         Table 4- 13 General comments of the respondents about HSOPSC       114         Table 4- 15 Number of participants identifying unclear items       117         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       122         Table 4- 17 Comparison of the questionnaires (12 interviews)       124         Table 4- 18 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 20 Changes made on the HSOPSC questionnaire       129         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       131         Table 5- 1 data collection of KFNGH       139         Table 5- 2 Data collection of KFNGH	Table 4-0 Characteristics of Teamwork and safety climate survey
Table 4- 8       Characteristics of Patient Safety Culture Survey on Health Care         Organisations	Table 4-7 Characteristics of Safety Attitude Questionnaire
Organisations	Table 4-8 Characteristics of Patient Safety Culture Survey on Health Care
Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient safety culture	Organisations
safety culture       96         Table 4- 10 Demographic information of participants (n=12)       104         Table 4- 11 Clarity and understanding of the HSOPSC       106         Table 4- 12 Number of participants identifying unclear items       107         Table 4- 13 General comments of the respondents about HSOPSC       114         Table 4- 14 Clarity and understanding of Teamwork and Safety Climate Survey       116         Table 4- 15 Number of participants identifying unclear items       117         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       122         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       122         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       122         Table 4- 18 Comparison of the questionnaires (12 interviews)       124         Table 4- 19 Advantages and disadvantages of questionnaires       127         Table 4- 20 Changes made on the HSOPSC questionnaire       129         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       131         Table 5- 1 data collection of KFNGH       139         Table 5- 2 Data collection of KFNGH       140         Table 5- 3 Data collection of the KFSH&RC       143         Table 5- 4 Dunlicate cases       147	Table 4-9 Steps of selection of an appropriate questionnaire for assessing patient
Table 4- 10 Demographic information of participants (n=12)	safety culture
Table 4- 11 Clarity and understanding of the HSOPSC       106         Table 4- 12 Number of participants identifying unclear items       107         Table 4- 13 General comments of the respondents about HSOPSC       114         Table 4- 14 Clarity and understanding of Teamwork and Safety Climate Survey       116         Table 4- 15 Number of participants identifying unclear items       117         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       112         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       122         Table 4- 17 Comparison of the questionnaires (12 interviews)       124         Table 4- 18 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 19 Advantages and disadvantages of questionnaires       127         Table 4- 20 Changes made on the HSOPSC questionnaire       129         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       131         Table 5- 1 data collection of KFNGH       139         Table 5- 2 Data collection of KFNGH       140         Table 5- 3 Data collection of the KFSH&RC       143         Table 5- 4 Duplicate cases       147	Table 4- 10 Demographic information of participants (n=12)104
Table 4- 12 Number of participants identifying unclear items107Table 4- 13 General comments of the respondents about HSOPSC114Table 4- 14 Clarity and understanding of Teamwork and Safety Climate Survey116Table 4- 15 Number of participants identifying unclear items117Table 4- 16 General comments of the respondents about Teamwork and Safety117Table 4- 16 General comments of the respondents about Teamwork and Safety117Climate Survey122Table 4- 17 Comparison of the questionnaires (12 interviews)124Table 4- 18 Comparison of the questionnaires in terms of clarity (n=12)126Table 4- 19 Advantages and disadvantages of questionnaires127Table 4- 20 Changes made on the HSOPSC questionnaire129Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire131Table 5- 1 data collection of KFNGH139Table 5- 2 Data collection of KFNGH140Table 5- 3 Data collection of the KFSH&RC143Table 5- 4 Duplicate cases147	Table 4- 11 Clarity and understanding of the HSOPSC    106
Table 4- 13 General comments of the respondents about HSOPSC       114         Table 4- 14 Clarity and understanding of Teamwork and Safety Climate Survey       116         Table 4- 15 Number of participants identifying unclear items       117         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       112         Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey       122         Table 4- 17 Comparison of the questionnaires (12 interviews)       124         Table 4- 18 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 19 Advantages and disadvantages of questionnaires       127         Table 4- 20 Changes made on the HSOPSC questionnaire       129         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       131         Table 5- 1 data collection of KFNGH       139         Table 5- 2 Data collection of KFNGH       140         Table 5- 3 Data collection of the KFSH&RC       143         Table 5- 4 Dunlicate cases       147	Table 4- 12 Number of participants identifying unclear items
Table 4- 14 Clarity and understanding of Teamwork and Safety Climate Survey	Table 4-13 General comments of the respondents about HSOPSC114
Table 4- 15 Number of participants identifying unclear items       117         Table 4- 16 General comments of the respondents about Teamwork and Safety       122         Climate Survey       122         Table 4- 17 Comparison of the questionnaires (12 interviews)       124         Table 4- 18 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 19 Advantages and disadvantages of questionnaires       127         Table 4- 20 Changes made on the HSOPSC questionnaire       129         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       131         Table 5- 1 data collection of KFNGH       139         Table 5- 3 Data collection of the KFSH&RC       143         Table 5- 4 Duplicate cases       147	Table 4-14 Clarity and understanding of Teamwork and Safety Climate Survey116
Table 4- 16 General comments of the respondents about Teamwork and Safety         Climate Survey       122         Table 4- 17 Comparison of the questionnaires (12 interviews)       124         Table 4- 18 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 19 Advantages and disadvantages of questionnaires       127         Table 4- 20 Changes made on the HSOPSC questionnaire       129         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       131         Table 5- 1 data collection of KFNGH       139         Table 5- 2 Data collection of KFMC       140         Table 5- 3 Data collection of the KFSH&RC       143         Table 5- 4 Duplicate cases       147	Table 4- 15 Number of participants identifying unclear items
Climate Survey       122         Table 4- 17 Comparison of the questionnaires (12 interviews)       124         Table 4- 18 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 19 Advantages and disadvantages of questionnaires       127         Table 4- 20 Changes made on the HSOPSC questionnaire       129         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       131         Table 5- 1 data collection of KFNGH       139         Table 5- 2 Data collection of KFMC       140         Table 5- 3 Data collection of the KFSH&RC       143         Table 5- 4 Duplicate cases       147	Table 4- 16 General comments of the respondents about Teamwork and Safety
Table 4- 17 Comparison of the questionnaires (12 interviews)       124         Table 4- 18 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 19 Advantages and disadvantages of questionnaires       127         Table 4- 20 Changes made on the HSOPSC questionnaire       129         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       131         Table 5- 1 data collection of KFNGH       139         Table 5- 2 Data collection of KFMC       140         Table 5- 3 Data collection of the KFSH&RC       143         Table 5- 4 Duplicate cases       147	Climate Survey
Table 4- 18 Comparison of the questionnaires in terms of clarity (n=12)       126         Table 4- 19 Advantages and disadvantages of questionnaires       127         Table 4- 20 Changes made on the HSOPSC questionnaire       129         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       131         Table 5- 1 data collection of KFNGH       139         Table 5- 2 Data collection of KFMC       140         Table 5- 3 Data collection of the KFSH&RC       143         Table 5- 4 Duplicate cases       147	Table 4-17 Comparison of the questionnaires (12 interviews)
Table 4- 10 Companison of the questionnaires in terms of clarity (n=12)         Table 4- 19 Advantages and disadvantages of questionnaires         127         Table 4- 20 Changes made on the HSOPSC questionnaire         129         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire         131         Table 5- 1 data collection of KFNGH         139         Table 5- 2 Data collection of KFMC         140         Table 5- 3 Data collection of the KFSH&RC         143         Table 5- 4 Duplicate cases	Table 4- 18 Comparison of the questionnaires in terms of clarity $(n=12)$ 126
Table 4- 19 Advantages and disadvantages of questionnaires       127         Table 4- 20 Changes made on the HSOPSC questionnaire       129         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       131         Table 5- 1 data collection of KFNGH       139         Table 5- 2 Data collection of KFMC       140         Table 5- 3 Data collection of the KFSH&RC       143         Table 5- 4 Duplicate cases       147	Table 4-10 Advantages and disadvantages of questionnaires (1~12)
Table 4- 20 Changes made on the HSOFSC questionnaire       129         Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       131         Table 5- 1 data collection of KFNGH       139         Table 5- 2 Data collection of KFMC       140         Table 5- 3 Data collection of the KFSH&RC       143         Table 5- 4 Duplicate cases       147	Table 4-79 Advantages and usadvantages of questionnaires
Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire       131         Table 5- 1 data collection of KFNGH       139         Table 5- 2 Data collection of KFMC       140         Table 5- 3 Data collection of the KFSH&RC       143         Table 5- 4 Duplicate cases       147	Table 4-20 Granges made on the hours of the HSOPSC questionnaire
Table 5- 1 data collection of KFNGH139Table 5- 2 Data collection of KFMC140Table 5- 3 Data collection of the KFSH&RC143Table 5- 4 Duplicate cases147	Table 4-21 Fallent safety culture dimensions of the HSOFSC questionnaire
Table 5- 1 data collection of KFNGH139Table 5- 2 Data collection of KFMC140Table 5- 3 Data collection of the KFSH&RC143Table 5- 4 Duplicate cases147	
Table 5- 2 Data collection of KFMC140Table 5- 3 Data collection of the KFSH&RC143Table 5- 4 Duplicate cases147	Table 5- 1 data collection of KFNGH    139
Table 5- 3 Data collection of the KFSH&RC       143         Table 5- 4 Duplicate cases       147	Table 5- 2 Data collection of KFMC    140
Table 5- 4 Duplicate cases 147	Table 5- 3 Data collection of the KFSH&RC    143
	Table 5- 4 Duplicate cases   147

Table 6-1 Work area of respondents (n=862) ......153 Table 6-2 Other work areas of respondents (n=164).....154 Table 6-4 Responses of the 24 positively worded safety climate items (n=862)....157 Table 6-5 Responses of the 18 negatively worded safety climate items (n=862)...158 Table 6- 6 Responses to the question of an overall grade on patient safety (n=862) Table 6-7 Responses to the guestion on the number of events reported in the past Table 6-8 Comparison between an overall grade on patient safety and number of event reports in the past 12 months (n=862)......160 Table 6-9 Descriptive analysis of the 24 positively worded safety climate items Table 6-10 Descriptive analysis of the 18 negatively worded safety climate items Table 6- 11 Descriptive analysis of an overall grade on patient safety item and Table 6-12 Responses of the HSOPSC items grouped by dimensions (n=862).....164 Table 6-13 Sample size requirements of factor analysis......169 Table 6-15 Generally agreed fit indices (parameters)......174 Table 6-16 Fit indices of CFA of USA 12 factors model (Sorra and Nieva, 2004)..175 Table 6-17 Results of reliability analysis for USA data and Saudi data ......176 Table 6-26 Structure of the 9 factors solution and its loading...... Table 6-27 Structure of the 8 factor solution and its item loadings......201 Table 6-28 Summary of the five solutions......202 Table 6-29 Low communalities and loading items in each solution......202 Table 6- 30 Communalities of the final 8 factor EFA solution (n=431).....206 Table 6-31 Total variance explained of the final 8 factor EFA solution (n=431).....207 Table 6- 32 Pattern matrix of the final 8 factor EFA solution (n=431) ......209 Table 6- 34 Output of CFA of 8 factors solution (n=431)......212 Table 6-36 Comparison of CFA OF USA, UK and the current study......213 Table 6-38 results of the reliability analysis of the 8 factor model (n=862) ......215 

 Table 7-1 Comparative CFA data from six countries
 226

Table 7-2 Comparative internal consistency data from six countries	230
Table 7-3 Comparison between the structures of the American version an	d the
English language Saudi version of eight factors of the HSOPSC	240
Table 7-4 Approaches of validation of the HSOPSC	244
Table 7-5 Common patient safety culture dimensions across different countries	246

## **List of Figures**

Figure 6-3 Scree plot of the	initial solution	
Figure 6-4 Scree plot of the	final 8 factor EFA solution	

## List of Appendices

Appendix 1 Consent form	A-1
Appendix 2 Participant Information Sheet	A-2
Appendix 3 Original Hospital Survey on Patient Safety Culture Survey	A-6
Appendix 4 Teamwork and Safety Climate	A-15
Appendix 5 University ethics approval 1	A-19
Appendix 6 University ethics approval 2	A-20
Appendix 7 Research ethics review	A-21
Appendix 8 King Fahad National Guard Hospital agreement	A-27
Appendix 9 Support letter from Associate Executive Director of Nursing dep	artment
	A-28
Appendix 10 Support letter from Chairman of Surgery department	A-29
Appendix 11 King Fahad Medical City agreement	A-30
Appendix 12 Support letter from Executive director of Medical Affairs	A-31
Appendix 13 King Faisal Specialist Hospital & Research Centre agreement	A-32
Appendix 14 King Faisal Specialist Hospital & Research Centre agreement	A-33
Appendix 15 Support letter from Deputy Executive Director Nursing Affairs	A-34
Appendix 16 Incomplete questionnaires	A-35
Appendix 17 Data management table (background items)	A-36
Appendix 18 Data management (safety climate items)	A-37
Appendix 19 Duple data entry (15%)	A-40
Appendix 20 Duple data entry for the rest of the questionnaires	A-42
Appendix 21 Cover letter	A-52
Appendix 22 Hospital Survey on Patient Safety Culture with changes	A-53
Appendix 23 Comparison between an overall grade of patient safety and nu	mber of
event reports	A-61
Appendix 24 Communalities of the initial solution (11 factors)	A-62
Appendix 25 Total Variance Explained of the initial solution (11 factors)	A-64
Appendix 26 Pattern Matrix of the initial solution (11 factors)	A-66
Appendix 27 Communalities of initial solution (11 factors) with safety climat	e items
(n=31)	A-68
Appendix 28 Total Variance Explained of the initial solution (11 factors) with	n safetv
climate (n=31)	
Appendix 29 Pattern Matrix of the initial solution (11 factors) with safety climate	te items
Appendix 30 Communalities of the 12 factors solution	A-74
Appendix 31 Total variance explained of the 12 factors solution	A-76
Appendix 32 Rotated Matrix of the 12 factors solution	A-78
Appendix 33 Communalities of the 10 factors solution	A-80
Appendix 34 Total variance explained of the 10 factors solution	A-82
Appendix 35 Rotated Factor Matrix of the 10 factors solution	A-84
Appendix 36 Communalities of the 9 factors solution	A-86
Appendix 37 Total variance explained of the nine factors solution	Δ_88
Appendix 38 Rotated Factor Matrix of the 9 factors solution	Δ_Q_
Annendix 39 Communalities of the 8 factors solution	Δ_07
Annendix 40 Total variance explained of the 8 factors solution	Δ_0Δ
Appendix 41 Rotated Factor Matrix of the 8 factors solution	Δ_06
Annendix 42 Pattern Matrix of the 8 factors solution	Δ_02

Appendix 43 Communalities of the step eight	A-100
Appendix 44 Total variance explained of the step eight	A-101
Appendix 45 Pattern Matrix of the 8 factors solution	A-102
Appendix 46 EFA of each scale of the optimal model (8 factors)	A-103
Appendix 47 Extra factor analysis investigations	A-104
Appendix 48 Actual texts of answers of open question in the HSOPSO	C questionnaire
	A-119

## Abbreviations

**HSOPSC:** Hospital Survey on Patient Safety Culture

MOII: Ministry of Health in Saudi Arabia

NHS: National Health Service

**IOM**: Institute Of Medicine

PIIC: Primary Health Care

JCAHO: Joint Commission on Accreditation of Healthcare Organisation

HROs: High Reliability Organisations

WHO: World Health Organisation

NPSF: National Patient Safety Foundation

NPSA: National Patient Safety Agency

NSCPS: National Steering Committee on Patient Safety

ACSQIIC: Australian Council for Safety and Quality in Health Care

CCHSA: Canadian Council on Health Services Accreditation

HSC: Health and Safety Commission

SLOAPS: Strategies for Leadership: An Organisational Approach to Patient Safety

**PSCHO**: Patient Safety Cultures in Healthcare Organisations

VIIAPSCQ: Veterans Administration Patient Safety Culture Questionnaire

CSS: Culture of Safety Survey

SAQ: Safety Attitudes Questionnaire

SCS: Safety Climate Survey

MSSA: Medication Safety Self Assessment

**HTSSCS**: Hospital Transfusion Service Safety Culture Survey

HSCQ: Hospital Safety Culture Questionnaire

MORMAQ: Modified Operating Room Management Attitudes Questionnaire

SCS: Safety Climate Scale

AIIRQ: Agency for Healthcare Research and Quality

FMASS: Flight Management Attitude Safety Survey

**CFA:** Confirmatory Factor Analysis

EFA: Exploratory Factor Analysis

 $\chi^2$ : Chi-square

Df: Degree of freedom

**CFI:** Comparative Fit Index

NNFI: Non-Normed Fit Index
RMSEA: Root Mean Squared Error of Approximation
SRMSR: Standardised Root Mean Square Residual
TLI: Tucker–Lewis Index
a: Cronbach's coefficient alpha value
KMO: Kaiser-Meyer-Olkin

## Glossary

Validity: refers to whether an instrument measures what it was designed to measure (Field, 2009).

Reliability: refers to the ability of a measure to produce similar results across different situations (Field, 2009).

**Internal consistency of an instrument**: is the most popular way for estimating reliability of an instrument and refers to how well the items in a particular dimension of a scale fit together (Pett et al, 2003). Within the validation literature this is commonly referred to as reliability but this differs from the use described above.

**Cronbach's alpha** ( $\alpha$ ): is widely used measure for assessing the internal consistency of a set of items (Pett et al, 2003).

**Factor analysis** can be used for theory and instrument development and assessing construct validity of an established instrument when administered to a specific population (Pett et al, 2003:3). Factor Analysis is a statistical technique widely used in psychology and social sciences. It is used for development and evaluation of instrument. There are two basic types of factor analysis: exploratory and confirmatory (Pallant, 2007).

**Exploratory Factor Analysis (EFA)**: is used to identify how many factors (dimensions) are necessary to explain the interrelationships among a set of items (Pett et al, 2003).

**Confirmatory Factor Analysis (CFA)**: is used to assess the extent to which the hypothesized structure of identified factors (dimensions) fits the data (Pett et al, 2003).

AMOS (Analysis of Moment Structures): is a popular statistical package for undertaking structural equation modelling. It is a more recent package, because of its user-friendly graphical (Arbuckle, 2005).

Goodness of fit tests determines if the theoretical model being tested should be accepted or rejected. These include:

Goodness-of-Fit Index (GFI) and Adjusted Goodness-of-Fit Index (AGFI): The goodness of fit index "is a measure of the relative amount of variances and covariances jointly accounted for by the model" (Joreskog & Sorborn, 1986: 41). The closer the GFI is to 1.00, the better is the fit of the model to the data.

**Chi-square test statistic**: A fundamental measure of fit used in the calculation of many other fit measures. Conceptually it is a function of the sample size and the difference between the observed covariance matrix and the model covariance matrix. In the case of the chi-square statistic, smaller rather than larger values indicate a good fit (Schumacker & Lomax, 2004).

**Root Mean Residual (RMR)**: is the mean absolute value of the covariance residuals. Its lower bound is zero but there is no upper bound, which depends on the scale of the measured variables. The closer RMR is to 0, the better the model fit. One sees in the literature such rules of thumb as that RMR should be < .10, or .08, or .06, or .05, or even .04 for a well-fitting model (Hu & Bentler, 1998).

Standardized Root Mean Square Residual, Standardized RMR (SRMR): SRMR is the average difference between the predicted and observed variances and covariances in the model, based on standardized residuals. The smaller the SRMR, the better the model fit. SRMR = 0 indicates perfect fit. A value less than 0.05 is widely considered a good fit and below .08 adequate fit. In the literature one will find rules of thumb setting the cut-off at < .10, .09, .08, and even .05 (Schreiber et al, 2006).

**Comparative Fit Index (CFI)**: Also known as the Bentler Comparative Fit Index. CFI compares the existing model fit with a null model which assumes the indicator variables (and hence also the latent variables) in the model are uncorrelated (the "independence model"). CFI should be equal to or greater than 0.90 to accept the model (Hu & Bentler, 1999).

Root mean square error of approximation (RMSEA): is also called RMS or RMSE or discrepancy per degree of freedom. There is good model fit if RMSEA is less than or equal to 0.05. There is adequate fit if RMSEA is less than or equal to 0.08. More recently, it has suggested RMSEA  $\leq$ 0.06 as the cut off for a good model fit (Hu and Bentler, 1999).

**Tucker-Lewis Index (TLI) or Non-Normed Fit Index (NNFI)**: a TLI close to 1 indicates a good fit. Some authors have used a cut-off as low as 0.80, while others suggested it should be greater than 0.9 for a good fit (Hu & Bentler, 1999).

**Principle Axis Factoring**: is "a method of identifying the relationship between these factors and the observed variables from the observed correlations between the variables" (Waterson et al, 2009:3).

Kaisers (eigenvalue >1) criterion: is "a method for deciding how many factors underlie the observed variables, based on extracting only factors that explain more variability than any single observed variable would" (Waterson et al, 2009:3).

**Cattell's screen plot**: is "a method for deciding how many factors underlie the observed variables, based on using a plot to identify at which point subsequent extracted factors explain only spurious extra variability, and hence should not be retained" (Waterson et al, 2009:3).

**Oblique factor rotation**: is used to aid interpretation of factors through rearranging the variance explained between them. It allows factors to be correlated when the expected underlying dimensions are probable to be related (Waterson et al, 2009:3).

**Kaiser-Meyer-Olkin (KMO)**: measure of sampling adequacy ranges from 0 to 1, with 0.6 suggested as the minimum value for a good factor analysis (Tabachnick and Fidell, 2007).

Factor loadings: are the correlations of the variables with the factors (Kline, 1994).

## **Chapter one: Introduction**

This thesis is concerned with investigating the measurement of patient safety culture and the aim of the research is to identify a suitable tool for assessing the patient safety culture in Saudi hospitals. This chapter provides an overview of the subject matter of the thesis and the current study. It outlines the study, including the background, research problem, research question, research aim and objectives and the significance of the study. It also offers insights into the Saudi healthcare system and patient safety in Saudi hospitals. The thesis structure is also outlined.

#### **1.1** Background

Healthcare organisations are considered to be high-risk environments in terms of safety incidents (Colla et al., 2005). There are many interrelated factors that make healthcare a complex and risky process. These are organisational factors (e.g. policies and procedures), human factors (e.g. workforce and patients), and material factors (e.g. medical equipment). All of these factors might increase the probability of risk and errors which can cause significant harm to patients in this kind of environment. Therefore, patient safety is a very important issue in healthcare organisations as a means of preventing potential harm to patients that may result from healthcare delivery.

The Institute of Medicine report (2000), To Err Is Human, estimated that between 44,000 to 98,000 Americans die each year because of medical errors; this makes medical errors the eighth leading cause of death in the USA. In terms of cost, these medical errors translated into \$29 billion annually in lost income, disability and health care costs. Since the publication of this report, patient safety has become an issue for discussion amongst healthcare providers. Healthcare organisations are generally showing a greater awareness of patient safety as a means of preventing the potential harm to the patients (Walshe and Boaden, 2006). Different interrelated factors influence patient safety improvement such as people (workforce), working conditions and the processes of the healthcare system. Most of these factors are related to safety culture as an important element of patient safety measurement and improvement.

Many researchers have discussed the importance of assessing safety culture as a tool to improve patient safety in healthcare organisations (Nieva and Sorra, 2003; Singer et al., 2003; 2007; Flin et al., 2006a). The UK Health and Safety Commission (1993) defined safety culture as:

"The product of individual and group values, attitudes, perceptions, competencies, and patterns of behaviour that determine the commitment to, and the style and proficiency of, an organisation's health and safety management. Organisations with a positive safety culture are characterised by communications founded on mutual trust, by shared perceptions of the importance of safety, and by confidence in the efficacy of preventive measures" (HSC, 1993:23).

An annual assessment of safety culture is considered a priority within patient safety goals of the USA's Joint Commission for Accreditation of Health Care Organisations (JCAHO) (Pronovost and Sexton, 2005). This indicates the importance of measuring and understanding safety culture to enhance patient safety in healthcare. A patient safety culture involves shared attitudes, values and norms of staff related to patient safety. It is characterized by a number of positive features such as open communication about safety problems, effective teamwork and management commitment to patient safety as top a priority (World Health Organisation, 2008; Sorra and Dyer, 2010). It is widely agreed that patient safety culture can be assessed using patient safety climate questionnaire to assess workers' perceptions of patient safety climate in their healthcare organisations (Sexton et al., 2006; Fleming and Wentzell, 2008).

However, although the measurement of patient safety climate by researchers and healthcare professionals is increasing (Kirk et al., 2006), there is limited research on the understanding of the measurement of patient safety climate and the usefulness of using different patient safety climate questionnaires in different contexts. There is a lack of provision of complete data relating to psychometric properties (validity and reliability) of patient safety climate questionnaires when applied in different contexts beyond their origin, such as USA healthcare contexts (Waterson et al., 2009). Therefore, this study focuses on the psychometric properties of the American Hospital Survey on Patient Culture (HSOPSC) (Sorra and Nieva, 2004) and its suitability for use within the Saudi healthcare context.

#### **1.2** Research problem

Patient safety problems are a major concern for healthcare organisations around the world in both rich and poor countries (Bodur and Filiz, 2010). Healthcare organisations aim to care for patients in a safe manner and have made patient safety a top priority in their healthcare processes. Patient safety culture is considered as important area in the field of patient safety. The Institute of Medicine (IOM, 2000) proposed that healthcare organisations should focus closely on improving their patient safety culture. A number of tools are available for assessing patient safety culture in hospitals, one of the most common of which is patient safety climate questionnaires.

Increasingly, there is an awareness of patient safety in Saudi Arabia; however, currently the problem is the lack of validated tools that are known to be suitable for assessing the patient safety culture in Saudi hospitals. Therefore this study is concerned with identifying a suitable tool for assessing patient safety culture in Saudi hospitals.

#### **1.3** Research question

Is there an existing patient safety culture measure that can be demonstrated to be a valid and reliable tool for use with the workforce in hospitals in Saudi Arabia?

#### **1.4** Research aim and objectives

This study aims to identify a suitable measure for assessing patient safety culture for use in hospitals in Saudi Arabia. The main objectives of the study are:

- 1. To select an appropriate questionnaire to assess hospital patient safety culture.
- 2. To evaluate the face validity of the selected patient safety climate questionnaire.
- 3. To assess the psychometric properties of the selected patient safety climate questionnaire in hospitals in Saudi Arabia.
- 4. To develop the most appropriate measure for assessing patient safety culture for use in hospitals in Saudi Arabia.

#### **1.5** Significance of the study

Saudi hospitals, like all hospitals worldwide, need to prioritise patient safety as an important issue and work towards improving it. However, there is limited research on patient safety culture in Saudi hospitals. In addition, there is a lack of patient safety climate questionnaires validated for use in the Saudi hospital context. This study seeks to contribute to the knowledge base by assessing the validity and reliability of an appropriate patient safety climate questionnaire for use in hospitals in Saudi Arabia. It is hoped that the knowledge generated will contribute to the scientific literature on the measurement of the patient safety culture, in particular in Saudi Arabia, thereby supporting the creation of a safer environment for patients in hospitals. Furthermore, this thesis is also intended to promote a greater understanding of the use of different patient safety culture tools such as the HSOPSC have been used in many counties outside their original context in which they were developed (Smits et al., 2008).

#### **1.6** The Saudi health care system

Saudi healthcare services are provided through a number of government and private agencies. Healthcare services in Saudi Arabia, which started with limited resources and very small clinics, have been the subject of many development plans. The Saudi government has worked to improve the healthcare system by delivering healthcare services and developing polices and plans which aim to improve those services in the country. The healthcare system in Saudi Arabia can be described as a universal and comprehensive system which operates through many independent government and private health agencies that deliver primary, secondary and tertiary healthcare services (Roemer, 1991).

The ministry of health operates as a national health service by delivering primary, secondary and tertiary care to the entire population. It is considered to be the lead government agency responsible for all aspects of the health care system. These responsibilities include financing, effective management, directing, planning and regulating of the entire health care system. It also supervises the healthcare services that are provided by the private sector (Statistics Directorate, 2009).

#### 1.6.1 Patient safety in hospitals in Saudi Arabia

Hospitals are considered to be the second line of treatment provision as they accept only referrals from primary care clinics or emergencies. Large numbers of hospitals are spread over the different parts of the country; more than 400 hospitals and a number of private clinics provide health care to around 26 million people (Statistics Directorate, 2009). These nationwide facilities employ around 500,000 people from more than 80 different countries (Alanazy, 2006).

Saudi Arabia has modern hospitals equipped with advanced medical technology, and qualified medical staff and many of these hospitals are operated to American and Western standards. Many Saudi hospitals have received accreditation by the Joint Commission on Accreditation of Healthcare Organisations (JCAHO), including the King Fahad Hospital National Guard in King Abdualsis Medical City, King Fasial Specialist Hospital and Research Centre and King Fahad Medical City. The Saudi hospitals are characterised by features such as:

- 1. The large size of the workforce.
- 2. The wide variety of employees from different countries.
- 3. The complexity of tasks that are performed especially advanced surgical procedures, for example open-heart surgery and organ transplant operations.
- 4. The increasing number of patients, which leads to an increased workload.

These factors may contribute to medical errors and, thus, threaten the safety of patients. Therefore, preventing medical error and keeping patients safe is a top priority in Saudi Arabia. Although the government of Saudi Arabia has not yet established a national patient safety organisation with responsibility for national patient safety policy, healthcare organisations are currently prioritising the provision of safe patient care (Alahmadi, 2010). On the whole, Saudi hospitals aim to provide safe and good quality healthcare. To this end, a number of patient safety and medical errors conferences have been held in various Saudi hospitals, for example, the Arab and Gulf Conference on Healthcare Quality and Safety 2009, under the slogan "Deep in the Challenges of Patient Safety"<sup>1</sup>. However, the research into patient safety, in particular patient safety culture is still in its early stages in the Saudi context.

<sup>&</sup>lt;sup>1</sup> www.hcqs.eu

#### **1.7** Thesis structure

This thesis comprises eight chapters and appendices. The current chapter has presented a background to the research and an overview of the current study. Chapter two provides a review of the literature related to safety in organisations, specifically patient safety in healthcare organisations, safety culture and safety climate, and measurement of patient safety climate. Chapter three describes the methodology employed in the study. Chapter four describes the process of choosing an appropriate patient safety climate questionnaire for use in the study. Chapter five describes the process of data collection by using the HSOPSC to collect the questionnaire data from the hospitals in Saudi Arabia. Chapter six reports the results of the psychometric properties of the HSOPSC questionnaire (Sorra and Nieva, 2004) in Saudi hospitals in Riyadh using Saudi data. Chapter seven discusses the research findings that emerged from this study whilst chapter eight presents an overall summary and the conclusion of the research.

#### 1.8 Summary

Patient safety is an important aspect of healthcare. Patient safety culture assessment is a new field in the Saudi context and few studies have been published in Saudi Arabia on this critical component of patient safety. Therefore, the current study focuses on the assessment of patient safety culture in Saudi hospitals to identify a suitable questionnaire for measuring patient safety climate for use in hospitals in Saudi Arabia. In the next chapter, the relevant literature of patient safety culture assessment will be reviewed.

## **Chapter two: Literature review**

#### 2.1 Introduction

The aim of this chapter is to provide a comprehensive review of the literature associated with organisational safety, safety culture and the measurement of patient safety climate. This chapter reviews relevant literature in relation to a patient safety culture assessment based on the measurement of patient safety climate using a patient safety climate questionnaire. The chapter is composed of five sections. After presenting a brief introduction, section 2.2 discusses organisational safety in order to provide the background of safety in organisations, safety in healthcare organisations and patient safety in healthcare organisations.

Section 2.3 discusses safety culture and safety climate. This section aims to provide an overview of safety culture, to discuss the debate on safety culture versus safety climate, and the assessment of safety culture with a focus on the patient safety culture in healthcare. Section 2.4 is concerned with the measurement of the patient safety climate in healthcare organisations including a discussion of the development and validation of patient safety climate questionnaires, existing patient safety climate questionnaires and their characteristics and the selection of a suitable questionnaire for measuring patient safety climate. Section 2.5 concludes with a brief summary of the literature review.

#### 2.2 Organisational safety

This section focuses on the review of safety in organisations including healthcare organisations and patient safety including the background to patient safety, the importance of patient safety and patient safety failure in healthcare organisations.

#### 2.2.1 Safety in organisations

Safety cannot be viewed as just the total lack of mistakes and an absence of errors, but that safety has multiple dimensions and many possible outcomes. Reason (1990) defines error as the failure of planned action to achieve an aim without an unexpected event happening or the

use of a wrong plan. This concept of error is a challenge for organisations (work environments) in terms of causality and consequence. Safety is a major concern for organisations and has attracted much attention across a wide range of industries, including the nuclear energy field, chemical processing, aviation and healthcare settings. The primary goal of safety within organisations is to prevent accidents and injuries. Safety is important to the welfare of the workers and customers of organisations and it is essential to a safe organisation (Dalling, 1997). Safety is an issue for organisations therefore, safety measures are particularly important to assess safety performance.

Considerable attention has been paid to the assessment of safety in high hazard industries such as aviation and nuclear energy. According to Weick et al. (1999) safety measures have moved from a focus on retrospective data of employee injuries and accidents to safety measures that focus on organizational, managerial and human factors rather than simply on technical failures that cause accidents in organisations. There are two main approaches to measuring safety performance: reactive (retrospective) and proactive (prospective). Reactive approaches are based on retrospective data (lagging indicators). In recent years there has been a movement away from relying on safety measures such as accident rate because it measures historical events of safety (past safety problems), towards proactive approaches that are based on the assessment of current safety culture (leading indicators) such as the measurement of safety climate (Flin et al., 2000; Choudhry et al., 2007). The proactive approach relies and focuses on current safety activities to determine system success rather than system failure (Cooper and Phillips, 2004). In this sense, the assessment of safety culture is categorized under the proactive approach of safety performance which relates to the measurement of safety climate. According to Cooper and Phillips (2004) both proactive and reactive approaches can help organisations to determine the effects of their safety activities.

There are two types of failure identified in the safety literature, active and latent failures. Active failures can be classified as unsafe acts by someone whose actions can have an immediate and serious effect, they include:

- 1. Slips or errors such as using the wrong tool.
- 2. Failures of a cognitive nature, such as lapses in memory, lack of concentration and mistakes through ignorance or in accurate assessment of situation.
- 3. Deviations from safe rules, operating practices, procedures, or standards.

A latent failure is seen as something that can remain inactive (invisible) for years before combining with an active failure to cause an accident. Many conditions lead to a latent failure, such as excessive workloads; lack of knowledge or experience; a stressful workplace; rapid change within an organisation; ineffective communication and poor leadership (Vincent et al., 1998).

Reason (1997, 1998) suggests that there are two kinds of accidents or errors that can happen in organisations: individual accidents and organisational accidents. There are two main approaches to investigate these accidents, the person approach and the system approach. The person approach places emphasis on blaming individuals for forgetfulness, loss of concentration, inattention, ethical weakness or other human attributes when they make errors. It concerns the actions of individual people when they do things wrongly. The system approach focuses on creating effective conditions under which individuals work and try to build defences in order to avoid errors or reduce their effects. In this approach, errors are viewed more as results of a problem rather than its causes. Moreover, this approach concerns not only the individual, but also the role of organisational factors (Reason, 2000).

In this sense, errors in organisations usually happen due to poorly designed systems rather than poorly functioning human beings (IOM, 2000). The system approach to error is widely used in organisations and is becoming increasingly used in healthcare (Currie and Watt, 2007). The best examples of the system approach are High Reliability Organisations (HROs). They expect the worst and prepare themselves to deal effectively with it throughout the organisation (Reason, 2000). HROs have professional team strategies to deal with problems when they arise in order to decrease the probability of error and manage unexpected events (Mckeon et al., 2006).

There has been a shift in focus within the safety literature away from individual level factors that might be responsible for accidents, such as error or non-compliance with safety procedures, towards organisational factors, such as safety climate (Reason, 1990). A literature review of safety identifies the factors that contribute to errors, such as lack of teamwork, communication, leadership and poor decision making (Flin and Yule, 2003; Yule et al., 2006). All these factors tend to be related to the organisational level rather than the individual level. A number of studies showed that the majority of errors are caused by organisational factors. For example, Vincent et al. (2000) reported that organisational

problems such as deficiencies in training and supervision, and poor communication emerged as likely causal factors. Dean et al. (2002) identified organisational factors such as: work environment, workload, lack of knowledge and lack of team communication as contributory factors.

The safety literature also emphasises the influence of organisational factors on measures of system safety, such as accidents and near misses (Hofmann et al., 1995; Hofmann and Stetzer, 1998). Tomas et al. (1999) identified organisational factors such as organisational climate as a critical element involved in safety, training and education programmes, management attitude to safety as a top priority in organisation (work environment). All these factors seem to be important to the understanding accidents and play an important role in improving safety in the organisation of work. They (Tomas et al. 1999) argued that accidents in organisations are caused by unsafe actions by workers due to organisational factors (e.g. ineffective communication, weak leadership role, and unclear safety regulations) rather than human factors (e.g. lack of concentration, inattention and careless).

A number of studies have also discussed organisational factors, such as safety climate. For example, Hofmann and Stetzer (1996) argued that safety climate is an organisational factor and it is proposed to influence safety performance by influencing the context within which workers and teams work. Neal et al. (2000) examined the effects of general organisational climate on safety climate as organisational factor and safety performance. They concluded that general organisational climate can influence perceptions of safety climate, and that these perceptions of safety climate influence safety performance. Zohar (2002) argued that leadership style, concern for safety and safety priority as organisational factors influence safety performance of groups of workers in an organisation. Studies have highlighted the identification of organisational, managerial and environmental factors that influence accident causation (Oliver et al., 2002; Flin et al., 1996; Cheyne et al., 1998). Other factors identified as having an impact on error rates in work environment such as teamwork, communications, work environment workload and training (Helmriech, 2000; Mearns et al. 2001; Dean et al. 2002). From the above it is clear that the organisational factors are important factors influencing safety performance within organisations. In general, safety is a very important issue in organisations, especially a high risk organisation such as healthcare organisations, where significant hazards often occur that could harm patients (Colla et al., 2005).

#### 2.2.2 Safety in healthcare organisations

Healthcare contains all the goods and services designed to promote health, including "preventive, curative and palliative interventions, whether directed to individuals or to populations" (World Health Organisation, 2000:6). Safety is a complex product consisting of the different integrated components of a system (people, job and organisation), particularly in complex environments such as healthcare organisations. Healthcare is concerned with preventing medical errors in order to provide safe and quality care for patients (Hudson, 2003). At the same time, healthcare is considered to be a high risk industry because it involves high risk of morbidity and mortality of patients (Colla et al., 2005).

Safety in healthcare differs from safety in other organisations in some aspects such as the nature of the services provided and the healthcare workers. Timely and effective use of healthcare services (preventive and treatment of illness) delivery can influence lives of patients. Healthcare workers are professionals who apply knowledge, adapt learned procedures, and use judgment at each step of the healthcare care process (Mary et al., 2005). The healthcare process focuses on a patient who is the recipient of healthcare services and is the person at the centred of most safety discussions in healthcare. Moreover, there are a number of factors such as workload (increasing number of patients), different specialities with different characteristics (e.g. surgery and intensive care) and complex technology which may lead to unsafe practice within healthcare organisations.

Incidents resulting from healthcare processes can harm both patients and staff, so safety in healthcare involves the safety of both healthcare workers (e.g. needle stick injuries, back injuries) and patients (safety events). Safety of staff and patients are very important in reducing the medical errors and injuries to both patients and workers (Flin, 2007). The IOM (2000) suggests that the healthcare system should be focused on patients in the right time, in an effective and fair way, with safety as the main aim. It suggests that organisations should creatively redesign their system to meet these aims. In particular, safety means that patients should not be injured by the healthcare they receive. It is thus necessary that healthcare providers should engage in patient safety. Patient safety outcomes are characterised by a decrease in medical errors and opportunities for risk, in other words, the result of minimizing medical errors and hazards (Flynn, 2004).

Unfortunately errors associated with healthcare cause significant harm to patients and increase the costs of healthcare (IOM, 2000). Consequently, patient safety is a very important issue and a goal for health care organisations in order to prevent harm happening to patients during the healthcare process. Patient safety will be examined in this study, in particular patient safety culture, and with a particular focus on the measurement of patient safety climate.

#### 2.2.3 Patient safety in healthcare

#### Background

Patient safety is a cornerstone of quality in healthcare (James and Thrall, 2004). Keeping patients safe is the responsibility of healthcare providers. Patient safety in healthcare organisations has developed in stages. By the late 1980s, the concept of patient safety emerged in Australia in 1987 with the establishment of the Australian Patient Safety Foundation (Runciman, 2002). Through the 1990s, researchers published a number of studies (for example, Brennan et al., 1991 and Leape et al., 1991) that highlighted the significance of patient safety to prevent errors or decrease their effects. Nevertheless, healthcare organisations did not seem to pay much attention to patient safety until the IOM published the report, 'To Err is Human: Building a Safer Health System in 2000'. Since then, patient safety has gained importance and global awareness. The World Alliance for Patient Safety (2004) states that healthcare organisations should concentrate their efforts on preventing potential harm to patients which may result from unsafe healthcare practices.

#### Patient safety definition and medical errors

There are a number of definitions of the concept of patient safety. For example, the IOM (2000) defined patient safety as the condition of not having to receive an accidental injury due to medical care, or medical errors. The National Patient Safety Agency (2003) considered patient safety to be any incident which may cause harm to patients. These definitions or other definitions include some key terms such as medical errors, adverse events and a near miss. Medical error is described as the failure of planned action to be completed as intended or the use of incorrect plan to achieve an aim (IOM, 2000). Reason (1990) defines error as the failure of planned action to achieve an aim without an unexpected event happening or the use of a wrong plan.

An adverse event is any injury whereby a patient is harmed by their care or treatment and is not due to the patient's disease or condition (Mohr et al., 2004). Some adverse events are preventable, while others are not. Adverse events that can be prevented and result in harm are classified as errors (Regenstein, 2004). A near miss is described as any unforeseen incident which could have potentially harmed a patient but was recovered in sufficient time before it did (Currie and Watt, 2007).

Leape et al. (1993) have characterised the different kinds of medical errors as:

- 1. Diagnostic, such as error or delay in diagnosis or misinterpretation of results.
- 2. Treatment, such as incorrect procedure of an operation and error in the dose or drug.
- 3. Preventive, such as ineffective follow-up of treatment plan.
- 4. Other, such as lack of communication or equipment failure.

These errors can lead to adverse events for patients and a waste of resources. Inevitably, the quality of healthcare is reduced and presents barriers to healthcare organisations in providing safe patient care.

#### Importance of patient safety

There is considerable evidence to show that errors in the healthcare field can cause significant harm to patients. For instance, in the USA it is estimated between 44,000 and 98.000 hospitalised patients die annually due to medical error (IOM, 2000). In the UK, Vincent et al. (2001) examined the feasibility of detecting adverse events through record review in two British hospitals. They found 10.8% of patients experienced an adverse event. In Australia, the Quality in Australia Health Care Study in 1995 showed that 16.6% of all patients admitted to hospital had adverse events of which 51% could be preventable (Wilson et al., 1995). This percentage was reduced to 6.8% (Ehsani et al., 2006). This evidence highlighted that healthcare is a high risk environment and requires a focus on ensuring patient safety. Moreover, the significance of patient safety has now been recognised by healthcare organisations.

In global terms, patient safety has become an important issue in many countries across the world. Patient safety is a worldwide problem affecting both rich and poor countries (Pittet and Donaldson, 2006). Consequently, a number of countries have established national patient safety agencies, for example, the National Patient Safety Agency (NPSA) in the U.K.,

the National Steering Committee on Patient Safety (NSCPS) and the Canadian Patient Safety Institute in Canada, the Australian Council for Safety and Quality in Health Care (ACSQHC) in Australia, and the National Patient Safety Foundation (NPSF) in the USA (Arah and Klazinga, 2004).

#### Patient safety failure and medical errors

As mentioned earlier, there are two approaches to investigate safety failure. In terms of patient safety failure, the person approach focuses on the actions of the individual and tends to blame them when things go wrong. This approach is based on a belief that freedom from error is possible. On the other hand, a system approach focuses on the chain of events leading up to an error. This approach is based on a belief that people are fallible so errors are expected to occur (Reason, 1997).

Johnson (2004) reported that a number of organisational factors can play a significant role in patient safety failure such as a lack of communication or miscommunication, lack of attention to safety procedures, poor supervision, deficiency of care and concern, excessive workload, and shortage of staff for specified tasks. Patient safety research addresses a number of important issues that cause a threat to patient safety, such as breakdown in communication, leadership, teamwork, lack of awareness level, poor planning and decision-making (Flin and Yule, 2003; Yule et al., 2006). In addition, improvement of patient safety requires focus on a number of elements, such as teamwork and organisational learning in the delivery of healthcare and concentrating on systems not just on individuals (Firth-Cozens, 2001).

Threats to patient safety may arise from organisational factors (Amanda, 2006). The majority of these organisational factors relate to safety culture (Singer et al., 2003). Healthcare organisations have been focused on the importance of safety culture as a way of improving the safety of patient care. The usefulness of safety culture assessment as a tool for improving safety of patient in healthcare organisations generated much discussion and research (Nieva and Sorra, 2003; Singer et al., 2003; 2007; Pronovost and Sexton, 2005; Flin et al., 2006 a). It appears that these studies have stressed in the importance of safety culture as a tool for enhancing patient safety in healthcare organisations. Safety culture refers to the shared employees' attitudes, beliefs, perceptions, and values in relation to safety (Cox and Cox, 1991). This concept is discussed in detail in section 2.3.

It is widely accepted that safety culture is an important element in patient safety. The IOM report "To Err Is Human" concluded that "Health care organisations must develop a culture of safety" (IOM 2000: 14). This is reflected in the way that national patient safety organisations are developing. For example, in the UK, the NPSA identifies building a safety culture as the first step of its seven steps guide to improving patient safety. Similarly, the Canadian Council on Health Services Accreditation (CCHSA) identifies safety culture as the first of five goals on patient safety (Fleming, 2005).

#### Summary

Safety is a concept that has attracted much attention across a wide range of industries. Organisational factors play a critical role in the state of safety in organisations. Those organisations with good safety may reflect organisational characteristics, such as good communication, management commitment to safety and organisational learning from errors. Safety in healthcare involves the safety of both healthcare workers and patients. Patient safety is increasingly recognised as an important issue and there is growing awareness of the importance of safety culture in order to improve patient safety in healthcare organisations. The next section will examine safety culture.

#### 2.3 Safety culture

The aim of this section is to provide an overview of safety culture, to discuss safety culture versus safety climate, and the assessment of safety culture. Lastly it focuses on patient safety culture in healthcare.

#### 2.3.1 Overview of safety culture

The term safety culture originated from industrial engineering and has more recently been applied to healthcare. The concept of safety culture first appeared in 1986 following the Chernobyl nuclear power accident (Gadd and Collins, 2002). Since that time, it has received considerable attention in theoretical and research literature (e.g. Cox and Cox, 1991; Pidgeon, 1998; Lee et al.; 2000; Cooper, 2000; Guldenmund, 2000). The work of these studies and others are reviewed below.

The literature includes a number of definitions of safety culture. For example, Cox and Cox (1991) proposed that safety culture refers to the shared employees' attitudes, beliefs, perceptions, and values in relation to safety. Similarly, Pidgeon (1991) indicated that safety culture is the set of employees' beliefs, attitudes and practices that are concerned with handling risk and hazards in the workplace. Lee (1996) proposed that safety culture is related to the product of individual and group of values, attitudes and behaviours that determine the commitment to safety in an organisation. Mearns et al. (1998) indicated that safety culture refers to the attitudes, values, norms and beliefs that workers share with respect to risk and safety.

These definitions point to common elements of safety culture. These commonalities include the shared attitudes and beliefs among members of an organisation, which impact on how workers perceive and act at work. This interpretation is supported by number of studies that indicated to a link between attitudes and beliefs in the workplace. Guldenmund (2000) and Hale (2000) proposed that the attitudes, beliefs and perceptions shared by workers which determine how they act and react in relation to safety and risk issues in organisations.

Recently, Choudhry et al. (2007) review numerous definitions of safety culture and they conclude that these definitions tend to reflect the view that safety culture refers to the beliefs, attitudes and values of members of an organisation regarding safety. Furthermore, they claim that safety culture is something in an organisation (the beliefs, attitudes and values of workers with respect to safety) rather than something an organisation has (the structure, policies and practices to improve safety). The UK Health and Safety Commission (1993) produced the most widely accepted definition of safety culture as:

"The product of individual and group values, attitudes, perceptions, competencies, and patterns of behaviour that determine the commitment to, and the style and proficiency of, an organisation's health and safety management. Organisations with a positive safety culture are characterized by communications founded on mutual trust, by shared perceptions of the importance of safety, and by confidence in the efficacy of preventive measures" (HSC, 1993:23).

This definition is comprehensive and covers a number of issues. Firstly, it includes the common elements of safety culture that were mentioned in other definitions of safety culture
which are workforce attitudes, perceptions and behaviours. Secondly, it highlights the characteristics of a positive safety culture (more details of the characteristics of a positive safety culture are discussed below). Most importantly, this definition can easily be adapted to the context of patient safety in healthcare and patient safety culture as it covers dimensions assessed by patient safety climate questionnaires (Nieva and Sorra, 2003).

It seems that the commonalities across most of definitions are to do with attitude and behaviour being important elements of safety culture in organisations. These common elements refer to the psychological aspect of safety culture in terms of how workers perceive and act towards safety and risk issues in organisation. This aspect refers to highly related concept known as safety climate. Safety climate is most commonly assessed by safety climate questionnaires to measure workforce attitudes and perceptions of safety (Gadd and Collins, 2002).

Although safety climate is a measurable aspect of safety culture, there are other aspects of safety culture. The literature focuses on the three aspects of safety culture as identified by Cooper (2000): situational, behavioural and psychological. A number of qualitative and quantitative tools are available for measuring these different aspects of safety culture. The situational aspect of safety culture focuses on the structure of organisations such as working policies and procedures, whilst the behavioural aspect can be evaluated through various measures, such as self-report measures, outcome measures and observations. Finally, the psychological aspect relates to people's norms, values, attitudes and perceptions of safety climate questionnaires. Similarly, Lee and Harrison (2000) proposed that safety culture is expressed as the product of multiple interactions between people (psychological), functions (behavioural) and organisations (contextual). Fernandez-Muniz et al. (2007) indicated that safety culture can be viewed as a part of the organisational culture that refers to the individuals, jobs, and organisational characteristics that affect employees' health and safety in workplace.

In conclusion, it can be seen that safety culture is a comprehensive concept that reflects the interaction between workers and activities with regard to safety issues. It consists of common elements referring to the shared employees' attitudes, beliefs, behaviours, and perceptions towards safety issues in the workplace. These elements refer to the psychological

aspect of safety culture in terms of how workers perceive and act towards the state of safety of an organisation which is called safety climate. The importance of safety culture in relation to improving safety in organisations is discussed below.

#### Importance of safety culture

Safety culture plays an important role in ensuring that the work environment is safe (Clarke, 2003). It can be regarded as a fundamental and important aspect of an organisation's ability to manage the safety of its operations (Cox and Flin, 1998; Cox and Cheyne, 2000). A series of major accidents such as the Chernobyl disaster, the King's Cross underground fire in London and the Piper Alpha oil platform explosion in the North Sea, highlighted the role of safety culture in improving workplace safety rather than concentrating on technical issues and individual human failures to prevent accidents. Poor safety culture was found to be a key contributory factor in all these accidents (Pidgeon, 1998).

A poor safety culture is characterised by a number of features, such as barriers to organisational learning from previous mistakes, absence of an incident reporting system and teamwork failure (e.g. lack of commitment and support from management to safety, breakdown in communication between workers and lack of coordination and communication at different levels of members of organisation) (Johnson, 2002). Helmreich (2000) pointed out that a breakdown in teamwork and communication has been identified as major factor in aviation accidents.

Mearns and Flin (1999) have identified important factors that contribute to accidents and near misses such as: employee experience; knowledge, skills and attitudes to safety; work environment and nature of tasks; safety culture and the safety management system. They also indicate that management commitment to safety has an effect on workers perceptions of safety issues. Cox and Flin (1998) argue that management's commitment to safety issues is essential to the overall state of safety in organisations.

The perceived importance of safety culture in preventing accidents and improving safety in complex and high-risk systems such as aviation, chemical and nuclear industries, and healthcare, has resulted in an increased numbers of studies examining safety culture (Cooper, 2000; Guldenmund, 2000). As a result, the World Health Organisation (2006) has

emphasized the importance of safety culture in decreasing medical errors and keeping patients safe in healthcare organisations.

A number of industries have shown an interest in safety culture as a means of reducing accidents in the work environment (Fernandez-Muniz et al., 2007). Safety culture is a very important issue in relation to safety in the work environment at all levels of an organisation, individuals, teams and management. The commitment and actions of management in relation to safety issues are important aspects of safety culture and have an effect on the perceptions of workforce with regard to safety. Jointly, management and the workforce can minimise the number of accidents and improve the state of safety within an organisation. Thus, safety culture can be considered an important management tool in identifying workforce beliefs, attitudes, and behaviours towards safety in the workplace (Beck and Woolfson, 1999).

Safety culture can be good or poor depending on the way in which safety is managed and to what extent the safety culture dimensions are operating in the work environment (Pidgeon, 1998). Ostrom et al. (1993) argue that the safety culture of an organisation may be influenced by values and beliefs of its leaders. Moreover, in an organisation with a good safety culture, the management encourages workers to report errors, pay attention to safety issues and to minimise risk in the work environment, and supports safe work practices. Overall, safety culture influences safety practices in the work place. A poor safety culture is a significant risk factor that can threaten the state of safety in organisations. A good (positive) safety culture is a key factor that can improve safety in organisations. The characteristics of positive safety culture will be discussed in the next section.

#### Characteristics of positive safety culture

There is a large body of research evidence to suggest that a positive safety culture may help to enhance the state of safety of an organisation (Gadd and Collins, 2002). Vecchio-sudus and Griffiths (2004) argued that developing and maintaining a positive safety culture can be an effective tool for improving the state of safety in an organisation.

A positive safety culture is defined as:

"A set of values, perceptions, attitudes and patterns of behaviour with regard to safety shared by members of the organisation; as well as a set of policies, practices and procedures relating to the reduction of employees' exposure to occupational risks. implemented at every level of the organisation, and reflecting a high level of concern and commitment to the prevention of accidents and illnesses." (Fernandez-Muniz et al., 2007:628).

According to Ostrom et al. (1993) the aim of a positive safety culture is to create a climate, in which employees are aware and concerned about the risks and safety in their workplace in order to maintain the state of safety, avoid any unsafe actions, and to prevent accidents. Cox and Cox (1991) argue that the workforce's attitudes and perceptions of safety constitute the most important aspect of safety culture.

A number of earlier studies have discussed the characteristics of positive safety culture. For example, Zohar (1980) mentioned a number of characteristics of positive safety culture such as the priority given to safety by management and workers; the importance of the reporting of incidents and open lines of communication between management and employees as being related to good safety practices and performance. According to Pidgeon (1991), positive attitudes to safety among the workforce are a reflection of good safety culture in the work environment.

Reason (1997) argued that the characteristics of a positive safety culture are justice (no blame), flexibility (to adopt changes for safety), learning (learn from errors) and systematic (have a system to manage hazards). Dalling (1997) reported a number of characteristics of a positive safety culture such as: encouraging workers to anticipate and manage risks and threats to safety; improving working practices and continuing to improve safety; concern and commitment to safety as a top priority at all levels of workforce; learning from mistakes and open communication. Clarke (1998) proposed that the key characteristics of positive safety culture which improve safety in organisations are: management's commitment and support for safety, encouraging workers to report incidents (no blame culture), improving communication based on open and honest communication, enhancing levels of trust between workers.

Clarke (1999) argued that a positive safety culture is a result of a number of factors such as: good communication between staff and management, agreement across all levels of the organisation on the importance of safety and confidence amongst workers that safety measures are adequate.

#### Chapter two

Recently, a number of studies discussed the characteristics of positive safety culture. For example, Leape and Berwick (2000) stated that leadership, communication and preventive safety measures are positive characteristics of safety culture. Hale (2000) identified a number of characteristics for a positive safety culture, these include safety as a top priority, importance of safety, participation of workers at all levels, trust and cooperation among staff, communication openness and continues safety improvement. Sorensen (2002) identifies a positive safety culture as including good communication, organisational learning and management's commitment to safety. According to Pronovost et al. (2003) the characteristics of a positive safety culture include management commitment to discuss and learn from errors, encouraging teamwork, reporting and analysing errors, documenting and improving safety and encouraging employees to improve safety. Harvey et al. (2004) refer to the importance of managements' commitment to safety culture. Kirk et al. (2006) identified a number of characteristics of a positive safety culture as follows:

- 1. The shared perceptions of workers regarding the importance of safety.
- 2. Good communication among workers based on mutual trust and openness.
- 3. The smooth flow of information within the team.
- 4. Leadership playing a role in giving direction for safe practice.
- 5. Effective preventive (safety) measures being applied.
- 6. Latent threats or causes that might lead to accidents being proactively identified.
- 7. Organisational learning.
- 8. Recognition that errors cannot always be avoided or stopped from happening.
- 9. The adoption of incident reporting and analysis instead of blame and a punitive culture.

Choudhry et al. (2007) noted that a positive safety culture compromises five characteristics which include management commitment to safety, management concerns for the workers, mutual trust between management and workers, workforce empowerment and continues improvement of safety in workplace. It is argued that an organisation's positive safety culture reflect a number of dimensions such as senior management commitment to safety, organisational learning (feedback and learning from mistakes), safety as a top priority and leadership (Pidgeon and O,leary, 2000).

In general, it seems that there is wide agreement across the majority of earlier and more recent studies regarding some of the common characteristics of positive safety culture. Review of these 13 studies (see footnote<sup>2</sup>) shows that there are some common characteristics:

- 1 Open and honest communication between workers (mentioned in ten studies).
- 2 Priority given to safety by management and workers of organisation (importance of safety in organisations) (mentioned in nine studies).
- 3 Management commitment to safety (mentioned in eight studies).
- 4 Importance of reporting incidents and analysing them and giving feedback (mentioned in six studies).
- 5 Organisational learning to learn from errors (mentioned in six studies).
- 6 Teamwork (trust and cooperation among staff) (mentioned in five studies).
- 7 No blame culture (mentioned in four studies).
- 8 Effective leadership role (mentioned in three studies).

The presence of these factors creates effective conditions which help workers to improve safety and avoid errors in work environment. In other words, all these characteristics create a positive safety culture that prevents accidents and maintains good safety within organisations (Clarke, 2003). It appears that a positive safety culture is characterised by organisational factors that play an important role in safety in organisations and most of these organisational factors have led to greater focus on safety climate as a related concept to the safety culture in organisations. Safety climate is an organisational factor that can influence safety performance (Hofmann and Stetzer, 1996; Flin et al., 2006a).

However, there is considerable debate in the literature regarding the relationship between safety culture and safety climate (Cox and Flin, 1998; Flin et al., 2000). A number of authors (e.g. Cox and Flin, 1998, Mearns and Flin, 1999) proposed that there is considerable overlap between the concepts, causing confusion for the reader, and this is evidenced by the fact that the two terms are highly related and often used interchangeably. According to Glendon and Stanton's (2000) argument that safety culture and safety climate are at an early development stage, it is appropriate to recognise the difference between the concepts of safety culture and

<sup>&</sup>lt;sup>2</sup> Zohar, 1980; Pidgeon, 1991; Reason, 1997; Daling, 1997; Clarke, 1998; Clarke, 1999; Leape and Berwick, 2000; Hale, 2000; Sorensen, 2002; Pronovost, 2003; Harvey et al., 2004; Kirk et al., 2006; Choudhry et al., 2007.

safety climate, which have elements in common. The following section is concerned with the relationship between safety culture and safety climate.

#### 2.3.2 Safety culture and safety climate

The concept of safety culture is derived from extensive research on organisational culture and climate. Therefore, terms such as culture, climate, organisational culture, and organisational climate play a role in defining and describing the relationship between safety culture and safety climate (Gonzalez-Roma et al., 1999). The following section discusses the distinction between the two concepts, culture and climate.

#### Culture versus climate

Although definitions of culture and climate tend to be similar, the term culture is generally seen as more comprehensive than climate. Culture can be seen as the set of values, beliefs and expectations that a group of people come to share (Van Maanen and Schein, 1979). Culture describes the shared corporate values within an organisation, which influence the attitudes and behaviours of its members. Safety culture is a part of the overall culture of the organisation and is seen as affecting the attitudes and beliefs of members in terms of health and safety performance (Cooper, 2000). Helmreich and Merritt (1998) describe culture as mixture of values and beliefs that guide the behaviours of members in an organisation. Pronovost and Sexton (2005:1) state that:

"Culture commonly refers to values, attitudes, norms, beliefs, practices and behaviours of personnel. In essence, culture is the way we do things around here, whereby the word here refers not to hospital, but rather to a particular work unit".

On the other hand, climate can be defined as the sum of workers' perceptions about organisation (James et al., 1978). Climate refers to the shared perceptions of workers of an organisation at a discrete point in time (Cox and Cheyne, 1999). Denison (1996) debated the differences and similarities between culture and climate. He concluded that culture refers to deep structure based on the values, beliefs and assumptions of the members of an organisation, whilst climate refers to a state that links to the thoughts, feelings and perceptions of the members of an organisation. Denison (1996) further claimed that culture must be measured using qualitative approaches, whereas climate can be measured using

quantitative methods such as questionnaires. Cooke and Rousseau (1998) reviewed a number of definitions of culture and climate and concluded that most of the culture definitions are concerned with values, beliefs and attitudes that members come to share. In contrast, climate definitions tend to take into consideration the perceptions of members regarding the organisation in which they work. According to Gonzales et al. (1999) culture contains values, beliefs and underlying assumptions, while climate refers to a descriptive measure reflecting the workforce's perceptions of the organisational atmosphere. Clarke (2003) argued that culture refers to the deeper values and norms that influence thinking and action, whereas climate is more superficial and concerned with perceptions of the work setting.

It appears that culture and climate are not clearly distinct and the terms are used interchangeably by a number of studies (Glendon and Stanton, 2000). Culture refers to the deep values and beliefs (deep structure of organisations) while climate refers to the employees' perceptions and norms and a measurable aspect of the work environment of organisations (policies, procedures and practices). The culture and climate of an organisation influence the practice of members in that organisation (Denison, 1996; Zohar, 2003). The relationship between organisational culture and organisational climate has also been debated. This is addressed in the following section.

#### Organisational culture and organisational climate

A number of studies have discussed the relationship between organisational culture and organisational climate. For example, Moran and Volkwein (1992) examined the distinctions between the two concepts and they conclude that organisational climate links to organisational culture as an element of organisational culture that is present when the members of an organisation act together when dealing with situations that may arise.

Guldenmund (2000) notes that organisational climate refers to member's attitudes, perceptions and behaviour of organisational features, whereas organisational culture has been defined as the values, beliefs, and assumptions shared by members of an organisation and which is more encompassing than organisational climate. Organisational culture is assessed through qualitative tools, such as observations and interviews, whereas organisational climate can be measured by using self-administered questionnaires. According to Glendon and Stanton (2000) organisational climate is usually regarded as being more superficial than organisational culture in that it addresses the current state of an organisation.

According to Neal et al. (2000:100):

"Safety climate is a specific form of organisational climate, which describes individual perceptions of the value of safety in the work environment"

It appears that organisational climate can influence perceptions of safety climate and these perceptions can influence safety performance. Arezes and Miguel (2003) argued that safety culture is deeply related to organisational culture. In general, organisational culture and organisational climate are considered to be closely related concepts and safety culture and safety climate are elements of these two integrated concepts. As mentioned earlier, there is considerable debate regarding the relationship between safety culture and safety climate in the relevant literature and the relationship between them is unclear.

Several researchers have pointed out that safety culture is used together with safety climate. For example, Cox and Flin (1998) argued that safety climate is often used together with safety culture, without making a clear distinction between the two concepts. They argued that there is a need to create a clear understanding of the relationship between two terms. Mearns and Flin (1999) argued that although the two terms are often interchangeable and related, they are not exactly the same and this should be considered therefore they should be used with caution. According to Fleming (2005) it is accepted that the two concepts are closely related and the safety climate consists of the surface elements of the safety culture and can be measured by using safety climate questionnaire.

It seems that the relationship between safety culture and safety climate is unclear because the two terms are highly related. However, it would appear that it is possible to distinguish between the two terms through reviewing their definitions, dimensions and assessment. The following subsections outline how safety culture and safety climate have been defined and what the dimensions of each concept are in the safety literature.

#### Safety culture definitions and dimensions

As mentioned earlier several definitions of safety culture can be found within the literature. Most of the safety culture definitions indicate that safety culture is part of organisational culture and that it concerns the shared values, beliefs and attitudes of members of an organisation to safety issues (Cox and Cox, 1991). It is argued that a positive safety culture is one in which all staff, from top level management to individual workers, are committed to work safely and to contribute positively to their own safety and that of others within an organisation (Dalling, 1997). The dimensions of safety culture are addressed in the following section.

#### Safety culture dimensions

The dimensions of safety culture have been described in a number of studies (e.g. Brown and Holmes, 1986; Cooper and Philips, 2004). The dimensions relate to the perceived significance of safety to job behaviour, perceived management commitment towards safety, perceived commitment of workers to safety, and importance of safety training. Sorensen (2002) appears to agree that the dimensions of safety culture include management's commitment to safety, good communications, organisational learning and effective leadership role. Singer and colleagues (2003) identified the following safety culture dimensions:

- 1 A high level of organisation's commitment to safety, this commitment is translated into shared values and beliefs amongst all members of organisation.
- 2 Availability of resources to support a commitment to safety.
- 3 Safety as a top priority: priority of safety versus production.
- 4 Communication between employees and across different levels and units of organisation.
- 5 Openness about problems and errors: reporting errors.
- 6 Frequency of unsafe practice is rare.
- 7 Organisational learning: learning from errors.

A number of studies indicates that safety culture is a multi dimensional concept for example, the importance of management commitment to safety, the actions to improve safety performance, the importance of communication between workers and management, good reporting system including reporting and analysing incidents, the policies and procedures of work and the importance of the participation of the workforce in safety (workers involvement in safety) (Parker et al., 2006; Fernandez-Muniz et al., 2007).

#### Safety climate definitions

Several definitions of safety climate can be found within the relevant literature. Zohar (1980:96) defines safety climate as:

"A summary of perceptions that employees share about their work environments"

According to Cooper and Philips (1994) safety climate refers to the shared perceptions and beliefs of the workers regarding the state of safety in their work environment. According to Mearns and Flin (1999) safety climate refers to the perceptions, attitudes and beliefs of workers related to hazards and safety issues in organisation. Safety climate refers to employee perceptions regarding their organisation's commitment to safety issue in order to maintain safe work practices (Robyn et al., 2000). Safety climate is a snapshot of the state of safety that can be viewed as an accurate indicator of the organisation's safety culture as perceived by workers (Flin et al., 2000).

It seems that there is a degree of overlap between the definitions of safety culture and safety climate (Mearns and Flin, 1999) and the two terms are highly related. Most definitions of safety culture present a view of shared attitudes and behaviours of employees about safety issues in an organisation. Jointly, most definitions of safety climate present a view of shared attitudes and perceptions of employees about safety issues in an organisation.

Safety culture is a part of the overall culture of the organisation and is seen as attitudes and behaviours of workers in terms of safety performance (Cooper, 2000). Safety climate is a part of the overall climate of the organisation and is seen as attitudes and perceptions of workers in terms of safety performance (Neal et al., 2000). Safety climate is considered as a measurable aspect of safety culture. It is argued that safety climate is related to the current surface features of safety culture which are determined by the employees' attitudes and perceptions at given point of time (Flin et al., 2000; Cox and Cheyne, 2000). These perceptions are derived from interactions with staff and different levels of management such as supervisors and top management level, policies and procedures and work practices (Mary et al., 2005).

In conclusion safety climate is a snapshot of the state of safety providing a clear picture of underlying safety culture of workplace (Mearns and Flin, 1999; Flin et al., 2000). Cox and Flin (1998) proposed that safety climate is the preferred term when using psychometric questionnaire studies as the measurement tool. Similarly Mearns et al. (1998) suggested that a safety climate questionnaire is appropriate for assessing workforces' perceptions regarding the surface features of safety culture at a given point in time as a snapshot of safety culture. Sexton et al. (2006) indicated that safety climate is a measurable aspect of safety culture and can be assessed by safety climate questionnaire to assess frontline workers perceptions

toward safety issues in organisations. Safety climate questionnaire can be used to assess workforce perceptions of procedures and behaviours in their work environment that indicate the priority given to safety relative to other organisational goals. Safety climate questionnaire can assesses a range of different dimensions (Zohar, 1980; Flin et al., 2006b).

#### Safety climate dimensions

The dimensions of safety climate have been examined in a number of studies. For example, Brown and Holmes (1986) found that management commitment to safety and safety activities are key dimensions of safety climate. Similarly, Dedobbeleer (1991) found management commitment to safety and workers involvement as a key dimension of safety climate across ten safety climate instruments. Flin et al. (2000) found that the most commonly assessed dimensions across 18 safety climate questionnaires concerned management commitment to safety, the safety system in relation to procedures and practices, probability of risk, work stress and professionalism of workers (workers' knowledge and skills). Robyn et al. (2000) identified a number of safety climate dimensions such as senior management support for safety, good communication among staff members, feedback and preventing barriers to safe work practices. Cheyne et al. (2002) identified that communication, safety rules and goals, workers involvement and physical work environment should be considered as safety climate dimensions.

Flin et al. (2006b) reviewed 12 studies of safety climate in healthcare and found common safety climate dimensions such as management support and commitment to safety, communication and feedback, reporting incidents, teamwork, rules and procedures of work, risk perception and safety attitudes of workers, organisational factors and safety system. It is worth noting that the safety climate dimensions emphasise the dimensions at a system level rather than an individual level. These include the importance of management support to safety, communication among employees and different units in workplace, feedback about errors, organisational learning, communication openness and reporting errors.

From the examples of safety climate definitions and dimensions above, safety climate refers to the shared perceptions of work members to improve safety and it may be regarded as a key indicator and measurable aspect of safety culture that arise from employees' perceptions of organisations. The concept of safety climate includes all levels of organisation: management, team and individual.

#### Chapter two

The most notable dimensions of safety climate are management commitment to safety, workers involvement, reporting system, organisational learning; communication openness, and feedback about error, and teamwork across different units in work environment. In short, safety climate dimensions enhance the state of safety in the work environment when most of these dimensions work in effective way (Mearns et al., 1998). Each dimension has a specific function and aim impacting on safety climate and safety performance.

There is a large body of research describing safety climate dimensions. For example, management commitment to safety is described as the perceptions of management commitment to safety issues in organisations (Abdullah et al., 2009). Management commitment to safety has been described as a management support and positive actions in order to enhance safety in organisation (Health and Safety Executive, 2007). Havold and Nesset (2008) described communication as interaction and an effective information exchange between management and workforce about safety and risk in the work environment. Open communication means exchanging information between units of organisation honestly and smoothly (Hsu et al., 2007). Reporting system describes the willingness of workers to give details about safety issues and those workers must be given feedback about errors for learning purposes (Health and Safety Executive, 2005).

Employee involvement refers to how employees in all levels participate in improving safety (Fernands-Muniz, 2007), while organisational learning is related to getting feedback and learning from errors that happened in organisation. Teamwork means workers look out for each other and trust each other and it consists of attitudes, knowledge and skills of workers towards safety and concerns (Kaissi et al., 2003). Non punitive culture means whether are workers willing to report errors without fear of punishment for making mistakes (Weingart et al., 2004). However, it is known that poor performance in relation to safety climate dimensions such as lack of communication, lack of attention to safety procedures, workload and breaks in continuity of care leads to increased errors in organisation (Yassi and Hancock, 2005).

In summary, safety climate may be regarded as a clear picture, a current surface picture and a measurable aspect of safety culture. It appears that using a safety climate questionnaire is a useful measurement of an organisation's safety climate to provide a clear picture of the current state of safety. The assessment of safety culture is addressed in the following section.

#### The assessment of safety culture

As mentioned earlier, safety culture consists of three components: psychological, situational and behavioural. It is important to mention that methods such as peer reviews, performance indicators and audit can be used to assess safety culture in a comprehensive way (Lee and Harrison, 2000). The psychological aspects of safety culture are concerned with workers' perceptions and actions toward safety issues in organisations (Kirk et al., 2006). This aspect of safety culture is a measurable aspect which is related to safety climate. In reality safety culture assessment tools that assess the psychological aspect of safety culture are measuring safety climate.

Zohar, (1980) suggested that safety climate questionnaires produce a clear picture of an individual's safety climate at a group or organisational level. Safety climate questionnaires assess surface features of safety culture to produce a clear picture of safety culture in terms of what the workforce do regarding safety, while the deeper layer of safety culture requires qualitative methods such as observations and in-depth interviewing to investigate deep features in relation to workforces' safety practices (Cooper, 2000). Cox and Cheyne (2000:114) proposed that:

#### "Assessment of safety climate is used as an indicator of overall of safety culture".

Flin et al. (2006b) argue that safety climate can be regarded as the surface (external) features of the underlying safety culture and it can be used to assess workforce perceptions of procedures and behaviours in their workplace that indicate the priority given to safety in relation to other organisational goals. Thus, an evaluation of safety culture can be based on measuring safety climate primarily through the use of questionnaire surveys.

Evidence from numerous studies demonstrates that questionnaires are a useful tool for measuring safety climate to provide a clear picture of the state of safety in organisations through assessing the perceptions of the workforce at a particular time (Mearns et al., 2003; Flin et al., 2006b; Guldenmund 2007). It is widely accepted that safety culture assessment is based on measuring safety climate by using safety climate questionnaire to assess workers' perceptions of safety climate in their organisations (Flin, 2007).

A number of studies identify benefits of using a safety climate questionnaire for assessing safety culture. For example, Baily and Petersen (1989) concluded that safety climate questionnaires are useful because measuring safety culture by traditional methods such as safety review and audits is not enough and not effective. Safety climate questionnaire is effective method to assess employee perceptions, to identify weakness and strengths areas of safety culture and by using the data of safety climate questionnaire can effectively identify improvements of safety culture.

Similarly, Ostrom et al. (1993) agreed with the conclusion of Baily and Petersen that the safety climate questionnaire is a valuable tool to assess safety culture. According to Kho et al. (2005) the self-administered questionnaire can help in understanding institutional perceptions of safety culture as an efficient way to ask standardized questions of all respondents concurrently and anonymously. Guldenmund (2007) agreed that the use of safety climate questionnaires is an effective approach in safety culture assessment in terms of assessing workforce perceptions towards safety issues in organisation. Safety climate questionnaires are used to assess safety culture in different organisations such as aviation, nuclear power, petrochemicals and Medicine (Helmreich and Merritt, 1998).

It is widely accepted that, the relationship between safety climate and safety culture is closely related and they are integrated concepts. After reviewing the relevant literature on safety culture and safety climate table 2-1 is presented to summarize some of the differences and similarities between safety culture and safety climate.

Differences	Safety culture	Safety climate		
and				
similarities				
Background	Has been discussed since Chernobyl disaster in 1986 (Cox and Flin, 1998).	Introduced in 1980 (Zohar, 1980).		
Definition	Safety culture refers to the shared employees' attitudes, beliefs, perceptions, and values towards safety (Cox and Cox, 1991).	Is a snapshot of the surface features of state of safety resulting from the workforce's perceptions attitudes and beliefs about risk and safety issues (Mearns and Flin, 1999).		
Components	Psychological (relates to people's norms, values, attitudes and perceptions), situational (relates to policies and procedures), and behavioural (qualification, knowledge and pattern of behaviour) components (Cooper, 2000).	Safety climate refers to workforces' perceptions of practices, policies and procedures relating to safety within organisation (Neal and Griffin, 2002).		
Dimensions (the most common dimensions)	<ol> <li>Such as:         <ol> <li>Importance of safety</li> <li>Management commitment to safety</li> <li>Communication between workers and across organisational levels.</li> </ol> </li> <li>Reporting system</li> <li>Organisational learning.</li> </ol>	<ul> <li>Such as:</li> <li>1 Management commitment to safety</li> <li>2 Workers involvement</li> <li>3 Communication and feedback about error</li> <li>4 Teamwork</li> <li>5 Organisational learning</li> <li>6 Reporting system</li> </ul>		
Assessment	It could be assessed by measure safety climate. Safety culture is usually measured in industry by workforce questionnaire survey to assess what is called safety climate (Flin et al., 2006 b).	Measured by workforce questionnaire to give a snapshot of the current state of safety (Mearns and Flin, 1999).		
Other considered points	Safety culture is a part of organisational culture. Attitudes are corner stone of culture (Mearns and Flin, 1999). Safety culture is more stable than safety climate and resistant to change (Mearns et al., 1998). Psychological aspect of safety culture is related to safety climate. Organisational safety culture is a measure of common thoughts, behaviours, and beliefs. Safety culture highlights deep features (Cox and Flin, 1998). Positive safety culture characterised by communication founded on mutual trust, by shared perceptions of the importance of safety, and by confidence in the efficacy of preventive measures (HSC,1993:23).	Safety climate is a part of organisational climate. Perceptions are basis of climate (Mearns and Flin, 1999). Safety climate is a measure of individual perceptions or feelings about organisation. Safety climate is a measurable aspect of safety culture (Wiegmann et al., 2002). Safety climate measurement is based on workforce perceptions. Safety climate highlights surface features of safety culture (Cox and Flin, 1998). Positive safety climate characterised by a shared commitment of care and concern towards safety (Cooper and Philips, 2004).		

## Table 2-1 Differences and similarities between safety culture and safety climate

In summary, safety culture reflects the deep values, attitudes and beliefs of members of organisation regarding safety and risk issues. It widely agreed that safety culture is normally assessed by workforce questionnaire to measure safety climate. Safety climate reflects members' perceptions of safety and risk issues in their organisation. Safety climate is usually measured by questionnaire and providing a picture of safety culture in terms of assessing workforce perceptions towards risk and safety issues at that point in time in an organisation.

Safety culture assessment is increasingly recognised as an important factor to improve patient safety in healthcare organisation (Pronovost et al., 2003; Pronovost and Sexton, 2005). Therefore, the current study focuses on patient safety culture assessment and its role in improving patient safety. The following section discusses patient safety culture in healthcare organisations.

#### 2.3.3 Patient safety culture

In healthcare organisations safety culture assessment is increasingly a critical element for improving patient safety (Nieva and Sorra, 2003). The significance of safety culture in preventing medical errors and improving patient safety has led to an increasing number of studies that discuss and assess safety culture in complex organisations such as healthcare and, in particular, in hospitals (Fleming and Wentzell, 2008). Healthcare organisations are increasingly aware of the importance of patient safety culture in order to enhance patient safety through the measurement of patient safety climate. Measuring patient safety climate in healthcare helps to diagnose the underlying patient safety culture of an organisation or work unit (Flin et al., 2006b). It is widely agreed that probably one of commonest ways to assess patient safety culture is based on using patient safety climate questionnaire to assess workers' perceptions of patient safety climate in their healthcare organisations (Sexton et al., 2006b); Fleming and Wentzell, 2008).

The importance of safety culture in healthcare is widely recognised. For example, Nieva and Sorra (2003) argue that the positive safety culture supports patient safety and a poor safety culture is considered as a significant risk factor that can affect patient safety negatively. Similarly, Parker et al. (2006) argue that safety culture is affected by organisational changes, such as the systems, processes and practices of the organisation. For example, an organisation with a poor safety culture often has limited safety systems, while an organisation

with a positive safety culture has many safety systems, processes and practices to promote patient safety. This argument suggests that it should be possible to assess the extent to which systems and processes promote a positive patient safety culture by evaluating organizational practices that influence the patient safety culture. Therefore, patient safety culture assessment tools were developed to assess a number of important organisational practices that may influence patient safety culture (Fleming and Wentzell, 2008).

Flin (2007) has argued that organisational factors of a positive safety culture influence patient safety culture and patient safety outcomes. Pronovost et al. (2003) argued that positive patient safety culture is characterized by clear commitment of leadership to discuss issues that threaten patient safety, learning from medical errors, continuous patient safety improvement, encouraging teamwork, using reporting systems to analyse adverse events and no blame culture.

A positive safety culture in healthcare can be seen as an environment that supports reporting, ends blame culture, effective leadership and focuses on systems. A poor safety culture contributes to medical errors, unsafe therapies and unintended injuries (Institute of Medicine, 2000). Poor safety culture includes barriers to organisational learning from previous incidents (Johnson, 2002). This argument is supported by Nieva and Sorra (2003) who agreed that a positive safety culture supports patient safety while a poor culture is a significant risk factor that can threaten patient safety. Similarly, Fleming and Wentzell (2008) suggested that a poor patient safety culture is a significant risk factor that can threaten patient safety.

The Institute of Medicine (2000) suggest that changing the patient safety culture from blaming culture in which healthcare workers are blamed for errors to an organisational learning culture in which people learn from errors to improve healthcare and prevent harm is the biggest challenge for moving towards patient safety. Similarly, the Institute for Healthcare Improvement (2005) suggests that a patient safety culture is characterised by an environment of mutual trust between staff members to allow them to talk and discuss freely about patient safety problems and work to solve them without fear of blame and punishment. Patient safety culture concerns how healthcare workers' values, attitudes and perceptions regarding their organisation's commitment to patient safety in order to maintain safe work practices in healthcare organisation (Handler et al., 2006).

A patient safety culture involves shared attitudes, values and norms of staff related to patient safety. It is characterized by a number of positive features such as open communication about safety problems, effective teamwork and management supports patient safety to make it a top priority (World Health Organisation, 2008). Sorra and Dyer (2010) argued that patient safety culture is an aspect of organisational culture. It refers to management and staff values, beliefs and norms about important issues in healthcare organisation, behaviour of members, acceptable and unacceptable actions and attitudes, and what process and procedures are in place with regard to patient safety.

All these definitions indicate that patient safety culture is characterised by number of dimensions. The healthcare literature mentioned a number of patient safety culture dimensions. Table 2-2 presents a number of patient safety culture dimensions.

Study			Patien	t safety culture dimensions				
Kirk	et	al.	1	Overall commitment to quality: The policies and procedures of				
(2007)				organisation should be aimed to provide safe and quality healthcare				
				services.				
			2	Priority given to patient safety: The issue of patient safety should be taken				
				in consideration at all levels of staff and management.				
			3	Communication about safety issues: Communication should be open to				
				discuss patient safety issues.				
			4	Staff education and training about safety issue: Availability and usefulness				
				of education and training programs.				
			5	Team working around safety issues: Managing and developing teamwork				
				based on effective cooperation and communication between staff.				
			6	Organisational learning following patient safety incidents: Learning from				
				patient safety incidents to improve patient safety.				
			7	Perceptions of the causes of patient safety incidents and their				
				identification: The mechanisms of reporting incidents.				
			8	Investigating patient safety incidents: How the incidents are investigated?				
			9	Personnel management and safety issues: Managing safety issues in work				
				environment.				

 Table 2- 2 Patient safety culture dimensions

Table 2-2 (continue	d)			
IOM (2004)	1	Healthcare workers share the view that healthcare is a high-risk		
		environment.		
	2	Management commitment to exposing and analysing mistakes (events) that		
		might harm patients.		
	3	A balance between the need for reporting of events and the need to take		
		action to improve patient safety.		
Weingart et al.	1	Leadership (patient safety is high priority to senior leaders)		
(2004)	2	Salience (responsibility to patient safety)		
	3	A non punitive environment (afraid of punishment)		
	4	Reporting and communication (reporting errors and getting feedback about		
		errors for learning from mistakes).		
Handler et al.	1	Non punitive response to error.		
(2006)	2	Teamwork within units.		
	3	Communication openness.		
	4	Feedback and communication about error.		
	5	Organisational learning.		
Amanda et al.	1	Communication,		
(2007)	2	leadership		
	3	Tendency of staff to report errors and teamwork.		
Sammer et al.	1	Leadership role to assure patients are safe from medical errors in		
(2010)		healthcare as a high-risk environment.		
	2	Teamwork in terms of the relationships among staff should be based on		
		cooperation and open communication.		
	3	Evidence-based which means patient care practices should based on		
		evidence and standardised to provide safe and high quality care to patients.		
	4	Communication in terms of an environment that helps start to speak up		
	~	when something is narm patients.		
	2	Organisational learning, stall learns from mistakes for performance		
	C	Improvement.		
	O	failures.		
	7	Patient-centred: Patient care is at the centred of healthcare process.		
		A		

.

It seems that it is not possible to include all these dimensions of patient safety culture at the same time in one patient safety climate questionnaire but some questionnaires such as the HSOPSC questionnaire (Sorra and Nieva, 2004) includes many of these common dimensions.

In summary, it appears that patient safety culture is concerned with shared values, attitudes and perceptions that determine how staff perceive and act upon patient safety within their healthcare organisations. Review of the studies in this section (2.3.3) (see footnote3) shows there are some common patient safety culture dimensions: management commitment to improve patient safety, communication among staff, teamwork, feedback communication about error, event reporting and organisational learning.

A large number of quantitative tools (e.g. Safety Climate Survey) and qualitative tools (e.g. Manchester Patient Safety Assessment Framework) have been developed to assess patient safety culture in healthcare organisations (Kirk et al., 2006). It is widely accepted that assessing patient safety culture in healthcare can be conducted by measuring patient safety climate through using patient safety climate questionnaire (Pronovost et al., 2003; Nieva and Sorra, 2003; Sorra and Nieva, 2004; Colla et al., 2005; Flin, 2007; Singla et al., 2006; Fleming and Wentzell, 2008). Measurement of patient safety climate seeks to explore the shared perceptions of healthcare workers regarding to safe performance in relation to patient care. The perceptions include key issues such as open and honest communication among staff, teamwork, organisational learning, management commitment of safety, feedback about errors and event reporting and analysing (Mary et al., 2005).

#### Summary

In summary, it can be seen that safety culture is a comprehensive concept that reflects the interaction between workers and processes in terms of the state of safety of the organisation. Safety culture refers to employees' shared attitudes, beliefs and perceptions towards safety issues in the workplace. The importance of safety culture in preventing accidents and improving safety in workplace is widely perceived. The characteristics of a positive safety culture are considered as important factors contributing to improve safety of organisations.

<sup>&</sup>lt;sup>3</sup> Kirk et al, 2007; Institute of Medicine, 2004; Wingart et al, 2005; Handler et al., 2006; Amanda et al, 2007; Sammer et al, 2010

Safety climate may be regarded as a clear picture, a current surface and measurable aspect of safety culture and it is most often assessed by workforce questionnaire to measure safety climate. Healthcare organisations are increasingly aware of the importance of patient safety culture in order to enhance patient safety, through measuring patient safety climate using patient safety climate questionnaires. There are a number of patient safety climate questionnaires available to assess patient safety culture. The next section discusses the measurement of patient safety climate and provides the details of patient safety climate questionnaires.

### 2.4 Measurement of patient safety climate

This section is concerned with the measurement of the patient safety climate in healthcare organisations. It examines the existing patient safety climate questionnaires and their characteristics. In addition, the development and validation of patient safety climate questionnaires and the selection of a suitable questionnaire for measuring patient safety climate are discussed in this section.

Patient safety culture can be assessed based on patient safety climate measurement using a patient safety climate questionnaire. The evaluation of patient safety climate generally refers to a measurable component of patient safety culture that is assessed by measuring the perceptions of healthcare staff toward patient safety through the use of patient safety climate questionnaires (Colla et al., 2005; Flin et al., 2006b). It is widely accepted that measuring perceptions of healthcare staff regarding patient safety issues assesses patient safety climate (Sexton et al., 2006b). Therefore, it is possible to measure healthcare professionals' perceptions of patient safety across a range of dimensions by using patient safety climate questionnaires (Fleming and Wentzell, 2008).

The current study aims to identify and validate a patient safety climate questionnaire for use in hospitals in Saudi Arabia. Initially, it is important to review the various patient safety climate questionnaires and their characteristics to select the most appropriate one. Section 2.4.1 identifies existing patient safety climate questionnaires. General characteristics, dimensions and psychometric properties of patient safety climate questionnaires are also discussed in this section.

Section 2.4.2 discusses the development and validation of patient safety climate questionnaires and their psychometric analysis and properties. Section 2.4.3 discusses selecting a suitable questionnaire for measuring patient safety climate. Psychometric testing entails the use of established psychometric assessment techniques to assess psychometric properties of questionnaires, such as exploratory factor analysis, confirmatory factor analysis and reliability analysis. These terms are provided and explained in a glossary (page number XIV).

#### 2.4.1 Existing patient safety climate questionnaires

There are a number of patient safety climate questionnaires available that are used to assess patient safety climate in healthcare. A search strategy was developed to identify existing patient safety climate questionnaires. A literature search was conducted utilising manual searching and electronic databases such as the Medline, Science direct and Pub med Central. The key words included patient safety climate, questionnaires, patient safety culture questionnaires/survey, safety climate questionnaires, measuring patient safety climate and healthcare. A combination of the key words, such as patient safety climate questionnaire in healthcare, and patient safety culture questionnaire in healthcare were used. No restriction on date or study type was used. However, the search was restricted to the English language publications. Websites relevant to patient safety were also used, and these included the website of the Agency for Healthcare Research and Quality (AHRQ), the National Patient Safety Agency (NPSA) and the Institute for Healthcare Improvement (IHI). A review of references from relevant studies was also conducted. The searches were originally conducted in December 2007.

The criteria used for identifying existing patient safety climate questionnaires were:

- 1 Use of questionnaire for measuring patient safety climate.
- 2 Use in healthcare settings.
- 3 Use for assessing the perceptions of healthcare staff.
- 4 Questionnaires are published in English.

The above strategy produced 18 patient safety climate questionnaires as follows:

- 1. Strategies for Leadership: An Organizational Approach to Patient Safety (SLOAPS) (Voluntary Hospital of America, 2000).
- 2. Veterans Administration Patient Safety Culture Questionnaire (VHA PSCQ) (Burr et al., 2002).
- 3. Hospital Survey on Patient Safety Culture (HSOPSC) (Sorra and Nieva, 2004).
- 4. Safety Attitudes Questionnaire (SAQ) (Sexton et al., 2004).
- 5. Safety Climate Survey (Sexton et al., 2000, 2003).
- 6. Hospital Transfusion Service Safety Culture Survey (Sorra and Nieva, 2002).
- Medication Safety Self-Assessment (MSSA) (Institute for Safe Medication Practices, 2000).
- 8. Culture of Safety Survey (CSS) (Weingart et al., 2004).
- 9. Teamwork and Patient Safety Attitudes Questionnaire (Kaissi et al., 2003).
- 10. Operating Room Management Attitudes Questionnaire (ORMAQ) (Helmriech et al., 1997).
- 11. Modified Operating Room Management Attitudes Questionnaire (MORMAQ) (Flin et al., 2003).
- 12. Safety Climate Scale (SCS) (Pronovost et al., 2003; Kho., 2005).
- 13. Stanford University Patient Safety Climate in Health Care Organisation Questionnaire (Singer et al., 2003).
- 14. Patient Safety Culture in Health Care Organisation (Modified Stanford instrument) (Singer et al., 2007).
- 15. An employee questionnaire for assessing patient safety (Carayon et al., 2005).
- 16. Team Work and Safety Climate Survey (Sexton et al., 2004).
- 17. Leiden Operating Theatre and Intensive Care Safety Scale (Beuzekom, 2007).
- 18. Patient Safety Culture Improvement Tool (Fleming and Wentzell, 2008).

A number of review papers concluded that the patient safety climate questionnaires varied according to the general characteristics, dimensions covered and psychometrics properties (e.g. Colla et al., 2005; Fleming, 2005; Flin et al., 2006b; Singla et al., 2006). Further details and examples of these characteristics are presented below in order to clarify this variation among patient safety climate questionnaires.

#### General characteristics of patient safety climate questionnaires

This section focuses on the general characteristics of patient safety climate questionnaires such as source, aim, settings suitable for use, respondents (who are supposed to complete a questionnaire), length of questionnaires and response scale.

A number of questionnaires have been adapted for healthcare from other industries such as aviation and nuclear because most of safety climate questionnaire were first developed in these industries (Helmreich and Merritt, 1998). For example, the SAQ (Sexton et al., 2004) was adapted from the Flight Management Attitude Questionnaire. The Modified Operating Room Management Attitudes Questionnaire (MORMAQ) was adapted from the Cockpit Management Attitudes questionnaire (Flin et al., 2003). Recently, a number of patient safety climate questionnaires have been developed specifically for healthcare such as the Hospital Survey on Patient Safety Culture (HSOPSC) (Sorra and Nieva, 2004). The majority of patient safety climate questionnaires originate from the USA (Colla et al., 2005). For example, the Agency for Healthcare Research and Quality (AHRQ) in USA developed the HSOPSC (Sorra and Nieva, 2004). The SAQ (Sexton et al., 2004) was developed at the University of Texas, Centre of Excellence for Patient Research and Practice.

In terms of the aims of patient safety climate questionnaires, each questionnaire has a specified aim. Most questionnaires are designed to measure attitudes and perceptions of staff about various aspects of patient safety. For example, the SAQ (Sexton et al., 2004) claims to measure attitudes of healthcare staff towards safety. The HSOPSC questionnaire (Sorra and Nieva, 2004) was developed to assess staff perspectives on patient safety culture (Handler et al., 2006). The SCS (Pronovost et al., 2003) was developed to assess perceptions of healthcare staff regarding the commitment to patient safety as a positive culture (Pronovost et al., 2003). In contrast, some questionnaires, such as the SLOAPS (Voluntary Hospital of America, 2000) and the MSSA (Institute for Safe Medication Practices, 2000) were developed to measure the extent to which patient safety improvement activities and concerns have been implemented (Colla, 2005). The SLOAPS (Voluntary Hospitals of America, 2000) aims to assess the extent to which safety is a strategic priority for a healthcare organisation. In other words, it is designed to assess attitudes of leadership towards patient safety (Pronovost et al., 2003).



Chapter two

Nieva and Sorra (2003) state that some questionnaires assess staff perspectives that focus on daily activities within organisation and their impact on patient safety. An example of a questionnaire that focuses on staff perceptions is the HSOPSC questionnaire (Sorra and Nieva, 2004). Some questionnaires assess managerial perspectives that focus on management assessment of patient safety policies and practices in healthcare organisations. An example of a self-assessment questionnaire designed for use by managers is the SLOAPS (Voluntary Hospital of America, 2000). However, the majority of patient safety climate questionnaires focus on the perceptions of healthcare staff while a minority of questionnaires are used at management level. Some questionnaires are designed to be completed as a team together such as the MSSA (Institute for Safe Medication Practices, 2000).

In short, patient safety climate questionnaires measure either staff or managerial perspectives, or combine perspectives of both. The questionnaires' respondents are healthcare staff including, senior managers, clinicians (front line staff such as physicians and nurses) and non clinicians (Nieva and Sorra, 2003). According to Singla et al. (2006) most patient safety climate questionnaires were designed to be completed by hospital staff including physicians, nurses, pharmacists and other caregivers.

The majority of patient safety climate questionnaires are appropriate for general use (general evaluation of patient safety climate in healthcare settings) such as the HSOPSC (Sorra and Nieva, 2004) and the Teamwork and Safety Climate Survey (Sexton et al., 2004). In contrast, others are designed for specific departments, for example the MSSA (Institute for Safe Medication Practices, 2000) for use in pharmacy. The Hospital Transfusion Service Safety Culture Survey (Sorra and Nieva, 2002) is used for transfusion services. Colla et al. (2005) reviewed nine patient safety climate questionnaires designed to measure patient safety climate. They found five of these questionnaires were used for general evaluation of patient safety climate across healthcare settings.

Flin et al. (2006b) reviewed 12 patient safety climate questionnaires designed to measure safety climate and diagnose the underlying safety culture of healthcare organisations. They found five of these questionnaires for general evaluation of patient safety climate in healthcare settings. Singla et al. (2006) reviewed 13 patient safety climate questionnaires. They found 9 questionnaires were designed for general use, for example the SAQ (Sexton et

al., 2004) and the HSOPSC questionnaire (Sorra and Nieva, 2004). Four questionnaires were designed for specific use, for example the Modified Operating Room Management Attitudes Questionnaire (MORMAQ) (Flin et al., 2003). It seems that the majority of the patient safety climate questionnaires are designed for general use.

The majority of patient safety climate questionnaires are designed to be self-administered (Singla et al., 2006). The majority of questionnaires use the 5 point Likert scale such as the HSOPSC questionnaire (Sorra and Nieva, 2004). Some questionnaires use the 6 point Likert scale, for example the Teamwork and Safety Climate Survey (Sexton et al., 2004). The majority of patient safety climate questionnaires use the five-point scale reflecting the level of agreement with statement about patient safety: from 'strongly disagree' (1) to 'strongly agree' (5), with a neutral category 'neither' (3). Other items can be answered using a five-point frequency scale from 'never' (1) to 'always'(5) (Colla et al., 2005).

The number of items to be completed ranges from 10 items in the Safety Climate Scale (Sexton et al., 2000, 2003) to 194 items in the MSSA (Institute for Safe Medication Practices, 2000) (Colla et al., 2005). The demographic information in most patient safety climate questionnaires includes data on gender, years of experience and job title.

#### Dimensions of patient safety climate questionnaires

Patient safety climate questionnaires differ from one another in the dimensions covered. These dimensions range from four dimensions, such as the Safety Climate Scale (Kho et al., 2005), while others include 12 dimensions, for example, the HSOPSC questionnaire (Sorra and Nieva, 2004). A number of studies reviewed patient safety climate questionnaires. Table 2-3 presents dimensions of patient safety climate questionnaires in these studies.

Study	Dimensions of Patient safety climate questionnaires
Colla et al.	1 Leadership
(2005)	2 Policies and procedures
	3 Staffing,
	4 Communication
	5 Reporting incidents.
Fleming (2005)	1 Management commitment to support patient safety.
	2 Importance of safety.
	3 Non punitive culture (no blame and shame response).
	4 Handoffs and transitions.
	5 Stress recognition.
	6 Staffing.
	7 Working condition relevant to patient safety.
	8 Reporting system.
	9 Teamwork (teamwork within units, teamwork across units).
	10 Communication openness.
	11 Organisational learning.
	12 Feedback and communication.
	13 Job satisfaction.
	14 Overall perception of safety.
	15 Supervisors' expectations and actions.
	16 Organisation/department.
Flin et al.	1 Management support and commitment to safety.
(2006b)	2 Supervisors' role.
	3 Communication and feedback.
	4 Reporting incidents.
	5 Teamwork.
	6 Rules and procedures of work.
	7 Personal resources.
	8 Job demands.
	9 Risk perception and safety attitudes of workers.
	10 Organisational factors.
	11 Safety systems.

Table 2- 3 Dimensions of patient safety climate questionnaires

Table 2-3 (continued)				
Singla	et	al.	1	Management commitment to safety.
(2006)			2	Non punitive culture (no blame and shame response).
			3	Handoffs and transitions.
			4	Working condition relevant to patient safety.
			5	Stress recognition.
			6	Risk taking (hazard detection).
			7	Frequency of event reporting.
			8	Teamwork.
			9	Staffing.
			10	Communication openness.
			11	Organisational learning.
			12	Feedback and communication about error.
			13	Beliefs about causes of errors such as human factors (e.g. fatigue and stress
				on patient safety).
			14	Job satisfaction.
			15	Overall perception of safety.

In general, it seems that patient safety climate questionnaires vary in the dimensions addressed. However, it appears that most of patient safety climate questionnaires cover a number of common dimensions of patient safety climate. A comparison of dimensions was conducted across number of review papers (e.g. Fleming, 2005; Colla et al., 2005; Flin et al., 2006b; Singla et al., 2006). These review papers include dimensions that related to patient safety climate in most of patient safety climate questionnaires that widely used and available in the literature. These dimensions are mentioned as common dimensions in most of these key review papers. Table 2-2 compares the 12 common patient safety climate dimensions that are mentioned in the four review papers.

Patient safety climate	Colla et al.,	Fleming,	Flin et al.,	Singla et al.,	
dimensions	(2005)	(2005)	(2006)	(2006)	
	Nine	four	12	13	
	questionnaires	questionnaires	questionnaires	questionnaires	
Overall perceptions of patient safety (Priority given to patient safety (patient safety is coming first and/safety system)		$\checkmark$	$\checkmark$	·······.	
Feedback and communication	$\checkmark$	$\checkmark$	$\checkmark$		
Incident reporting	$\checkmark$	$\checkmark$	$\checkmark$		
Staffing	$\checkmark$	$\checkmark$			
Leadership/supervisors role	$\checkmark$	$\checkmark$	$\checkmark$		
Policies and procedures of work/working condition relevant to patient safety	$\checkmark$	$\checkmark$	$\checkmark$		
Organisational learning (Learning from patient safety incidents)		$\checkmark$	$\checkmark$		
Management commitment to support patient safety		$\checkmark$	$\checkmark$	$\checkmark$	
Teamwork		$\checkmark$	$\checkmark$	$\checkmark$	
Communication openness		$\checkmark$		$\checkmark$	
Perception and understanding of the causes of patient safety incidents				$\checkmark$	
Non punitive responses to error		$\checkmark$			

# Table 2-4 A comparison of the common patient safety climate dimensions that are mentioned in four review papers

These questionnaires in table 2-4 are included in the list of existing patient safety climate questionnaires in section 2.3.1 as the most widely used questionnaires that are representative of the patient safety climate questionnaires currently available in the literature.

From table 2-2 it seems that management commitment to support patient safety, feedback and communication, incident reporting, leadership/supervisors role, teamwork, and policies and procedures of work/working condition relevant to patient safety are common dimensions across three of the review papers. The dimensions for priority given to patient safety, staffing, organisational learning, and communication openness are common across two of the review papers. The dimensions for preception and understanding of the causes of patient safety incidents and non punitive responses to error are mentioned in one of the review papers. These dimensions were discussed in the section of safety culture and safety climate (2.3.2) in the current chapter.

It is important that a patient safety climate questionnaire should be a valid and reliable tool and it should measure the common dimensions of patient safety culture (Nieva and Sorra, 2003). Although the reviewed literature revealed that in recent years there has been increasing attention towards measuring patient safety climate, it has been reported that there is a lack of information reporting related psychometric properties of patient safety climate questionnaires (Flin et al., 2006b). The psychometric properties of patient safety climate questionnaires are discussed below.

#### Psychometric properties of patient safety climate questionnaires

The term psychometric can be described as the science of the measurement of mental functions, theoretically, when a concept does not lend itself to being measured directly, a set of questions which explore various aspects of the same concept can be asked and then tested for their reliability and validity (Bowling, 2005). Psychometric analysis is concerned with statistical techniques that help to assess the validity and reliability of the items of patient safety climate questionnaires (Singla et al., 2006). The validity and reliability of instruments are very important aspects of psychometric measurement. Instrument validity refers to what extent a measurement instrument measures what it is intended to measure, while reliability is based on the measuring tool which yields the same results on repeated occasions (Carmines and Zeller, 1979).

Reliability is also referred to as consistency. The more common way of estimating reliability is by assessing internal reliability which means to what extent a group of items relates to a specific dimension of instrument (Pett et al., 2003). It can be assessed by using a statistical test called Cronbach's alpha (Cronbach, 1951).

A number of statistical analyses are used to assess the psychometric properties of questionnaires. For example, item analysis is used to provide descriptive item responses. Factor analysis is the most frequently used statistical technique and is employed to assess the psychometric properties of survey instruments (Colla et al., 2005; Flin et al., 2006b). It can be used to define a number of basic dimensions, each of which contains items which can be grouped together in a consistent and structured way (Bowling, 2005). The more popular choice for estimating reliability is internal consistency (Etchegaray and Fischer, 2006).

Although there is agreement about the importance of assessing psychometric properties (validity and reliability) of patient safety climate questionnaires, there is a lack of reporting the data of the psychometric properties of questionnaires. The psychometric properties of patient safety climate questionnaires are not always provided in published studies, for instance, the Stanford University Patient Safety Climate in Health Care Organisation (Pronovost et al., 2003; Fleming, 2005). Colla et al. (2005) examined the psychometric properties of nine patient safety climate questionnaires using a number of psychometric tests as review criteria, including item analysis, Exploratory Factor Analysis (EFA), Confirmatory Factor Analysis (CFA), Cronbach's alpha, correlation across dimensions, test/retest reliability and analysis of variance to test for variance between groups of a sample. They found the psychometric testing varied considerably across nine patient safety climate questionnaires and there was a limitation on reporting psychometric data from most patient safety climate questionnaires. For example, while psychometric properties were reported for three questionnaires: the HSOPSC (Sorra and Nieva, 2004), Hospital Transfusion Service Safety Culture Survey (HTSCS) (Sorra and Nieva, 2002) and the SAQ (Sexton et al., 2004), other questionnaires had a lack of reporting of their psychometric properties.

Flin et al. (2006b) reviewed the psychometric properties of 12 patient safety climate questionnaires including: construct validity which refers to the factor (dimension) structure of a questionnaire and content validity which refers to the degree to which elements of a measure are relevant to a construct. Content validity of patient safety climate questionnaire

refers to the degree to which items of a patient safety climate questionnaire are relevant to patient safety culture (items related to dimensions). Internal consistency is assessed by Cronbach's alpha. Finally, factor structure is assessed by factor analysis (Pallant, 2007). In this review (Flin et al., 2006b) only six studies were found to report the results of factor analysis and Cronbach's alpha. Furthermore, only one study provided a comprehensive report of scale development and psychometric properties, relating to the HSOPSC questionnaire (Sorra and Nieva, 2004).

Flin et al. (2006b) found there was a lack of clear theoretical reinforcement for most questionnaires and psychometric properties of some questionnaires were not reported where this information was available. Also, they concluded that there was a lack of standard psychometric analysis in several studies. It seems that the main limitation of Flin et als' review was the lack of reporting psychometric data of patient safety questionnaires.

Similarly, Singla et al. (2006) examined the psychometric properties of 13 patient safety climate questionnaires. Psychometric testing was reported for two questionnaires, the SAQ (Sexton et al., 2004) and the HSOPSC questionnaire (Sorra and Nieva, 2004). Some psychometric testing was reported for three questionnaires. No psychometric was reported for eight of the questionnaires.

In general, the authors of these review papers agreed that there is a lack of information reporting psychometric properties of a number of patient safety climate questionnaires. Moreover, it was noted that most of the above studies used general words such as adequate psychometric properties and good psychometric properties in relation to the psychometric evaluation of patient safety climate questionnaires. Unfortunately, this could increase the ambiguity of psychometric literature. Therefore, it is recommended that more consideration should be given to psychometric properties in the design of healthcare safety climate instruments and health services researchers need to assess the validity and reliability of their measures correctly (Flin et al., 2006; Etchegaray and Fischer, 2006).

Psychometric data should be reported in a clear way by using standardised psychometric tests. For example, as a minimum psychometric analysis should include EFA, CFA and reliability analysis and the correlations between questionnaire dimensions (factors) to ensure a rigorous scientific approach (Flin et al., 2006b). Factor analysis is commonly used as a measure of the construct validity of a questionnaire by using EFA to test factor structure (grouping a number of items together into one extracted factor) and CFA to test the proposed factor structure of the questionnaire (Hutchinson et al., 2006). In terms of reliability, internal consistency is used to assess questionnaire reliability by calculating the Cronbach's alpha ( $\alpha$ ) coefficient alpha value for each factor of a questionnaire. It is generally agreed that questionnaire validity is good when items load strongly to one factor (item loading >0.4) and reliability being good when coefficient alpha value >0.7 under these conditions, all items of a questionnaire are assumed to be correlated (Wet et al., 2010).

To conclude, it is seen that patient safety climate questionnaires vary in a number of ways, such as general characteristics, content (dimensions) and psychometric properties (validity and reliability). Although there are differences between the questionnaires most of them have important similarities. For instance, they are designed to be used in particular settings such as hospitals and they are mainly intended for front line respondents (e.g. physicians, nurses). Furthermore, there are common dimensions across a number of patient safety climate questionnaires and psychometric analysis includes factor analysis and reliability analysis.

In general, a patient safety climate questionnaire should be a valid and reliable questionnaire and it should measure the common dimensions of patient safety culture (Nieva and Sorra, 2003). The following section aims to discuss the development and validation of patient safety climate questionnaires including their psychometric analysis and properties.

# 2.4.2 Development and validation of patient safety climate questionnaires

The development of patient safety climate questionnaires is increasingly recognised as an important tool for assessing and improving patient safety culture in healthcare organisations (Nieva and Sorra, 2003; Colla et al., 2005; Prononvost and Sexton, 2005). Recent studies have focused on developing patient safety climate questionnaires either by adapting a questionnaire from other industries or by developing new questionnaires for healthcare organisations to measure patient safety climate in order to assess and improve patient safety culture. A number of questionnaires that were either adapted or developed have been validated in a number of countries (either original country of questionnaire or outside its country) (e.g. Sexton et al., 2000, 2003, 2004; Sorra and Nieva, 2004; Singer et al., 2003, 2007).

The first development approach has been to adapt patient safety climate questionnaires from other industries (adapting approach). A number of questionnaires were adapted for healthcare from other industries such as aviation as early stage of developing patient safety climate questionnaires. For example, the Safety Attitude Questionnaire (SAQ) (Sexton et al., 2004) was adapted from the Flight Management Attitude Questionnaire (FMAQ). The Safety Climate Scale (SCS) (Kho et al., 2005) was adapted from a widely used questionnaire in aviation called the Flight Management Attitudes and Safety Survey which is designed to measure attitudes toward stress, status hierarchies, leadership and interpersonal interaction. The SCS questionnaire is used to assess safety culture among staff (Pronovost et al., 2003). Helmreich et al. (1997) extended their work with the aviation industry to examine the attitudes of hospital operating rooms staff to teamwork and safety. They adapted the Cockpit and Flight Management Attitudes Questionnaires (CMAQ) to produce the Operating Room Management Attitudes Questionnaire (ORMAQ) (Helmreich et al., 1997). The Teamwork and Safety Climate Survey was adapted for use in healthcare settings (Sexton et al., 2004).

A number of patient safety climate questionnaires originated from the USA (Colla et al., 2005). Therefore, most of them were validated in the USA healthcare settings. For example, the SAQ was developed at the University of Texas, Centre of Excellence for Patient Research and Practice by Sexton et al. in 2004 with new items generated by a focus group of healthcare providers, review of the literature and consultations with safety experts. The items were then

evaluated through pilot testing. The SAQ contains 60 items and covers six dimensions, which were identified in psychometric testing: teamwork, safety climate, perceptions of management, job satisfaction, stress recognition and work condition. Internal consistency by Cronbach's alpha coefficient ranged from 0.74 to 0.93. It demonstrated good psychometric properties. There are several different versions with slight modifications of its items based on different departments (e.g. emergency, pharmacy, intensive care unit, operating rooms and laboratory) (Sexton et al., 2006).

In addition, some adapted questionnaires have been applied and empirically evaluated in different countries. For example, the SAQ (Sexton et al., 2004) was translated into Norwegian and validated in Norway (Norwegian university hospital) (Deilkas and Hofoss, 2008), 1306 staff members completed and return the questionnaire (68% response rate). Reliability analysis and confirmatory factor analysis were performed. The factor structure was tested by confirmatory factor analysis and this identified 36 items representing seven dimensions: teamwork, safety climate, perceptions of hospital management, perception of unit management, job satisfaction, stress recognition and work condition. They used goodness of fit indices: the Adjusted Goodness of-Fit Index (AGFI), and the Root Mean Square Error of Approximation (RMSEA). CFA showed acceptable goodness of fit values. The reliability analysis was acceptable for the seven dimensions (Cronbach's alpha coefficient ranged from 0.68 to 0.85). The SAQ Norwegian version showed satisfactory psychometric properties (Deilkas and Hofoss, 2008).

Norden-Hagg et al. (2010) conducted a validation study of the SAQ (Sexton et al., 2004) for use in Swedish pharmacies. The SAQ data from 828 community pharmacies in Sweden, including 6,683 eligible pharmacists were received. The psychometric properties of the translated questionnaire of the SAQ (Sexton et al., 2004) were analysed using Cronbach alpha and inter correlations among the scales. CFA was conducted. The coefficient alpha value for each of the SAQ scales ranged from 0.72 to 0.89. The confirmatory factor analysis results, demonstrate that the Swedish translation of the SAQ has acceptable, to good, psychometric properties. Perceptions of the pharmacy (teamwork climate, job satisfaction, perceptions of management, safety climate, and working conditions) were moderately to highly correlated with one another whereas attitudes about stress (stress recognition) had only low correlations with other factors. Norden-Hagg et al. (2010) conclude that the Swedish
translation of the SAQ demonstrates acceptable construct validity, for assessing the frontline perspective of safety culture in community pharmacy staff.

The three examples above of using the SAQ showed that the psychometric results are consistent. In other words, using the SAQ (Sexton et al., 2004) in different healthcare settings and other languages showed that the SAQ was valid and reliable. The development and characteristics of SAQ is provided in section 4.2 of chapter four.

A number of patient safety climate questionnaires have been validated in different countries. For example, the Modified Operating Room Management Attitudes Questionnaire (MORMAQ) was adapted by Flin et al. (2003) to use in hospitals in the United Kingdom. The MORMAQ was adapted from the Cockpit Management Attitudes questionnaire. It contains 60 items and covers seven dimensions. The proposed factor structure measures attitudes of staff to leadership, communication, teamwork, stress and fatigue, work values, human error and organisational climate. The questionnaire was completed by 222 anaesthetists. A reliability analysis of the proposed structure based on Cronbach's alpha scores for each of the proposed factors showed low values ( $\alpha = 0.18-0.54$ ) for internal consistency (Flin et al., 2003).

Hutchinson et al. (2006) examined the factor structure, reliability and potential usefulness of the 27 item Teamwork and Safety Climate Survey (Sexton et al., 2004) for use in the UK healthcare, based on a sample of n=897. Factor analysis and reliability analysis were carried out. EFA was carried out on a random 50% sample of respondents (the "construction" half of the data) using principal components extraction. Table 2-3 shows the psychometric information of Hutchinson et al's study (2006).

Exploratory factor analysis to identify optimal model	Five factors were identified
Confirmatory factor analysis on 11 factors	CFA indicated adequate fit. The CFI
	and RMSEA took values of 0.93 and
	0.08 for teamwork, and 0.94 and 0.07
	for safety climate
Reliability analysis on five factors	Internal consistency reliabilities were
	satisfactory to good, with Cronbach's
	alpha 0.69 or above in all five factors
Optimal model	five factors and 22 items

# Table 2- 5 Psychometric properties of Hutchinson et al's study (2006)Psychometric analysis performedPsychometric properties

Table 2-3 shows that five factors and 22 items were identified. CFA on the remaining 50% (the "validation half") of the dataset indicated an almost adequate fit of the model to the data under the widely applied fit indices criteria. The CFI and RMSEA values were: 0.93 and 0.08 for teamwork, and 0.94 and 0.07 for safety climate. Internal consistency reliabilities were satisfactory to good, with Cronbach's alpha ( $\alpha$ ) 0.69 or above in all five factors.

To summarise, it appears that the early stage of the measurement of patient safety climate in healthcare was an immature stage of development as compared to other industries such as aviation. Therefore, adapting commonly used questionnaires from other industries was the common approach as healthcare organisations started to work with patient safety culture and with the measurement patient safety climate. Therefore, learning from other industries experiences such as aviation was helpful in understanding the concepts and how they were measured (Helmreich, 2000; Sexton et al., 2000). Although, validation studies in healthcare (e.g. Hutchinson et al., 2006; Norden-Hagg et al. 2010) showed that some questionnaires such as the SAQ (Sexton et al., 2004) and the Teamwork and Safety Climate Survey (Sexton et al., 2004) had acceptable psychometric properties across different countries, there was a lack of psychometric data for early questionnaires, for example, psychometric properties and analysis of the SCS had not been reported (Pronovost et al., 2003).

The reason for not reporting psychometric data in some studies might be because they were concerned with developing new item questionnaires to use in healthcare rather than evaluating psychometric properties. In addition, there was a little experience in healthcare of safety culture assessment but currently there are now a number of questionnaires for assessing safety culture in healthcare (Fleming, 2005).

The situation of measuring patient safety climate has changed to an approach in which a number of questionnaires have been developed specifically for healthcare organisations (Nieva and Sorra, 2003). For example, the SLOAPS was developed by the Volunteer Hospital Association (2000) to assess the extent to which safety is a strategic priority for healthcare leaders (Pronovost et al., 2003), 23 clinical and administrative leaders completed the SLOAPS. No psychometric testing has been reported. The Teamwork and Patient Safety Attitudes Questionnaire consists of 24 items across 4 dimensions. The questionnaire was completed by 261 staff members. Psychometric analysis was performed and revealed four dimensions: perceived of teamwork, support for team communication and decision making, teamwork in department and leadership role. Internal consistency results were reported for each dimension (Kaissi et al., 2003).

Singer et al. (2003) developed a patient safety climate questionnaire based on a review of existing patient safety climate questionnaires. They identified a list of key dimensions. A draft questionnaire consisting of 122 items was tested in pilot study to get feedback and to modify the questionnaire. The questionnaire was completed by 21,496 (response rate 51%). The principle factor analysis was conducted on the responses of 82 questions and this identified five factors: organisation, department, production, reporting/seeking help and shame/self-awareness. The initial psychometric properties and analysis were not reported (Fleming, 2005). After that the revised version consisting of 38 items was used to collect data. Analysis of the psychometric properties of the questionnaire was performed by splitting the sample into derivation sample and validation sample, and then assessed by using exploratory factor analysis to responses in derivation sample. They identified nine factors (dimensions) and 38 items, Cronbach's alpha coefficients ranged from 0.50 to 0.89 (Singer et al., 2007).

Fleming and Wentzell (2008) developed the Patient Safety Culture Improvement Tool (PSCIT) to assess patient safety culture in healthcare organisations. The early development stage of the PSCIT involved reviewing the literature on patient safety culture assessment and patient safety culture instruments. Five interviews with patient safety experts across Canada were conducted to assess the content and face validity of the original version of the PSCIT questionnaire. This questionnaire covers five patient safety culture dimensions such as leadership, risk analysis, workload management, sharing and learning and resource management. However, Fleming and Wentzell (2008) conclude that this questionnaire should be used with caution because the psychometric properties (validity and reliability) have not been determined.

It seems that although some patient safety climate questionnaires were developed to assess patient safety culture in healthcare organisations, there is limited reporting of psychometric data for some questionnaires (e.g. SLOAPS (Volunatry Hospital of America, 2000); Teamwork and Patient Safety Attitudes Questionnaire (Kaissi et al., 2003); Patient Safety Culture Improvement Tool (Fleming and Wentzell, 2008). More recently, patient safety climate questionnaires have been developed specifically for healthcare such as the HSOPSC questionnaire (Sorra and Nieva, 2004) (Flin et al., 2006b).

For example, the Agency for Health Care Quality and Research (AHQR) developed the HSOPSC questionnaire. The original items were validated by the Agency for Healthcare Research and Quality (AHRQ) for the USA hospital settings (Sorra and Nieva, 2004). The HSOPC questionnaire is based on a set of pilot studies carried out in 21 different hospitals involving 1461 hospital staff across the USA. The HSOPSC questionnaire (Sorra and Nieva, 2004) is considered valid, reliable, and the most efficient questionnaire for assessing patient safety culture (Colla et al., 2005; Flin et al., 2006b). The HSOPSC questionnaire (Sorra and Nieva, 2004) was also modified to assess patient safety culture in the nursing home setting. The internal consistency of the individual dimensions was generally similar to that of the original HSOPSC questionnaire (Sorra and Nieva, 2004). Cronbach's alpha values ranged from 0.50 for staffing to 0.84 for teamwork across units (Handler et al., 2006). Table 2-6 shows the original analysis of the HSOPSC questionnaire (Sorra and Nieva, 2004).

# Table 2- 6 The original analysis of the HSOPSC (Sorra and Nieva, 2004) Sorra and Nieva's original analysis

- 1 Cognitive interviews were conducted to assess the face validity of the questionnaire
- 2 Pilot study is conducted.
- 3 4983 questionnaires were administered across 21 hospitals in USA, with 1437 responses received (29%response rate)
- 4 The psychometric analysis consisted of analytical techniques:
- Reliability analysis: Cronbach's alpha values ranged from 0.63 to 0.84.
- Item analysis
- Content analysis
- Exploratory factor analysis.
- Confirmatory factor analysis.
- Correlation analysis.
- Analysis of variance.

The HSOPSC questionnaire (Sorra and Nieva, 2004) is used increasingly in different countries such as the UK, Netherland, Turkey and Germany. This might indicate that this questionnaire reflects stable psychometric properties when used in different contexts. A number of studies have validated the HSOPSC questionnaire (Sorra and Nieva, 2004) to measure patient safety culture in different health care settings across different countries. Some illustrative examples of recent validation studies in healthcare in different countries are provided below.

Waterson et al. (2009) report on the assessment of the psychometric properties and suitability of the American HSOPSC questionnaire (Sorra and Nieva, 2004) for use within the UK. The questionnaire was completed by 1437 staff members. EFA, CFA and reliability analyses were carried out to assess the psychometric performance of this questionnaire. The results of reliability analysis of the items within each original scale showed that more than half the scales failed to achieve satisfactory internal consistency (Cronbach's alpha <0.7). Furthermore, a CFA carried out on the UK data set achieved a poor fit when compared with the original American model. An optimal measurement model was then constructed via exploratory and confirmatory factor analyses with split half sample validation and consisted of nine dimensions. The sample was split randomly into two halves; on one "construction" half, EFA was used to construct a measurement model for the items; the other "validation" half of the data was then used to test this model via CFA.

#### Chapter two

The results of a reliability analysis on the original dimensions (12 groupings of items) showed that seven dimensions (overall perceptions of safety, supervisor/manager expectations, organisational learning—continuous improvement, communication openness, non-punitive responses to error, staffing, hospital management support) fell short of an adequate level of internal consistency (Cronbach's alpha <0.70), with staffing exhibiting an extremely poor level of reliability ( $\alpha$ =0.58). Only two of the dimensions achieved alpha values >0.80 (frequency of error reporting, and feedback and communication about error). A CFA of the original model was then run ( $\chi^2$  =1907, df =674); the full range of fit indices suggested a level of fit with minor adequacy; specifically Comparative Fit Index (CFI) =0.91, Non-Normed Fit Index (NNFI) =0.89, RMSEA=0.04, Standardised Root Mean Square Residual (SRMSR) =0.05.

Waterson et al. (2009) performed an EFA, using principal axis factoring as the extraction method and assessing the number of factors to be extracted by a combination of Kaiser's criterion and Cattell's screen plot method. An oblique rotation was carried out to aid interpretation of the resulting factors. Having examined a series of possible models, and gradually removing 13 items that were either severely cross-loaded or had very low loadings and communalities, the evidence pointed most strongly towards a nine-factor model for the remaining 27 items. Then, they tested the fit of this model to the other "validation" half of the data set using CFA ( $\chi^2$  =588, df =288). The fit indices suggested an adequate fit to the data, with CFI =0.95, Tucker-Lewis Index (TLI) =0.93, RMSEA=0.04, SRMSR=0.04.

Finally, using the whole sample, reliability analyses were performed for each of the groups of items defined by this factor structure. This indicated suitable internal consistency, with Cronbach's alpha >0.7 for seven of the nine dimensions. Only two dimensions fell below this level. Table 2-4 summarises the Psychometric analysis performed and psychometric properties provided of this validation study (Waterson et al., 2009).

Psychometric analysis performed	Psychometric properties
Reliability analysis on the original HSOPSC model	More than half failed to achieve
	Satisfactory internal consistency
Confirmatory factor analysis on the original HSOPSC model	A poor fit (CFI=0.91, NNFI=0 89
· · · · · · · · · · · · · · · · · · ·	RMSEA=0.04, SRMSR=0.05)
Exploratory factor analysis to identify optimal model	Nine factors were identified
Confirmatory factor analysis on nine factors	An adequate fit CFI= $0.95$ , TLI = $0.93$ ,
	RMSEA=0.04, SRMSR =0.04
Reliability analysis on nine factors	Suitable internal consistency, with
	Cronbach's $\alpha > 0.7$ for seven of the nine
	dimensions. Two dimensions $< 0.7$

# Table 2-7 Psychometric properties of Waterson et al's study (2009)

optimal model

Nine factors and 27 items

Another example of validation studies is provided by Bodur and Filiz (2010). They translated the HSOPSC questionnaire (Sorra and Nieva, 2004) to assess patient safety culture in Turkish hospitals. The aim of the study was to assess the validity and reliability of the translated form of the questionnaire. The questionnaire was completed by 309 staff members. The data analysis strategy was summarized as percentages, means, and standard deviation values.

Factor analysis, correlation coefficient, Cronbach's alpha ( $\alpha$ ), analysis of variance (ANOVA), and t-tests were employed in statistical analyses. Items on patient safety were categorized into 10 factors. Factor loadings and internal consistencies of dimension items were high. They (Bodur and Filiz, 2010) concluded that the Turkish version of HSOPSC questionnaire was sufficiently valid and reliable to assess patient safety culture. The internal consistency of the Turkish items was lower for each factor than the original items in the AHRQ study except for frequency of events reported ( $\alpha = 0.86$ ) and teamwork within units ( $\alpha = 0.83$ ). The internal consistency of two factors was poor: staffing ( $\alpha = 0.19$ ) and non-punitive response to error ( $\alpha = 0.31$ ). The CFA results for the original showed a poor fit (fit indices not provided).

The EFA identified 10 factors. Factor loadings were between 0.36 and 0.87. The factors jointly explained 62.1% of the variance in the responses. The internal consistency of the 10 factors was calculated with Cronbach's alpha coefficient was between 0.57and 0.86, except in two dimensions. Internal consistency reliability for all items was high ( $\alpha = 0.88$ ). Construct validity for each of the 10 factors, were calculated by obtaining the mean of the item scores within one factor for every respondent and correlations between the scale scores were calculated. The scale scores showed there was correlation between factors. Table 2-5 summarises the psychometric properties of Bodur and Filiz's study (2010).

Psychometric analysis performed	Psychometric properties (details )
Reliability analysis on the original HSOPSC model	Internal consistency: 8 factors lower
	than original model. One factor $\alpha$ =0.86. Two factors were poor ( $\alpha$ =0.10
	0.31).
Confirmatory factor analysis on the original HSOPSC model	A poor fit (fit indices not provided).
	· · ·
Exploratory factor analysis to identify optimal model	Ten factors were identified
Confirmatory factor analysis on ten factors	Not performed
Reliability analysis on nine factors	Internal consistency: Cronbach's alpha
	coefficient was between 0.57 and 0.86,
	except in two dimensions.
Construct validity	There was correlation between 10
	factors
	· · · ·
Optimal model	10 factors and 42 items

#### Table 2- 8 Psychometric properties of Bodur and Filiz's study (2010)

Smits et al. (2008) examined the psychometric properties of the HSOPSC questionnaire (Sorra and Nieva, 2004) in Dutch hospitals. The aim of this study was to examine the underlying dimensions and psychometric properties of the questionnaire in Dutch hospital settings, and to compare these results with the original questionnaire used in American hospital settings. The questionnaire was completed by 583 staff members. CFA was performed to examine the applicability of the factor structure of the American questionnaire to the Dutch data. EFA was performed to examine whether another composition of items and factors would fit the data better. Supplementary psychometric analyses were performed, including internal consistency and construct validity.

The result of the CFA (principal component analysis with Varimax rotation) showed that the factor structure of the original American questionnaire was a poor fit to Dutch data. Exploratory factor analysis identified 11 factors (factor loadings >0.40). Two items were removed from the questionnaire. Smits et al. (2008) concluded that the Dutch translation of the HSOPSC consists of 11 factors with acceptable reliability and good construct validity. It is similar to the original HSOPSC factor structure (Sorra and Nieva, 2004). The construct validity was studied by calculating scale scores for every factor and subsequently calculating Pearson correlation coefficients between the scale scores. The construct validity of each factor is reflected in scale scores that are moderately related. The construct validity was satisfactory for all factors; the moderate correlations of the factors show that there are no two factors measuring the same construct. For each factor, the internal consistency of the Dutch items was lower than the original HSOPSC items in the AHRO study, except for communication openness, which was the same. The internal consistency of three factors was poor or even unacceptable: organisational learning and continuous improvement ( $\alpha = 0.57$ ), staffing ( $\alpha = 0.49$ ) and teamwork across hospital units ( $\alpha = 0.59$ ). The internal consistency of 11 factors was acceptable (0.64 <  $\alpha$  < 0.79), except for factor 10, adequate staffing was doubtful (0.58). Table 2-6 shows the psychometric properties of Smits et al's study (2008).

Psychometric analysis performed	Psychometric properties
Reliability analysis on the original HSOPSC model	Internal consistency was lower than the original HSOPSC items in the AHRQ study. One factor was the same. Three factors was poor or even unacceptable (0.57-0.49-0.59)
Confirmatory factor analysis on the original HSOPSC model	A poor fit (fit indices not provided).
Exploratory factor analysis to identify optimal model	11 factors were identified
Confirmatory factor analysis on 11 factors	Not performed
Reliability analysis on 11 factors	The internal consistency of 11 factors was acceptable ( $0.64 < \alpha < 0.79$ ), but factor 10, Adequate staffing was doubtful (0.58)
Construct validity	There was correlation between 11 factors. The construct validity was satisfactory for all factors
Optimal model	11 factors and 40 items

# Table 2-9 Psychometric properties of Smits et al's (2008) study

A more recent example is provided in the literature of the development and validation of a new version of patient safety climate questionnaire. Pfeiffer and Manser (2010) developed the German version the HSOPSC. The original HSOPSC version (Sorra and Nieva, 2004) was translated into German. They assessed the dimensionality and psychometric properties of the questionnaire. Two new dimensions (unit management support and unit handoffs and transitions) were added to the questionnaire (six items more and the total items of the HSOPSC became 50 items). The questionnaire was completed by 568 staff members. Confirmatory and exploratory factor analyses (CFA, EFA) and Reliability analysis were applied. This strategy of analysis revealed eight factors instead of twelve factors for the original HSOPSC model. The CFA results indicated that the fit is not sufficiently good to confirm the original factor structure proposed by Sorra and Nieva (2004). The EFA results indicated eight factors) showed satisfactory to good internal consistency. Table 2-7 shows the psychometric properties of Pfeiffer and Manser's study (2010).

Psychometric analysis performed	Psychometric properties (details )
Reliability analysis on the original HSOPSC model	Internal consistency was lower than the original HSOPSC items in the AHRQ study
Confirmatory factor analysis on the original HSOPSC model	The overall fit was not consistently satisfactory: three criteria indicate an adequate fit (RMSEA = .047, PCLOSE = .91,CMIN/df = 2.271). While GFI = .878, NFI = .859, and TLI = .901
Exploratory factor analysis to identify optimal model	8 factors were identified
Confirmatory factor analysis on 8 factors	Not performed
Reliability analysis on 11 factors	The internal consistency of 8 factors was satisfactory to good internal consistency (alpha was ranged from 0.61 to 0.88) Seven out of twelve scales showed a lower internal consistency than was reported for the HSOPSC
Construct validity	There was correlation between 8 factors. The construct validity was satisfactory for all factors (correlation coefficient)
Optimal model	Eight factors

#### Table 2- 10 Psychometric properties of Pfeiffer and Masner's study (2010)

On the whole, the results of validation studies (e.g. Hutchinson et al., 2006; Smits et al., 2008, Waterson et al., 2009; Bodur and Filiz, 2010; Pfeiffer and Manser's study, 2010) are similar in using agreed fit indices and identifying optimal models and they are different in the values for internal consistency (Cronbach's alpha coefficient) and the number of factors in each optimal model.

In summary, it seems that the early stage of development of patient safety climate questionnaires which had been adapted provided little data about their psychometric properties. After that a number of questionnaires were specifically developed for healthcare but they provided limited data about the psychometric properties of the questionnaires. More recent validation studies provide comprehensive data about the psychometric properties of questionnaires. However, it seems that there is still a need for studies to assess the psychometric properties of patient safety climate questionnaires when used in new contexts. Therefore, the current study aims to provide detailed information on psychometric analysis and properties of the questionnaire used in this investigation.

Although, the development and validation of patient safety climate questionnaires has been carried out in a wide range of countries, there was no validated questionnaire for use in Saudi Arabia. There are very few studies of patient safety culture in Saudi Arabia, in particular, there are no validation studies and consequently no valid and reliable questionnaire for measuring patient safety climate. For example, Alahmadi, Talal (2009) assessed patient safety culture in Saudi hospitals by using the original USA HSOPSC questionnaire (Sorra and Nieva, 2004) without examining the reliability and validity of the questionnaire in the Saudi context. Similarly, a study by another author (Alahmadi, Hanan, 2010) used the original USA HSOPSC questionnaire (Sorra and Nieva, 2004) to assess patient safety culture in a number of Saudi hospitals without performing a detailed assessment of the validity and reliability of the questionnaire before its application in the Saudi context.

These two examples of Saudi studies (e.g. Alahmadi, Talal 2009; Alahmadi, Hanan 2010) used the HSOPSC questionnaire to assess patient safety culture without validation of the questionnaire. They did not assess the validity and reliability of the questionnaire before applying it in a new context as recommended by Smits et al. (2008). These recent studies (Alahmadi, Talal, 2009; Alahmadi, Hanan, 2010) in Saudi hospitals were published after the current study began. However, both studies adopted a different approach to assessing patient safety culture than the current study. They aimed to assess patient safety culture in Saudi hospitals by using the original HSOPSC questionnaire without prior validation of the questionnaire. Whereas the current study using Saudi data performed extensive validation in order to identify a suitable measure for assessing patient safety culture in Saudi hospitals. Therefore, the current study aims to address an important gap in the literature.

In general, more recent validation studies since 2004 have reported psychometric properties of validated patient safety climate questionnaires (e.g. Sorra and Nieva, 2004; Hutchinson et al., 2006; Smits et al., 2008, Waterson et al., 2009). This may be because these studies followed similar recommended practice in terms of assessing psychometric properties of patient safety climate questionnaires and publishing information regarding psychometric analysis including factor analysis and reliability analysis.

# 2.4.3 Selecting a suitable questionnaire for measuring patient safety climate

A large and growing body of literature addresses ways of measuring patient safety climate either by developing or selecting an established a suitable tool. It is recommended that using a valid and reliable questionnaire may be better than developing a new questionnaire (Pett et al. (2003). Indeed, Nieva and Sorra (2003:6) recommended that:

"Healthcare organisations should first examine the suitability of existing tools to their needs before embarking on an effort to develop a new tool".

The selection of a suitable patient safety climate questionnaire depends on the purpose and the needs of the study. In addition, the choice of questionnaire depends on the intended use, the target population, reliability and validity of the tool and dimensions. Colla et al. (2005) mentioned that selecting a suitable patient safety climate questionnaire depends on its purpose. Singla et al. (2006) argued that choice of patient safety climate questionnaire questionnaire depends on a number of issues such as the intended use, the target population, psychometric properties including validity and reliability.

A number of studies used selection criteria to find a suitable patient safety culture assessment tool. For example, Hutchinson et al. (2006) used criteria for the selection of an instrument that included: the instrument measured safety climate and was short enough for busy healthcare professionals. Nieva and Sorra (2003) proposed the following criteria: (1) the dimensions of patient safety culture that are evaluated; (2) the staff members who will complete the tool; (3) the settings for which the tool was developed; and (4) the availability of reliability and validity evidence about the tool. Flin et al. (2006b) used criteria such as: use of questionnaire to measure safety climate in healthcare setting, availability of details of the questionnaire, questionnaires tested on sample of over 50 respondents and finally, English

#### Chapter two

language questionnaire. A valid and reliable patient safety climate questionnaire is the most important criteria beside the patient safety culture dimensions that are covered in a questionnaire (Fleming, 2005; Pronovost and Sexton, 2006).

However, it is important to note that the selection of an appropriate patient safety climate questionnaire does not mean that one survey questionnaire is better than others but it depends on the purpose of study, taking into account some very important issues related to psychometric properties and dimensions of the questionnaire.

In conclusion, this study needs a patient safety climate questionnaire that covers a comprehensive range of patient safety culture dimensions, and is a valid and reliable tool to assess patient safety culture in Saudi hospitals. Therefore, there are many questionnaires for specific aims and use that are not suitable for this study such as the Modified Operating Room Management Attitudes Questionnaire (Flin et al., 2003) and the Teamwork and Patient Safety Attitudes Questionnaire (Kaissi et al., 2003). Further details of the selection of an appropriate patient safety climate questionnaire for use in the current study are provided in section 4.2 of chapter four.

#### Summary

There are a number of patient safety climate questionnaires available to assess patient safety culture in healthcare. They are varied in terms of the general characteristics; the dimensions of patient safety culture covered in the questionnaires and psychometric properties. The development of patient safety climate questionnaires is increasingly recognised as an important tool for assessing and improving patient safety culture in healthcare organisations. The selection of the patient safety climate questionnaire depends on the purpose and the needs of the study, reliability and validity of the questionnaire and dimensions that are addressed in a questionnaire.

### 2.5 **Overall summary of chapter two**

The literature review for this study focuses on the key elements of safety in organisations, patient safety, and the assessment of patient safety culture in healthcare. Safety is a concept that has attracted much attention across a wide range of industries, particularly high risk settings such as nuclear energy field, chemical processing, aviation, and healthcare. Organisations with good safety may reflect specific characteristics such as good communication, management commitment to safety and organisational learning from errors. Safety is a particularly important issue in high-risk organisations such as healthcare, where significant hazards are present.

Safety in healthcare involves the safety of both healthcare workers and patients. The Institute of Medicine (2000) defines patient safety as freedom from accidental injury due to medical care or medical errors. It is widely accepted that safety culture is an important element in patient safety improvement. Safety culture refers to the shared attitudes, beliefs, and values of employees of an organisation in relation to safety issues. It is normally assessed by means of workforce questionnaire surveys to measure safety climate. Safety climate can be seen as clear picture, a current surface and measurable aspect of safety culture, which is recognized, from the employee's perceptions (Flin et al., 2000).

It is widely agreed that safety climate questionnaires are used to assess safety culture. The common approach for identifying safety culture dimensions is through analysing the psychometric properties of highly structured questionnaires (Cox and Flin, 1998). For that reason, patient safety climate measurement studies should focus on the psychometric properties of patient safety culture questionnaires that measure patient safety climate. The real test of the safety climate questionnaire is validation, in terms of its power to expose safety climate dimensions (Flin et al., 2000).

Assessing patient safety culture in healthcare is based on measuring patient safety climate through the use of a patient safety climate questionnaire. A number of patient safety climate questionnaires with differing characteristics have been developed to assess patient safety culture in healthcare organisations. These questionnaires are varied in terms of the general characteristics, psychometric properties and the dimensions of patient safety culture that are addressed by the questionnaires.

#### Chapter two

il.

It is also very important to take into account that a patient safety climate questionnaire should be a valid and reliable tool and it should measure the common dimensions of patient safety culture (Nieva and Sorra, 2003).

It seems that few studies have provided a full report of the development of measurement scales and their psychometric properties at an early stage of development of the patient safety climate questionnaires. In addition, there was no agreement about how to assess the validity and reliability of patient safety climate questionnaires. Most studies at that early stage adapted questionnaires from other industries or reviewed a number of questionnaires to develop new questionnaires. However, the development and validation studies are varied in terms of how they develop and validate the patient safety climate questionnaires and how they assess psychometric properties. The validation of the patient safety culture questionnaire construct has been performed by using factor analysis to uncover the latent structures of the safety culture dimensions and Cronbach's alpha for internal consistency as an accepted method for assessing reliability. Importantly, that more recent validation studies have provided a clearer view of psychometric analysis and psychometric properties of patient safety climate questionnaires (e.g. Sorra and Nieva, 2004; Hutchinson et al., 2006; Waterson et al., 2009).

Selecting a suitable patient safety climate questionnaire depends on its intended use, the target population, length of questionnaire, dimensions of patient safety culture, and reliability and validity of questionnaire. However, it is important to review the various questionnaires to select the most appropriate one. The process of selecting an appropriate questionnaire involves establishing methodological criteria which can be employed as guidelines in this process.

The reviewed literature revealed a number of gaps. Firstly, is in relation to a lack of reporting the psychometric properties of patient safety climate questionnaires provided in relevant literature. Although the reviewed literature revealed that in recent years there has been increasing attention to measuring patient safety climate, there remains a lack of reported information on the psychometric properties of patient safety climate questionnaires (Flin et al., 2006b).

Most importantly for this current study is the lack of a suitable tool for assessing patient safety culture in Saudi hospitals. This topic is still at an early stage of development in Saudi Arabia. Although much patient safety research has been published, in particular in patient safety culture assessment, the patient safety climate measurement in non-Western countries such as Saudi Arabia is less advanced. There is no paper that provides a unique measurement of patient safety climate in healthcare organisations in Saudi Arabia (Walston et al., 2010). This means there is a little known about patient safety culture in Saudi hospitals (Alahmadi, 2010). However, it is clear that research on patient safety culture in the Saudi healthcare setting is needed. This in turn, increased the need for appropriate tools to measure patient safety climate in Saudi Arabia. Therefore, the current study aims to identify a suitable measure for assessing patient safety culture in hospitals in Saudi Arabia. In the next chapter, the research methodology used to address the aim of the current study is presented.

# **Chapter Three: Methodology**

# 3.1 Introduction

This chapter outlines the methodological procedures that were followed in addressing the aim of the present study. This study employed qualitative and quantitative approaches to identify a suitable patient safety climate questionnaire for use in hospitals in Saudi Arabia. Details of the research ethics process are presented in this chapter.

# **3.2** Research aim and objectives

This study aims to identify a suitable measure for assessing patient safety culture for use in hospitals in Saudi Arabia. The current study investigates whether there is an existing patient safety climate questionnaire that would be suitable for assessing patient safety culture in hospitals in Saudi Arabia. In this investigation four main objectives need to be met in order to achieve the aim of the study.

The main objectives of the study are:

- 1. To select an appropriate questionnaire to assess hospital patient safety culture.
- 2. To evaluate the face validity of the selected patient safety climate questionnaire.
- To assess the psychometric properties of the selected questionnaire in hospitals in Saudi Arabia.
- 4. To develop the most appropriate measure for assessing patient safety culture for use in hospitals in Saudi Arabia.

## 3.3 Study design

The present study is focused on the validation and development of an appropriate patient safety climate questionnaire for use in Saudi hospitals. The selected patient safety climate questionnaire was identified to use in this investigation. Then the face validity of the selected questionnaire was established. The selected questionnaire was used to collect data to assess its psychometric properties (validity and reliability) and to develop an optimal model for assessing patient safety culture in Saudi hospitals. The settings for the study were three hospitals in Riyadh, Saudi Arabia.

The study comprised four main methods to address the study objectives:

1. Literature review process to select appropriate quantitative tools for assessing patient safety culture.

This stage of research aimed to select appropriate patient safety culture tools for assessing the perceptions of the workforce towards patient safety culture in hospitals for use in the investigation. A comprehensive literature search was conducted in this stage. This stage consisted of four steps and a number of selection criteria at each step. This stage identified two questionnaires, the HSOPSC questionnaire (Sorra and Nieva, 2004) and the Teamwork and Safety Climate questionnaire (Sexton et al., 2004). Further details of the approach taken can be found in chapter four, section 4.2.

2. Qualitative interviews to assess face validity of the selected patient safety climate questionnaires.

This stage aimed to establish the face validity of the HSOPSC questionnaire (Sorra and Nieva, 2004) and the Teamwork and Safety Climate questionnaire (Sexton et al., 2004) and compared them in order to select the more appropriate patient safety climate questionnaire for the study. Semi-structured interviews were conducted with 12 medical and nursing staff to explore the face validity of each questionnaire. At the end of this stage the HSOPSC questionnaire (Sorra and Nieva, 2004) was identified as the most appropriate questionnaire to use in this investigation. The wording of the selected patient safety climate questionnaire (HSOPSC questionnaire) was modified to suit the Saudi settings. This method and its results are provided in detail in chapter four, section 4.3.

i

3. Data collection using the HSOPSC questionnaire in hospitals in Saudi Arabia. In this stage of the study the HSOPSC questionnaire was applied in order to provide the data for psychometric assessment. This stage was concerned with the distribution and collection of the HSOPSC questionnaires. This method is described in detail in chapter five. The data management process is also provided in chapter five.

4. Data analysis to assess the psychometric properties of the HSOPSC questionnaire in hospitals in Saudi Arabia.

The main aim of the analytical strategy is to assess the psychometric properties (validity and reliability) of the HSOPSC questionnaire in hospitals in Saudi Arabia. Validity is defined as the extent to which any instrument measures what it is intended to measure. Reliability concerns the extent to which an experiment, test, or any measuring procedure yields the same results in repeated trials (Carmines and Zeller, 1979). To achieve this aim a number of analytical techniques were used such as factor analysis, correlation and reliability analysis.

Factor analysis is a technique for examining the interrelationships among the items of the HSOPSC questionnaire in order to determine the structure (important underlying patient safety culture dimensions) in the Saudi data. Factor analysis aims to assess construct validity of an established instrument when administered to a specific population (Pett et al., 2003). Correlation refers to the association between two factors (dimensions) using a correlation coefficient ranging from -1 to +1. A positive correlation means both factors increase, while a negative correlation means one factor increases as the other decreases (Campbell et al., 2007). Reliability refers to the ability of the measure to produce similar results across different situations, Cronbach's alpha ( $\alpha$ ) is the most common measure of internal consistency (Field, 2009). Internal consistency of instrument is the most popular way for estimating how well the items of a particular dimension are related to each other (Pett et al., 2003).

In the current study, the analytical strategy of assessing the psychometric properties of the original HSOPSC questionnaire is as follows:

 Confirmatory Factor Analysis (CFA) to test the fit of the original model (original HSOPSC questionnaire) to the whole data. If the original model fits satisfactorily this means the HSOPSC questionnaire is valid. 2. Reliability analysis is carried out to assess internal consistency of each dimension of the original HSOPSC questionnaire.

The second probable result is if the original HSOPSC questionnaire data does not fit satisfactorily, in this situation:

- 1. The data was split randomly into two halves.
- 2. Exploratory Factor Analysis (EFA) was used on one half to derive optimal model.
- 3. Confirmatory Factor Analysis (CFA) was used on the other half of the data to test the fit of the optimal model.
- 4. Correlation of factors of the optimal model was performed to assess construct validity (to ensure the factors of the optimal model are related to each other).
- 5. Reliability analysis was used on the whole data to assess the internal consistency of the optimal model in hospitals in Saudi Arabia by calculating the Cronbach alpha coefficients for the items in each dimension emerging from the factor analysis (EFA and CFA).

There are a number of issues that should be considered in performing CFA and EFA:

Firstly, CFA was used to test the original model fit to the data. Amos software was used to do CFA (Arbuckle, 2005). The guidelines for adequacy of fit in CFA suggests a comparative fit index (CFI) value at least 0.95. The adequacy of the model fit was determined by multiple fit indices. This includes the chi-square test statistic, which is the most commonly used fit statistic. Root-mean-square error of approximation (RMSEA) and incremental fit index (IFI) are also considered. The Normed fit index (NFI) is a rescaled chi-square value (0 = no fit, 1.0 = perfect fit) that is based on a comparison of the proposed model with a model that contains no relations, or a null model (Raykov and Marcoulides, 2000). Several authors (e.g. Bone, et al., 1989; Hair et al., 1998; Joreskog and Sorbom, 1993; Schumacker and Lomax, 2004; Nusair and Hua, 2010) agreed that the overall model fit is evaluated using goodness-of-fit indices including x/df ratio, CFI, NFI, PNFI, RFI, IFI and RMSEA. Table 6-15 in page 174 provides the general agreed fit indices (parameters).

Secondly, Exploratory Factor Analysis (EFA) is an important and useful tool for refining measures and evaluating construct validity. In other words, EFA is used for creating and refining the instrument's scales (Conway and Allen, 2003). In EFA the aim is to discover the main constructs or dimensions (Kline, 1994).

A number of studies (e.g. Ford et al., 1986; Floyd and Widaman, 1995; Gorsuch, 1997; Fabrigar et al., 1999) suggested that there are three important decisions should be taken in consideration before using EFA as follows:

- (a) The factor extraction model used.
- (b) The number of factors retained.
- (c) The method used to rotate factors (assuming more than one factor is retained).

#### Selection of the factor extraction model:

A variety of factor extraction models are available, most can be categorized as either a common factor model or a components model (Gorsuch, 1983). The most popular components model is Principle Component Analysis (PCA). Among common factor models, maximum likelihood and principal axis factoring with estimated communalities are popular. The main difference between the two models is in their purposes. The purpose of common factor models is to understand unobserved (latent) factors that account for relationships among measured variables (items). The goal of PCA is simply to reduce the number of variables by creating linear combinations that retain as much of the original measures' variance as possible (without interpretation in terms of constructs). The results of two models may very closely resemble common factor results in some cases (Conway and Allen, 2003).

In other words, if a study's purpose is to understand the latent structure of a set of variables and interpret results then the use of a common factor model such as principal axis factoring represents the best decision. If a study's purpose is to be purely reduction of variables without interpreting the resulting variables in terms of latent constructs, then use of PCA represents the best decision (Conway and Allen, 2003). Snook and Gorsuch (1989) pointed out that common factor analysis tends to produce more sensible and accurate results than PCA.

Generally, the selection of the factor extraction model is based on the purpose of the study. The current study aims to identify a suitable measure for assessing patient safety culture for use in hospitals in Saudi Arabia. That means exploring and understanding the patient safety culture dimensions (factors) that could be used in Saudi hospitals for assessing patient safety culture. Therefore, the aim here is not just purely reducing the variables but the aim is to understand and interpret the resulting factors in terms of the structure of the patient safety

culture tool. Generally, the purpose of the current study involves latent constructs (important underlying patient safety culture dimensions) in order to determine the structure of patient safety culture tool in the Saudi data. Therefore, Principle Axis Factoring was selected to use as the extraction method of EFA.

#### The number of factors retained:

ļ

The second important decision is the criterion for the number of factors to retain. A number of studies (e.g. Zwick and Velicer, 1986; Fabrigar et al., 1999; Conway and Allen, 2003) clearly show that different criteria often lead to different numbers of factors being retained such as "Eigenvalues greater than one" rule, the Scree test and parallel analysis. Research shows that the Eigenvalues greater than 1 rule is commonly used but tends to produce too many factors. Therefore, in general there is wide agreement that using multiple techniques such as Eigenvalues and Scree plot have become more common that gives the most interpretable solution (Zwick and Velicer, 1986; Gorsuch, 1997; Fabrigar et al., 1999). Eigenvalue is the sum of the squared of all items of each factor of extracted factors. Eigen value represents the amount of variation contributed by the factor. The Scree plot is used to identify a number of extracted factors (Pallent, 2007). Therefore, in order to achieve the most interpretable solution in terms of a number of factors retained the combination of techniques of Eigenvalues greater than 1 rule and Scree plot were used in the current study.

#### The method used to rotate factors (assuming more than one factor is retained):

Rotation of factors usually aims to find a more interpretable solution, given a number of factors greater than one, (Conway and Allen, 2003). Fabrigar et al. (1999) described simple structure of rotation factors in that each factor has a subset of variables (items) with high loadings, and the rest with low loadings, and that each variable has high loadings on only some of the factors and low loadings on the rest. Factor loadings are the correlations of items with factors. The factors deduced by their loadings (Kline, 1994).

There are two type of analytical rotations can be used to get a more interpretable solution. The first type is called orthogonal rotation, such as varimax. The second type is called oblique rotation, such as direct oblimin. Orthogonal rotation decreases the number of factors and uses uncorrelated factors that mean this rotation gives some high loadings and some low loadings for each factor. Oblique rotation is allowing correlated factors and gives one high loading and other loadings near zero for each factor (Conway and Allen, 2003).

#### Chapter three

In this type of rotation the covariance between elements on factors is minimized (Kim and Mueller, 1978). Oblique rotation provides a more interpretable solution than orthogonal rotations therefore; an oblique rotation is preferred (Ford et al., 1986; Gorsuch, 1997; Fabrigar et al., 1999). In this study varimax was used at the initial solution (first step of EFA) to produce the initial idea of factor loadings and to examine the clarity of the initial solution. After that, the oblique rotation was used to force factors of patient safety culture to be correlated. In summary the following criteria are used in EFA:

- 1. Principle Axis Factoring was selected to use as the extraction method of EFA.
- 2. The combination technique of eigenvaluesgreater-than-1 rule and scree plot were used.
- 3. Varimax then oblique rotation (direct oblimin) were used.

The data in this study was analysed according to the above strategy to assess the psychometric properties of the HSOPSC questionnaire in Saudi hospitals. More details of the method and the results are presented in chapter six.

## **3.4** Study setting and sample

The location of the study was in Riyadh, Saudi Arabia. The HSOPSC questionnaire was administered at three hospitals in Riyadh from June to mid of September 2009. The study sample was selected from all doctors and nurses in the three selected hospitals. Details of study settings and sample are provided in chapter five.

## **3.5** Research Ethics Review

Research ethics review was required from the School of Health and Related Research at the University of Sheffield. Thereafter, ethics review and agreements were sought from the three hospitals in Riyadh in Saudi Arabia. The first stage involved ethical approval from the University of Sheffield. The current study received approval from the ethics committee in the School of Heath and Related Research at the University of Sheffield (see copies of ethics approval in appendix 5 and 6).

The second stage of ethics review was the hospitals' approval to conduct the study. Communication was established with the research centres in the three hospitals in Riyadh for ethics application forms and regulations (instructions) for research requirements to obtain their agreement to undertake the study.

- 1. Institutional Review Board Committee of the Research Centre, King Fahad Hospital (submitted in April 2009).
- 2. Office of research affairs in King Faisal Specialist Hospital and Research Centre (submitted in April 2009).
- 3. Institutional Review Board of the King Fahad Medical City (submitted in June 2009).

The proposal of the current study and relevant forms were submitted to the three research centres above in hospitals in Riyadh in Saudi Arabia. The proposal was approved by the Chairman of the Research Committee in the King Fahad Hospital (see appendix 8 for the copy of the approval letter from Institutional Review Board of King Abdullah International Medical & Health Sciences Research Centre in King Fahad Hospital). The proposal of the study was reviewed and approved by two committees, Research Ethics Committee (REC) and Clinical Research Committee (CRC) in the King Faisal Specialist Hospital and Research Centre. After that the proposal was approved by the Research Advisory Council (see appendices 13 and 14 for the copies of the approval letter from the Research Advisory Council, King Faisal Specialist Hospital and Research Centre). The proposal was reviewed by the External Research Review Committee after that the proposal was approved by the Institutional Review Board of the King Fahad Medical City). This research ethics review process is described in detail in appendix 7.

### **3.6** The fieldwork and plan

The fieldwork commenced in early 2009 for a period of six months in Riyadh in Saudi Arabia. Firstly, there were interviews with a number of participants. The approximate duration of the interview process including obtaining agreements of the hospitals was two and a half months. Secondly, the questionnaires were distributed in the three hospitals. The approximate duration of data collection process including distributing and collecting the questionnaires in the three hospitals was three months.

This period of time was particularly convenient because there were no holidays within this period of time that might have interrupted the field work. Table 3-1 shows the timetable of the fieldwork in Saudi Arabia.

Table 5- 1 Fleluwork	umetable		
Stage	Start	End	,
Interviews	10-4-2009	25-6-2009	
Questionnaires	28-6-2009	1-10-2009	

Table 3- 1 Fieldwork timetal	bl	l	e
------------------------------	----	---	---

#### 3.7 **Summary**

In summary, the methodology used for undertaking the research was based on the aim and objectives of the research. In the current study, both quantitative and qualitative methods were used to identify patient safety climate questionnaire that would be a suitable for assessing patient safety culture in hospitals in Saudi Arabia. The next chapter provides the details of choosing an appropriate patient safety climate questionnaire for use in Saudi hospitals. The details of the data collection process can be found in chapter five. The results of the psychometric analysis of the HSOPSC questionnaire properties (Sorra and Nieva, 2004) in Saudi hospitals using Saudi data can be found in chapter six of this thesis.

# Chapter 4: Choosing an appropriate patient safety climate questionnaire for use in Saudi hospitals

# 4.1 Introduction

I

The current chapter aims to describe the process of choosing an appropriate patient safety climate questionnaire for use in the study. This chapter is divided into four sections. Following this introduction, section 4.2 describes the selection of appropriate questionnaires for assessing the perceptions of the workforce towards patient safety culture in hospitals. Section 4.3 describes the face validity testing of the two patient safety climate questionnaires and reports the findings. Section 4.4 provides the overall summary of the chapter.

# 4.2 Selection of appropriate patient safety climate questionnaires

This stage seeks to select appropriate patient safety climate questionnaires for assessing the perceptions of the workforce towards patient safety culture in hospitals for use in the investigation. This stage consists of four steps. Table 4-1 describes the criteria used in each step of the selection of appropriate patient safety climate questionnaires for assessing patient safety culture.

Step	Criteria
Step 1: Identifying existing patient	1. Quantitative instruments
safety climate questionnaires	2. Use for healthcare organisations
	3. Patient safety culture /climate questionnaire
Step 2: Assessment of psychometric	1. A valid and reliable questionnaire
properties and dimensionality of questionnaires	2. A questionnaire that encompasses the important patient safety culture dimensions
Step 3: Assessment of general	1. For assessing perceptions of hospital staff
characteristics of questionnaires	2. Self- reported questionnaires
	<ol> <li>A questionnaire is short enough for use by health professionals who are busy</li> </ol>
	4. Availability of versions of questionnaires
Step 4: Selection of appropriate patient safety climate questionnaires	The selected questionnaires should be fitted with all criteria above

# Table 4-1 Steps of selecting appropriate patient safety climate questionnaires

# 4.2.1 Step1: Identifying existing patient safety culture questionnaires

This step was based on the following criteria:

- 1. Patient safety culture /climate measures.
- 2. Use for healthcare organisations.
- 3. Quantitative instruments.

#### Search strategy

An extensive and comprehensive literature search was conducted utilising hand (library) searching and electronic databases such as the Medline, Science direct, Pubmed Central and the Psycho review of the published literature from 1966 to July 2008, Psycho from 1985 to July 2008. Using the search terms "healthcare organisation", "safety culture", "safety climate", "patient safety culture"," patient safety climate", "guestionnaire" and "measuring". A combination of the key words, such as patient safety climate questionnaire in healthcare were used. No restriction on

date or study type was used. However, the search was restricted to the English language. Websites relevant to patient safety were also used, and these included the website of the Agency for Healthcare Research and Quality (AHRQ), the National Patient Safety Agency (NPSA) and the Institute for Healthcare Improvement (IHI). Moreover, bibliographic references were also searched. The searches were conducted in 2007.

The above strategy and criteria yielded 14 questionnaires that are used to measure patient safety culture /climate in healthcare organisations as follows:

- 1. Hospital Survey on Patient Safety Culture (Sorra and Nieva, 2004).
- 2. Safety Climate Survey (Sexton et al., 2000, 2003).
- 3. Team Work and Safety Climate Survey (Sexton et al., 2004).
- 4. Safety Attitudes Questionnaire (Sexton et al., 2004).
- 5. Safety Climate Scale (Pronovost et al., 2003; Kho et al., 2005).
- 6. Leiden Operating Theatre and Intensive Care Safety scale (Beuzekom, 2007).
- 7. Veterans Administration Patient Safety Culture Questionnaire. (Burr et al., 2002).
- 8. Strategies for Leadership: An Organisational Approach to Patient Safety (Voluntary Hospital of America, 2000).
- 9. Hospital Transfusion Service Safety Culture Survey (Sorra and Nieva, 2003).
- 10. Patient Safety Culture in Health Care Organisation (Singer et al., 2007).
- 11. Medication Safety Self-Assessment (Institute for Safe Medication Practices, 2000).
- 12. Culture of Safety Survey (CSS) (Weingart et al., 2004).
- 13. Stanford University Patient Safety Climate in Health Care Organisation Questionnaire (Singer et al., 2003).
- 14. An employee questionnaire for assessing patient safety (Carayon et al., 2005).

ŀ

# 4.2.2 Step 2: Assessment of psychometric properties (validity and reliability) and dimensionality of patient safety climate questionnaires

This step was related to reviewing the validity and reliability of questionnaires, an important step in selection process (Pronovost and Sexton, 2005). The dimensionality of the questionnaires was also considered. This step addressed the following criteria:

- 1. A valid and reliable patient safety climate questionnaire.
- 2. A questionnaire that encompasses the important patient safety culture dimensions.

These criteria were applied to the list of 14 questionnaires; four questionnaires were excluded. Table 4-2 shows the excluded patient safety climate questionnaires in this step.

<u>ruoto</u> + 2 Excluded patient safety enmate questionnanes in step2		
Patient safety climate questionnaire	Reason for exclusion	
Stanford University Patient Safety Climate in Health Care Organisation (Singer et al., 2003)	Reliability not mentioned (Fleming, 2005)	
Strategies for Leadership: An Organisational Approach to Patient Safety(Voluntary Hospital of America, 2000)	Cronbach's alpha not mentioned (Colla et al., 2005)	
Safety Climate Scale (Pronovost et al., 2003; Kho et al., 2005)	Small number of dimensions	
Culture of Safety Survey (CSS) (Weingart et al., 2004)	Cronbach's alpha (internal consistency scores) were poor	

Table 4-2 Excluded patient safety climate questionnaires in step2

The remaining ten patient safety climate questionnaires as follows:

- 1. Hospital Survey on Patient Safety Culture (Sorra and Nieva, 2004).
- 2. Safety Climate Survey (Sexton et al., 2000, 2003).
- 3. Team Work and Safety Climate Survey (Sexton et al., 2004).
- 4. Safety Attitudes Questionnaire (Sexton et al., 2004).
- 5. Leiden Operating Theatre and Intensive Care Safety scale (Beuzekom, 2007).
- 6. Veterans Administration Patient Safety Culture Questionnaire (Burr et al., 2000).
- 7. Hospital Transfusion Service Safety Culture Survey (Sorra and Nieva, 2003).
- 8. Patient Safety Culture in Health Care Organisation (Singer et al., 2007)

- 9. Medication Safety Self-Assessment (Institute for Safe Medication Practices, 2000)
- 10. An employee questionnaire for assessing patient safety (Carayon et al., 2005)

# 4.2.3 Step 3: Assessment of general characteristics of patient safety climate questionnaires

The third step in the process of selecting appropriate patient safety climate questionnaires for assessing patient safety culture was based on the following criteria:

- 1. An instrument that can be utilised to assess the perceptions of patient safety culture of doctors and nurses in hospitals.
- 2. Self-reporting questionnaires that staff could complete themselves without any assistance.
- 3. Instrument short enough for use health workers who are busy such as doctors and nurses (Hutchinson et al., 2006) as time to complete questionnaire is an important consideration.
- 4. The availability of the questionnaires. Full copies of questionnaires giving descriptions, such as number of questions (items), type of questions, scale used the intended sample population and the setting where it has been used.

These criteria above were applied to the list of 10 questionnaires, five questionnaires were excluded. Table 4-3 shows the excluded patient safety climate questionnaires in this step.

L

Patient safety climate questionnaire	Reason for exclusion
Leiden Operating Theatre and Intensive Care	It contains 99 items. It is used for assessing system
Safety scale (Beuzekom, 2007)	failure in operating room and intensive care
Medication Safety Self-Assessment (Institute for Safe Medication Practices, 2000)	Too long questionnaire contains 194 items
Hospital Transfusion Service Safety Culture Survey(Sorra and Nieva, 2003)	Use for transfusion service. In addition the version of questionnaire is not available
Veterans Administration Patient Safety	Relatively long, it contains 71 items. In addition the
Culture Questionnaire (Burr et al., 2000)	version of questionnaire is not available
An employee questionnaire for assessing patient safety (Carayon et al. 2005	The version questionnaire is not available
(ourujon or un, 2005	

### Table 4- 3 Excluded patient safety climate questionnaires in step 3

The remaining five patient safety climate questionnaires were:

- 1 Hospital Survey on Patient Safety Culture (HSOPSC) (Sorra and Neiva, 2004).
- 2 Safety Climate Survey (SCS) (Sexton et al., 2000, 2003).
- 3 Team Work and Safety Climate Survey (Sexton et al., 2004).
- 4 Safety Attitudes Questionnaire (SAQ) (Sexton et al., 2004).
- 5 Patient Safety Culture Survey in Health Care (PSCSHC) (Singer et al., 2007).

-

### 4.2.4 Step 4: Select appropriate patient safety climate questionnaires

The five patient safety climate questionnaires mentioned above were further reviewed in detail for their general characteristics, dimensions covered and psychometric properties of each questionnaire (Colla et al., 2004; Singla ., 2006).

#### **Hospital Survey on Patient Safety Culture (HSOPSC)**

Sorra and Neiva developed this tool in 2004 funded by the Agency for Health Care Quality and Research (AHRQ). This survey aims to enable hospitals to assess their patient safety culture. It was developed by reviewing the literature and existing safety culture surveys followed by interviews with hospital staff. The resulting questionnaire was piloted and the data analysed using factor analysis to identify which scales and items to retain. It contains 44 items on a 5-point Likert Scale, and it is appropriate for general use by healthcare staff. The HSOPSC questionnaire is considered as a valid and reliable tool when tested in the USA to assess the perceptions of the workforce towards the patient safety culture in hospitals. It covers 12 dimensions of safety culture, which are:

1 Good channels of communication openness.

2 Supervisor/manager actions promoting patient safety.

3 No punitive response to error.

4 Staffing.

5 Hospital management support for patient safety.

6 Teamwork within units.

7 Teamwork across hospital units.

8 Effective processing of organisational learning.

9 Regular feedback and communication about errors.

10 Hospital handover and transitions (information handover, exchange of information).

11 Overall perceptions of safety.

12 Frequency of event reporting.

There are four outcomes, which include overall perceptions of safety, frequency of events reported, number of events reported and overall patient safety grade. This patient safety climate questionnaire shows good psychometric properties (Sorra and Neiva, 2004). Table 4-4 shows the characteristics of the original HSOPSC questionnaire.

Authors and	Sorra and Nieva
development date	(2004)
Country	U.S.A
Aim	To enable hospitals to assess their patient safety culture.
Number of items	44
Scale	On a 5-point Likert Scale
Setting	Hospital
Staff	Health care staff
Dimensions	1. Communication openness
	2. Supervisor/manager actions promoting patient safety
	3. No punitive response to error
	4. Staffing
	5. Hospital management support for patient safety
	6. Teamwork within units
	7. Teamwork across hospital units
	8. Organisational learning processing
	9. Feedback and communication about error
	10. Hospital handover and transitions (information handover, exchange of information)
	11. Overall perceptions of safety
	12. Frequency of event reporting
Psychometric	1 Cood noushamatria propartian
	Crombash's alpha range from 0.63.0.84
assessment	2 Cronbach's alpha range from 0.03-0.84
	3 Tested on large sample
	Statistical analysis such as item analysis, exploratory factor analysis, confirmatory factor analysis and correlated composite scores across dimensions performed to assess psychometric properties. It has a strong content validity and
	has been validated at all levels (Sorra and Nieva, 2004; Colla et al., 2004).

# Table 4- 4 Characteristics of the original Hospital Survey on Patient Safety CultureCharacteristicsHospital Survey on Patient Safety Culture

The version of the questionnaire can be found on www.ahrq.gov/qual/hospculture
As mentioned in section (2.4) in chapter two, the HSOPSC questionnaire is very widely used in different countries, for example, the study in Belgium by Hellings and colleagues (2007) selected the HSOPSC questionnaire for their survey because they found it met more of their psychometric criteria than the other instruments they reviewed (Hellings et al., 2007). Overall, the questionnaire examines patient safety culture from a hospital staff perspective. The HSOPSC questionnaire measures a number of important dimensions and includes a number of outcomes of patient safety culture. It is a reliable and a valid questionnaire in measuring patient safety climate.

#### Safety Climate Survey (SCS)

Sexton and colleagues (2000, 2003) developed the Safety Climate Survey (SCS). It aims to allow an organisation to be able to assess staff perceptions of safety culture and to monitor the success of the patient safety initiatives. It is derived from the Flight Management and Attitudes Safety Survey (FMASS). The Safety Climate Survey contains 19 items covering seven safety culture dimensions, which are:

- 1. Communication.
- 2. Priority given to patient safety.
- 3. Perception and understanding of the cause of patient safety incidents.
- 4. Learning from patient safety incidents.
- 5. Incident reporting.
- 6. Leadership.
- 7. Error management.

It has been reported that this tool has a good test-retest reliability and internal consistency and that it is predictive of performance and accident rates (Sexton et al., 2000; Pronovost et al., 2003). Table 4-5 shows the characteristics of the SCS.

	Safety Chimate Survey
Authors and	Sexton et al.
development date	(2000,2003)
Country	U.S.A
Aim	To allow organisations to be able to assess staff perceptions of safety culture and to monitor the success of the patient safety initiatives
Number of items	19
Scale	On a 5-point Likert Scale
Setting	Acute hospital settings
Staff	Health care staff
Dimensions	1. Communication
	2. Priority given to patient safety
	3. Perception of the cause of patient safety incidents
	4. Learning from patient safety incidents
	5. Incident reporting
	6. Leadership
	7. Error management
Psychometric	Cronbach's alpha is good.
assessment	Partial item analysis, partial confirmatory factor analysis, partial test /retest reliability and analysis of variance across services performed (Colla et al., 2005)

## Table 4- 5 Characteristics of Safety Climate Survey Characteristics Safety Climate Survey

### Teamwork and Safety Climate Survey

Sexton et al. (2004) developed this questionnaire. It contains 27 items covered in two sections teamwork and safety climate. Table 4-6 shows the characteristics of the Teamwork and Safety Climate Survey.

Table 4- 6 Charac	teristics of Teamwork and safety climate survey
Characteristics	Teamwork and safety climate survey
Authors and	Sexton et al
development date	(2004)
Country	U.S.A
Aim	To measure teamwork and safety climate in healthcare organisations
Number of items	27
Scale	6-point Likert scale
Setting	General primary and secondary health care settings
Staff	Hospital and primary care staff
Dimensions	1. Safety climate domain
	Attitudes to safety within own team, capacity to learn from errors, overall confidence in safety organisation and perception of management attitudes
	2.Team work domain
	Communication and collaboration with other staff, information handover (exchange of information)
Psychometric	Psychometric analysis:
assessment	1 Cronbach's alpha range from 0.69-0.84
	2 Face validity is good
	Factor analysis and reliability analysis performed

### Safety Attitudes Questionnaire (SAQ)

The Safety Attitudes Questionnaire (SAQ) is a refinement of the Intensive Care Unit Management Attitude Questionnaire (ICUMAQ), which has been taken from a questionnaire widely used in commercial aviation, the Flight Management Attitudes Questionnaire (Sexton et al., 2004). This questionnaire was developed by Sexton et al. (2004) to measure caregiver attitudes about the following dimensions: teamwork, safety climate, job satisfaction, stress recognition, perception of management and working condition. It contains 60 items and demographic information such as sex, age, nationality and experience. It uses a Likert scale and some items are negatively worded. It is used for multiple units in an organisation. The SAQ was designed to take a snapshot of culture in clinical areas. The SAQ is used for assessing the attitudes of front-line healthcare providers across a number of dimensions of patient safety. This questionnaire has shown adequate psychometric properties and was tested on a large sample. Cronbach's alpha ranged from 0.65-0.83 (Sexton et al., 2006). Table 4-7 shows the characteristics of the SAQ.

Characteristics	Safety Attitude Questionnaire
Authors and	Sexton et al.
development date	(2004)
Country	U.S.A
Aim	To measure the attitudes of front-line healthcare providers about six patient safety-related domains
Number of items	60
Scale	On a 5-point Likert scale
Setting	Multiple units
Staff	Health care staff
Dimensions	1. Teamwork
	2. Safety climate
	3. Job satisfaction
	4. Stress recognition.
	5. Perception of management
	6. Working conditions
Psychometric assessment	Adequate psychometric properties. Tested on large sample Cronbach's alpha range from 0.65 - 0.83. Exploratory factor analysis, confirmatory factor analysis, correlated composite scores across dimensions and test /retest reliability (Colla et al., 2005)

### Table 4- 7 Characteristics of Safety Attitude Questionnaire

Note: Questionnaires (Safety Attitudes Questionnaire, Safety Climate Survey and Team Work and Safety Climate) are available at the University Of Texas Centre Of Excellence for Patient Safety Research and Practice website: http://www.uth.tmc.edu/schools/med/jatientsafety/survey&tools.htm.

### Patient Safety Culture Survey in HealthCare (PSCSHC)

Singer and colleagues developed the Patient Safety Culture Survey (Stanford instrument) in 2003. It was adapted from five questionnaires in aviation and health care to assess safety culture. It contains 82 items and covers 16 topics but reliability statistics have not been published and it is relatively long. It was modified into 38 items, which is now the Patient Safety Culture Survey (modified Stanford instrument) and it is used to measure workforce perception of hospital safety culture. Cronbach's alpha ranged from 0.50-0.89 (Singer et al., 2007). Table 4-8 shows the characteristics of the PSCSHC.

Organisations				
Characteristics	Patient Safety Culture Survey on Health Care Organisations			
Authors and	Singer et al.			
Development date	(2007)			
Country	U.S.A			
Aim	To assess workforce perception of hospital safety culture			
Number of items	38			
Scale	Likert scale			
Setting	General			
Staff	Healthcare staff			
Dimensions	1. Senior manager's engagement			
	2. Organisational resources for safety			
	3. Overall emphasis on safety			
	4. Unit safety norms			
	5. Unit recognition and support for safety effort			
	6. Fear of shame.			
	7. Provision of safety care			
	8. Learning			
	9. Fear of blame			
Psychometric	1. Pilot tested			
assessment	2. Alpha's range from 0.50 - 0.89			
	3. It is a valid and reliable tool that is used to measure features of hospital safety climate			
	4. Exploratory factor analysis performed			

# Table 4- 8 Characteristics of Patient Safety Culture Survey on Health Care Organisations

To sum up the process of selecting appropriate patient safety climate questionnaires, the decision was made to include the HSOPSC questionnaire (Sorra and Nieva, 2004) as an appropriate questionnaire to be used in the present study because of the following reasons:

- 1. The tool is developed for hospitals.
- 2. It is used internationally such as in the USA and Europe.
- 3. It is a valid and reliable instrument when it was tested in USA hospitals.
- 4. It examines patient safety culture from a hospital staff perspective.
- 5. It is new tool, based on the review of many instruments.
- 6. It is a comprehensive tool; that encompasses 12 dimensions of patient safety culture.
- 7. The HSOPSC questionnaire freely available. It has no copyright restriction, this is considered as a good point especially as time is limited in the present study.

The second appropriate questionnaire selected was the Teamwork and Safety Climate Survey because of the following reasons:

- 1. It is a valid and reliable instrument when it was tested in USA hospitals.
- 2. It examines patient safety culture from a hospital staff perspective.
- 3. It is relatively short questionnaire (27 items).
- 4. It is used internationally such as in the USA and Europe.
- 5. The copy of the questionnaire freely available. It has no copyright restriction, this is considered as a good point especially as time is limited in the present study.

Table 4-9 shows the steps of selecting appropriate patient safety climate questionnaires for assessing patient safety culture that were used in this study.

Step		Description	Outcomes
1.	Identifying an existing patient safety climate questionnaire	An extensive and comprehensive literature search	14 questionnaires
2.	Assessment of psychometric properties (validity and reliability) and dimensionality of the questionnaires	Review the psychometric properties and dimensions of the tools	10 questionnaires
3.	Assessment of general characteristics of the questionnaires	Review the general characteristics	Five questionnaires
4.	Select appropriate patient safety climate questionnaires	A more detailed review and compares general characteristics, dimensions covered and the psychometric assessment of each questionnaire	Two questionnaires

 Table 4- 9 Steps of selection of an appropriate questionnaire for assessing patient safety culture

The next section discusses the comparison of the two questionnaires.

### 4.2.5 Comparison of the two questionnaires

The aim of the comparison of the two questionnaires is to select the most appropriate one of them to use in the study. The first suitable patient safety climate questionnaire for this study was the HSOPSC questionnaire (Sorra and Nieva, 2004) and the second suitable questionnaire was the Teamwork and Safety Climate Survey (Sexton et al., 2004). Therefore, in order to confirm that the final selected questionnaire was an appropriate instrument for the study, it was useful to make a comparison between the two questionnaires. In addition this comparison might provide important data for the study based on the answers of the participants.

Theoretically it is important to consider the characteristics of patient safety climate questionnaires to compare and select an appropriate patient safety climate questionnaire. Therefore, the methodology of the current study started with comparing the patient safety climate questionnaires based on their characteristics as an important step to select appropriate patient safety climate questionnaires does not work because it is not feasible to the participants to ask them to compare between more than two questionnaires, this would take a long time and consume their time especially they are not experts in the measurement of patient safety climate to make a decision for which questionnaire is suitable for this study. Moreover, most of the other questionnaires are too long or too short or for specific settings such as pharmacy or operation rooms and not for the whole hospital. The section on selecting an appropriate questionnaire in this chapter has provided more details related to the comparison of the patient safety climate questionnaires therefore there was no point in comparison of the questionnaires in the interviews unless to confirm that the patient safety climate questionnaire that has been selected is the most appropriate questionnaire for the current study.

A number of doctors and nurses from one hospital were selected to ask them to read the two tools and compare them to select the most clearly and understandable patient safety culture questionnaire. The comparison was between the following patient safety culture questionnaires:

- 1. The HSOPSC questionnaire [44 items] (Sorra and Neiva, 2004).
- 2. Teamwork and Safety Climate Survey [27 items] (Sexton et al., 2004).

The next section provides more details of the face validity interviews.

# 4.3 Face validity testing of two patient safety climate questionnaires

### 4.3.1 Introduction

This step was concerned with establishing and comparing the face validity of the HSOPSC questionnaire and the Teamwork and Safety Climate questionnaire in order to select the more appropriate patient safety climate questionnaire for the study. The main aim was to ensure that the selected questionnaire was understandable and easy to complete by the respondents without any difficulties in the language and wording of questionnaire. Fink (1995: 51) points out that" Face validity is concerned with how a measure appears". To achieve this aim a maximum of 12 semi-structured interviews on a one-to-one basis were conducted to investigate perceptions of doctors and nurses towards the HSOPSC questionnaire and the Teamwork and Safety Climate questionnaire.

The cognitive interviewing technique was used in the face validity testing of the survey instrument to learn how the respondents understand the questionnaires and to identify the problems that might have arisen regarding the language and wording of the questionnaire. The cognitive interview is an interviewing technique based on memory retrieval for testing and improving the wording of questions in questionnaires. Therefore, the goal of cognitive interviewing is to make questionnaire items clear and understandable for respondents (Collins, 2003). There are two types of cognitive interview: concurrent and retrospective. Concurrent encourages the respondents to give a verbal account of their thinking. Retrospective involves a respondent answering a draft questionnaire (Drennan, 2003). Therefore the participants in the interviews were asked to give their verbal and written responses of their thinking in answering the HSOPSC questionnaire and the Teamwork and Safety Climate questionnaire.

The interviews aimed to investigate the perceptions of the interviewee, both doctors and nurses, towards both questionnaires. The process lasted from 11/4/2009 to 25/6/2009. It was concerned with conducting the interviews in one selected hospital because the data from one hospital would be sufficient to identify the potential changes related to the wording of the questionnaire, making it unnecessary to conduct the interviews in a second hospital. The interviews were conducted in hospital where the first ethics approval was obtained.

Twelve physician and nurse volunteers (Saudi and non Saudi) were recruited from one targeted hospital to participate in the interviews. The language of the questionnaires used in the interviews was English, for both Saudi and non Saudi physicians and nurses.

### 4.3.2 **Objectives of interviews**

The interviews aimed to:

- 1. To establish the face validity of the HSOPSC questionnaire.
- 2. To establish the face validity of the Teamwork and Safety Climate questionnaire.
- 3. To compare the two questionnaires.
- 4. To select the most appropriate patient safety climate questionnaire for the study.

### 4.3.3 Setting of interviews

The interviews were conducted in the hospital that gave ethical approval and agreement first, because there is no big difference in terms of bed capacity and workforce. Also, the three hospitals are considered as tertiary hospitals and they are accredited by the Joint Commission of American Hospital Organisation (JCAHO), and thus the views of staff in one hospital were expected to be represented of the views of staff in other two hospitals in terms of the clarity and ease of understanding the language of the questionnaires. The first hospital providing an approval letter and agreeing to conduct the study was King Fahad National Guard Hospital in Riyadh Saudi Arabia on 1/5/2009.

### 4.3.4 Sample of interviews

The participants in the interviews were selected based on convenience sampling technique. 12 doctors and nurses volunteers (Saudi and non-Saudi) were recruited from one targeted hospital to participate in the interviews.

### 4.3.5 **Recruitment (interview process)**

The workforce of the three hospitals consists of Saudi and non-Saudi staff. Therefore, the interviews were conducted with Saudi and non-Saudi doctors and nurses.

1 Identification: six physicians and six nurses to be interviewed were approached from the different categories of physicians and nurses (resident, specialist, consultant

Chapter four

physicians, head nurse and nurse), Saudi and non Saudi, in King Fahad National Guard hospital. In addition letters from medical and nursing directors were sent to the physicians and nurses in medical and nursing departments in the hospital indicating that the research was being undertaken and that staff who would like to participate may do so (see appendices 9 and 10 for the copies of letters from Associate Executive Director of Nursing Services and from Chairman of Surgery department in King Fahad National Guard Hospital). These letters were very helpful and motivated the staff to participate in the study.

- 2 Approach: the researcher sent an invitation letter with participant information sheet to 12 doctors and 12 nurses to invite them to participate in the interviews. The participant information sheet explains important points such as the purpose of this study and the confidentiality of data (see appendix 2). A period of 5-7 days was given to read the information sheet and to decide whether or he/she wishes to participate. Invitees who expressed their willingness to participate in the study were recruited until the target number of participants was reached, which was six physicians and six nurses. Each doctor/nurse who decided to participate was asked to sign a consent form and choose a suitable time for conducting the interview. If the invitees refused to take part in the study other participants were identified for recruitment. Any invitee was free to decline to participate at any point.
- 3 Recruitment: the doctors and nurses that agreed to participate; were asked by the researcher to sign a consent form (see appendix 1). Then the researcher organised an interviews schedule. Those who took part in the interviews were not asked to take part in completing the questionnaires.

In brief, the participants received the invitation letter to participate in the interviews and information sheet about the study. Approximately one week later, the researcher contacted the invited doctors and nurses to ensure their willingness to participate and to arrange a convenient time and place for the interviews. Participation in the study involved no harm or discomfort to participants. The participants received no direct benefit from participating in this study, but they were informed that the data they provided would contribute to the knowledge base by investigating the suitability of an appropriate patient safety climate questionnaire for use in hospitals in Saudi Arabia.

### 4.3.6 The structure of the interview

The purpose of the interviews was to make sure that the respondents had no difficulties in the language and wording of the selected patient safety culture questionnaire when administered in the selected hospitals. The language of the interviews was English because the common language of work in Saudi hospitals is English. The content of the interviews helped to identify the difficulties in understanding the language and wording of the HSOPSC questionnaire (see appendix 3) and the Teamwork and Safety Climate questionnaire (see appendix 4). Therefore, it was important to create a structure for the interview, which helped to achieve this. The structure of the interview, from the opening words to the final thanking of the respondents is as follows:

- 1- The first contact between the researcher and the participants involved greetings, welcoming, and thanking the participant in advance, for his/her participation. Then, the researcher told the participant about the purpose of interview, the content of the interview and how the confidentiality of the information will be protected. The interview lasted 50 to 60 minutes. The content of the interview was recorded on the interview record sheet.
- 2- The HSOPSC questionnaire and Teamwork and Safety Climate questionnaire with a cover sheet including the title and the aim of the study were provided to the participant.
- 3- The researcher asked the participant to read and complete the two questionnaires carefully regarding to the clarity of the language, understanding the meaning of the words and understanding of the instructions and items in each section of the questionnaires.
- 4- The following three questions were asked for each item of the two questionnaires:

Q1: Is the wording of the item number (A1) clear for you?

Answer: 1 -YES 2 -NO

This question aims to ensure the item is clear to the participants.

Q2: What does item number (A1) mean to you?

This question aims to ensure that the participants understand the meaning of the items straight away without getting confused by different possible meanings coming in their minds. In other words direct to the point.

Q3: Do you find the item number (A1), easy or difficult to understand? Comments:

This question aims to make sure that the participants understand and answer the items easily, without any difficulty.

- 5- The researcher discussed the comments of the participant regarding the clarity of the language and understanding the meaning of the words, terminology used for each item in the questionnaires.
- 6- An interview record sheet was completed for each interview to save the content of each interview and for analysing the data of the interviews.
- 7- The interview record sheet consists of four parts, firstly participant information such as name, job, nationality, department and hospital. Secondly, Part A: HSOPSC questionnaire. Thirdly, Part B: Teamwork and Safety Climate questionnaire. Finally, General comments.
- 8- In closing the interview, the researcher checked that the questions and comments in the interview record sheet are completed and thanked the respondent for his/her interest and effort.

In summary, 12 semi-structured interviews on a one-to-one basis were employed to investigate perceptions of doctors and nurses towards the HSOPSC questionnaire and the Teamwork and Safety Climate. The interviews started at different times during the hospital working hours and in different places of the hospital according to the participants' convenience. The participants were asked to carefully read and complete the two questionnaires, with focus on the clarity of the language and understanding of the items of the questionnaires. Therefore the participants in the interviews were asked to give verbal and written responses of their thinking in answering the two questionnaires.

Generally, the same procedure was followed for both questionnaires, however, to avoid potential bias associated with the order of presenting the questionnaires, six interviews started with the HSOPSC questionnaire and the other six interviews started with the Teamwork and Safety Climate Survey. As mentioned earlier the answers of these three questions were recorded in the interview record sheet for each interview. These sheets were analysed. The next section provides the interview data analysis.

### 4.3.7 Analysis of interviews data

The main aim of the interviews was to establish the face validity of the selected questionnaires (HSOPSC and Teamwork and Safety Climate Survey) in order to ensure that the final selected questionnaire was understandable and easy for the respondents to complete. with no language and wording difficulties. The language of the two questionnaires was English. The interviews focused on investigating the wording and language of the questionnaires to ascertain which one of them was the most appropriate tool to use in the study in terms of the clarity and ease of understanding. The 12 interviewees provided a number of comments related to the meaning of the words, the clarity of the language, the structure of the sentence and the scale used in the questionnaire and the overall understanding of the language of each questionnaire. Generally, the interviews focused on the respondents' perceptions toward the clarity of the language and understanding the items in the two questionnaires. A number of points and issues were observed during the interviews in relation to the respondents' perceptions of the HSOPSC and Teamwork and Safety Climate Survey. This section reports the interview findings of the 12 semi-structured interviews. It begins with demographic information then the results of the HSOPSC questionnaire, followed by the results of the Teamwork and Safety Climate, after that, comparison of the tools, followed by the decision and changes for a choice of questionnaire. Finally, the summary of the interviews is provided.

### 4.3.8 Characteristics of participants

Table 4-10 shows the characteristics of the participants of the interviews.

Interview	Job title	Department	Gender	Nationality	Duration of
No					interview
					(hours/minutes)
1	Doctor	Paediatrics	Male	Saudi	1:20
2	Nurse	Outpatient clinics	Male	Philippine	1:25
3	Nurse	Cardiology	Female	British	1:13:
4	Nurse	Nursing	Male	Dutch	1:15
5	Nurse	Surgery	Female	Saudi	1:05
6	Nurse	inpatient wards	Female	South Africa	1:00
7	Doctor	Rehabilitation	Male	Saudi	1:00
8	Nurse	Inpatient wards	Male	Philippine	1:00
9	Doctor	Medicine	Male	Saudi	00:55
10	Doctor	Surgery	Male	Saudi	00:53
11	Doctor	Paediatrics	Female	Saudi	00:50
12	Doctor	Medicine	Female	Saudi	00:50

 Table 4- 10 Demographic information of participants (n=12)

Table 4-10 shows that the interviews included nurses and doctors of both genders across a variety of departments and nationalities. The duration of the interviews ranged from 50 minutes to 80 minutes. The next section reports the interview findings of the face validity testing of the HSOPSC questionnaire.

### 4.3.9 Results of the face validity testing of the Hospital Survey on Patient Safety Culture (HSOPSC)

### Instructions and definitions:

The instructions, definitions and the structure of the questionnaire were regarded as considered to be clear and organised, as the following participants commented:

"I think it is fine and well organised and structured" (Interview 1)

"The instructions and definitions are very clear to me" (Interview 5)

"The first page is organised with clear instructions and definitions are provided" (Interview 10).

The majority of the participants suggested using square brackets instead of circles in all the instructions in order to be consistent. (Mark one answer by filling in the brackets). Under section F only one respondent suggested adding "SECTION F": Your hospital (continued) and using the same scale of answering to avoid confusion on page six. However, the instruction under each section of the questionnaire was very clear and easy to understand. The definitions used in the questionnaire were clear and the respondents were familiar with them. Thus it was concluded that, the instructions and definitions used in the questionnaire were clear and there was no need to change them.

#### First question:

The respondents to the study were doctors and nurses working in medical and nursing departments only. Therefore, pharmacy, laboratory and radiology were removed as there was no staff from those departments in the hospital. The first question in the questionnaire was changed (see a copy of the HSOPSC questionnaire in appendix 3).

### The wording of the questionnaire and the clarity of the items:

The total number of items in the HSOPSC questionnaire is 54 including background items. Table 4-11 shows how clear and easily understood items of the HSOPSC were, based on the participants' answers.

Interview	Clarity of terms		Understan	ding of items	Unclear items	
No						
	Number of Yes answers	Number of No answers (%)	Number of easy items	Number of difficult items (%)	-	
1	52	2 (3.7%)	52	2 (3.7%)	A5-A7	
2	54	0	54	0		
3	52	2 (3.7%)	52	2 (3.7%)	A5-B4	
4	53	1 (1.9%)	53	1 (1.9%)	A5	
5	50	4 (7.4%)	50	4 (7.4%)	A5-A7-A14-F3	
6	52	2 (3.7%)	52	2 (3.7%)	A15-F3	
7	47	7 (13.0%)	47	7 (13.0%)	A5-A7-B4-F3-	
					F11-H1-H8	
8	54	0	54	0	-	
9	52	2 (3.7%)	52	2 (3.7%)	A5-A7	
10	50	4 (7.4%)	50	4 (7.4%)	A5-A7-B4-H1	
11	54	0	54	0	-	
12	51	3 (5.6%)	51	3 (5.6%)	A5-A7	
_						

Table 4 11 CI UCODCC ... e ...

Three respondents perceived that the HSOPSC was very clear. Nine respondents agreed that nine items of the HSOPSC were not clear and were difficult to understand. The words identified as difficult to understand are underlined. Table 4-12 shows these items and the number of participants who found them unclear and difficult.

Table 4- 12 Number of participants identifying unclear items	
Items	Number of
	participants
A5: Staff in this unit work longer hours than is best for patient care	8
A7: We use more agency/temporary staff than is best for patient care	6
A14: We work in " <u>crisis mode</u> " trying to do too much, too quickly	1
A15: Patient safety is never sacrificed to get more work done	1
B4: My supervisor/manager <u>overlooks</u> patient safety problems that happen over and over	3
F3: : Things " <u>fall between the cracks</u> " when transferring patients from one unit to another	2
F11: Shift changes are problematic for patients in this hospital.	1
H1: What is your <u>staff position</u> in this hospital? Mark ONE answer that best describes your <u>staff position</u> .	3
H8: In your <u>staff position</u> , do you typically have direct interaction or contact with patients?	2

Table 4-12 shows that the participants agreed that items A5-A7-A14-A15-B4-F3-F11-H1-H8 were unclear and difficult to understand. In particular A5 and A7 appear most problematic.

It was observed that the majority of the items of the HSOPSC were clear and easy to understand except a few items, as the following participants indicated:

"In general it is clear, the language is simple and clear just two phrases (crisis mode and fall between the cracks) so it is important to check those with other staff. The meaning of the items is clear, so it is easy to understand the key word of each item quickly and easily. Most terms used related to patient safety, medical errors and incident reports. Reading one time enough to understand and get the meaning of the items. I think time of completion this is fair and reasonable (around 10 minutes)." (Interview 6)

"Terms of the questionnaire are more related to patient care. A5 and A7 are not clear in good way because of (than is best)." (Interview 1)

"The majority of the items of the HSOPSC are simple and short. Generally, it is clear and all the items are understandable." (Interview 2)

"Generally the HSPOPSC was very clear although, the item number A5 was simple but complex in same time." (Interview 3)

"The language of the questionnaire is clear and easy to understand except A5 is very ambiguous question." (Interview 4)

"Most of items were clear and self explanatory. There are few confusing items that might carry more than one meaning such as A5-A7-B4-F3-F11. Open question at the end is good and it is considered as advantage." (Interview 7)

"It is very clear questionnaire and easy to understand so I do not have comments about all the items." (Interview 8)

"It is written in clear and excellent language except A5 and A7." (Interview 9)

"It is very clear and easy to complete. The language of items is simple except A5 and A7." (Interview 10)

"The language of the questionnaire is excellent and clear item." (Interview 11)

Although, the above examples showed that the majority of the items were clear it was observed that a number of items were not clear enough such as A5-A7-A14-A15-B4-F3-F11, as the following comments indicated:

Item A5: Staff in this unit work longer hours than is best for patient care. The participant commented:

"It seems difficult to understand because (than is best) confuses me, do you mean good or it is not good for patient care." (Interview 5)

"Generally the questionnaire was very clear and the items were easy to understand except item A5 sort of complexity." (Interview 3)

"It is not clear because of (than is)." (Interview 7)

Similarly, other participants stated:

"I do not know the meaning exactly because of (than is) so it is difficult I read it 3 times." (Interview 9)

"It is not clear item because of (than is)." (Interview 10)

Item A7: We use more agency/temporary staff than is best for patient care. Similarly, the participants commented:

"Also this item was not clear enough because of (agency and than is best), these words make the question difficult to understand, here we just use temporary staff I do not know what do you mean by agency." (Interview 5)

"It is not clear and difficult to understand because of (than is) and we do not use agency here" (Interview 7).

"Also than is not clear it is difficult." (Interview 9)

"It is not clear because of (than is)." (Interview 10)

Item A14: We work in "crisis mode" trying to do too much, too quickly. The respondent stated:

"Do you mean doing many things quickly, to be onset it is difficult to understand because of (crisis mode)." (Interview five)

Item A15: Patient safety is never sacrificed to get more work done. The participants stated:

"I think this item is clear in general but I would wonder about the word sacrificed? So maybe it needs change." (Interview 6)

"It is difficult to understand the item because of (sacrificed)" (Interview 12). Item B4: My supervisor/manager overlooks patient safety problems that happen over and over. The participants commented:

"I do not know the meaning of (overlook) here exactly, do you mean ignore." (Interview 10)

"No comments. But it is better to change overlook (ignore)." (Interview 3)

"Overlook carry two meaning, it should be changed." (Interview 7) Item F3: Things "fall between the cracks" when transferring patients from one unit to another. The participants stated:

"It is easy but the (fall between the cracks) I do not know, if is it easy to understand to the staff or not so it should be checked." (Interview 6)

"I do not understand the meaning of fall between the cracks so it was no meaning." (Interview 5)

Item F11: Shift changes are problematic for patients in this hospital. The participant stated:

#### Chapter four

"It needs more clarification because of problematic." (Interview 7)

Item H1: What is your staff position? The participants commented:

"If you mean my current position so it should be changed to what is your current position? This is easy and direct question because your staff position could be meant my colleague position." (Interview 7)

"Your staff position means the staff work with you, it is difficult to understand so I think it should be your current position." (Interview 9)

"Use current position instead of staff position." (Interview 10)

Similarly, item H8: In your staff position, do you typically have direct contact with patients? The participant stated:

"What do you mean by staff position? Is it my position or my colleague position?" (Interview 7)

Thus it was decided that, items number A5-A7-A14-A15-B4-F3-F11-H1-H8 should be reworded to make them clearer and easier to understand.

Some inconsistencies in the wording of the questionnaire were highlighted, for example, item A1 and A4 use the term people while other items use such as A8, A16 and C6 the term staff. Participants agreed that it would be better to use one term instead of two terms, as the following participants stated:

"Use staff instead of people." (Interview 1)

"Staff instead of people in terms of consistency." (Interview 4)

"Use one term for consistency staff instead of people." (Interview 7)

As a result of this, it was decided to change the word people to staff in items A1 and A4.

#### Scale:

An important issue highlighted by the participants was the scale used in the questionnaire. The 5-point Likert scale that reads: (Strongly disagree -Disagree -Neither- Agree- Strongly agree-). The perception of most participants was that the scale of the questionnaire was good, as the following examples indicated:

"It is clear and I can select my answer." (Interview 4)

"The scale used in this questionnaire was good." (Interview 10) In general, it was observed that the most of the respondents had no difficulty linking their answer with this scale. Furthermore, it was observed that most of the respondents read the items of the HSOPSC questionnaire only one time before linking their answers with the scale. This indicates the clarity of the items and the scale was easy to use.

#### Positive and negative items:

It is worth noting that several respondents understood the negative items and could differentiate between the positive and negative items, as the following examples indicated: Item B1: My supervisor/manager says a good word when he/she sees a job done according to established patient safety procedures.

"I think the meaning of this item is positive and good attitude of the supervisors, really it is good to feel satisfy and somebody has recognised good job." (Interview 3)

Item B2: My supervisor/manager seriously considers staff suggestions for improving patient safety.

"This item means take suggestion of staff in positive way." (Interview 3)

On the other hand, in the case of the negative items, for example:

Item B3: Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it means taking shortcuts

"This indicate to negative things because they do not care, just ask me to shortcut to do a job." (Interview three)

Item B4: My supervisor/manager overlooks patient safety problems that happen over and over

"We aware there is a problem but they do not mange it in satisfactory way to avoid happening a problem. It is clear that means they pay no attention to problems that happen. This affects patient safety negatively." (Interview 3)

Item A5: Staff in this unit work longer hours than is best for patient care. The participant commented:

"It is not good for patient safety because it means we work more than regular hours which affect patient safety negatively." (Interview 8)

It was clear that the negative items were recognised by the respondents without any difficulty. Therefore, it was decided that there was no need to direct participants' attention towards negative items by using a mark such as underlining. In general, the participants distinguished between positive and negative items.

### General comments on the Hospital Survey on Patient Safety Culture:

Table 4-13 shows the general comments of the respondents about the HSOPSC questionnaire.

Interview	General comments
No	
1	The questionnaire is clear and easy to understand except items A5 and A7. Use 'staff' instead of 'people'. Use same term either 'mistakes' or 'errors'.
2	The majority of the items are simple and short. Generally it is clear and all the items are understandable.
3	The questionnaire is very clear. Item number A5 is complex in terms of clarity and understanding. Generally, all items are easy to understand.
4	The questionnaire is clear but a few items are ambiguous should be changed. Use what is your position instead of what is your staff position?. The scale of the questionnaire is good.
5	Some items were not clear. The majority of the items were easy to understand.
6	It is clear. Most of the terms used were related to patient safety, medical errors and reporting. Time for completion is fair. It is easy to understand the key word of each item therefore; you can get the idea of each item quickly.
7	Most of the items were clear and self explanatory. There were few confusing items that might carry more than one meaning such as A5, A7, F11, and B4. Use one term for consistency such as staff or people. Use that instead of like in item number A8. Use current position instead of staff position.
8	It is very clear and easy to understand.
9	It is clear and excellent language except A5-A7.
10	It is very clear and easy to complete. The language of items is simple except two (A-A7). Use current position instead of staff position. The scale of the questionnaire is good.
11	The language of the questionnaire is excellent. Easy format. Clear items.
12	In general it is clear. Use one style of instruction in all sections. Clarity of language is good.

Table 4-	· 13 General	comments of	f the res	pondents	about HS	SOPSC
Table 4-	· 13 General	comments o	f the res	pondents	about HS	SOPS

In summary, the language of the HSOPSC questionnaire was found to be simple and understandable. The meaning of the words was found to be clear and direct and the structure of the sentences simple. The items were easy to understand. However, A5-A7-A14-A15-B4-F3-F11-H1-H8 needed rewording and more clarification. The participants found the scale in the questionnaire easy to use. The positive and negative items were clear and the participants were able to distinguish between them. Generally, the HSOPSC is clear and easy to understand and complete. The next section reports the interview findings of the Teamwork and Safety Climate Survey.

### 4.3.10 Results of the face validity testing of the Teamwork and Safety Climate Survey

#### Instructions:

It was observed that the participants found the instructions in the questionnaire were not clear enough, as the following participants commented:

> "Instructions were not clear on the first page because using (with respect to your specific unit or clinical units), I am working in many different units not in specific unit. This cause confused." (Interview 11)

> "I think the instructions of the questionnaire were very short and it needs more clarification but still acceptable." (Interview 12)

### The wording of the questionnaire and the clarity of the items:

The total number of items in the Teamwork and Safety Climate Survey is 35 including background items. Table 4-14 shows how clear and easily understood the items in the Teamwork and Safety Climate Survey were.

Interview	Clarity of ite	Clarity of items Understanding of items		Items	
No.	Number of	Number of	Number of	Number of	-
	Yes	No answers.	easy items	difficult	
	answers.			items.	
1	30	5 (14.3%)	30	5 (14.3%)	5-11-12-24-25
2	31	4 (11.4%)	31	4 (11.4%)	11-18-24-25
3	32	3 (8.6%)	32	3 (8.6%)	1-15-25
4	32	3 (8.6%)	32	3 (8.6%)	1-5-25
5	29	6 (17.1%)	29	6 (17.1%)	5-11-12-18-24-25
6	34	1 (2.9%)	34	1 (2.9%)	11
7	33	2 (5.7%)	33	2 (5.7%)	11-15
8	35	0	35	0	
9	32	3 (8.6%)	32	3 (8.6%)	11-15-24
10	34	1 (2.9%)	34	1 (2.9%)	24
11	35	0	35	0	
12	33	2 (5.7%)	33	2 (5.7%)	1-11-15

Table 4- 14 Clarity and understanding of Teamwork and Safety Climate Survey

The participants agreed that eight items were not clear and were difficult to understand. There was a wide range of views from the respondents as to which items were ambiguous and difficult to understand, which makes the decision to make changes extremely difficult. The number of items that were not clear was eight of 27 items which means large number of the items of Teamwork and Safety Climate were somewhat ambiguous as far as the participants were concerned. Table 4-15 shows these items and how many respondents found them unclear and difficult.

Table 4- 15 Number of participants identifying unclear items				
Item No	Number of			
	participants			
1: Nurse input is well received in this clinical area	3			
5: Disagreement in this clinical area resolved appropriately (i.e. not who is right, but what is best for the patient)	3			
11: Briefings personnel before the start of a shift (i.e. to plan for possible contingencies) is important for patient safety	7			
12: Briefings are common in this clinical area	2			
15: The levels of staffing in this clinical area are sufficient to handle the number of patients	4			
18: Personnel frequently disregard rules or guidelines (e.g. hand wash, treatment protocols/clinical pathways, sterile field, etc) that are established for this clinical area	2			
24: Hospital management does not knowingly compromise the safety of patients	5			
25: This situation is doing more for patient safety now, than it did one year ago	5			

There was agreement that items number 1, 5, 11, 12, 15, 18, 24, and 25 are ambiguous and difficult to understand. It was observed that the participants found some of the items of the questionnaire clear and easy to understand while they found others ambiguous and difficult to understand, as the following participants indicated:

"Generally it is ok but there is some editing will assist future respondents to answer the questions more accurately such as using words with specified meaning rather than using comprehensive words such as input in item 1, levels of staffing in item 15 and institution in item 25, also there are number of long sentences so I read most of items more than once because of these comprehensive words." (Interview 3)

"It is clear and easy to understand but some items need to read more than one time to understand." (Interview 8)

"Some items were not to clear to me and also there are too long items so it is not clear such as item 11-18-24-25." (Interview 2)

"Items 24 it is difficult to understand because (knowingly compromise). The word (institution) in item 25, this word more general should be used healthcare organisation or hospital. The word briefings in items 11 and 12, it was not specified because we use endorsement instead of briefings. Using i.e. in items 11-5 makes these items unclear such as the word (contingencies) in item 11 it was not clear to me." (Interview 1)

Similarly, some items used abbreviation such as (i.e.) were not easy to understand, as the participants stated:

"Items with i.e.in brackets are not easy to understand such as item 5. It is difficult to understand item 11 because (i.e.to plan for possible contingences)" (Interview 5)

"I.e. makes some items not clear, I get confused because of i.e." (Interview 7)

Some participants found words that have wider more general meaning which confusing and other words are ambiguous, as the following examples indicated:

"I think you have to write organisation or hospital instead of institution in item 25. Endorsement instead of briefings because briefings could be carried for many meanings, in general, some words were too comprehensive in meaning such as nurses input in item1, levels of staff in item 15, institution in item 25." (Interview 5)

"What do you mean by levels of staffing in item 15? I think this could be mean number of staff." (Interview 5)

"Knowingly compromise makes item 24 not clear and difficult to understand the meaning of this item so I am unable to answer it." (Interview 5)

"I do not understand item 24 and what the purpose of this item is." (Interview 10)

"It is difficult to understand item 24, I think this item need reworded because of (Knowingly compromise)." (Interview 9)

"Some items were too long such as 5-11-18 this make understanding too difficult. Some items included ambiguous words such as input, contingences, briefings and knowingly compromise." (Interview 5)

"The word contingences were not clear in item 11." (Interview 7)

"The item 11 is clear but the word contingences was not clear and it is difficult to understand its meaning exactly." (Interview 9) "I have to read most of the questions twice to be clear." (Interview 4)

"Some items are little long, the content of the most items is more comprehensive, most of items need reading twice to understand and to connect with suitable answer in the scale. Several items are not clear without i.e. and examples." (Interview 6) Scale:

As mentioned earlier the scale used in the questionnaire was an important issue. The questionnaire used a 6-point Likert scale (Disagree Strongly-Disagree Slightly-Neutral-Agree Slightly-Agree Strongly-Not Applicable). The participants experienced significant difficulty linking their answers to this scale, as the following comments indicated:

"I am not comfortable with that scale, because more chances (six answers), need more concentration and focus, do I have disagreement or do I have agreement, so this scale put you off balance in answering the questions, so I need go back to read the question again which takes long time to read and understand the questions. What is the difference between disagree and slightly disagree, if I am just disagree what can I select, this scale confused me so the chance of just disagree is lost." (Interview 6)

"The scale used is not easy to use and not comfortable and not clear to express your agreement." (Interview 10)

"It is clear and appropriate length but the scale is not good." (Interview 11)

Similarly, another participant noted that the scale was not appropriate:

"The scale is not appropriate use the previous one because I want just agree, which one should I select, in this questionnaire it is difficult to answer many questions because many ambiguous questions in this questionnaire and because of the scale." (Interview 4)

A major concern was related to understanding their preferences between 5-point and 6-point Likert scales. It was very clear that the respondents did not express a preference for the 6-point Likert scale. A reason for this because strongly and slightly to agree prevented the participants from expressing their feeling correctly, hence they spent more reading the items of the Teamwork and Safety Climate Survey and debating over their responses. Also there was no option to simply agree or disagree. Furthermore, because of the addition of a sixth possibility the participant may not have understood under what circumstances not applicable should be used.

It was noted that many items carrying different meanings and could be interpreted in different ways for example:

"What do you mean by institution in items 25." (Interview 4)

"It was quite difficult to understand some items because of the wording of the questionnaire such as the words contingencies in item11 and levels in item 15. It is short questionnaire." (Interview 7)

"It is clear and short questionnaire. Ambiguous words such as contingencies in item 11."(Interview 9)

### General comments of the Teamwork and Safety Climate Survey:

Table 4-16 shows the general comments of the respondents about the Teamwork and Safety Climate Survey.

Interview	General comments	
No		
1	The questionnaire is clear except some items. Some items needed further explanation, hence the use of i.e.in brackets. Some items were not direct to the point.	
2	Generally the questionnaire is clear.	
3	Generally, the questionnaire is ok but some editing would assist future respondents to answer the questions more accurately. Some words were too comprehensive in meaning such as situation (item 25), levels of staffing (item 15), nurses input (item 1).	
4	The scale used is not appropriate so the questions were difficult to answer. Ambiguous items because of difficult word used such as contingencies, knowingly compromise.	
5	Some items were too long, this makes understanding too difficult. Some items included ambiguous words such as levels of staff, contingences and briefing. Items with i.e. in brackets are not easy to understand.	
6	You need to think more before answering the questions. The items themselves were more comprehensive. The two items use i.e.in bracket that means is not easy to understand then need more concentration and focus and therefore, reading many times. The scale used is less user friendly as it is likely to make respondent confused.	
7	Generally, it was clear except some items. It is short questionnaire.	
8	It is clear questionnaire but some items need to be read many times to be understood.	
9	It is clear and short questionnaire.	
10	It is clear to some extent. Scale used is not comfortable and not clear to express your agreement. Current position instead of staff position.	
11	It is very clear and easy to understanding. Appropriate length. Instructions were not clear on the first page. Scale was different (duration) in item 31 and 32.	
12	It is clear. Duration was different in items 31 and 32 need for consistency.	

## Table 4- 16 General comments of the respondents about Teamwork and Safety Climate Survey

In summary, it was clear that the participants found the language in the Teamwork and Safety Climate Survey more complex at times than the HSOPSC, such as in items 1-5-11-12-15-18-24-25, in which non-specific words, for example "institution", ambiguous words, for example "contingencies"; and words with different meanings, for example, "input" were used. Therefore, the wording of the questionnaire needs clarification to be clear and easy for participants to understand. Participants found the scale used in the questionnaire difficult to apply. All these points are considered as disadvantage of this questionnaire especially in multi-cultural settings such as Saudi hospitals that contains staff from Saudi and other different nationalities. Therefore, simplifying the language of a questionnaire is an important consideration.

# 4.3.11 Results of the comparison of the two patient safety climate questionnaires

In this comparison, it is not just the number of clear and easy to understand answers that will be taken into consideration but also the degree of clarity and difficulty of items based on the participants' comments above. Table 4-17 show the comparison of the questionnaires.

Patient safety climate	Clarity of items			
questionnaires	Number of clear	Number of not	Total of the items	
	and easy to	clear and difficult		
	understand items.	to understand items		
The first interview		······································		
HSOPSC	52	2 [A5-A7]	54	
Teamwork and Safety Climate	30	5 [5-11-12-24-25]	35	
Survey				
The second interview			<u></u>	
HSOPSC	54	0	54	
Teamwork and Safety Climate	31	4 [11-18-24-25]	35	
Survey				
The third interview				
HSOPSC	52	2 [A5-B4]	54	
Teamwork and Safety Climate	32	3 [1-15-25]	35	
Survey				
The fourth interview		<u></u>	<u> </u>	
HSOPSC	53	1.[A5]	54	
Teamwork and Safety Climate	32	3 [1-5-25]	35	
Survey				
The fifth interview	· · · · · · · · · · · · · · · · · · ·			
HSOPSC	50	4 [A5-A7-A14-F3]	54	
Teamwork and Safety Climate	29	6 [5-11-12-18-24-	35	
Survey		25]		
The sixth interview			<u> </u>	
HSOPSC	52	2 [A15-F3]	54	
Teamwork and Safety Climate	34	1 [11]	35	
Survey				
The seventh interview	······			
HSOPSC	48	6 [A5-A7-B4-F11-	54	
		H1-H8]		
Teamwork and Safety Climate	33	2 [11-15]	35	
Survey				

.

### Table 4- 17 Comparison of the questionnaires (12 interviews)
The eighth interview			
HSOPSC	54	0	54
Teamwork and Safety Climate	35	0	35
Survey			
The ninth interview			
HSOPSC	52	2 [A5-A7]	54
Teamwork and Safety Climate	32	3 [11-15-24]	35
Survey			
The tenth interview	· · · · · · · · · · · · · · · · · · ·		<u>.</u>
HSOPSC	50	4 [A5-A7-B4-H1]	54
Teamwork and Safety Climate	34	1 [24]	35
Survey			
The eleventh interview			
HSOPSC	54	0	54
Teamwork and Safety Climate	35	0	35
Survey			
The twelfth interview			
HSOPSC	52	2 [A5-A7-]	54
Teamwork and Safety Climate	32	3 [1-11-15]	35
Survey			

Table 4-18 shows the comparison of the questionnaires in terms of clarity of items based on number of no answers (unclear and difficult to understand).

Interview	v Number of NO answers(unclear and difficult to		Result (based on
No	understand)	inderstand)	
	IISOPSC	Teamwork and Safety	lowest number of NO
		Climate	answers).
1	2	5	HSOPSC
2	0	4	HSOPSC
3	2	3	HSOPSC
4	1	3	HSOPSC
5	4	6	HSOPSC
6	2	1	Teamwork and Safety Climate
7	6	2	Teamwork and Safety Climate
8	0	0	Both
9	2	3	HSOPSC
10	4	1	Teamwork and Safety Climate
11	0	0	Both.
12	2	3	HSOPSC

Table 4- 18 Comparison of the questionnaires in terms of clarity (n=
--

Seven out of twelve respondents agreed that the HSOPSC questionnaire was clearer than the Teamwork and Safety Climate questionnaire, while only three respondents agreed that the Teamwork and Safety Climate was clearer. Two respondents agreed that the both tools were clear. It was observed that the majority of the respondents read and completed the HSOPSC questionnaire easily and smoothly whilst the majority of the participants had to read Teamwork and Safety Climate questionnaire many times before completing the items. Table 4-19 summarises the advantages and disadvantages of the HSOPSC questionnaire and the Teamwork and Safety Climate questionnaire in terms of the clarity of language.

Questionnaire	Advantages	Disadvantages
HSOPSC	1. Clear instructions	1. Language of some items was
	2. Good scale of answer	difficult to understand such as items
	3. Clear language	number A5, A7.
	4. Short items	2. It used different terms such as
	5. Straight forward items	people, staff, mistakes and errors.
	6. Easy to understand	3. Use short forms like don't.
	7. Items consistency	All these could be solved without changing
	8. It has open question	the meaning.
	9. Easy format.	
Teamwork and	1. Language clear to some	1. Less friendly user scale.
Safety Climate	extent	2. Use of comprehensive words.
	2. Short questionnaire	3. Use of ambiguous words.
		4. Long sentences
		5. Need more focus and concentration.
		6. Has to be read many times to be
		clear and understood
		7. Complex structure of sentences (not
		direct)

## Table 4-19 Advantages and disadvantages of questionnaires

# 4.3.12 Decision and changes (making a choice of questionnaire and changes to questionnaire's language and structure)

Two main decisions needed to be made regarding which questionnaire should be selected to use in the study and which items should be changed in the selected questionnaire. The decision was made to use the HSOPSC questionnaire as the most appropriate questionnaire for the study because its language is clear and easy to understand, straightforward items and the scale is easy to use.

In terms of changes to be made, it became apparent that were three possibilities in relation to the items of the HSOPSC questionnaire, could remain unchanged, be revised or be omitted. It is important to mention that the proposed changes were based on the comments and suggestions of participants. These changes were checked with supervisors of the research. Moreover, the changes were checked with one of the authors of the original HSOPSC questionnaire to confirm that the changes would not alter the meaning of the items. The decision was based on the clarity of the items to all participants. Table 4-20 shows the changes made to the number of items of the HSOPSC questionnaire.

Item	Before change	After change
number		
Al	People support one another in this unit	Staff support one another in this unit
A4	In this unit, <u>people</u> treat each other with respect	In this unit, staff treat each other with respect
A5	Staff in this unit work longer hours <u>than is</u> best for patient care	Staff in this unit work longer hours than they should which is not good for patient care.
A7	We use more <u>agency</u> /temporary staff <u>than</u> is best for patient care	We use more temporary staff than we should which is not good for patient care
A14	We work in " <u>crisis mode</u> " trying to do too much, too quickly	We work in a hurry, trying to do too much, too quickly
A15	Patient safety is <u>never sacrificed</u> to get more work done	Patient safety never takes second place to get more work done
F3	Things <u>fall between cracks</u> when transferring patient from one unit to another	Some things do not happen or get missed when transferring patients from one unit to another
B4	My supervisor/manger <u>overlooks</u> patient safety problems that happen over and over	My supervisor/manger ignores patient safety problems that happen over and over
F11	Shift changes are <u>problematic</u> for patients in this hospital.	Shift changes cause problems for patients in this hospital
HI	What is your <u>staff position</u> in this hospital? Mark ONE answer that best describes your staff position	What is your current position in this hospital? Mark ONE answer that best describes your staff position
H8	In your <u>staff position</u> , do you typically have direct interaction or contact with patients?	In your current position, do you typically have direct interaction or contact with patients?

Table 4- 20 Changes made on the HSOPSC questionnaire

In conclusion, the HSOPSC questionnaire has been selected for use in the study. This questionnaire includes 12 dimensions of patient safety culture divided into nine sections and including a background information section. The questionnaire consists of five pages in addition to the covering letter. There were many changes in section H (background information) which included as follows: three questions (question number two, three and four) about gender (Male or Female), nationality (Saudi or non Saudi) and educational levels of respondents were added to this section. These questions are important to describe the characteristics of respondents in the study. In question number one, the other staff positions were omitted, except doctor and nurse positions, because the target population of the study was doctors and nurses. The changes were made to the HSOPSC items that were mentioned in table 4-20 (see appendix 22 for a copy of the Hospital Survey on Patient Safety Culture questionnaire). The workforce of the three hospitals is English. Therefore, the language of the HSOPSC questionnaire is English. Table 4-21 provides the patient safety culture dimensions of the HSOPSC questionnaire.

Patient safety culture	Definition	Number
dimension [items]		of items
Communication openness [C2-C4-C6]	Staff will freely speak up if they see something that may negatively affect patient care, and feel free to question those with more authority	3
Feedback and communication about error[C1-C3-C5]	Staff are informed about errors that happen, given feedback about changes put into place based on event reports, and discuss ways to prevent errors	3
Frequency of events reported [D1-D2-D3]	<ul><li>Mistakes of the following types are reported:</li><li>1) mistakes caught and corrected before affecting the nationt.</li></ul>	3
	<ol> <li>2) mistakes with no potential to harm the patient,</li> <li>3) and 3) mistakes that could harm the patient, but do not</li> </ol>	
Handoffs & transitions [F3-F5-F7-F11]	Important patient care information is transferred across hospital units and during shift changes	4
Management support for patient safety [F1- F8-F9]	Hospital management provides a work climate that promotes patient safety and shows that patient safety is a top priority	3
No punitive response to error [A8-A12-A16]	Staff feel that their mistakes are not held against them, and mistakes are not kept in their personnel file	3
Organizational learning- Continuous improvement [A6-A9- A13]	Mistakes have led to positive changes and changes are evaluated for their effectiveness	3.
Overall perceptions of patient safety [A15- A18-A10-A17]	Procedures and systems are good at preventing errors and there is a lack of patient safety problems	4
Staffing [A2-A5-A7- A14]	There are enough staff to handle the workload and work hours are appropriate to provide the best care for patients	4
Supervisor/manager expectations and actions promoting safety [B1-B2-B3-B4]	Supervisors/managers consider staff suggestions for improving patient safety, praise staff for following patient safety procedures, and do not overlook patient safety problems	4
Teamwork across units [F2-F4-F6-F10]	Hospital units cooperate and coordinate with one another to provide the best care for patients	4
Teamwork within units [A1-A3-A4-A11]	Staff support one another, treat each other with respect, and work together as a team	4

## Table 4- 21 Patient safety culture dimensions of the HSOPSC questionnaire

#### Summary

The participants understanding of the patient safety climate questionnaires were varied because of the way that they were written. Some participants understood the items in a certain way while others did not understand them at all. Therefore, an initial assessment is very important before full application of any patient safety culture tool is used in different contexts because of language and cultural variations.

The interviews revealed variation of English language skills amongst participants. Therefore, it is very important to ensure using clear language, simple and clear words, simple structure of the sentences and simple and clear scale. The simple and clear items could be answered by all participants regardless of their level of English. Therefore, changes should be made to make all the items easy for all participants to understand.

In terms of the language of the two questionnaires, the participants perceived that the language and wording of the HSOPSC questionnaire was clearer than the Teamwork and Safety Climate questionnaire. The respondents indicated that they understood the HSOPSC questionnaire better than the Teamwork and Safety Climate Survey because the language of the HSOPSC is simpler and the sentences are shorter and more straightforward than those of the Teamwork and Safety Climate Survey. The HSOPSC questionnaire is the most appropriate patient safety climate questionnaire to use for the study.

## 4.4 **Overall summary**

The first stage of the process of choosing an appropriate patient safety culture questionnaire for use in the study identified two questionnaires (the HSOPSC and the Safety Climate Survey). The next stage entailed an assessment of the face validity of the two questionnaires. The results of the comparison between these two questionnaires showed that the HSOPSC questionnaire (Sorra and Nieva, 2004) was the most suitable patient safety climate questionnaire for the current study. In the next chapter, the process of data collection by using the HSOPSC questionnaire will be described.

## Chapter Five: Data collection: Using the Hospital Survey on Patient Safety Culture questionnaire

## 5.1 Introduction

In this part of study the amended HSOPSC questionnaire was used to obtain the data for further validity assessment. The current chapter aims to describe the process of data collection using the HSOPSC to collect the questionnaire data from three hospitals in Saudi Arabia. This chapter is divided into seven sections. In section 5.2 describes the study settings. Section 5.3 concerns the study population. Section 5.4 addresses the sample size of the study. Section 5.5 describes the distribution and collection of the HSOPSC questionnaires. Section 5.6 describes the data management process and section 5.7 provides the summary of the chapter.

## 5.2 Study setting

The location of the study was in Riyadh, Saudi Arabia. The HSOPSC questionnaire was administered at three hospitals in Riyadh from June to October 2009. Three separate hospitals in Riyadh city, which has the largest number of hospitals in Saudi Arabia, have been selected for the purpose of this study:

- 1. King Fahad National Guard Hospital (KFNGH) at King Abdul-Aziz Medical City
- 2. King Fahad Medical City (KFMC)
- 3. King Faisal Specialist Hospital and Research Centre (KFSH&RC)

These hospitals represent different types of the government providers in the Saudi health care system. These hospitals have been chosen for this study because they are the largest and most modern hospitals. Bed capacity is large with a large number of staff especially physicians and nurses, therefore, the three hospitals would provide sufficient potential respondents for the present study.

#### Chapter five

In addition, those hospitals have expertise and experiences in communicating with international healthcare organisations. Moreover, the three hospitals are accredited by the Joint Commission on Accreditation of Healthcare Organisations (JCAHO).

## 5.3 **Population of study**

The population of the study consists of physicians and nurses from the three hospitals. The staff in these hospitals are Saudi and non Saudi from a wide range of different nationalities. The common language is English. The study sample was selected from all doctors and nurses in the three selected hospitals.

## 5.4 Sample size

The HSOPSC questionnaires were distributed to a sample of the doctors and nurses (Saudi and non Saudi) in the three hospitals. The ideal sample would be derived from random sampling, as this technique is relatively simple to apply and to represent the target population of the study but this technique takes a long time to obtain the required number of questionnaires. Therefore, the alternative sample entailed convenience sampling of physicians and nurses at all levels within the three hospitals to achieve the target number of completed questionnaires. This technique is a practical way to increase the return rate in a short time because it depends on readily available physicians and nurses in the three selected hospitals (Fink, 2003). There are several factors to take into consideration to estimate sample size of the study in order to determine the number of questionnaires should be distributed. The first factor is the minimum number of completed questionnaires required to obtain enough participants to perform the strategic plan of data analysis.

In general a major concern was to obtain enough participants to perform factor analysis. Therefore, the required number of questionnaires is a minimum of 300 completed questionnaires to satisfactorily undertake factor analysis. It has been argued that a minimum sample size is at least 300 for factor analysis (Guadagnoli and Velicer, 1988). Secondly, anticipated return rate, the expected return rate of this study was 20%. Thirdly, the cost (printing cost of questionnaires) and available resources were taken into account.

The workforce of doctors and nurses in the three selected hospitals is approximately 4630 (Statistics Directorate, 2009) with a target response rate is a minimum 300 completed questionnaires and the expected return rate was 20%. Also, according to Sorra and Nieva (2004) the sample size should be at least double the number of responses that are required. The number of questionnaires distributed was 1700 in the three hospitals.

### 5.5 Data collection

The data collection process was concerned with the distribution and collection of the HSOPSC questionnaires from three hospitals in Riyadh in Saudi Arabia. The data collection process aimed to collect at least 300 completed questionnaires (target data) for analysis. The study utilized primary data that was collected from doctors and nurses in the three hospitals in order to provide the data for assessment. The data collection was done during daily working hours in the three hospitals to avoid any physical or emotional harm to the patients and the staff.

After getting the final approval and permission from the University of Sheffield (see the copy in appendices 5 and 6), 1700 HSOPSC questionnaires and cover letter as one document were printed in Saudi Arabia. The appearance of the questionnaires contributes to stimulating the respondents to complete them. Therefore, the questionnaires were printed on good quality paper with quality envelopes to motivate the staff to reply. Envelopes were used for maintaining respondents' confidentiality as an important factor for increasing response rate (Bourque and Fielder, 1995).

The principal aim of the distribution process was to ensure getting the target number of questionnaires returned. The data collection process was based on distributing as many questionnaires as possible to maximize the return rate in the limited time available in this very important stage of the study. The process was supervised and followed up carefully from the beginning to ensure the flow of distribution and collection of questionnaires was smooth and flexible without any difficulties. The proposed strategy was based on posters and signs posted in the nurses' stations and doctors' meeting rooms throughout the hospitals to inform them of the survey. The questionnaires were distributed by the researcher in wards and other specialist areas such as out-patient departments during staff breaks. The questionnaires were returned to the boxes located within the hospital or to a designated hospital contact person (primary contact person through the internal mail system in each hospital).

In fact, this method was not effective because it did not work in terms of practicality and nobody had time to track the questionnaires. Moreover, with this strategy it was taking too long to get the data because during the first two weeks just 10 doctors and 25 nurses returned completed questionnaires. Consequently, at that point in time it was considered that there was a need to find a more effective method. Therefore, the questionnaires were distributed in the three hospitals by establishing a data collection plan for each hospital to capture the target data by using different range of methods based on the current situation of each hospital such as the response of hospital management towards the study, the number of staff available, workload, awareness of patient safety and time limitations such as holidays. All these issues were taken into consideration in devising the best data collection plan for each hospital.

## 5.5.1 King Fahad National Guard hospital at King Abdul-Aziz Medical City

The first hospital to give the ethics approval letter was King Fahad National Guard Hospital (see copy in appendix 8). Once the approval letter was obtained from the hospital's research committee, the planning of distributing the questionnaires was started immediately in order to save time. The new proposed strategy commenced with an introductory week in the hospital in order to gain good support for the data collection and to inform the staff about the research. The introductory week included the meeting with the chairman of research centre of King Abdullah International Medical Research Centre (KARC). The advice was given to submit the proposal and the ethical approval and discuss it with hospital management and medical and nursing directors to seek their help and support for the study by encouraging the staff to complete the questionnaires.

A number of appointments with several medical directors in the hospital such as the director of surgery, medicine and paediatric department were made. In these meetings they read the proposal and ethical approval. The protocol of the study was presented and the outlines of the study were discussed. All of them were interested and very keen to help and support this research because they believe the research was important and useful for the patients and staff and for healthcare as well. Generally, they were interested in this project because they are concerned about patient safety in their departments and wish to improve it. In fact, these meetings were very helpful and they provided generous offers to distribute and collect the questionnaires through sending memos and they asked their secretaries to distribute the questionnaires in the doctors' mail boxes and the head nurses were requested to distribute and collect the questionnaires from their nursing staff during daily shift.

This strong support demonstrates the interest of the hospital management, the medical directors and the nursing directors in looking after patient safety as a very important issue in healthcare. Patient safety is clearly considered to be top priority. Further, this support is recognised as the most important factor for maximizing the return rate of doctors' and nurses' contributions (see appendices 9 and 10 for the copies of their support letters). Consequently, all of them sent internal memos to the staff to ask them to contribute to the study as much as they can. A number of anonymous questionnaires were distributed through medical secretaries in each medical department based on the number of doctors available at that time.

137

Two weeks were given for doctors to get their responses. The first reminder was issued after two weeks and the final reminder was issued after another two weeks because they are very busy staff and also owing to holidays. It is important to mention that three important strategies were used to improve data collection process.

There was a meeting with the Associate Executive Director of the nursing department related to approaching the nursing staff in the hospital. In this meeting the proposal and ethical approval were provided and the protocol of the study was presented and discussed. The researcher accepted the generous suggestion of the Executive Director of the nursing department of the hospital to distribute the questionnaires to the rest of the departments in the hospital through the head nurses. This support from the director of nursing was helpful in increasing the return rate in this period of time (summer holidays) because during the summer there is a shortage of staff because of holidays. In addition the nurses are very busy in their clinical areas which hinder the communication between the researcher and the nursing staff. The follow up process of distributing and collecting the questionnaires through administrative assistant of nursing services started, approximately one week after the first meeting with director to get the responses of nursing staff in patient areas and clinical areas such as wards and clinics of the hospital (see appendix 9 for the copy of the letter of the Associate Executive Director of the nursing department). Ten days was given for nurses to get their responses. The first reminder was issued after the ten days and the final reminder was issued after two weeks because they are very busy staff.

To ensure the smooth flow of the data collection regular visits were made to doctors' meeting rooms; head nurses offices and medical secretaries in order to remind them about the survey. The main purpose of the follow up process was to emphasise the importance of their response and purpose of the study. A tracking record was used for distributing the questionnaires and handling the returned questionnaires from the medical secretaries and head nurse. The follow up process was continued every day and this tracking also is recognised as an important factor for maximizing the return rate of nurses' contribution. This process lasted more than one month (45 days) to complete. Table 5-1 shows the number of these questionnaires distributed to the medical and nursing staff of the hospital.

Table 5-1 data conection of KrivGri				
Staff	Questionnaires distributed	Questionnaires returned	Return rate	
Medical	250	107	42.8%	
Nursing	650	440	67%	
First reminder	Two weeks for doctors	Ten days for nurses		
Final reminder	Two weeks for doctors	Two weeks		
Duration of time	45 days			
Total	900	547	60%	

1 Jaka - H. Alen - CIZENCH

King Faisal Specialist Hospital and Research Centre (KFSH&RC) was intended to be the second hospital. However, there was a delay in getting the ethical approval because of the establishment of a new research committee there. Therefore, the second hospital was King Fahad Medical City (KFMC) instead of KFHRC.

#### 5.5.2 King Fahad Medical City

The proposal and relevant forms were submitted to the External Research Review Committee. a subcommittee of Institutional Review Board for consideration and approval. The proposal was reviewed by the External Research Review Committee after that the principal investigator (the researcher) was informed that the proposal has approved by the Institutional Review Board of King Fahad Medical City (see the copy in appendix 11). Once the hospital's permission was obtained the introductory week was started immediately. The introductory week included meetings at management level of the hospital in order to gain support in the data collection. First meeting was done with Executive Director of Medical Affairs and Medical Consultant Council of the hospital. The protocol of the study and the ethical approval were presented and the main outlines of the study were discussed. As a result of this meeting internal memo was sent to all medical directors to encourage the doctors to participate in completing the attached questionnaire (see the copy of this letter in appendix 12), which reflects the keenness of the hospital administration to support patient safety and their conviction in this research.

139

In addition, meetings were conducted with medical directors such as paediatrics and medicine department to seek their support and make this study successful. The overall outcome of these meetings was positive and effective in terms of cooperation and participation in the study.

A further meeting was conducted with the nursing director of the hospital. The protocol of the study and the ethical approval were presented and the main outlines of the study were discussed. There was agreement for help and support to encourage nursing staff to participate in this study. The advice was given that to present the study would be useful for seeking nursing cooperation for conducting this study. Therefore, the main outlines of the study were presented for the target audiences head nurses in the King Fahad Medical City. At workforce level a number of meetings were done with a number of doctors and nurses in the hospital in order to inform the medical and nursing staff of the survey. In each meeting the proposal and ethical approval were provided.

After that, 400 questionnaires were distributed for both doctors and nurses in the hospital based on the discussion with the medical directors and head nurses that this number was reasonable as first distributing to their available staff at that time because of holidays. Two weeks were given for doctors to get their responses. The first reminder was issued after two weeks and the final reminder was issued after another two weeks because they are very busy staff and also owing to holidays. Ten days was given for nurses to get their responses. The first reminder after two weeks because they are very busy staff. Table 5-2 shows the number of these questionnaires distributed to the medical and nursing staff of the hospital.

Staff	Questionnaires distributed	Questionnaires returned	Return rate
Medical	150	46	30%
Nursing	250	164	65.6%
First reminder	Two weeks for doctors	Ten days for nurses	
Final reminder	Two weeks for doctors	Two weeks	
Total	400	210	52.5%

 Table 5- 2 Data collection of KFMC

At this point in time (three weeks before the start of the holiday there), the proposal of the study had been approved by the new research committee of KFHRC. There were two options either to stay in the second hospital or work in parallel and start the data collection in the third hospital. The decision was made to include the third hospital in the study. Therefore, the data collection was started in the third hospital before the holiday. In this strategy (decision) time was saved to start in the third hospital before the holiday. Moreover, it was aiming to increase the probability of getting large number of respondents from the two hospitals rather than one hospital especially to increase the return rate of doctors responses because they are usually very difficult to recruit them to this type of questionnaire survey. Lastly, that it was worthwhile going to the third hospital (KFSH&RC) because three hospitals improve and enhance the value of the results of the study. According to this strategy 800 questionnaires were distributed in the second (KFMC) and the third hospital, 400 questionnaires for each hospital.

#### 5.5.3 King Faisal Specialist Hospital and Research Centre

The King Faisal Specialist Hospital and Research Centre (KFSH&RC) was the third hospital of this study. According to the KFSH&RC guidelines and instructions for submission of research proposals, the researcher must have successfully completed the National Institute of Health (NIH) Web based training course "Protecting Human Research Participants" which was done on 22/3/2009. Then the principal investigator was identified. Subsequently, the research proposal format and the application form were completed. The principal investigator submitted the proposal with all relevant forms to the Office of Research Affairs (ORA). This office screened the proposal for compliance with submission guidelines, then forwarded it for peer review and sent it to the appropriate research committee for evaluation. The final decision was the study has approved by Research Advisory Council (RAC) (see the copy of ethical approval in appendices 13 and 14).

The introductory week was started by making appointments with the principle investigator in the hospital and the Deputy Executive Director of Nursing Affairs of the hospital. In that week the proposal and ethical approval were provided to the directors of medical departments such as surgery, medicine and paediatric to give them chance to read it before the meetings. All of them were interested and very keen to help and support the research because they believe the research was important and useful for patients. The principal investigator coordinated the process of data collection in Family Medicine and Polyclinic, Surgery, Medicine and Paediatrics. A number of questionnaires were distributed in each department based on available number of doctors working in each department.

There was a meeting with the Deputy Executive Director of Nursing Affairs of the hospital. In this meeting the proposal and ethical approval were provided and the protocol of the study was presented and discussed. As a result of this meeting internal memo was send to all head nurses to encourage the nurses to contribute in completing the questionnaire as important study for patient safety as much as they can(see the copy of this letter in appendix 15). Also there was a meeting with the Senior Clinical Research Coordinator of King Faisal Heart Institute. There was support for data collection by sending covering letter with the questionnaires to encourage doctors to participate in the study. In this way 400 questionnaires were distributed in different medical departments and clinical areas and patient areas such as wards and clinics. To ensure the smooth flow of the data collection regular visits were made with the principal investigator; head nurses and Senior Clinical Research Coordinator in order to remind them about the survey.

Two weeks were given for doctors to get their responses. The first reminder was issued after two weeks and the final reminder was issued after another two weeks through the communication between principal investigator and medical secretaries. Ten days was allowed for nurses to get their responses. The first reminder was done through head nurses after ten days and the final reminder after another two weeks because they are very busy staff. All returned data collected from the medical secretaries, head nurse and the Senior Clinical Research Coordinator. The time required to complete data collection was approximately (six months). Therefore, at that point of time in the beginning of October 2009 this process was completed. Table 5-3 shows the number of these questionnaires was distributed to medical and nursing staff of the hospital.

KI SHAKC			
Staff	Distributed questionnaires	Returned questionnaires	Return rate
Medical	150	40	26%
Nursing	250	116	46%
First reminder	Two weeks for doctors	Ten days for nurses	
Final reminder	Two weeks for doctors	Two weeks	
Total	400	156	39%

#### Table 5-3 Data collection of the KFSH&RC

VESUP.DC

In summary, 1700 questionnaires and cover letter were distributed as one document. Different methods were used in each hospital to approach potential respondents and these methods produced 913 returned questionnaires from the three hospitals in Riyadh in Saudi Arabia which was very successful data collection as a good number of the questionnaires for data analysis. The data collection process was based on the following main principles:

- 1. The data collection started by submitting the study proposal and hospital's requirements for getting the ethical approval.
- 2. Making a data collection plan for each hospital in order to save time.
- 3. Using different methods of gathering the data to achieve the minimum target data for each hospital.
- 4. Approximately two weeks were given for doctors and nurses to return questionnaires because staff differ in terms of time they need to complete the questionnaires which might be completed at one sittings or over two sittings or may they completed at home because they are busy staff with patients (Bourque and Fielder, 1995).
- 5. Follow up and assessment of the progress of the data collection.
- 6. There was no need to redistribute the questionnaires because a large number of questionnaires were returned.
- 7. At the end of the data collection process all the people who provided support to get this data were thanked for their helps and support, such as medical and nursing directors, principal investigator, head nurses and medical secretaries in all three hospitals.

The data collection included the following activities to maximize return rate:

- 1. An introductory week was done in each hospital before distributing the questionnaires included a number of meetings at both management level and workforce level, this week was very helpful and supportive in terms of informing the medical and nursing staff of survey and to make plan for data collection.
- 2. The meetings were held with medical director and nursing directors in each hospital to seek support and facilitate the data collection process.
- 3. A number of presentations were given in order to inform staff of the outlines of the study.
- 4. A number of meetings with doctors and nurses in nurses' stations and doctors meeting rooms at each hospital were held to inform them of the survey.
- 5. The questionnaires were returned to the head nurses and medical secretaries in wards, clinics and other specialist areas such as medical centres in the hospitals. The questionnaire and cover letter were handed to the participants to be completed. The cover letter explained the purpose of this study. The cover letter also assured the participants that the questionnaires were totally anonymous. In addition how to return the questionnaires was also stated in the cover letter clearly.
- 6. Working every day in the hospital and following up the progress by regularly visiting the doctors and head nurses in order to motivate them to reply is considered very important factor in increasing the response rate (Bourque and Fielder, 1995).
- 7. The strategy of data collection was changed many times and depends on the situation in each hospital.
- 8. Doctors and nurses were motivated to complete the questionnaires because patient safety is an important issue and one of the main responsibilities of their job and because they believe that they are responsible for the safety of patients in their hospitals therefore they were very keen to contribute and complete the questionnaire.
- 9. Handling the return questionnaires was based on collecting the returned questionnaires from the head nurses and the medical secretaries in all three hospitals as a reception point for the returned questionnaires.

In short, all these activities helped to approach potential participants for the study. There were number of factors that increased the response rate. Firstly, there were factors related to the subject of the study, because assessment of patient safety culture is a very important element for improving patient safety in hospitals.

#### Chapter five

Furthermore, the study protocol was carried out carefully to achieve this good data return. Secondly, factors related to the questionnaire, included the simple and clear language of the questionnaire and easy questionnaire formats and structures to make it easy for respondents to complete. Finally, factors related to the practicality of the gathering the data such as using different effective methods of data collection. All returned questionnaires (913) were kept in a secure location for confidentiality during all stages of the data collection in Saudi Arabia as well as in the university. The returned questionnaires were brought to the UK to start data management and cleaning before performing data analysis.

### 5.6 Data management

#### 5.6.1 Introduction

Reviewing and cleaning of the data was conducted to make it ready for entry into SPSS for analysis. In this stage of the research each questionnaire was examined for completeness, to make sure that the respondents had completed the questionnaires properly prior to coding and entering the data into an electronic data file using statistical software (SPSS).

The data management process is concerned with preparing, entering and editing the questionnaire responses from the returned questionnaires. The data management aims to prepare a clean data file for analysis. The process started with revision, checking and cleaning of the returned questionnaires to make them ready for entry into the data file in order to perform the analysis. The statistical packages SPSS version 16.0 and STATA V.8 were used for data processing and analysis. At this stage identifying complete and incomplete questionnaires is very important in calculating the return rate of the study.

#### 5.6.2 Data Preparation

Data management was started with sorting the returned questionnaires in terms of completed and incomplete questionnaires. Thus each questionnaire was thoroughly reviewed to check whether it was completed or not. Among 1700 questionnaires that were administered at three hospitals, there were 862 completed questionnaires with a return rate of 50.7% and 51 incomplete questionnaires. At this level the completed questionnaires (862) were sufficient to perform the analysis according to the statistical requirements (Guadagnoli and Velicer, 1988) and therefore the incomplete questionnaires were excluded from the data entry in this stage. The incomplete questionnaires were examined later in order to know which items were not completed and the reasons for incompleteness. The completed questionnaires were provided with serial numbers, started from number 1 to number 862. This numbering helped in data entry and also facilitated tracking and queries during different stages of data management; it was also useful for the questionnaires' archiving and sorting. Each questionnaire was addressed by the name of the hospital where it was completed, work area and the current position of the respondent in order to enable the researcher to identify participants by this information during data entry.

## 5.6.3 Data entry

The variables of the HSOPSC questionnaire were clearly numbered and defined in the SPSS data file before entering the data. This stage consisted of two steps. The first step was coding and defining the variables' numeric values in the "Variable view" in SPSS. The second step was entering data into SPSS. The data entry step was started with variables definition by identifying the variables of the HSOPSC questionnaire which includes the number of the variable, variable label, type and measurement. The variables (44 variables). The first set of variables (11 variables) and patient safety culture variables (44 variables). The first set of variables provides information about the respondents' identity and background characteristics (Demographic items include: A-H1-H2-H3-H4-H5-H6-H7-H8-H9-H10). The second set of variables includes the variables of patient safety culture (44 variables) (Non demographic items include A1-A2-A3-A4-A5-A6-A7-A8-A9-A10-A11-A12-A13-A14-A15-A16-A17-A18-B1-B2-B3-B4-C1-C2-C3-C4-C5-C6-D1-D2-D3-E-F1-F2-F3-F4-F5-F6-F7-F8-F9-F10-F11-G). The total coding of the variables is 55 data variables plus the ID of questionnaire (see appendices 17 and 18).

The second step in this stage was entering data into SPSS (through data view). With the variables defined, data entry started according to the serial number sequence of the questionnaires from number one to number 862. During the data entry each questionnaire was subject to double check and matching with its value label in order to ensure the precision of the data entry and to minimize the probability of data entry errors. This process was continued up to the end of the data entry. The data entry record was used to write comments

during data entry e.g. record of latest entry every day, poor handwriting, etc. It is worth to mentioning that the data was entered in reasonable blocks of time (approximately 20 questionnaires each time) to minimize data entry mistakes due to fatigue and tiredness rather than entering it in one large period of time.

#### 5.6.4 Data editing (validation and checking)

This stage was intended to check and validate the data by carefully reviewing and screening the entered data in order to identify duplicate cases, to check missing values, and to identify implausible (unlikely) values outside the range. In this stage the following strategy was used:

#### 1. Identify duplicate cases

The option of identifying duplicate cases (questionnaires) in SPSS was used to check if there was any duplicate case (sorting cases by serial number). This step identified five duplicate cases. These cases were 166, 334, 600, 674 and 675. Each case was reviewed carefully. Initial data entry error with code for 196 was being wrongly input as 166. Number 334 was entered wrongly with number 344 as 334. Number 600 was entered twice. Number 674 was entered in place of 574. Number 675 was entered instead of number 575. All duplicate cases were sorted out or deleted. Table 5-4 shows the duplicate cases and the action taken to these cases.

ĪD	Reason	Action taken
166	196 wrongly input as 166	ID changed to 196
334	344 wrongly input as 334	ID changed to 344
600	Same questionnaire input twice	The second entry was removed
674	Number 674 was entered in place of 574	ID (674) changed to 574
675	Number 675 was entered instead of number 575	ID (675) changed to 575

**Table 5-4 Duplicate cases** 

#### 2. Identify Missing values

The option of identifying missing values in SPSS was used to check if there was any missing value. The frequencies tables show blank records in 34 questionnaires. All these questionnaires were reviewed carefully to double check the data entry of these questionnaires. All were found to be due to errors in the data entry. Therefore all these blank records were completed using the responses from the questionnaires. All the items of each questionnaire above were reviewed and not only specific items. After that the data was checked again and the frequencies tables showed no missing values.

#### 3. Identify implausible values outside the range:

The frequencies tables show number of implausible values outside the range as follows:

Table 5- 5 Outliers			
Number of questionnaire	Number of item	Action taken	
601	H4	H4 was changed from 6 to 4	
606	G-H2-H3	Corrected G=4, H2=2, H3=2	
608	H2	H2 was changed from 5to 1	
686	A5	A5 was changed from 45to 4	
710	A17	A17 was Changed from 15to 4	

the frequencies tables show number of implausible values outside the range as for

All the above questionnaires were checked and all the outliers were corrected. Generally, in this stage 39 questionnaires were reviewed carefully in order to clean the data continuously. Errors could happen at any stage of entering the data from the HSOSPC questionnaires into SPSS, therefore ensuring the accuracy of data entry is very important to achieve the best quality of data to obtain a realistic analysis. Double entry of data is recommended as a method for validating the accuracy of the data entry and an effective method for reducing data entry errors (Day, 1998).

#### 5.6.5 Double entry

In order to have complete confidence in the final data set, 15% double random check entry was done as follows:

- 1. 15% of the cases were randomly selected (130 cases) entitled DF1A1 (original file)
- 2. The new file was created including 130 cases entitled DF1A2 (re entering data)
- 3. Re entering the same 130 cases into DF1A2
- 4. STATA V.8 was used to compare the two files to check entry data errors (see the result of the comparison
- 5. The values of the two files were compared with the values in the questionnaires to detect the place of error (see the comparison table in appendix 19).

In summary, the comparisons showed that there were 55 data items (ignoring the ID) and 130 cases ( $55 \times 130$ ), which make 7150 records. There were 51 errors (the number of errors found in the re-entering file (DF1A2) was 33 errors; while in the original file (DF1A2) was 18 errors) in 23 questionnaires in total which is an error proportion of 0.7%. Therefore, double entry was made for the rest (732 questionnaires).

Double entry was done as follows:

- 1. First data entry file was created containing 732 cases
- 2. The new file was created including 732 cases entitled double entry data file (DF1A2)
- 3. Re entering the same 732 cases into DF1A2
- 4. STATA V.8 was used to compare the two files to check entry data errors.
- 5. The values of the two files were compared with the values in the questionnaires to detect the place of error (see appendix 20).

In summary, the comparisons showed that there were 55 data variables (ignoring the ID) and 732 cases (questionnaires)  $(55 \times 732)$ , which make 40260 records. There were 320 mismatches (errors) in 118 questionnaires in total which is an error proportion of 0.8%. The comparison was performed between these errors which showed that the number of errors found in double data entry file was 183 errors; while in the first data entry (original file) was 137 errors in 66 questionnaires. This means, an error proportion of the first data entry original file is 0.34%, less than one per variable. In fact the output of the double data checking from STATA that shows for each variable there is a mismatch is a good example of the value of double entry that was made in the study. This is a good return in terms of

accuracy of data entering. All entry data errors that found in first data file entry were corrected in the original file. All the questionnaires were reviewed and checked carefully as a final stage of cleaning data to detect any errors of the data in order to resolve them before starting data analysis.

#### 5.6.6 Incomplete questionnaires

The incomplete questionnaires were checked to clarify whether the respondents faced difficulty with completion in common items or sections or whether it was due to other reasons. Also, the incomplete questionnaires were examined to know whether the demographic information or non demographic information was incomplete. The number of incomplete questionnaires were returned from the three hospitals was 51 (25 from KFNGH, 14 from KFMC and 12 from KFSH&RC). The questionnaires were provided with serial numbers, started from number 1 to number 51 and entered into SPSS. Frequencies tables were used to examine these incomplete questionnaires. These tables showed incomplete item or section in each questionnaire (see appendix 16). Table 5-6 summarises these incomplete questionnaires in five groups.

In complete section or items		Number of returned	%
		questionnaires	
1.	Demographic items	16	31.4
2.	More than half of demographic items	9	17.6
3.	Section B	5	9.8
4.	Different sections (D-C-E-F-G)	14	27.5
5.	Different items (A3-A4-A7-A17-D2-D3-	7	13.7
	F6 F7-H4-H5-H8)		
Total		51	100%

#### Table 5- 6 Incomplete questionnaire (n=51)

From the above table the demographic items were incomplete in 16 questionnaires and more than half of demographic items were incomplete in nine questionnaires. Section B was incomplete in five questionnaires. Different sections such as D, C, E, F and G were incomplete in 14 questionnaires. Different items such as A3, A4, A7, A17, D2, D3, F6, F7, H4, H5, and H8 were incomplete in seven different questionnaires. As mentioned earlier incomplete questionnaires were examined to clarify whether the respondents faced difficulty with completion in common items or sections and to identify if there is any trend or interest issues in these questionnaires.

There are several interpretation of these findings: for example as a few respondents may have felt that the questionnaire was too long so they left many sections blank, or they did not have time to complete it. For section B it could be that respondents do not like to use the word supervisor or manager in section B or were unsure which was more appropriate word, thus section B was left incomplete in five questionnaires.

Generally, the aim of analysing incomplete questionnaires was to examine whether the incomplete questionnaires were valid to be included in data analysis or not. Due to incomplete demographic information and the fact that considerable numbers of sections in these questionnaires were missing and moreover, given the availability of the large number of completed questionnaires (862). Hence, it was felt that it was better to use just the completed data set rather than including missing data as well as, because of the uncertainty of the seriousness of the respondents of incomplete questionnaires. Therefore the decision was made that the incomplete questionnaires were excluded from data analyses. At the end of the data management process the clean data file was ready for the next stage, data analysis.

## 5.7 Summary

The HSOPSC questionnaire was used to collect data, and 1700 questionnaires were administered at three hospitals in Saudi Arabia to collect the questionnaire data. In terms of return rate 913 questionnaires was returned. The data was cleaned and prepared for data analysis, 862 fully completed questionnaires were returned from three hospitals with a return rate of 50.7%. In the next chapter the data analysis will be presented.

## Chapter 6: Assessing the psychometric properties of the Hospital Survey on Patient Safety Culture

## 6.1 Introduction

This chapter reports the results of the psychometric properties of the HSOPSC questionnaire (Sorra and Nieva, 2004) in Saudi hospitals in Riyadh in Saudi Arabia using Saudi data (862 completed questionnaires). The chapter is composed of seven sections. Section 6.2 presents descriptive analysis including the sample characteristics and the item responses (item analysis). Section 6.3 concerns assessing the psychometric properties of the HSOPSC questionnaire (Sorra and Nieva, 2004). This section presents the results of the assessment of the suitability of the Saudi data for factor analysis. Section 6.4 provides the results of testing the original HSOPSC questionnaire (Sorra and Nieva, 2004). Section 6.5 reports the results of identifying the optimal model, including EFA, CFA and reliability analysis. Section 6.6 provides the proposed optimal model of the study. Finally, section 6.7 presents a brief summary of the chapter.

152

## 6.2 Descriptive Analysis

This section provides the demographic characteristics of the study respondents. It also presents an analysis of the responses to individual items of the HSOPSC questionnaire. Frequency tables are used to show the responses and the percentages of the respondents to safety climate items of the questionnaire.

#### 6.2.1 Demographic information

This section reports the characteristics of the sample. Table 6-1 shows work area of respondents (doctors and nurses).

Table 0-1 Wolk area of respondents (n 002)					
Number (%)					
164	(19.0)				
164	(19.0)				
148	(17.2)				
117	(13.6)				
106	(12.3)				
69	(8.0)				
43	(5.0)				
26	(3.0)				
16	(1.9)				
6	(0.7)				
3	(0.3)				
	Nun 164 164 148 117 106 69 43 26 16 6 3				

 Table 6- 1 Work area of respondents (n=862)

Table 6-1 shows that the variety of work areas of the respondents. Among the total respondents (862) the highest proportion of respondents worked in surgery (19.0%) and other departments (19.0%) followed by intensive care units (17.2%), medicine (13.6%), and paediatrics (12.3%). The detail of other departments is provided in table 6-2. The lowest percentage of respondents worked in obstetrics (8%), followed by emergency (5%), rehabilitation (3%), many different hospital units (1.9%), anaesthesiology (0.7%) and psychiatry /mental health (0.3%). It was important to identify other work areas because it was the highest percentage. Table 6-2 shows the other departments.

Work area unit Number		
Cardiac	30 (18.3)	
In patient wards	28 (17.0)	
Endoscopy	23 (14.0)	
Operation room	20 (12.2)	
Oncology	15 (9.0)	
Ophthalmology	13 (8.0)	
Neonate	10 (6.0)	
Quality and safety	9 (5.5)	
Orthopaedic	6 (4.0)	
Nephrology	5 (3.0)	
Endocrinology	5 (3.0)	

Table 6- 2 Other work areas of respondents (n=164)

The demographic variables of the sample included: current position; gender; nationality; highest educational level; training outside Saudi Arabia; years of work in this hospital; years of work in current hospital work area/unit; years of work in current speciality or profession; hours of work per week; and direct contact with patients. Table 6-3 provides the sample characteristics of the study.

.

N	Variable		Number in eac	ch unit (%)
1	Current position	Nurse	661	(76.7)
		Doctor	201	(23.3)
2	Gender	Female	632	(73.3)
		Male	230	(26.7)
3	Nationality	Non Saudi	700	(81.2)
		Saudi	162	(18.8)
4	Highest qualification	PhD	129	(15.0)
		Master	34	(3.9)
		Bachelor	525	(60.9)
		Diploma	160	(18.6)
		Other	14	(1.6)
5	Training outside S.A	Yes	668	(77.5)
		No	194	(22.5)
6	Years of working in this hospital	Less than 1 year	123	(14.3)
		1 to 5 years	464	(53.8)
		6 to 10 years	181	(21)
		11 to 15 years	48	(5.6)
		16 to 20 years	26	(3.0)
		21 years or more	20	(2.3)
7	Years of working in current hospital	Less than 1 year	119	(13.8)
	work area/unit	1 to 5 years	499	(57.9)
		6 to 10 years	161	(18.7)
		11 to 15 year	45	(5.2)
		16 to 20 years	23	(2.7)
		21 years or more	15	(1.7)
8	Years of working in your current	Less than 1 year	42	(4.9)
	speciality or profession	1 to 5 years	235	(27.3)
		6 to 10 years	220	(25.5)
		11 to 15 yea	162	(18.8)
		16 to 20 years	99	(11.5)
		1 years or more	104	(12.0)
9	Working hours per week	Less than 20 hours per week	9	(1.0)
		20 to 39 hours per week	46	(5.3)
		40 to 59 hours per week	661	(76.7)
		60 to 79 hours per week	105	(12.2)
		80 to 99 hours per week	37	(4.3)
		100 hours per week or more	4	(0.5)
10	Direct interaction or contact with	Yes	850	(98.6)
	patients	No	12	(1.4)

## Table 6- 3 Sample characteristics of the study (n=862)

The sample of the study consisted of 201 doctors (23%) and 661 nurses (77%). The majority of respondents were female (73.3%). Most of respondents were non Saudi staff (81.2%) from 24 different nationalities. These nationalities were: British, Australian, Canadian, American, South African, Philippine, Pakistani, Indian, French, Egyptian, Netherlands, Jordanian, Norwegian, Finland, Yemeni, Kuwaiti, Chinese, Scottish, New Zealand, Malaysian, Sudan, Syria, Ireland, and Bahraini.

The majority of staff (60.9%) have a bachelor's degree and 18.9% of staff have postgraduate qualifications (PhD, master's degree). Over 18.6% have a diploma degree. The majority of staff had training outside Saudi Arabia (77.5%) in many countries such as UK, USA, Australia, and Canada. The majority of the respondents (53.8%) worked from one to five years in their hospital and 57.9% worked from one to five years in their specific hospital unit or work area. The majority of the respondents (52.8%) have experience from one to ten years in their current speciality or profession. The most common category of work hours per week (76.7%) was 40 to 59 hours. Almost all respondents (98.6%) had direct contact with patients.

#### 6.2.2 Items analysis

The HSOPSC questionnaire consists of 42 items (24 positively worded items and 18 negatively worded items) plus two items on an overall patient safety grade (E) and number of events reported (G). These 42 items measure 12 dimensions of patient safety culture. Table 6-4 shows the responses of the positively worded safety climate items.

Item	Strongly Disagree	Disagree	Neither	Agree	Strongly Agree
	%	%	%	%	%
A1 .	1.3	6.7	8.4	62.9	20.8
A2	8.4	25.4	14.4	42.3	9.5
A3	0.7	7.5	11.0	58.7	22.0
A4	2.0	8.2	15.2	55.7	18.9
A6	0.5	1.4	4.6	57.0	36.5
A9	2.9	10.3	21.7	54.4	10.7
A11	4.1	18.3	13.0	51.4	13.2
A13	1.2	5.2	9.5	65.5	18.6
A15	4.9	15.7	13.0	46.8	19.7
A18	2.0	9.5	16.1	60.2	12.2
B1	4.1	7.0	12.9	56.4	19.7
B2	3.4	8.4	13.6	57.1	17.6
C1	2.0	9.3	34.5	36.0	18.3
C2	1.2	8.8	31.0	38.5	20.5
C3	1.6	7.7	23.3	37.4	30.0
C4	9.6	23.1	32.9	25.3	9.0
C5	1.3	5.8	16.5	43.3	33.2
D1	3.5	18.1	24.7	31.8	21.9
D2	4.1	16.0	25.4	33.8	20.8
D3	3.1	10.0	20.3	36.5	30.0
F1	1.3	6.7	11.3	64.2	16.6
F4	1.4	12.9	22.6	55.9	7.2
F8	2.2	4.2	9.0	49.1	35.5
F10	0.9	7.2	13.9	55.3	22.6

Table 6-4 shows that the highest percentage of all positive safety climate items is between the responses agree and strongly agree. The response of neither ranges from 4.6% to 34.5%. The lowest percentage is between disagree and strongly disagree. This means the majority of the respondents tended to agree and strongly agree with these positive items. Table 6-5 shows the responses of the negative safety climate items.

Item	Strongly Disagree	Disagree	Neither	Agree	Strongly agree
	%	%	%	%	°/0
A5 ®	6.1	35.2	18.3	26.6	13.8
A7 ®	21.0	48.1	14.8	11.1	4.9
A8 ®	4.8	26.2	22.3	35.7	11.0
A10 ®	9.3	35.8	18.7	31.6	4.6
A12 ®	4.3	26.7	20.5	37.4	11.1
A14 ®	5.9	34.2	18.6	30.3	11.0
A16 ®	1.9	10.0	17.3	50.9	20.0
A17 ®	8.4	37.1	19.4	29.7	5.5
B3 ®	12.9	57.0	19.6	8.6	2.0
B4 ®	33.4	56.4	6.0	2.91	1.3
C6 ®	15.9	26.0	40.5	13.2	4.4
F2 ®	7.2	46.5	21.2	21.3	3.7
F3 ®	2.6	30.9	24.9	39.0	2.7
F5 ®	11.4	53.9	18.9	14.3	1.5
F6 ®	6.4	45.9	26.7	17.4	3.6
F7 ®	2.9	34.2	30.4	29.4	3.1
F9 ®	10.9	47.7	15.2	20.6	5.6
F11 ®	11.3	52.3	22.7	11.9	1.7

Table 6-5 Responses of the 18 negatively worded safety climate items (n=862)

R indicates negatively worded items

The above table shows the responses for the 18 negatively worded safety climate items. The highest percentage of all items is between disagree and strongly disagree except items A16 and F3. The response of neither ranges from 6% to 40.5%. The smallest percent is between agree and strongly agree. The responses of disagree and strongly disagree of negatively worded items count as positive response because negative response of negative items means positive response. This means the majority of the respondents disagreed and strongly

disagreed with these negative items. Table 6-6 provides the responses to the question of an overall grade on patient safety of 862 respondents.

Table 6- 6 Responses to the question of an overall grade on patient safety (n=862)							
Item	Excellent	Very good	Acceptable	Poor	Failing		
	%	%	%	%	%		
Е	14.0	48.1	35.7	1.9	0.3		

Among the 862 respondents 48.1% perceived that the grade of patient safety was very good and 14% perceived that the grade of patient safety was excellent. A sizable proportion of the respondents 35.7% indicated that patient safety was acceptable. Few respondents (1.93%) perceived that patient safety was poor and failing.

Table 6-7 provides the responses to the question on the number of events reported in the past 12 months for all respondents.

months (n=862)								
ltem	No event reports %	1 to 2 event reports %	3 to 5 event reports %	6 to 10 event reports %	11 to 20 event reports %	21 event reports or more %		
G	40.8	38.9	12.1	5.5	1.6	1.2		

Table 6-7 Responses to the question on the number of events reported in the past 12 months (n=862)

The highest percentage of respondents (40.8%) did not report events followed by one to two event reports. Over 12 % reported three to five event reports. Five and half percent (5.5%) reported six to ten event reports. Around 1.6% reported 11 to 20 event reports. The lowest percent was 1.2% reported more than 21 event reports.

It was useful to determine if those reporting poor/failing were the same individuals who reported lots of events. Table 6-8 shows the comparison between an overall grade of patient safety and number of event reports in the past 12 months.

Item		G: Number of event report in the past 12 months				
		No event reports and 1 to 2 event reports	3 to 5 event reports and 6 to 10 event reports	11 to 20 event reports and 21 event reports or more	Total	
E: An overall grade of patient safety	Very good and Excellent	433 (80.7%)	88 (16.4%)	15 (2.7%)	536	
	Acceptable	242 (78.6%)	58 (18.8%)	8 (2.5%)	308	
	Poor and Failing	12 (66.7%)	5 (27.8%)	1 (5.5%)	18	
	Total	687	151	24	862	

Table 6-8 Comparison between an overall grade on patient safety and number of event reports in the past 12 months (n=862)

Table 6-8 shows for those reporting poor/failing the higher proportion (66.7%) reported no event reports, while the lowest proportion (5.5%) reported 11 to 20 event reports. This means those reporting poor/failing were not the same individuals who reported lots of events. Table 6-9 shows the descriptive analysis of the positively worded safety climate items.
Table 0- 9 Descriptive analysis of the 24 positively v	vorueu salety chim	ate items (n=862)
Item	Mean (Sd)	Median (IQR)
A1: Staff support one another in this unit	3.9 (0.8)	4.0 (1-5)
A2: We have enough staff to handle the workload	3.1 (1.2)	4.0 (1-5)
A3: When a lot of work needs to be done quickly, we work together as a team to get the work done	3.9 (0.8)	4.0 (1-5)
A4: In this unit, people treat each other with respect	3.8 (0.9)	4.0 (1-5)
A6: We are actively doing things to improve patient care	4.2 (0.6)	4.0 (1-5)
A9: Mistakes have led to positive changes here	3.6 (0.9)	4.0 (1-5)
A11: When one area in this unit gets really busy, others help out	3.5 (1.1)	4.0 (1-5)
A13: After we make changes to improve patient safety, we evaluate their effectiveness	3.9 (0.8)	4.0 (1-5)
A15: Patient safety never takes second place to get more work done	3.6 (1.1)	4.0 (1-5)
A18: Our procedures and systems are good at preventing errors from happening	3.7 (0.9)	4.0 (1-5)
<b>B1</b> : My supervisor/manager says a good word when he/she sees a job done according to established patient safety procedures	3.8 (0.9)	4.0 (1-5)
<b>B2</b> : My supervisor/manager seriously considers staff suggestions for improving patient safety	3.7 (0.9)	4.0 (1-5)
C1: We are given feedback about changes put into place based on event reports	3.5 (0.9)	4.0 (1-5)
C2: Staff will freely speak up if they see something that may negatively affect patient care	3.6 (0.9)	4.0 (1-5)
C3: We are informed about errors that happen in this unit	3.8 (0.9)	4.0 (1-5)
C4: Staff feel free to question the decisions or actions of those with more authority	3.0 (1.1)	3.0 (1-5)
C5: In this unit, we discuss ways to prevent errors from happening again	4.0 (0.9)	4.0 (1-5)
<b>D1</b> : When a mistake is made, but is caught and corrected before affecting the patient, how often is this reported?	3.5 (1.1)	4.0 (1-5)
<b>D2</b> : When a mistake is made, but has no potential to harm the patient, how often is this reported?	3.5(1.1)	4.0 (1-5)
<b>D3</b> : When a mistake is made that could harm the patient, but does not, how often is this reported?	3.8(1.1)	4.0 (1-5)
F1: Hospital management provides a work climate that promotes patient safety	3.8(0.8)	4.0 (1-5)
F4: There is good cooperation among hospital units that need to work together	3.5(0.8)	4.0 (1-5)
<b>F8:</b> The actions of hospital management show that patient safety is a top priority	4.1(0.9)	4.0 (1-5)
F10: Hospital units work well together to provide the best care for patients	3.9 (0.8)	4.0 (1-5)

#### Table 6-0 Descriptive analysis of the 24 positively worded safety elimeteritems . .041

Table 6-9 shows that the mean scores of positive items were divided into three groups, the first group was between 3.0 and 3.5. The second group was between 3.6 and 3.9. The third group was between 4.0 and 4.2. The median score of all items was 4.0 except item number C4 was 3.0. Table 6-10 shows the descriptive analysis of the negative safety climate items.

Item	Mean (S d)	Median (IQR)
A5: Staff in this unit work longer hours than they should, which is not good for patient care	3.0 (1.2)	3.0 (1-5)
A7: We use more temporary staff than we should, which is not good for patient care	2.3 (1.1)	2.0 (1-5)
A8: Staff feel like their mistakes are held against them	3.2 (1.1)	3.0 (1-5)
A10: It is just by chance that more serious mistakes are held against them	2.8 (1.1)	3.0 (1-5)
A12: When an event is reported, it feels like the person is being written up, not the problem	3.2 (1.1)	3.0 (1-5)
A14: We work in a hurry, trying to do too much, too quickly	3.0 (1.1)	3.0 (1-5)
A16: Staff worry that mistakes they make are kept in their personnel file	3.7 (0.9)	4.0 (1-5)
A17: We have patient safety problems in this unit	2.8 (1.1)	3.0 (1-5)
<b>B3</b> : Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it means taking shortcuts	2.3 (0.9)	2.0 (1-5)
<b>B4</b> : My supervisor/manager ignores patient safety problems that happen over and over	1.8 (0.8)	2.0 (1-5)
C6: Staff are afraid to ask questions when something does not seem right	2.6 (1.0)	3.0 (1-5)
F2: Hospital units do not coordinate well with each other	2.6 (1.0)	2.0 (1-5)
F3: Some things do not happen or get missed when transferring patients from one unit to another	3.0 (0.9)	3.0 (1-5)
F5: Important patient care information is often lost during shift changes	2.4 (0.9)	2.0 (1-5)
F6: It is often unpleasant to work with staff from other hospital units	2.6 (0.9)	2.0 (1-5)
F7: Problems often occur in the exchange of information across hospital units	2.9 (0.9)	3.0 (1-5)
F9: Hospital management seems interested in patient safety only after an adverse event happens	2.6 (1.1)	2.0 (1-5)
F11: Shift changes cause problems for patients in this hospital	2.4 (0.9)	2.0 (1-5)

Table 6-10 shows that the mean scores of negative items were divided into three groups. The first group was between 1.8 and 2.6. The second group was between 2.8 and 3.2. The third group was 3.7. The median scores of all items were also divided into three groups. The first group scored 2. The second group scored 3. The third group scored 4. Table 6-11 shows the descriptive analysis of an overall grade on patient safety and number of event reports in the past 12 months.

Item	Mean (S d)	Median (IQR)		
E: An overall grade on patient safety	3.7 (0.7)	4.0 (1-5)		
G: Number of event reports in the past 12 months	1.9 (1.0)	2.0 (1-5)		

Table 6- 11 Descriptive analysis of an overall grade on patient safety item and number of event reports in the past 12 months (n=862)

Table 6-11 shows that the mean score of item (E) was 3.7 and its median score was 4. The mean score of item (G) was 1.9 and its median score was 2. The HSOPSC questionnaire consists of 42 items grouped by 12 dimensions. Table 6-12 shows the responses of the HSOPSC questionnaire items grouped by dimensions.

Item	Dimensions	Strongly	Disagree	Neither	Agree	Strongly
Ν		disagree	%	%	%	agree
		%				%
	1. Teamwork within units (4 ite	ms)		····		
A1	Staff support one another in this unit	1.3	6.7	8.4	62.9	20.8
A3	When a lot of work needs to be done					
	quickly, we work together as a team to get the work done	0.7	7.5	11.0	58.7	22.0
A4	In this unit, people treat each other with respect	2.0	8.2	15.2	55.7	18.9
A11	When one area in this unit gets really busy, others help out	4.1	18.3	13.0	51.4	13.2
	2. Supervisor/Manager expectat	ion and actio	n promoting	patient safe	ty (4 items	)
B1	My supervisor/manager says a good word when he/she sees a job done according to established patient safety procedures	4.1	7.0	12.9	56.4	19.7
B2	My supervisor/manager seriously considers staff suggestions for improving patient safety	3.4	8.4	13.6	57.1	17.6
B3 ®	Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it means taking shortcuts	12.9	57.0	19.6	8.6	2.0
B4 ®	My supervisor/manager ignores patient safety problems that happen over and over	33.4	56.4	6.0	2.91	1.3
	3. Management support for patie	nt safety (3 i	tems)			
F1	Hospital management provides a work climate that promotes patient safety	1.3	6.7	11.3	64.2	16.6
F8	The actions of hospital management show that patient safety is a top priority	2.2	4.2	9.0	49.1	35.5
F9 ®	Hospital management seems interested in patient safety only after an adverse event happens	10.9	47.7	15.2	20.6	5.6
	4. Organizational learning-contin	uous improv	ement (3 iten	ns)		
A6	We are actively doing things to improve patient care	0.5	1.4	4.6	57	36.5
A9	Mistakes have led to positive changes here	2.9	10.3	21.7	54.4	10.7

Table 6- 12 Responses	of the HSOPSC items	grouped by dimensions	(n=862)
		8	( ···)

•

A13	After we make changes to improve					
	patient safety, we evaluate their	1.2	5.2	9.5	65.5	18.6
	effectiveness					
	5. Overall perceptions of patient	safety (4 iter	ms)			
A15	Patient safety never takes second place	4.9	15.7	13.0	46.8	19.7
4 1 0	to get more work done					
A18	at preventing errors from happening	2.0	9.5	16.1	60.2	12.2
A10	It is just by chance that more serious	0.2	25.0	10 7	21.6	
®	mistakes do not happen around here	9.5	33.0	10.7	51.0	4.6
A17	We work in a hurry, trying to do too much,	84	37.1	19.4	29.7	5 5
®	too quickly	0.11	0,11	17.1	27.1	5.5
	6. Feedback and communication	about error	(3 items)			
C1	We are given feedback about changes put into place based on event reports	2.0	9.3	34.5	36	18.3
C3	We are informed about errors that happen in this unit	1.6	7.7	23.3	37.4	30.0
C5	In this unit, we discuss ways to prevent	1.3	5.8	16.5	43.3	33.2
	7 Communication openness (3 if	ems)				,
C?	Staff will freely sneak up if they see	emsy				
	something that may negatively affect	1.2	8.8	31.0	38.5	20.5
	patient care					20.5
C4	Staff feel free to question the decisions					
	or actions of those with more authority	9.6	23.1	32.9	25.3	9.0
C6 ®	Staff are afraid to ask questions when something does not seem right	15.9	26.0	40.5	13.2	4.4
	8. Frequency of event reported (	3 items)				
D1	When a mistake is made, but is caught					
	and corrected before affecting the	3.5	18.1	24.7	31.8	21.9
	patient, how often is this reported?					
D2	When a mistake is made, but has no		14.0			
	potential to harm the patient, how often	4.1	16.0	25.4	33.8	20.8
D1	is this reported?					
D3	when a misiake is made that could have the national but does not how					
	often is this reported?	3.1	10.0	20.3	36.5	30.0

E4	9. Teamwork across units (4 iten	ns)				
<b>F4</b>	There is good cooperation among			•• •		
	nospital units that need to work together	1.4	12.9	22.6	55.9	7.2
F10	Hospital units work well together to					
	provide the best care for patients	0.9	7.2	13.9	55.3	22.6
F2 ®	Hospital units do not coordinate well	7.2	46.5	21.2	21.3	3.7
F( @	with each other					
LO K	from other hospital units	6.4	45.9	26.7	17.4	3.6
	10. Staffing (4 items)					
A2	We have enough staff to handle the workload	8.4	25.4	14.4	42.3	9.5
A5 ®	Staff in this unit work longer hours					
	than they should, which is not good for	6.1	35.2	18.3	26.6	13.8
	patient care					
A7®	We use more temporary staff than we					
	should, which is not good for patient	21.0	48.1	14.8	11.1	4.9
	care					
A14	We work in a hurry, trying to do too					
®	much, too quickly	5.9	34.2	18.6	30.3	11.0
		,				
F3 @	11. Handolis and transitions (4 iter	ns)				
	missed when transferring patients from	26	30.9	24.9	39.0	27
	One unit to another	2.0	50.9	21.7	09.0	2.,
F5 ®	Important patient care information is					
	often lost during shift changes	11.4	53.9	18.9	14.3	1.5
F7 ®	Problems often occur in the exchange	• •	24.0	20.4	20.4	2 1
	of information across hospital units	2.9	34.2	30.4	29.4	3.1
F11	Shift changes cause problems for	11.2	52.2	22.7	11.0	17
®	patients in this hospital	11.5	52.5	22.1	11.9	1.7
	12. No punitive response to error (3	items)			•	
A8 ®	Staff feel like their mistakes are held	4.8	26.2	22.3	35.7	11.0
	against them					
A12	When an event is reported, it feels like					
(R)	the person is being written up, not the	4.3	26.7	20.5	37.4	11.1
A 1 4	problem					
АТО (R)	are kept in their remained file	1.9	10.0	17.3	50.9	20.0
	are kept in their personnel file					

Table 6-12 shows the responses tending to positive response (strongly agree and agree) toward positively worded safety climate items and dimensions, while they tend towards negative response (strongly disagree and disagree) toward negatively worded items and dimensions. The next section will report the results of the psychometric analysis of the HSOPSC questionnaire in Saudi hospitals.

## 6.3 Psychometric Analysis of the Hospital Survey on Patient Safety Culture

#### 6.3.1 Introduction

This study aims to examine the underlying dimensions and the psychometric properties of the selected patient safety climate questionnaire in hospitals in Saudi Arabia. This stage was concerned with assessing the psychometric properties (validity and reliability) of the HSOPSC questionnaire using the Saudi data and comparing this with the original HSOPSC model (Sorra and Neiva, 2004). Factor analysis is a method for examining the interrelationships amongst the items of the HSOPSC questionnaire in order to determine the structure (important underlying patient safety culture dimensions) in the questionnaire data. Furthermore, factor analysis can be used to assess the construct validity of an established instrument when administered to a specific population (Pett et al., 2003).

Therefore, it is important to examine the suitability of the data for factor analysis and to check the inter-item correlations of the HSOPSC questionnaire before performing factor analysis in order to ensure that the data is sufficient to be used for factor analysis (Smits et al., 2008). The following section presents the assessment of the suitability of the data for factor analysis.

# 6.3.2 Assessment of the suitability of the data for factor analysis

This step aimed to assess the suitability of the Saudi data for factor analysis. There are two main issues when considering whether a particular data set is suitable for factor analysis. The first issue is sample size (adequacy of sample). The second issue is the strength of the inter-correlation amongst the items (correlation between the items) (Pallant, 2007).

Sample size is considered as an important requirement for factor analysis (Field, 2009). Therefore, before measuring the adequacy of the sample for factor analysis, it is important to report sample requirements such as the sample size of the study, the ratio of the sample size to the number of variables, the ratio of the number of variables to the number of factors in

168

order to assess whether the sample size is sufficient to satisfactorily undertake factor analysis (Ford et al., 1986; Tinsley and Tinsley, 1987). Table 6-13 shows the sample size requirements.

Table 6-13 Sample size requirements of factor analysis				
	Number			
Sample size of the study	862			
Number of variables	42			
Ratio of sample size to number of variables:	Greater than 20:1			
Ratio of number of variables to number of factors:	3:1			
	4:1			

~ ~

The sample size of the data (862) is large and more than enough to conduct factor analysis with greater than 20 observations to each variable according to Coakes and Steed (2003). Comrey and Lee (1992) suggest that 300 cases are good, 500 cases are very good and 1000 cases or more are excellent because the large sample is useful and support the high confidence in the results. Also they report that normality is not necessary, because although normally distributed variables make the solution stronger, this is not necessary with a large data set. It is recognised that the assumption of normality may not be fully met (Helen, 2010). It is important to mention that the factor analysis assumed that the 5-point response scale for the HSOPSC items was a continuous response scale (Pett et al., (2003).

The ratio of sample size to number of variables is greater than 20:1, which is sufficiently large according to Everitt (1975). Costello and Osborne (2005) empirically tested the effect of sample size on the results of factor analysis and reported that larger samples tend to produce more accurate solutions. Moreover, 70% of the samples with the largest ratio (20:1) produced correct solutions. They also report that the number of misclassified items was also significantly affected by the size of a sample. MacCallum et al. (1999) suggest that increasing the sample size is one means of overcoming these problems. They argue that, as the sample size increases, sampling error is reduced, factor analysis solutions become more stable and more reliably produce the factorial structure of the population (MacCallum et al. 1999). It is clear that a large data set is very useful in factor analysis in terms of producing a robust factor solution. From these results we can conclude that the sample of the study is sufficient for undertaking the factor analysis.

169

In terms of correlation between the items, looking at the inter-correlation between items is also an important requirement when conducting factor analysis. There are two potential problems, firstly, correlations that are not high enough (below 0.3) and secondly, correlations that are too high (greater than 0.8) (Field, 2009). Tabachnick and Fidell (2007) recommend an inspection of the correlation matrix for evidence of coefficients greater than 0.3.

Two statistical measures help to assess the suitability of the data for factor analysis: Bartlett's test of sphericity (Bartlett, 1954) and the Kaiser-Meyer-Olkin (KMO) measure of sampling adequacy. Bartlett's test of sphericity should be significant (P<.05) for the factor analysis to be considered appropriate and the KMO should be greater than 0.6 as the minimum value for a good factor analysis (Tabachnick and Fidell, 2007). Therefore KMO and Bartlett's test were used to assess factorability of anti image to correlate (items to correlate as factors). Table 6-14 shows the results of KMO and Bartlett's test.

Table 6-	14	KMO	and	Bartlett's test
----------	----	-----	-----	-----------------

Kaiser-Meyer-Olkin Measure o	0.905		
Bartlett's Test of Sphericity	Approx. Chi-Square	6606.872	
	df	861	
	Sig.	.000	

If KMO >0.6 that means the correlation matrix is valid (acceptable) for factor analysis. Pett et al. (2003) report that a KMO above 0.90 is very good. In this data set KMO=0.905, which means that the degree of homogeneity of the sample that was obtained was sufficient and the sample is adequate for factor analysis.

The Bartlett's test was applied to determine the presence of the relationship with variables (items) of the HSOPSC questionnaire. The Bartlett's test was highly significant (P<0.001) for the data set which is considered appropriate and factor analysis is applicable according to Coakes and Steed (2003) and Field (2009).

It is important to mention that if Bartlett's test is significant (P<0.05) that means the correlation matrix is not an identity matrix (Coakes and Steed, 2003; Pett et al., 2003).

"Identity matrix is a matrix which 1s appear on the diagonal and 0s on the offdiagonal. An example of such an identity matrix would be a correlation matrix in which all items on the off-diagonal are completely uncorrelated. A correlation matrix that is an identity matrix is not a welcome sight in factor analysis because its presence would imply that there are no interrelationships among the items" (Pett et al., 2003:63).

Moreover, according to Field (2009), the Bartlett's test examines whether the correlation matrix resembles an identity matrix (i.e. off-diagonal components are zero). Therefore, if Bartlett's test is significant that means the correlations between items are significantly different from zero. This means good news for conducting factor analysis because there are correlations between the items. If Bartlett's test is not significant that means the correlation matrix resembles an identity matrix. This means items correlate very badly with all other items (i.e. all correlation coefficients are close to zero) (Field, 2009). If the correlation is greater than 0.3 this means factor loading is greater than 0.3, which is good for the data.

In summary both the KMO test statistic and Bartlett's test of Sphericity indicate that the Saudi data is suitable to be subjected to factor analysis.

### 6.3.3 Psychometric Data Analysis Strategy

The psychometric data analysis strategy used in this stage was based on three main steps:

#### Step 1:

The first step aimed to investigate whether the factor structure of the original USA HSOPSC questionnaire (Sorra and Nieva, 2004) can be used with the Saudi data, in which both Confirmatory Factor Analysis ( $CFA^4$ ) and reliability analysis used. This step is described in section 6.4.

#### Step 2:

The second step aimed to examine whether an alternative structure of factors and items would fit the Saudi data better. In the other words, the objective is to check whether the items of the HSOPSC questionnaire formed different factors in the Saudi data, using exploratory factor analysis in order to develop the optimal model that can be used for assessing patient safety culture in Saudi hospitals. This step is described in section 6.5.1.

#### Step 3:

This step of psychometric analysis was concerned with undertaking the CFA and reliability analysis (internal consistency). It was concerned with testing the fit of the model that emerged from the Exploratory Factor Analysis (EFA<sup>5</sup>) using CFA and reliability analysis (section 6.5.2). In this step, the comparison between the CFA output of resulting models and the CFA output of USA (Sorra and Nieva, 2004) and UK (Waterson et al., 2009) studies was presented. Factor correlations of the optimal mode were performed (section 6.5.3). Finally, the reliability analysis (internal consistency) was performed to assess how well the items within each dimension relate to each other of the optimal model of the study (section 6.5.4). Each step is presented below:

<sup>&</sup>lt;sup>4</sup> For full definition of CFA, see glossary

<sup>&</sup>lt;sup>5</sup> For full definition of EFA, see glossary

## 6.4 Testing the original USA HSOPSC model

This step aimed to test the fit of the original USA HSOPSC model (Sorra and Nieva, 2004) to the Saudi data. This step of analysis is concerned with undertaking the CFA and Reliability Analysis.

#### 6.4.1 Confirmatory Factor Analysis

This step aimed to test the fit of the original USA HSOPSC model (Sorra and Nieva, 2004) to the Saudi data using CFA which is the best procedure for investigating the factor structures of questionnaire (Hoyle, 1995). CFA<sup>6</sup> was performed in order to examine the applicability of the factor structure of the original USA HSOPSC questionnaire to the Saudi data. AMOS software was used to undertake CFA (Arbuckle, 2005). Various measures of fit (fit indices<sup>7</sup>) examine whether the model provides a satisfactory fit to the data. This analysis includes the chi-square test statistic ( $\chi^2$ ), which is the most commonly used fit statistic, and the degree of freedom (df) ratio. Comparative Fit Index (CFI) value, Root-Mean-Square Error of Approximation (RMSEA), Incremental Fit Index (IFI), Root mean square residuals (RMR), Standardised Root Mean Square Residual (SRMR) and Tucker-Lewis Index (TLI) are also considered (Hu and Bentler, 1999). There is wide agreement that the overall model fit is evaluated using goodness of fit indices such as CFI-RMSEA-RMR-SRMR-TLI (March and Balla, 1994; Sanislow et al., 2002; Lievens and Anseel, 2004; Kline, 2005).

Hu and Bentler (1999) examine a range of cut-offs for many of these measures under different conditions (different sample size and model structure) and suggest using a combination of fit indices such as CFI and SRMR and RMSEA. They also suggest the CFI is used, with a value >0.90 indicating a good model. SRMR is used, with a value <0.08 indicating a good model. RMSEA is used, with a value  $\leq 0.06$  indicating a good model. Use of more fit indices been worthwhile such as  $\chi^2/df$ , RMR, and TLI. Therefore,  $\chi^2/df$ , CFI, RMSEA, RMR, SRMR and TLI were also used to evaluate the adequacy of the model fit. Table 6-15 shows the general agreed fit indices (parameters).

<sup>&</sup>lt;sup>6</sup> For full definition of CFA, see glossary

<sup>&</sup>lt;sup>7</sup> For more detail of fit indices and cut offs for fit indices, see glossary

r ( muex(parameter)	Cut off level (value)
Chi square $(\chi^2)$	CMIN/DF should be 2 or less reflects good
AMOS lists chi-square as CMIN/DF	fit. In the case of the chi-square statistic, smaller rather than larger values indicate a good fit.
Degree of freedom (df)	Should be small
Comparative Fit Index (CFI)	CFI should be equal to or greater than 0.90 to accept the model.
Root-Mean-Square Error of Approximation (RMSEA)	RMSEA ≤0.06 as the cut-off for a good model fit
Root mean square residuals (RMR)	RMR should be < .10, or .08, or .06, or .05, or even .04, for a well-fitting model
Standardized root mean square residual (SRMR)	A value less than 0.05 is widely considered good fit and below 0.08 adequate fit.
Tucker-Lewis index (TLI)	It should be greater than 0.9 for a good fit

## Table 6- 15 Generally agreed fit indices (parameters)

Note: The usual words that are used in assessing model fit are: good model fit (for good model), adequate model fit (when the values of fit indices are at borderline), not satisfactory (when the values of fit indices are outside the generally agreed parameters).

#### Practical steps in the CFA process:

The following steps describe the process of CFA:

- 1 AMOS Graphics was used to open up a blank diagram window.
- 2 Opening the data (the whole data was used, 862 questionnaires).
- 3 Building the model (drawing the 12 factors by using model drawing area in step one).
- 4 Adding the variable names to the model.
- 5 Running the model and using the output window.

Table 6-16 shows the fit indices of CFA when applying the original USA model to the Saudi data compared with USA data (Sorra and Nieva, 2004).

	to rit mu		AU USA I	a lacturs	model (SU		icva, 2004)		
Model	Chi-	df	CMIN/DF	CFI	RMSEA	RMR	SRMR	TLI	•
	square								
USA data	2064	746	_	0.94	0.040	-	-	-	•
Saudi data	4995	805	6.2	0.64	0.067	0.178	0.193	0.617	

Table 6- 16 Fit indices of CFA of USA 12 factors model (Sorra and Nieva, 2004)

The fit indices from CFA indicated a weak fit to the Saudi data, while the values of fit indices in USA indicated a good fit to the USA data. For example the CFI in the Saudi data was less than 0.9 while in the USA data it was 0.94 (good model > 0.90). RMSEA in the Saudi data was 0.067 while in the USA data it was 0.040 (good model < 0.06). CMIN/DF in Saudi data was greater than 2 (good model  $\leq$ 2). Overall, the values of the fit indices of the Saudi data (12 factors) are outside the general agreed parameters in table 6-15. Therefore, this model fit was not satisfactory.

#### 6.4.2 Reliability Analysis of the original USA HSOPSC model

The internal consistency of the dimensions of the original USA HSOPSC model (Sorra and Nieva, 2004) to the Saudi data was calculated with Cronbach's alpha ( $\alpha$ ). When different items are supposed to measure the same dimension (concept), it is recommended that internal consistency should be greater than or equal to 0.6 (Field, 2000). Since the HSOPSC questionnaire contains both positively worded items (n=24) and negatively worded items (n=18), the negatively formulated items were first recoded to make sure that a higher score always means a more positive response (Pallant, 2007). The internal consistency was calculated for every factor according to the original dimensions and items were then compared with the internal consistency found in the American study (Sorra and Nieva, 2004). The results of the reliability analysis for USA and Saudi data are presented in table 6-17.

HSOPSC dimension	Cronbach's alpha	ι (α)
	American data	Saudi data
Overall perceptions of safety	0.74	0.31
Frequency of error reporting	0.84	0.83
Supervisor/manager expectations and actions promoting patient safety	0.75	0.75
Organisational learning-continues improvement	0.76	0.63
Teamwork within units	0.83	0.75
Communication openness	0.72	0.67
Feedback and communication about error	0.78	0.74
Non punitive response to error	0.79	0.72
Staffing	0.63	0.57
Hospital management support for patient safety	0.83	0.65
Teamwork across hospital units	0.80	0.69
Hospital handoffs and transitions	0.80	0.59

Table 6- 17 Results of reliability analysis for USA data and Saudi data

The internal consistency of the Saudi data (n=862) for each dimension was greater than 0.6 except for three dimensions that were poor or unacceptable with values less than 0.6: overall perceptions of safety ( $\alpha = 0.31$ ), staffing ( $\alpha = 0.57$ ) and hospital handoffs and transitions ( $\alpha = 0.59$ ). The internal consistency of the Saudi data was lower for each dimension than the American data except dimension of supervisor/manager expectations and actions promoting patient safety.

Thus the results of the confirmatory factor analysis and reliability analysis showed that when applying the original USA HSOPSC model (Sorra and Nieva, 2004) to the Saudi data, the fit indices from CFA and internal consistency result indicated that this model fit was not satisfactory. This prompted an exploratory factor analysis in order to investigate if there is a factor structure that better fits the Saudi data.

## 6.5 Identifying an optimal model

This step aimed to construct an optimal model for assessing patient safety culture in Saudi hospitals. Exploratory factor analysis (EFA) is an important and useful tool for refining measures and evaluating construct validity. It is used for creating and refining the instrument's scales (Conway and Allen, 2003) and to explore the field and to discover the main constructs or dimensions (Kline, 1994). The analysis strategy in this step was based on validating the HSOPSC questionnaire construct by using EFA to uncover the latent structures of the safety culture dimensions (factors) and Cronbach's alpha for internal consistency, using the Saudi data set.

The Saudi data set was split randomly into two halves. Two files for each split half includes 431 cases, the first one called the exploratory data file (EFA) and the second one called the Confirmatory data file (CFA). EFA was used on one half to produce optimal model followed by CFA of resulting measurement model on validation half of data to test the fit of the resulting optimal factor structure. In other words, the first random half of the Saudi data set was used for model construction by exploring a number of factors (patient safety climate dimensions). The second random half of the Saudi data set was used for model validation by confirming a number of factors resulting from model construction.

The output of the first and final step EFA including tables of communalities, total variance explained and rotated Factor matrix and pattern matrix are presented in this section. The rest of the output of the other steps (from 2 to 10) of EFA are provided in the appendices as they are large tables and many that are difficult to present in this chapter because of space. Section 6.5.1 is concerned with a step by step strategy of EFA and reports its outputs.

#### 6.5.1 Exploratory Factor Analysis (EFA)

#### Step by step strategy and output of Exploratory Factor Analysis (EFA) is as follows:

EFA consists of two basic stages. The first stage, extraction, is concerned with how many factors should be extracted to represent the HSOPSC (safety climate items). The second stage is concerned with interpreting the meaning of the extracted factors and representing them in terms of theoretical structures of the patient safety climate dimensions. The strategy was based on running an initial EFA, using Principle Axis Factoring and Varimax rotation, as well as both Kaiser's criterion and the Scree plot, to assess how many factors should be extracted. This strategy produced a number of solutions. All these solutions were then examined by extracting the number of factors and changing the rotations to enable interpretation of the factors.

Each possible solution was examined in order to determine the number of extracted factors. A number of points were taken into consideration for each of the possible solutions included identifying items with no loading, low loading, low communalities, items which cross-load and the theoretical structure of items. The theoretical structure refers to the extracted factors (related items grouped in one factor) that are expected to be theoretically related. This process was continued until a satisfactory solution was reached.

Generally, each solution was examined and analysed carefully in relation to all these points before making a decision to either accept or reject it. Factor analysis is used as a data exploration technique and therefore, in order to achieve a satisfactory solution, the interpretation and the guidelines used are up to researcher judgement, rather than any hard and fast statistical rules (Pallant, 2007).

#### Step 1: EFA on all safety climate items of HSOPSC (42 items)

In order to run an initial EFA, Principal Axis Factoring was the chosen extraction method. Varimax was the chosen rotation method to provide an initial idea of items loading on each factor of the initial solution, thus providing a view about the underlying factors (latent structure). Eigenvalues greater than one (Eigenvalues > 1), and Scree plot were used to determine the number of factors to be extracted.

#### Output of the initial solution (11 factor solution):

As mentioned earlier, factor analysis depends on correlation between variables. Therefore, examining the correlations between the safety climate items (42 items) is considered to be a good idea in order to examine the correlations between the items before starting exploratory factor analysis. The correlation matrix shows the blocks of larger correlations running roughly diagonally down the matrix, indicating that the patient safety climate items are indeed closely related and that factor analysis is applicable. In order to examine the clarity of this solution the communalities, total variance explained, and rotated factors were investigated. Table 6-18 shows the communalities of the initial solution.

## Table 6-18 Communalities of the initial solution (n=431)

Safety climate items	Initial	Extraction
A1 Staff support one another in this unit	.52	.57
A2 We have enough staff to handle the workload	.25	.26
A3 When a lot of work needs to be done quickly, we work together as a team to get	.50	.59
A4 In this unit, people treat each other with respect	.50	.69
A5 Staff in this unit work longer hours than they should, which is not good for patient	.25	.28
A6 We are actively doing things to improve patient care	.41	.47
A7 We use more temporary staff than we should, which is not good for patient care	.25	.31
A8 Staff feel like their mistakes are held against them	.46	.66
A9 Mistakes have led to positive changes here	.35	.39
A10 It is just by chance that more serious mistakes do not happen around here	.13	.09
A11 When one area in this unit gets really busy, others help out	.38	.40
A12 When an event is reported, it feels like the person is being written up, not the	.44	.53
A13 After we make changes to improve patient safety, we evaluate their effectiveness	.47	.53
A14 We work in a hurry, trying to do too much, too quickly	.36	.37
A15 Patient safety never takes second place to get more work done	.20	.19
A16 Staff worry that mistakes they make are kept in their personnel file	.31	.35
A17 We have patient safety problems in this unit	.18	.20
A18 Our procedures and systems are good at preventing errors from happening	.42	.42
B1 My supervisor/manager says a good word when he/she sees a job done according	.60	.77
B2 My supervisor/manager seriously considers staff suggestions for improving	.63	.71
B3 Whenever pressure builds up, my supervisor/manager wants us to work faster,	.38	.46
B4 My supervisor/manager ignores patient safety problems that happen over and over	.42	.60
C1 We are given feedback about changes put into place based on event reports	.42	.44
C2 Staff will freely speak up if they see something that may negatively affect patient	.46	.52
C3 We are informed about errors that happen in this unit	.44	.48
C4 Staff feel free to question the decisions or actions of those with more authority	.41	.56
C5 In this unit, we discuss ways to prevent errors from happening again	.56	.57
C6 Staff are afraid to ask questions when something does not seem right	.28	.34
D1 When a mistake is made, but is caught and corrected before affecting the patient,	.54	.58
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?	.66	.88

Table 6-18 (continued)	A	
D3 When a mistake is made that could harm the patient, but does not, how often is	.54	.55
F1 Hospital management provides a work climate that promotes patient	.41	.41
F2 Hospital units do not coordinate well with each other	.45	.46
F3 Some things do not happen or get missed when transferring patients from one unit	.15	.14
F4 There is good cooperation among hospital units that need to work together	.47	.56
F5 Important patient care information is often lost during shift changes	.39	.46
F6 It is often unpleasant to work with staff from other hospital units	.37	.39
F7 Problems often occur in the exchange of information across hospital units	.42	.48
F8 The actions of hospital management show that patient safety is a top priority	.45	.47
F9 Hospital management seems interested in patient safety only after an adverse event	.36	.36
F10 Hospital units work well together to provide the best care for patients	.53	.61
F11 Shift changes cause problems for patients in this hospital	.34	.37

Extraction Method: Principal Axis Factoring.

The extraction column of the communalities table (6-18) above shows the proportion of variance of each variable explained by the initial solution. Five items, specifically A2, A10, A15, A17, and F3 are very poorly explained (very poor communalities, less than 0.3). A5 (2.84) was close to 0.3. This item (A5) was retained initially, but it was a candidate for removal later. Table 6-19 shows the total variance explained by the initial solution.

	Total Variance Explained										
Factor	In	itial Eigenv	values	Extract	ion Sums	of Squared	Rotatio	on Sums c	of Squared		
	Total	% of	Cumulative	Total	% of	Cumulative	Total	% of	Cumulative		
		Variance	%		Varianc	%		Varianc	%		
1	9.78	23.9	23.9	9.3	22.7	22.7	2.20	5.4	5.4		
2	2.32	5.7	29.5	1.8	4.5	27.1	2.18	5.1	10.4		
3	2.19	5.3	34.9	1.7	4.2	31.4	2.10	5.0	15.5		
4	1.73	4.2	39.1	1.2	3.0	34.4	1.20	4.9	20.3		
5	1.62	3.9	43.1	1.2	2.9	37.3	1.20	4.7	25.0		
6	1.50	3.7	46.8	1.0	2.5	39.8	1.92	4.5	29.6		
7	1.26	3.1	49.8	.8	1.8	41.6	1.85	4.4	33.9		
8	1.18	2.8	52.7	.6	1.5	43.2	1.55	3.7	37.7		
9	1.08	2.6	55.4	.6	1.5	44.7	1.42	3.4	41.1		
10	1.07	2.6	58.0	.4	1.1	45.8	1.40	3.3	44.5		
11	1.00	2.5	60.5	.4	1.0	46.8	0.95	2.3	46.8		
12	.96	2.3	62.9				<u> </u>				
13	.90	2.1	65.1								
14	.84	2.0	67.1								
15	.77	1.8	69.0								
16	.76	1.8	70.9								
17	.73	1.7	72.6	<b></b>							
18	.72	1.7	74.4								
19	.68	1.6	76.1								
20	.65	1.5	77.7								
21	.63	1.5	79.2								
22	.61	1.5	80.7								
23	.60	1.4	82.2								
24	.58	1.4	83.6								
25	.56	1.3	85.0								
26	.53	1.2	86.3								
_ 27	.51	1.2	87.6								
28	.47	1.1	88.7								
29	.46	1.1	89.8								
30	.43	1.0	90.9								
31	.43	1.0	92.0								
32	.41	1.0	93.0								

## Table 6- 19 Total variance explained by the initial solution (n=431)

Table (	Table 6-19 (continued)									
33	.39	.9	93.9					1		
34	.37	.9	94.9							
35	.36	.9	95.8							_
36	.34	.8	96.6							
37	.34	.8	97.4							
38	.30	.7	98.2							_
39	.27	.6	98.9							
40	.25	.6	99.5							
41	.19	.4	100							

Extraction Method: Principal Axis Factoring.

The three columns above show the Eigenvalues and respective cumulative percentages of variance accounted of all possible extracted factors. The next three columns indicate how many factors have been extracted determined by Kaiser's Eigenvalues > 1 criterion. The final three columns represent the percentages of variance explained by each factor after rotation. The total variance explained by this initial solution shows that 11 factors account for 46.8% of total variance. This number of factors was determined by Kaiser's Eigenvalues >1 criterion. Table 6-20 shows the rotated factor matrix of the initial solution.

Safety climate items Factor											
	1	2	3	4	5	6	7	8	. 9	10	11
F7 Problems often occur in the exchange of information across hospital units	.65							1			
F5 Important patient care information is often lost during shift changes	.57						18-11		e		
F6 It is often unpleasant to work with staff from other hospital units	.55	1.18							2		
F11 Shift changes cause problems for patients in this hospital	.50	4-73 1							£		
F9 Hospital management seems interested in patient safety only after an adverse event happens	.31				-		27				
F3 Some things do not happen or get missed when transferring patients from one unit to another	.30				14				and a		
A17 We have patient safety problems in this unit								33			
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?		.91						3.4	TT:		
D3 When a mistake is made that could harm the patient, but does not, how often is this reported?	5.5	.70	ing t	6. J.	hes	4244		13 13 P	eV.	5	
D1 When a mistake is made, but is caught and corrected before affecting the patient, how often is this reported?	17 S 17 S	.69		2004 2007	ante ante	g K dog	altat acar	-349 3 - 5	na vi		
A4 In this unit, people treat each other with respect	er da.	349	.75	ligasi.	ho	1 92	£	10132	19.		
A1 Staff support one another in this unit	£.,		.68		sheet.	30	2.08	14	12		1
A3 When a lot of work needs to be done quickly, we work together as a team to get the work	6.5.4	1 24	.60	130 H	daris.		spile	(ma)	14		
A6 We are actively doing things to improve patient care				.55							
A9 Mistakes have led to positive changes here				.52							
A18 Our procedures and systems are good at preventing errors from happening				.38			.32				
F1 Hospital management provides a work climate that promotes patient safety	30			.36			.30				
A15 Patient safety never takes second place to get more work done											
C3 We are informed about errors that happen in this unit					.54						
C1 We are given feedback about changes put into place based on event reports					.52			100	- 14		
A13 After we make changes to improve patient safety, we evaluate their effectiveness	1.00			.44	.46				-		
C5 In this unit, we discuss ways to prevent errors from happening again					.46					.30	
A8 Staff feel like their mistakes are held against them						.77					
A12 When an event is reported, it feels like the person is being written up, not the problem					-	.64		-			
A16 Staff worry that mistakes they make are kept in their personnel file						.51		- e feit is			
F10 Hospital units work well together to provide the best care for patients							.63	a			
F4 There is good cooperation among hospital units that need to work together							.58				
F8 The actions of hospital management show that patient safety is a top priority				.36			.45	Second Second			

## Table 6- 20 Rotated Factor Matrix of the initial solution (n=431)

Table 6-20 (continued)			nitizin	12.15	1410	<u>a_</u>		
F2 Hospital units do not coordinate well with each other	.41			42	1025			
B1 My supervisor/manager says a good word when he/she sees a job done according to established patient safety procedures	14				.77	4		
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety					.66			
A5 Staff in this unit work longer hours than they should, which is not good for patient care						.48		
A7 We use more temporary staff than we should, which is not good for patient care						.44		
A14 We work in a hurry, trying to do too much, too quickly				.38		.38		
B3 Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it			30	81.		.38		.32
A2 We have enough staff to handle the workload						36		
All When one area in this unit gets really busy, others help out		.32				36		
C4 Staff feel free to question the decisions or actions of those with more authority							.63	
C2 Staff will freely speak up if they see something that may negatively affect patient care			.34				.50	
C6 Staff are afraid to ask questions when something does not seem right							45	
B4 My supervisor/manager ignores patient safety problems that happen over and over		e cinero i		incury.	32	n		.56
A10 It is just by chance that more serious mistakes do not happen around here	1	201-1	1864	11-09	24	31		

Extraction Method: Principal Axis Factoring. Rotation Method: Varimax with Kaiser Normalization

a. Rotation converged in 17 iterations.

The rotated factor matrix shows the loadings of the initial solution (11 factors) with multiple items loading on each factor. Items A10, A15, A17 did not load upon any factor. Item cross loadings were: A11, A13, A14, A18, B3, B4, C2, C5, F1, F2, and F8. A11 cross loaded on factors three and nine. A13 cross loaded on factors four and five. A14 cross loaded on factors six and nine. A18 cross loaded on factors four and seven. B3 cross loaded on factors five and nine and eleven. B4 cross loaded on factors eight and 11. C2 cross loaded on factors one and four and seven. F2 cross loaded on factors one and seven. F8 cross loaded on factors four and seven. Table 6-21 shows the summary of the 11 factor solution and the factor loadings.

Factor	Items	loading (<.4)	Strongly loading( >.4)
number			
1	F1-F2-F3-F5-F6-F7-F9-F11	F1-F3-F9	F5-F2-F6-F7-F11
2	D1-D2-D3		D1-D2-D3
3	A1-A3-A4-A11	A11	A1-A3-A4
4	A6-A9-A13-A18-F1-F8	A18-F1-F8	A6-A9-A13
5	A13-B3-C1-C2-C3-C5	A13-B3-C2	C1-C3-C5-
6	A8-A12-A14-A16	A14	A8- A12 -A16
7	A18-F1-F2-F4-F8-F10	A18-F1	F2-F4-F8-F10
8	B1-B2-B4	B4	B1-B2
9	A2-A5-A7-A11-A14-B3	A2-A11-A14-B3	A5-A7
10	C2-C4-C5-C6	C5	C2-C4-C6
11	B3-B4	B3	B4

Table 6-21 Structure of the initial solution (11 factors) and its loading

The 11 factor solution contained several items with loading less than 0.4. These items were F1-F3-F9-A11-A18-A14-A2-B3. Some items cross loaded (had loading < 0.4 on one factor but loadings > 0.4 on another factor such as A13-C2-C5-F8-B4). Factor number 11 has only one item loading heavily (which might mean too many factors were extracted). In summary this solution provides ten clear factors with quite strong multiple items loading (> 0.4) on each factor (Pallant, 2007). It is usual to regard factor loadings as high if they are greater than 0.6 (the positive or negative sign is irrelevant) and moderately high if they are above 0.3. Other loading can be ignored (Kline, 1994).

In summary, the initial solution was unclear because of cross loadings of many items. In addition, several items had loading less than 0.4, factor number 11 has only one item loading heavily, two factors (factor one and seven) were not theoretically related and the 11 factors solution accounts for 46.8% of total variance. Because of these findings and the lack of clarity of this initial solution, EFA on 42 items with 11 factors and oblique rotation was undertaken to aid interpretation of the items in each factor of the solution. The following pattern matrix shows the structure and loadings of the 11 factor solution.

## Table 6- 22 Pattern matrix of the initial solution (11 factors) (n=431)

Safety climate items						Factor					
	1	2	3	4	5	6	7	8	9	10	11
C3 We are informed about errors that happen in this unit	.45						191620	-		12.55	
C1 We are given feedback about changes put into place based on event reports	.42						01409 1. 0.5			And a second	
A13 After we make changes to improve patient safety, we evaluate their effectiveness	.41						to by			Trail (	100
A8 Staff feel like their mistakes are held against them		.79						and a	lone	turner.	12 64
A12 When an event is reported, it feels like the person is being written up, not the problem		.65				16.21		Arts	a. 6		
A16 Staff worry that mistakes they make are kept in their personnel file		.49						r la la	of he		
A14 We work in a hurry, trying to do too much, too quickly		.34					Renta Renta	31	be vi		
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?	11000		97			101 21	smel.	12 11	2ando Tipis		12.815
D3 When a mistake is made that could harm the patient, but does not, how often is this			74		AN I I			a den		1.201	
D1 When a mistake is made, but is caught and corrected before affecting the patient, how often			- 73				- Charles	1 Seice	(be		
is this reported?	othe	ate	15	- ha	17	Nº S	2014	19 Edit			
F7 Problems often occur in the exchange of information across hospital units			1.591	.64				181	CK3	e lare	
F5 Important patient care information is often lost during shift changes	1010		n ne	.57	10.53	2.5	and the	in the second	112	2 hees	
F6 It is often unpleasant to work with staff from other hospital units			is a	.52		1000					
F11 Shift changes cause problems for patients in this hospital	11-		199	.46		100	Ser-	m	199		
F3 Some things do not happen or get missed when transferring patients from one unit to another	2,81		261	312		- Chart	ites Corre				
A17 We have patient safety problems in this unit	803		281			EX.					
A4 In this unit, people treat each other with respect	187		82	68	.77	40	-14		14		
A1 Staff support one another in this unit				PIL C	.67				i Chai		
A3 When a lot of work needs to be done quickly, we work together as a team to get the work					.57						
C4 Staff feel free to question the decisions or actions of those with more authority						.69					
C2 Staff will freely speak up if they see something that may negatively affect patient care						.55					
C6 Staff are afraid to ask questions when something does not seem right						41					
C5 In this unit, we discuss ways to prevent errors from happening again	.30					.32					
B1 My supervisor/manager says a good word when he/she sees a job done according to							- 87		1		
established patient safety procedures											
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety							73			-	
A5 Staff in this unit work longer hours than they should, which is not good for patient care								48			
A7 We use more temporary staff than we should, which is not good for patient care								46			

Table 6-22 ( continued)											
B3 Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it means taking shortcuts						- 51 		33			
A2 We have enough staff to handle the workload					2.00		~	.32			
A11 When one area in this unit gets really busy, others help out		-		- feelin				.30			
B4 My supervisor/manager ignores patient safety problems that happen over and over			-				.33		.56		
A10 It is just by chance that more serious mistakes do not happen around here				in the			1				
F10 Hospital units work well together to provide the best care for patients						1.6.2	3			69	
F4 There is good cooperation among hospital units that need to work together						1				63	
F8 The actions of hospital management show that patient safety is a top priority						32				49	
F2 Hospital units do not coordinate well with each other	in.									.43	
A18 Our procedures and systems are good at preventing errors from happening										31	
F1 Hospital management provides a work climate that promotes patient safety				1					1.45		
A6 We are actively doing things to improve patient care				1			100 100			× 6. 4.	48
A9 Mistakes have led to positive changes here						Just v		in a s	5.471	No. 19 March	46
F9 Hospital managemnet seems interested in patient safety only after an adverse event happens		a les rects	1		3 	21.1 32:043		15.3	dit se te const		
A15 Patient safety never takes second place to get more work done	25.). 19	il. Sezel			1.000		in stra Stancesi	i fa mai	ary mas	*	

Extraction Method: Principal Axis Factoring. Rotation Method: Oblimin with Kaiser Normalization Rotation converged in 20 iterations

The pattern matrix above indicated that items A10-A15-A17-F3-F1-F9 did not load upon any factor. Items A11-A14-A18-A2-B3-C5 had low loadings. The next step excluded A10-A15-A17-F3-F1-F9 because they did not load upon any factor.

Step 2: EFA was then run excluding items A10-A15-A17-F3-F1-F9, with 11 factors and oblique rotation to aid interpretation of the items in each factor of the solution (Waterson et al., 2009).

The extraction column of the communalities table (see appendix 24) shows that item A2 was very poorly explained (0.243), while item A5 was now above 0.3. The table of the Total Variance Explained of the initial solution shows that the 11 factors account for 50.8% of total variance (see appendix 25). The pattern matrix of the initial solution in this step indicated that items A2-A11 did not load upon any factor. Items number A14-A18-C5 had low loadings (see appendix 26). The next step excluded A2-A11-A14-A18-C5.

Step 3: After excluding items A10-A15-A17-F3-F1-F9-A2-A11-A14-A18-C5 EFA was then run again with 11 factors and an oblique rotation was carried out to aid interpretation of the items in each factor of the solution.

The communalities table shows that all the 31 safety climate items had high communalities (see appendix 27). Item A7 was close to 0.3 (0.285). The total variance explained by this step shows that the 11 factors account for 52.9 % of total variance (see appendix 28). The pattern matrix of the initial solution shows that this solution included 31 safety climate items in 11 factors while 11 items were excluded (A2-A10-A11-A14-A15-A17-A18-C5-F1-F3-F9). Item loadings were between 0.37 and 0.97. There were only small shifts among items across factors. For instance, A13 moved to 'Feedback and communication about error', F6 moved to 'Handoffs and transitions', and F8 moved to 'Teamwork across units'. All dimensions consisted of two to four safety climate items (see appendix 29). Table 6-23 shows the final structure of the initial solution.

190

Factor	Factor	Items loading(0.37-	Number
number		0.97)	of items
1	Feedback and communication about error	C1-C3-A13	3
2	Frequency of event reported	D1-D2-D3	3
3	No punitive response to error	A8-A12-A16	3
4	Handoffs and transitions	F5-F6-F7-F11	4
5	Teamwork within units	A1-A3-A4	3
6	Communication openness	C2-C4-C6	3
7	Supervisor expectation and action promoting patient safety	B1-B2	2
8	Staffing	A5-A7	2
9	Teamwork across units •	F2-F4-F10-F8	4
10	Organisational learning- continuous improvement	A6-A9-	2
11	Supervisor expectation and action promoting patient safety negatively attitude	B3-B4	2

Table 6-23 Final structure of the initial solution (11 factors)

The composition of the factors of the initial solution was very similar to that of the USA HSOPSC questionnaire (Sorra and Nieva, 2004) because the main part of the factor structure was unchanged. Although the initial solution was clear and appropriate, use of a combination of Kaiser's criterion and Cattell's screen plot (Scree plot) method was also required to assess the number of factors to be extracted (Conway et al., 2003). Thus, the Scree plot was examined in order to identify the optimal number of factors that should be extracted, to explore potential improvements by examining a series of possible solutions and to identify common factor structure across different solutions of the exploring approach. Figure 6-3 shows the Scree plot of the initial solution.





Scree Plot

Cattell (1966) recommended retaining all factors above the elbow or break in the plot, as these factors contribute the most to the explanation of the variance in the data set. The breaks in the Scree plot suggest three break points. Only factors to the left of the point of the break should be retained (Field, 2009). The first break point corresponds to factor number 12. Counting back one factor means that the actual break point was 12, so the 11 factor solution should be tested. The second break point corresponds to factor number 11. Counting back one factor means that the actual break point was 11 then a 10 factor solution should be tested. The third break point corresponds to factor number 9. Counting back one factor means that the actual break point should be also tested.

In fact, the Scree plot in figure 6-3 was not very clear because there was more than one obvious break and it seems just a gradual curve. Therefore it was very important to examine all possible realistic solutions such as, 12 factors because it is the original structure of the questionnaire, and also 9 factors for reasons of completeness and also because it was suggested by another validation study (Waterson et al., 2009). All of the possible solutions should be investigated because EFA is an exploratory approach and experimentation with different number of factors should be continued until a satisfactory solution is found (Pallant, 2007).

The decision was made that all these possibilities; 12-11-10-9-8 factor solutions should be investigated in order to find how many factors should be extracted as the most appropriate solution and also to explore loading of each item and variance explained in each solution. The best solution is:

- 1 One where it makes theoretical sense (items in one factor theoretically related), it explains the most total variance, each item is explained well by this number of factors, the rotated solution has no low loading items.
- 2 All items loading upon factors.
- 3 No cross- loading items or cross loading items are heavily in one factor and low on another.
- 4 No free- standing items (one item in one factor).
- 5 The most items loading should be quite strongly (close or above 0.4) in the final model.
- 6 Three items and above load up on each factor is the best but two items are acceptable especially if they are very high loadings items (above 0.5) and theoretically related.

All these possible (11-12-10-9-8) factor solutions were investigated. Each solution is presented below: Step 4 investigated the 11 factor solution, step 5 investigated the 12 factor solution, step 6 investigated the 10 factor solution, step 7 investigated the 9 factor solution and step 8 investigated the 8 factor solution.

#### Step 4: The 11 factor solution

EFA was run on all safety climate items of HSOPSC (42 items), Principal Axis Factoring was the chosen extraction method. Varimax was the chosen rotation method and 11 factors were used instead of eigenvalues > 1.

#### Output of the 11 factor solution:

The output of the 11 factors including the communalities, the total variance explained table and rotated factor matrix were the same output of the initial solution above in step 1 (pages: 181-186). To avoid repetition it is not presented twice.

#### Step 5: The 12 factor solution

EFA was run on all safety climate items (42 items) with using varimax rotation and number of factors (12).

#### **Output of the 12 factor solution:**

The output of the 12 factor solution includes the communalities, the total variance explained table and rotated factor matrix. The extraction column of the communalities table of the 12 factor solution shows the proportion of variance of each variable explained by 12 factors solution. Six items, specifically A2, A5, A10, A15, A17, and F3 are very poorly explained (very poor communalities, < 0.3) (see appendix 30). The total variance explained table shows that 12 factor solution accounts for 48% of total variance (see appendix 31).

The rotated factor matrix of the 12 factor solution shows the loadings of 12 factors with multiple items on each factor. Items A17 and F3 did not load upon any factor. Items cross loadings were A2, A11, A13, A14, A18, B4, C2, F1, F2, F8, and F9. A2 cross load on factor four and factor nine. A11 cross load on factors four, nine and 11. A13 cross loaded on factors three and seven. A14 cross loaded on factors six and nine. A18 cross loaded on factors five and seven. B4 cross loaded on factors five and seven. F2 cross loaded on factors one and five. F8 cross loaded on factors five and seven. F9 cross loaded on factors one and seven (see appendix 32). Table 6-24 shows the structure of the 12 factor solution and item loadings.

Factor	Items	loading (<0.4)	Ileavily loading( >0.4)
number			·····
1	F2-F5-F6-F7-F9-F11	F2-F9	F5-F6-F7-F11
2	D1-D2-D3		D1-D2-D3
3	C1-c2-C3-C5-A13	C2	C1-C3-C5-A13
4	A1-A2-A3-A4-A11	A2-A11	A1-A3-A4
5	F1- F2-F4-F8 -F10-A18	F1-A18	F2-F4-F8-F10
6	A8- A12 - A14-A16	A14	A8- A12 -A16
7	A6-A9-A13-A15-A18-F1-F8-F9	A15-A18-F1-F8-F9	A6-A9-A13
8	B1-B2-B4	B4	B1-B2
9	A5-A7-A14-B3-A11-A2-	A2-A11-A14	А5-А7-В3
10	C2-C4-C6		C2-C4-C6
11	A10-A11-B4	A10-A11-B4	
12	B4	B4	· · · · · · · · · · · · · · · · · · ·

Table 6-24 Structure of the 12 factors solution and item loadings

The 12 factor solution contained several items with loading less than 0.4. These items were F1-F9-A2-A10-A11-A14-A15-A18-B4. Factors number 11 and 12 had no items loading >0.4, which means there are too many factors in this solution. In summary, this solution provided ten clear factors with multiple items loading heavily on each factor.

#### Step 6: The 10 factor solution

EFA was run on all safety climate items (42 items) using varimax rotation and number of factors (10).

#### Output of the 10 factor solution:

The output of the 10 factors including the communalities; the total variance explained table and rotated factor matrix are provided in appendix. The extraction column of the communalities table of the 10 factor solution shows the proportion of variance of each variable explained by the 10 factors solution. Seven items, specifically A2, A5, A7, A10, A15, A17, and F3 are very poorly explained (very poor communalities, <0.3) (see appendix 33).

The total variance explained table shows the 10 factors solution accounts for 45.5% of total variance (see appendix 34). The rotated factor matrix shows the loadings of the 10 factors with multiple items on each factor. Items A10, A15, A17 and F3 did not load upon any factor. Items cross loadings were A3, A11, A13, A14, B3, C2, C3, C5, F2 and F10.

A3 cross loaded on factor one and factor four. A11 cross loaded on factors four and seven. A13 cross loaded on factors one and eight. A14 cross loaded on factors five and seven. B3 cross loaded on factors seven and ten. C2 cross loaded on factors eight and nine. C3 cross loaded on factors one and eight. C5 cross loaded on factors one and eight. F2 cross loaded on factors one and factor two. F10 cross loaded on factors one and factor two (see appendix 35). Table 6-25 shows the structure of the 10 factor solution and item loadings.
Factor	Items	loading (<0.4)	IIeavily loading( >0.4)
number			
1	A6-A13-A18-F8-F10-F1-F4-A9-	A3-C3-F2-F9	A6-A13-A18-F8-F10-F1-
	C5-A3-C3-F2-F9		F4-A9-C5
2	F2-F5-F6-F7-F11-F10	F10	F2-F5-F6-F7-F11
3	D1-D2-D3		D1-D2-D3
4	A1-A3-A4-A11	A11	A1-A3-A4
5	A8- A12 - A14-A16	A14	A8- A12 -A16
6	B1-B2-		B1-B2
7	A5-A7-A14-B3-A11-A2-	A2-A11-A14-B3	A5-A7
8	C3-C1-C5-A13-C2	C2	C3-C1-C5-A13
9	C2-C4-C6		C2-C4-C6
10	B3-B4	B3	B4

Table 6-25 Structure of the 10 factor solution and its loading

The 10 factor solution contained several items with loading less than 0.4. These items were F9-A11-A14-A2-B3. Some items had loading <0.4 but loadings >0.4 on different factors such as A3-C3-F2-F10-C2. Factor number 10 had one item loading >0.4. In summary this solution provided nine clear factors with multiple items loading heavily on each factor. Factor number one included nine items that had loadings >0.4. This factor might be divided into two factors meaning this solution included 10 factors.

EFA was re run on all safety climate items (42 items) with using oblique rotation and number of factors (10). In this step the rotation was changed into oblique to make sure that factor number 10 was loading by one item. The result was given by oblique rotation indicated that the factor number 10 was loading heavily by one item. Therefore, this solution provides nine factors with multiple items loading heavily on each factor.

#### Step 7: The 9 factors solution

EFA was run on all safety climate items (42 items) using varimax rotation and number of factors (9).

#### Output of the 9 factor solution:

The output of the 9 factors includes the communalities, the total variance explained table and rotated factor matrix. The extraction column of the communalities of the 9 factors solution table shows the proportion of variance of each variable explained by 9 factors solution. Seven items, specifically A2, A5, A7, A10, A15, A17, C6, and F3 are very poorly explained (very poor communalities, < 0.3) (see appendix 36).

The Total variance explained table shows the 9 factor solution accounts for 44.2% of total variance (see appendix 37). The rotated factor matrix table of the 9 factor solution shows the loadings of 0 9 factors with multiple items on each factor. Items A10, A15, A17 and F3 did not load upon any factor. Items cross loadings were A3, A11, A14, B1, B2, B3, C2, C5, F1, F4, F9 and F10. A3 cross loaded on factors one and four. A11 cross loaded on factors one and six. A14 cross loaded on factors five and six. B1 cross loaded on factors one and eight. B2 cross loaded on factors one and eight. B3 cross loaded on factors six and nine. C2 cross loaded on factors one and seven. C5 cross loaded on factors one and seven. F1 cross loaded on factors one and two. F4 cross loaded on factors two and nine. F9 cross loaded on factors one and two. F10 cross loaded on factors one and two (see appendix 38). Table 6-26 shows the structure of the 9 factor solution and item loadings.

Factor	Items	loading (<0.4)	Heavily loading(>0.4)
number	······		
1	A3-A6-A9-A11-A13-A18-B1-	A3-A11-B1-B2-B4-C2-	A6-A9-A13-A18-C1-
	B2-B4-C1-C2-C3-C5-F1-F8-F9-	F1-F9	C3-C5-F8-F10
	F10		
2	F1-F2-F4-F5-F6-F7-F9-F10-F11	F1-F4-F9	F2-F5-F6-F7-F10-F11
3	D1-D2-D3		D1-D2-D3
4	A1-A3-A4		A1-A3-A4
5	A8-A12-A14-A16	A14	A8-A12-A16
6	A2-A5-A6-A7-A11-A14-B3	A2-A11-	A5-A7-A14-B3
7	C2-C4-C5-C6	C5	C2-C4-C6
8	B1-B2		B1-B2
9	B3-F4	B3	F4

Table 6-26 Structure of the 9 factors solution and its loading

The 9 factor solution contained several items with loading less than 0.4. These items were A2-A5-A11-B3-C5-F1-F9. Some items had loading <0.4 but loadings >0.4 on different factors such as A3-A14-F4-C2. Factor number 9 have one item loading >0.4. In summary this solution provided nine factors with multiple items loading heavily on each factor. Factors number one consists of nine items loading >0.4. This factor might be divided into two factors. Factors number two consists of six items loading >0.4. This factor might be divided into two factors.

EFA was re run on all safety climate items (42 items) using oblique rotation and number of factors (9) to check whether the number of extracted factors had changed or not. In this step the rotation was changed into oblique to make sure that factor number 9 was loading by one item. The result given by oblique rotation indicated that factor number 9 was still loading heavily by one item. However, this solution provided eight factors with multiple items loading heavily on each factor.

#### Step 8: The 8 factor solution

EFA was run on all safety climate items (42 items) using varimax rotation and number of factors (8).

#### Output of the 8 factor solution:

The output of the 8 factors solution includes the communalities, the total variance explained table and rotated factor matrix. The extraction column of the communalities table of the 8 factors solution shows the proportion of variance of each variable explained by 8 factors solution. Seven items, specifically A2, A5, A7, A10, A15, A17, C6, and F3 are very poorly explained (very poor communalities, < 0.3) (see appendix 39). The Total variance explained table of the 8 factors solution shows 8 factors solution accounts for 42.4% of total variance (see appendix 40).

The rotated factor matrix of the 8 factor solution shows the loadings of 8 factors with multiple items on each factor. Items A10, A15, A17 and F3 did not load upon any factor. Items cross loadings were: A3, A11, A14, B1, B2, B3, B4, C2, C5, F1, F2, F4, F9 and F10. A3 cross loaded on factors one and four. A11 cross loaded on factors one and four and seven. A14 cross loaded on factors five and seven. B1 cross loaded on factors one and eight. B2 cross loaded on factors one and eight. B3 cross loaded on factors seven and eight. B4 cross loaded on factors one and eight. C2 cross loaded on factors one and six. C5 cross loaded on factors one and two. F1 cross loaded on factors one and two and four. F9 cross loaded on factors one and two structure of the 8 factor solution and its item loadings.

Factor	Items	loading (<.4)	IIeavily loading( >.4)
number		·····	
1	A3-A6-A9-A11-A13-A18-B1-	A3-A11-B2-B4-C2-F1-	A6-A9-A13-A18-B1-
	B2-B4-C1-C2-C3-C5-F1-F2-F4-	F2-F4-F9	C1-C3-C5-F8-F10
	F8-F9-F10		
2	F1-F2-F4-F5-F6-F7-F9-F10-F11	F1-F4-F9	F2-F5-F6-F7-F10-F11
3	D1-D2-D3		D1-D2-D3
4	A1-A3-A4-A11-F4	A11-A14	A1-A3-A4
5	A8-A12-A14-A16	A14	A8-A12-A16
6	C2-C4-C5-C6	C5-C6	C2-C4
7	A2-A5-A7-A11-A14-B3	A2-A11	A5-A7-A14-B3
8	B1-B2-B3-B4-		B1-B2-B3-B4

Table 6-27 Structure of the 8 factor solution and its item loadings

It is clear that the 8 factor solution contained several items with loading less than 0.4. These items were A2-A11-C5-C6-F1-F9. Some items had loading <.4 but loadings >0.4 on different factors such as A3-A14-B2-B4-C2. In summary this solution provided eight clear factors with multiple items loading heavily on each factor. Factor number one consisted of ten items loading > 0.4. This factor might be divided into two factors. Factor number two consists of six items loading > 0.4. This factor might be divided into two factors.

EFA was re run on all safety climate items (42 items) with using oblique rotation and number of factors (8) to check whether the number of extracted factors had changed or not. The result was given by oblique rotation indicated that the same number of extracted factors (8 factors). Therefore, this solution provided eight clear factors with multiple items loading heavily on each factor.

#### Conclusion from the different factor solutions (11-12-10-9-8)

The investigation of the five solutions in steps number 4-5-6-7-8 aimed to identify the number of factors that could be extracted. The 11 factors solution shows that the 11 factors seems the most logical fit to this set of items. The 12 factors solution shows that the 10 factors seems the most logical fit to this set of items. The 10 factors solution shows that the nine factors seems the most logical fit to this set of items. The 9 and 8 factor solutions show that the 8 factors seems the most logical fit to this set of items. Items number A10-A15-A17-F3 had no loading upon any factor. Table 6-28 shows the summary of the five solutions.

Steps of	Solution	Number of	Total variance explained %	Items not loading
E.F.A		factors		
Step 4	11	11	46.8	A10-A15-A17
Step 5	12	10	48	A17-F3
Step 6	10	9	45.5	A10-A15-A17-F3
Step 7	9	8	44.2	A10-A15-A17-F3
Step 8	8	8	42.4	A10-A15-A17-F3

Table 6-28 Summary of the five solutions

Table 6-29 shows the low communalities and loading items in each solution.

Steps	of Solution	Items with low communalities	Items with low loading (< 0.4)
E.F.A		(< 0.3)	
Step 4	11	A2-A5-A10-A15-A17-F3	F1-F3-F9-A11-A18-A14-A2-B3
Step 5	12	A2-A5-A10-A15-A17-F3	F1-F9-A11-A10-A14-A15-A18-A2-
			B4
Step 6	10	A2-A5-A7-A10-A15-A17-F3	F9-A11-A14-A2-B3
Step 7	9	A2-A5-A10-A15-A17-F3-C6	A2-A5-A11-B3-C5-F1-F9
Step 8	8	A2-A5-A7-A10-A15-A17-C6-F3	A2-A11-C5-C6-F1-F9

Table 6- 29 Low communalities and loading items in each solution

The items that did not load upon any factor were A10-A15-A17-F3 across most of the proposed solutions and items A2-A5-A10-A15-A17-F3 had low communalities across most the proposed solutions. Items A2-A11-F1-F9 had low loading (< 0.4). It is important to notice that A10-A15-A17 were all from one dimension (overall perceptions of patient safety) while A2-A5 were from another dimension (staffing), F1-F9 were from another dimension (management support for patient safety) and F3 was from different dimension (handoffs and transitions).

It is important to mention that extra investigation (EFA and CFA) was performed in order to explore potential improvements and because exploratory factor analysis is based on exploration approach. Therefore, this investigation was performed by exploring the comparison between the 11-12-10-9-8 factors solutions that produced the common factor structure across these solutions. Although the eight factors seems the most appropriate number in terms of how many factors that should be extracted, it seems that the common factor structure which consists of 10 factors should be examined. However, the result of this extra investigation also confirmed that the eight factor solution is the optimal model of the current study (see appendix 47).

In conclusion the results of the comparison above between the five solutions suggested 10 or 9 or 8 factor solution. However, the 8 factors seems the most appropriate number in terms of how many factors that should be extracted because it was indicated in the Scree plot and it was recommended by the 9 factors solution and the 8 factors solution. Each solution has a number of cross loadings items; hence orthogonal rotation was used to produces more easily interpretable results (Costello and Osborne, 2005). However, rotation cannot improve the basic aspects of the analysis, such as the amount of variance extracted from the items (Costello and Osborne, 2005). Therefore, the 8 factors were investigated by using oblique rotation to obtain the optimal solution.

## Step 9:

EFA was run on all safety climate items (42 items) with using an oblique rotation and number of factors (8).

## **Output:**

As mentioned earlier the output of the 8 factors including the communalities, the total variance explained were the same output in step 8. The pattern matrix of the 8 factors solution shows the loadings of 8 factors with multiple items on each factor. Items A10, A15, A17, F3 and F4 did not load upon any factor (see appendix 42). The 8 factor solution contained several items with loading less than 0.4. These items were A2-A11-A14-A18-B3-B4-C1-C3-C6-F1-F9-F10. Therefore, these 17 items were excluded on the next step.

## Step 10:

EFA was run excluding A10-A15-A17-F3-F4-A2-A11-A14-A18-B3-B4-C1-C3-C6-F1-F9-F10 by using oblique rotation and number of factors (8).

## **Output:**

The output of the 8 factors included the communalities, the total variance explained table and pattern matrix. The extraction column of the communalities table of the 8 factors solution shows the proportion of variance of each variable explained by 8 factors solution. Item A7 was very poorly explained (very poor communalities, < 0.3) (see appendix 43). The total variance explained table of the 8 factors solution shows 8 factor solution accounts for 51.2% of total variance (see appendix 44).

The pattern matrix shows the loadings of the 8 factors with multiple items on each factor. Item number C5 cross loaded on factors one and six. Item number F2 had loading less than 0.4 (see appendix 45). Therefore, these two items (C5-F2) were excluded on the next step.

#### Step 11: The final 8 factor EFA solution

EFA was run excluding 19 items (A10-A15-A17-F3-F4-A2-A11-A14-A18-B3-B4-C1-C3-C5-C6-F1-F9-F10-F2) with using oblique rotation and number of factors (8).

#### Output of the final 8 factor EFA solution

The output of the final 8 factor EFA solution included the communalities, the total variance explained table, Scree plot and pattern matrix. Table 6-30 shows the communalities of the final 8 factor EFA solution.

Safety climate items	Initial	Extraction
A1 Staff support one another in this unit	.48	.66
A3 When a lot of work needs to be done quickly, we work together as a	.41	.51
A4 In this unit, people treat each other with respect	.45	.60
A5 Staff in this unit work longer hours than they should, which is not good	.15	.28
A6 We are actively doing things to improve patient care	.37	.47
A7 We use more temporary staff than we should, which is not good for	.17	.24
A8 Staff feel like their mistakes are held against them	.41	.61
A9 Mistakes have led to positive changes here	.31	.34
A12 When an event is reported, it feels like the person is being written up,	.40	.54
A13 After we make changes to improve patient safety, we evaluate their	.38	.49
A16 Staff worry that mistakes they make are kept in their personnel file	.26	.35
B1 My supervisor/manager says a good word when he/she sees a job done	.57	.70
B2 My supervisor/manager seriously considers staff suggestions for	.60	.77
C2 Staff will freely speak up if they see something that may negatively	.40	.51
C4 Staff feel free to question the decisions or actions of those with more	.31	.60
D1 When a mistake is made, but is caught and corrected before affecting the	.52	.56
D2 When a mistake is made, but has no potential to harm the patient, how	.64	.89
D3 When a mistake is made that could harm the patient, but does not, how	.50	.52
F5 Important patient care information is often lost during shift changes	.34	.43
F6 It is often unpleasant to work with staff from other hospital units	.30	.37
F7 Problems often occur in the exchange of information across hospital	.36	.57
F8 The actions of hospital management show that patient safety is a top	.29	.34
F11 Shift changes cause problems for patients in this hospital	.31	.40

## Table 6- 30 Communalities of the final 8 factor EFA solution (n=431)

Extraction Method: Principal Axis Factoring

		Initial Eigen	values	Extra	Extraction Sums of Squared					
Factor	Total	% of	Cumulative %	Total	% of	Cumulative %	Total			
1	5.65	24.6	24.6	5.21	22.6	22.6	3.0			
2	2.04	8.8	33.4	1.67	7.2	29.9	2.8			
3	1.91	8.3	41.8	1.41	6.1	36.1	1.9			
4	1.47	6.4	48.2	.98	4.2	40.4	2.3			
5	1.29	5.6	53.8	.89	3.9	44.3	2.5			
6	1.20	5.2	59.0	.65	2.8	47.1	2.3			
7	1.02	4.4	63.5	.56	2.4	49.6	2.9			
8	0.93	4.0	67.6	.44	1.9	51.5	1.3			
9	.79	3.4	71.0							
10	.72	3.1	74.1							
11	.70	3.0	77.2							
12	.63	· 2.7	79.9							
13	.56	2.4	82.4							
14	.53	2.3	84.7							
15	.51	2.2	86.9							
16	.47	2.0	89.0							
17	.45	1.9	91.0							
18	.44	1.9	92.9							
19	.42	1.8	94.7		i.					
20	.38	1.7	96.5							
21	.33	1.6	97.1							
22	.25	1.2	· 99.4							
23	.22	.5	100							

Table 6- 31 Total variance explained of the final 8 factor EFA solution (n=431)

The total variance explained table shows the 8 factors solution accounts for 51.5% of total variance. Figure 6-4 shows the Scree plot of the final 8 factor EFA solution.





The break in this Scree plot suggested the eight factor solution. The following table shows the pattern matrix of the final 8 factor solution.

Control States

#### Factor Safety climate items 1 2 3 4 5 6 7 8 A6 We are actively doing things to improve patient care .69 A13 After we make changes to improve patient safety, we evaluate their effectiveness .65 F8 The actions of hospital management show that patient safety is a top priority .40 A9 Mistakes have led to positive changes here .42 D2 When a mistake is made, but has no potential to harm the patient, how often is this reported? .98 D1 When a mistake is made, but is caught and corrected before affecting the patient, how often is this reported? .72 D3 When a mistake is made that could harm the patient, but does not, how often is this reported? .77 A8 Staff feel like their mistakes are held against them .73 A12 When an event is reported, it feels like the person is being written up, not the problem .60 A16 Staff worry that mistakes they make are kept in their personnel file .46 F7 Problems often occur in the exchange of information across hospital units .75 F6 It is often unpleasant to work with staff from other hospital units .56 F5 Important patient care information is often lost during shift changes 58 F11 Shift changes cause problems for patients in this hospital .46 A1 Staff support one another in this unit -.71 A4 In this unit, people treat each other with respect -.71 A3 When a lot of work needs to be done quickly, we work together as a team to get the work done -.50 C4 Staff feel free to question the decisions or actions of those with more authority .75 C2 Staff will freely speak up if they see something that may negatively affect patient care .55 B2 My supervisor/manager seriously considers staff suggestions for improving patient safety -.76 B1 My supervisor/manager says a good word when he/she sees a job done according to established patient safety -.71 procedures A5 Staff in this unit work longer hours than they should, which is not good for patient care A7 We use more temporary staff than we should, which is not good for patient care

#### Table 6- 32 Pattern matrix of the final 8 factor EFA solution (n=431)

Extraction Method: Principal Axis Factoring. Rotation Method: Oblimin with Kaiser Normalization a. Rotation converged in 11 iterations.

Factor	Factor	Items	Number
number		(loading > 0.4)	oritems
1	Organisational learning- continuous improvement	A6-A9-A13-F8	4
2	Frequency of event reported	D1-D2-D3	3
3	No punitive response to error	A8-A12-A16	3
4	Handoffs and transitions	F5-F6-F7-F11	4
5	Teamwork within units	A1-A3-A4	3
6	Communication openness	C2-C4	2
7	Supervisor expectation and action promoting patient	B1-B2	2
	safety		
8	Staffing	A5-A7	2

Table 6-33 shows the structure and factor loadings of the final EFA solution (8 factors).

	Fable	e 6-	33	Str	ucture	of	the	final	<b>EFA</b>	solution
--	-------	------	----	-----	--------	----	-----	-------	------------	----------

The final EFA solution (8 factors) accounted for 51.5% of total variance and consisted of eight factors representing 23 safety climate items. The solution included 23 safety climate items in 8 factors while 19 items were excluded (A2-A10-A11-A14-A15-A17-A18-B3-B4-C1-C3-C5-C6-F1-F2-F3-F4-F9-F10). Factor loadings were between 0.43 and 0.97. There were only small shifts among items across factors for instance, F6 moved from "teamwork across units" to handoffs and transitions. F8 moved from "management support for patient safety" to organisational learning-continuous improvement. All dimensions consisted of two to four safety climate items. The structure of the 8 factor final EFA solution makes theoretical sense, as all 23 safety climate items that are theoretically related were grouped together as a factor. There was no cross-loading. There were no free - standing items (just one item in one factor). All the 23 safety climate items were loading quite strongly (loading > 0.4).

The best solution is one where it makes theoretical sense (items in one factor theoretically related), it explains the most total variance, each item is explained well by this number of factors, the rotated solution has no low-loading items and all items load upon factors. There is no cross-loading items (item is heavily in one factor and low on another). There is no free -standing item (one item in one factor).

The most items loading quite strongly (above 0.4) in the final model. Three items and above upon each factor is the best but two items are acceptable especially if they are very high loading items (above 0.5) and theoretically related. The next step investigated each factor of the final 8 factor EFA solution.

#### EFA of each potential scale of the final solution (8 factors solution)

This step aimed to check the final solution (8 factors) derived from these exploratory factor analyses. Therefore, conducting an exploratory factor analysis on each scale (factor) to check each set of items is satisfactorily explained by a single factor as structured in the final solution in table 6-33. In the other words this step aimed to make sure that these items for the scale only load onto one scale. In general, the reason for this extra EFA is to check the overall (final) EFA of the final solution. The output of this step showed that each set of items was loaded on one factor as they were structured in the final solution in table 6-33. Overall, all items (23 items) and 8 factors were satisfactorily explained by a single factor (the output of this step is provided in appendix 46). The next section concerns with testing the fit of the optimal model (8 factors).

## 6.5.2 Testing the fit of the optimal model: CFA

This step aimed to test, using the validation half of the data, the fit of the optimal model (eight factors) that emerged from the EFA that was performed on the construction half of the data.

The process of CFA analysis of the eight factors solution was followed:

- 1 AMOS Graphics was used to open up a blank diagram window
- 2 Opening the data (the validation half of the data, 431 questionnaires)
- 3 Building the model (drawing the eight factors by using the model drawing area in step one)
- 4 Adding the variable names to the model
- 5 Running the model and using the output window

Table 6-34 shows the output of CFA of 8 factors solution.

Table 6-	Table 6- 34 Output of CFA of 8 factors solution (n=431)											
Model	Chi-	DF	CMIN/DF	CFI	RMSEA	RMR	SRMR	TLĪ				
	square											
Eight	407	202	2	0.93	0.04	0.04	0.04	0.92				
Factors	(good)	(good)	(good)	(good)	(good)	(good)	(good)	(good)				

The output in table 6-34 above shows that this model achieved good fit, Chi-square ( $\chi^2$ ) (202) = 407, P< 0.001; CMIN/DF = 2 (good model  $\leq 2$ ), CFI = 0.93 (CFI > 0.90); RMSEA = 0.049 (RMSEA < 0.06); SRMR = 0.047 (SRMR < 0.08). Overall, the fit indices from CFA of eight factors indicated that this model fit was good. Furthermore, the fit of the optimal model (eight factors) was tested for whole data (862) to confirm it as optimal model of the current study. Table 6-35 shows the output of the CFA testing the fit of optimal models for whole data (862 questionnaires).

Chapter six

#### Testing the fit of optimal model for whole data

Model	Chi-	DF	CMIN/DF	CFI	RMSEA	RMR	SRMR	TLI
Eight	553	202	2	0.94	0.04	0.03	0.04	0.93
factors	(good)	(good)	(good)	(good)	(good)	(good)	(good)	(good)

 Table 6- 35 Output of the CFA testing for whole data (n=862)

The values of the fit indices of the Saudi data (8 factors) meet the general agreed parameters in table 6-15 page 174. Therefore, this model fit was a good fit. Consequently this model was acceptable as the optimal model of the current study. Moreover, the comparison of CFA of USA and UK models and the optimal model (8 factors) of the current study was performed in order to check the results of the CFA of the optimal model of the current study. Table 6-36 presents the comparison of CFA of the USA (Sorra and Nieva, 2004) and the UK (Waterson et al., 2009) models and the optimal model (8 factors) of the current study.

Model	Chi-	DF	CMIN/DF	CFI	RMSEA	RMR	SRMR	TLI
	square							
USA	2064	746	-	0.94	0.04	-	-	-
UK	587	288	-	0.94	0.04	<b>-</b>	0.04	0.93
The current study	553	202	2	0.94	0.04	0.03	0.04	0.93

 Table 6- 36 Comparison of CFA OF USA, UK and the current study

The results of the CFA of the optimal model of the current study are in line with other results of CFA in USA and UK studies. It is clear that the optimal model that fits the Saudi data is the eight factor solution.

# 6.5.3 Construct validity (factor correlations of the optimal model)

The factor correlations of the optimal model were examined to find out whether the eight factors related to each other or not. The construct validity was studied by calculating correlations between the scale scores for every factor and subsequently calculating Pearson correlation coefficients between the scale scores. Table 6-37 shows the inter-correlations of the 8 dimensions (correlation coefficient).

N	Dimension	1	2	3	4	5	6	7
1	Organisational learning- continuous improvement							
2	Frequency of event reported	.15	-					
3	No punitive response to error	.25	.35	-				
4	Handoffs and transitions	.31	.38	.29	-			
5	Teamwork within units	.41	.38	.62	.45	-		
6	Communication openness	.29	.50	.46	.39	.58	-	
7	Supervisor expectation and action promoting patient safety	.25	.52	.25	.51	.35	.33	-
8	Staffing	.23	.36	.48	.31	.65	.56	.29

Table 6-37 Inter-correlation of the 8 dimensions

Note: All correlations are significant, P<.001

Table 6-37 shows that inter-correlation between the eight factors range between .154 up to .658 supporting that the factors are not independent from each other. The construct validity of each factor is reflected in scale scores that are moderately related. High correlations (r > 0.7), however, would indicate that factors measure the same concept (Smits et al., 2008). The highest correlation were those between teamwork within units and staffing(r = 0.658), but no correlation was exceptionally high. The construct validity was satisfactory for all factors; the moderate correlations of the factors show that there are no two factors measuring the same construct. The eight factors (dimensions of patient safety culture) of the optimal model are related to each other. The next section concerns reliability analysis of the optimal model.

## 6.5.4 Reliability analysis (internal consistency)

This step was concerned with reliability analysis (internal consistency) of the optimal model. The negatively formulated items were first recoded on the whole sample (862) to ensure that all items are coded in the same conceptual direction. The internal consistency was calculated for every factor according to the optimal model dimensions and items structure (8 factor solutions). Table 6-38 shows the results of the reliabilities of each scale of the eight factor model.

Table 0-36 results of the reliability analysis of the 6 factor model (n=802)						
Factor	Items	Cronbach's Alpha				
Factor 1	A6-A9-A13-F8	0.68				
Factor 2	D1-D2-D3	0.84				
Factor 3	A8-A12-A16	0.72				
Factor 4	F5-F6-F7-F11	0.69				
Factor 5	A1-A3-A4	0.79				
Factor 6	C2-C4	0.69				
Factor 7	B1-B2	0.83				
Factor 8	A5-A7	0.41				

 Table 6- 38 results of the reliability analysis of the 8 factor model (n=862)

The reliability analysis indicated that Cronbach's alpha values of the factors are above 0.68, except for factor 8 which is low. This factor consists of two items. However, this factor is still acceptable because the various fit indices in confirmatory factor analysis of the model meet the agreed parameters. Furthermore, the theoretical structure of this factor is related to the staffing dimension. It is possible that a set of items will be below 0.7 on Cronbach's alpha, yet various fit indices in confirmatory factor analysis will be above the cut off levels. Alpha coefficient may be low because of lack of homogeneity of variances among items, for instance, and it is also lower when there are fewer items in the scale/factor (Kline, 2005).

In summary, the 8 factor model (23 safety climate items) is the optimal model for the Saudi data. The next section provides the dimensions and items comprising the eight structure of the optimal model.

## 6.6 **Proposed optimal model**

#### Table 6-39 Proposed optimal model

#### Factor 1: Organisational learning-continues improvement (4 items)

A6: We are actively doing things to improve patient care

A9: Mistakes have led to positive change here

A13: After we make changes to improve patient safety, we evaluate their effectiveness

F8: The actions of hospital management show that patient safety is a top priority

#### Factor 2: Frequency of events reported (3 items)

D1: When a mistake is made, but is caught and corrected before affecting the patient, how often is this reported?

D2: When a mistake is made, but has no potential to harm the patient, how often is this reported?

D3: When a mistake is made that could harm the patient, but does not, how often is this reported?

#### Factor 3: No punitive response to error (3 items)

A8: Staff feel like their mistakes are held against them®

A12: When an event is reported, it feels like the person is being written up, not the problem®

A16: Staff worry that mistakes they make are kept in their personnel file®

#### Factor 4: Handoffs and transition (4 items)

F5: Important patient care information is often lost during shift changes®

F6: It is often unpleasant to work with staff from other hospital units®

F7: Problems often occur in the exchange of information across hospital units®

F11: Shift changes cause problems for patients in this hospital®

#### Factor 5: Teamwork within units (3 items)

A1: Staff support one another in this unit

A3: When a lot of work needs to be done quickly, we work together as a team to get the work

A4: In this unit, people treat each other with respect

#### Factor 6: Communication openness (2 items)

C2: Staff will freely speak up if they see something that may negatively affect patient care

C4: Staff feel free to question the decisions or actions of those with more authority

#### Factor 7: Supervisor expectation and action promoting patient safety (2 items)

B1: My supervisor/manager says a good word when he/she sees a job done according to established patient safety procedures

B2: My supervisor/manager seriously considers staff suggestions for improving patient safety

#### Factor 8: Staffing (2 items)

A5: Staff in this unit work longer hours than they should, which is not good for patient safety®

A7: We use more temporary staff than we should, which is not good for patient care®

R indicates negatively worded items

Table 6-39 shows the structure of the optimal model consists of 23 items (14 positively worded items and nine negatively worded items). The optimal model (eight factors) includes five positively worded factors and three negatively worded factors.

## 6.7 Summary of overall findings

The validity and reliability of the HSOPSC questionnaire was rigorously tested using Saudi data (862 completed questionnaires). The results of descriptive analysis show that the majority of respondents agree and strongly agree with positively worded items. The majority of respondents disagree and strongly disagree with negative items. The pre-analysis showed that the Saudi data was suitable to be subjected to factor analysis.

The initial results of the CFA and reliability analysis showed that when applying the original USA HSOPSC model (Sorra and Nieva, 2004) to the Saudi data, the fit indices from CFA and values of Cronbach's alpha ( $\alpha$ ) indicated that this model fit was not satisfactory. The results of EFA, CFA, correlation and reliability analysis showed that the optimal model to the Saudi data consists of eight factors (23 safety climate items).

The findings of this study indicated that the original USA HSOPSC model (Sorra and Nieva, 2004) was not valid and reliable in Saudi hospitals. The optimal model could be used in Saudi hospitals to assess patient safety culture. In the next chapter, the findings are discussed.

## **Chapter Seven: Discussion**

## 7.1 Introduction

This chapter discusses the research findings that emerged from this study. The purpose of this chapter is to discuss the findings of this thesis with reference to the literature of patient safety culture measurement that was reviewed in chapter two. This chapter addresses the key findings in relation to three topics: face validity (section 7.2), testing the original English language HSOPSC model (Sorra and Nieva, 2004) (section 7.3) and developing an optimal model in a Saudi context (section 7.4). The methodological issues as well as the strengths and limitations of the study are addressed. Section 7.5 discusses some methodological issues related to negatively worded items and reporting psychometric properties of patient safety climate questionnaires. Section 7.6 presents the strengths and limitations of the study. Finally, section 7.7 summarises the findings and impact of the study.

## 7.2 Face validity findings

This section aims to discuss the findings of the face validity testing of the original HSOPSC questionnaire (Sorra and Nieva, 2004) and the Teamwork and Safety Climate survey (Sexton et al., 2004) in the Saudi context. Face validity is a measure of whether a tool seems reasonable, and produces reasonable data from the viewpoint of its respondents (Carmines and Zeller, 1979). Face validity refers to whether the test seems to measure what it is intended to measure. In other words, face validity is an initial assessment of how a test looks (Streiner and Norman, 1995).

In terms of the importance of face validity, a number of studies demonstrated that establishing the face validity of a patient safety climate questionnaire is an essential element before full application. For example, Hutchinson et al. (2006) asked a number of staff members to complete the Stanford University Patient Safety Climate in Healthcare Organisations questionnaire (Singer et al., 2003) and the Teamwork and Safety Climate Survey (Sexton et al., 2004) for comparison. A number of changes were made to the Teamwork and Safety Climate Survey (Sexton et al., 2004) to make it clear and accessible for

frontline staff. Hutchinson et al., (2006) demonstrated a more general need for thorough validation of safety climate questionnaires before widespread usage because environmental differences might exist at the national level. Waterson et al. (2009) discussed the original HSOPSC questionnaire (Sorra and Nieva, 2004) with NHS staff members before full application. As a result a number of changes were made to the wording of the questionnaires with respect to terminology used within the UK settings. Waterson et al., (2009) emphasised that there is a need for caution in using patient safety climate questionnaires in different settings and they stress the importance of appropriate validation before widespread use in different contexts.

It seems that assessing face validity is a critical issue whenever patient safety climate questionnaires are used. According to Pronovost and Sexton (2005) it is important to validate patient safety climate questionnaires before extensive use in a new context. Therefore, attention should be paid to establish face validity before extensive use, especially in a new context.

The approach taken to test the face validity of a patient safety climate questionnaire varies between studies. The first approach is used where patient safety climate questionnaires have been translated and validated in other languages for use in different countries. For example, Deilkas and Hofoss (2008) translated the SAQ (Sexton et al., 2004) to Norwegian and validated the translated version. The original HSOPSC questionnaire (Sorra and Nieva, 2004) was translated and validated into Dutch for application in the Netherlands (Smits et al., 2008), into German (Pfeiffer and Manser, 2010) and into Turkish (Bodur and Filiz, 2010). These studies used their national language by using forward-backward translation (Sperber, 2004). They used a translation to their most commonly spoken language and validated the translated questionnaire: this might be because English was not used as the professional language in their hospital settings.

However, a translation of a patient safety climate questionnaire is based on the common professional language used in each country while the validation of a patient safety climate questionnaire should be performed before full application in different contexts. The current study used the same language as the original questionnaires, which is English, because the common language in the Saudi hospitals is English as the professional language.

The face validity testing of the two patient safety climate questionnaires was conducted in order to select the most appropriate one of them to use in the current study. Moreover, testing two questionnaires provided new information in terms of how staff can understand the content of different questionnaire items.

The second approach is testing the face validity of a patient safety climate questionnaire when using its original language without translation of the questionnaire to another language. For example, in the study conducted by Hutchinson et al. (2006) minor adaptations were made to the Teamwork and Safety Climate Survey (Sexton et al., 2004) before it was used in their study. Similarly, in Waterson et al. (2009) a number of changes were made to the wording of the HSOPSC questionnaire (Sorra and Nieva, 2004) before full application with respect to terminology used within the UK settings. These studies keep the same language but they changed some words because of different terminology.

The face validity stage (an initial assessment stage) of the current study highlighted a number of issues. For example, the current study indicated that there is a range of spoken language ability and understanding of the written language in the questionnaires, because of the ethnic cultural mix of staff in Saudi settings. The interviews revealed variation of English language skills amongst participants. This variation is expected because they use English as a professional language but it is not their first language since they are from different nationalities. Therefore, it seems plausible that the participants might vary in their understanding of the questionnaires due to variation in language ability among participants. However, it was observed that simple and clear items in the questionnaires could be understood by participants who speak English regardless of their nationalities. Overall, this suggests that in a multi cultural context the cultural differences should be taken into account whenever patient safety climate questionnaires are applied in different cultural settings, such as Saudi hospitals settings. Therefore, rigorous validation should be performed to reduce the impact of the complexities of cultural and language issues which might exist, and to enhance the quality and validity of the research (Sperber, 2004).

In the current study, face validity testing indicated that there were problems and difficulties of comprehension with some wordings of both questionnaires. For example, some words of the original HSOPSC questionnaire (Sorra and Nieva, 2004) were difficult to understand such as "than is", "crisis mode", and "fall between the cracks". Similarly, some words of the Teamwork and Safety Climate questionnaire (Sexton et al., 2004) were difficult to understand such as "briefings", "contingencies" and "knowingly compromise". These problems might be because the variation of clarity of language and the wording of both questionnaires which led to some participants understanding the items in a certain way while others did not understand them at all. However, with regard to comprehension and competence of the language and wording of the patient safety climate questionnaires, it was observed in the interviews that simple, clear and straightforward wording of items in the questionnaires could be understood by participants.

For the purpose of assessing the comprehensibility of items and identifying problems with the wording of questionnaires, face validity testing is an essential element of the application of patient safety climate questionnaires. For example, Bodur and Filiz (2010) tested the comprehensibility of the HSOPSC questionnaire (Sorra and Nieva, 2004) by discussing the wording of the questionnaire with 5 doctors and 8 nurses and their opinions were used to make the final version of the questionnaire clear, simple and easy for respondents to complete.

The face validity testing in the current study highlighted a very important issue regarding the scale that is used in the questionnaires. It was observed that the majority of the participants had no difficulty linking their answers with the 5-point Likert responses scale, ranging from strongly disagree to strongly agree with a midpoint called "neither", of the original HSOPSC questionnaire (Sorra and Nieva, 2004). Furthermore, it was observed that most of the respondents read the items of the original HSOPSC questionnaire (Sorra and Nieva, 2004). only once before linking their answers with the scale. This indicates the clarity of the items and that the scale was easy to use with answers. On the other hand, the participants experienced significant difficulty linking their answers with the 6-point Likert responses scale of the Teamwork and Safety Climate Survey (Sexton et al., 2004). The participants believed it was not easy to use because more chances prevented them from expressing their feeling correctly. Observation of how participants approached answering the questionnaires was valuable in identifying where they were encountering difficulties.

The current study highlighted another important issue which was observed regarding negatively worded items. It is worth noting that the positive and negative items were clear and the participants were able to distinguish between them. However, this issue will be discussed in detail in section (7.5) in the current chapter.

The current study indicated that the face validity testing was a helpful tool for identifying the most appropriate patient safety climate questionnaire for use in the study. The first method of the research was concerned with selecting appropriate patient safety climate questionnaires for assessing the perceptions of the workforce towards patient safety culture in hospitals, for use in the investigation (see chapter four, section 4.2 for more details). This method identified two questionnaires, the HSOPSC questionnaire (Sorra and Nieva, 2004) as a first appropriate questionnaire and the Teamwork and Safety Climate Survey (Sexton et al., 2004) as a second appropriate questionnaire.

The HSOPSC and the Teamwork and Safety Climate Survey were used in a number of studies because they are valid and reliable questionnaire (Sorra and Nieva, 2004; Hutchinson et al., 2006). However, the first method of the research was helpful to identify appropriate questionnaires but it did not tell us about how staff understand the meaning of the items of patient safety climate questionnaires. Therefore, there was a need to test and assess the wording of individual items to select the most appropriate patient safety climate questionnaire to use in the current study before the full application.

The participants perceived that the language and wording of the original HSOPSC questionnaire (Sorra and Nieva, 2004) was clearer and easier to understand than the Teamwork and Safety Climate questionnaire (Sexton et al., 2004) because the language and wording of the HSOPSC questionnaire is simple and clear and more straightforward than that of the Teamwork and Safety Climate Survey. Moreover, the terminology of the original HSOPSC questionnaire is directly related to patient safety, medical errors and reporting incidents. This might be because of the simplicity of the wording of individual items of the original HSOPSC questionnaire and because it was developed specifically for healthcare settings (Sorra and Nieva, 2004). The wording of the Teamwork and Safety Climate Survey (Sexton et al., 2004) was perceived as more complicated and difficult for participants to understand easily. In addition, the participants found the scale used in the questionnaire difficult to apply.

Chapter seven

However, these differences might be because of the two questionnaires were from different sources, the Teamwork and Safety Climate Survey (Sexton et al., 2004) was adapted from industry and the original HSOPSC questionnaire (Sorra and Nieva, 2004) was developed for specifically for healthcare settings.

Selecting an appropriate patient safety climate questionnaire should be based on not only on reviewing available questionnaires but also an initial assessment of patient safety climate questionnaires (face validity testing) is very important and an essential element in validation studies. The current study has highlighted that face validity is a helpful tool in terms of identifying changes that should be made to questionnaires to be clear and easy to understand by respondents. However, while only minor changes needed to be made to the original HSOPSC questionnaire (Sorra and Nieva, 2004), the Teamwork and Safety Climate Survey (Sexton et al., 2004) needed many changes because a number of its items were ambiguous and difficult to understand, which makes the decision to make changes extremely difficult as it may alters the meaning of the items.

A number of studies made changes to questionnaires as a result of face validity testing. For example, Hutchinson et al. (2006) made minor changes to the questionnaire wording based on face validity results. Waterson et al. (2009) made a number of changes to the original HSOPSC questionnaire (Sorra and Nieva, 2004) based on feedback from group discussion with staff members with respect to terminology used within UK. Changes should be made to make items easier for the respondents to understand them.

#### Summary

The current study found that staff varied in terms of their understanding and interpretation of the wording of the items of the two patient safety climate questionnaires. Therefore, an initial assessment is a very important element before the full application of any patient safety climate questionnaire in a different context, such as Saudi hospitals, because of language and cultural variation due to different cultural and ethnic groups. The current study highlights that face validity is an essential element before the widespread application of a patient safety climate questionnaire in different settings environmentally and culturally. It seems that the face validity of the current study is more than just identifying ambiguous words in the questionnaires that should be changed to make them clear and easy to understand (minor adaptations). It also takes account of the range of language ability among participants, and comprehensibility and competence of the language of the patient safety climate questionnaire, the applicability of the scale, and the ability to recognise negatively worded items. When comparing two questionnaires it is also a helpful tool in terms of selecting the appropriate patient safety climate questionnaire. On the basis of the above issues and important benefits, it is worthwhile and valuable to undertake face validity as good practice and an essential element in validation studies, especially where environmental differences might exist. Therefore, attention should be paid to face validity before extensive use especially in a new context (Hutchinson et al., 2006; Waterson et al., 2009).

The face validity in the current study was aimed at more than how the two patient safety climate questionnaires look. The additional value is to adapt an appropriate questionnaire in a culturally relevant and comprehensible form while maintaining the meaning of the original items of the selected patient safety climate questionnaire. The original HSOPSC questionnaire (Sorra and Nieva, 2004) was identified as the most appropriate patient safety climate questionnaire to use for the study. Testing the validity of the factor structure of the original HSOPSC questionnaire (Sorra and Nieva, 2004) with Saudi data will be discussed in the next section.

## 7.3 Testing the original model findings

The pre-analysis (Bartlett's test and KMO) was performed to assess the suitability of the Saudi data for factor analysis. The results showed that the Saudi data was sufficient for performing factor analysis. Pallant (2007) highlights the importance of undertaking pre-analysis in order to assess the suitability of data for factor analysis prior to assessing the psychometric properties of patient safety climate questionnaires and more recent studies are adopting this as good practice (e.g. Smits et al., 2008; Waterson et al., 2009; Bodur and Filiz, 2010).

One of the main findings of the current study relates to the applicability of using the original English language HSOPSC questionnaire (Sorra and Nieva, 2004) for assessing patient safety culture in Saudi hospitals. In the current study the first step of psychometric analysis aimed to investigate whether the factor structure of the original USA HSOPSC questionnaire (12 factors and 42 items) can be used with the Saudi data by using CFA and reliability analysis.

The current study demonstrated that the original USA HSOPSC questionnaire (Sorra and Nieva, 2004) did not fit the Saudi data well (poor fit). Therefore, the factor structure of the original USA HSOPSC questionnaire cannot be effectively used in Saudi hospitals. The results of CFA and reliability analysis are discussed and compared with the results of the key studies below.

Firstly with regard to CFA in the current study, the fit indices were used to evaluate the adequacy of the original model fit and their values were as follows: (Chi-square=4995, DF=805, CFI=0.642, RMSEA=0.678, RMR=0.178, SRMR=0.193 and TLI=0.617). These fit indices from CFA indicated that when applying the original USA HSOPSC model (Sorra and Nieva, 2004) to the Saudi data, this model fit was not satisfactory because overall, the values of the fit indices of the Saudi data were outside the general agreed parameters. In other words, the fit indices did not meet the goodness of fit criteria (see table 6-15, page 174). In contrast, the fit indices were used in the original USA HSOPSC questionnaire (Sorra and Nieva, 2004) as follows: (Chi-square=2064, DF=746, CFI=0.94, RMSEA=0.040). Although they used just four fit indices, they achieved the general agreed parameters. However, the relatively high values of fit indices of CFA achieved in the original model (Sorra and Nieva, 2004) may be due to their use of the same sample for the EFA and CFA, in other words the split half validation was not undertaken and they tested the fit of their model by using the same data (Waterson et al., 2009). In general, it is recommended that use of a wide range of fit indices is the best means of evaluating the degree of fit (Hu and Bentler, 1999), as has been done in the current study.

The current study indicated that the fit indices of CFA of the original HSOPSC model (Sorra and Nieva, 2004) using Saudi data were unsatisfactory. Other studies found similar results, for example, Smits et al. (2008) examined the applicability of the factor structure of the HSOPSC questionnaire (Sorra and Nieva, 2004) to the Dutch data (Dutch translated version of the HSOPSC questionnaire). The fit indices from CFA indicated that this model fit was not satisfactory. Similarly, Waterson et al. (2009) tested the original HSOPSC model (Sorra and Nieva, 2004) in the UK context and they found that the fit indices of CFA were a poor fit when they compared them with the original American model (Sorra and Nieva, 2004).

Bodur and Filiz (2010) performed CFA to investigate whether the factor structure of the original USA HSOPSC questionnaire (Sorra and Nieva, 2004) could be used with Turkish data (Turkish translated version of the HSOPSC questionnaire). They (Bodur and Filiz, 2010) concluded that the original model was not a satisfactory fit with the Turkish settings. In a study by Pfeiffer and Manser (2010) they examined the applicability of the original HSOPSC questionnaire (Sorra and Nieva, 2004) to assess patient safety climate in a German speaking hospital setting in Switzerland (German translated version of the HSOPSC questionnaire). They found the values of fit indices of CFA indicated that the fit is not sufficiently good to confirm the proposed factor structure of the HSOPSC questionnaire (Sorra and Nieva, 2004). It appears that the overall fit was consistently unsatisfactory for the above studies when they compared with Sorra and Nieva's (2004) original USA HSOPSC questionnaire.

It seems, based on the evidence from the results of CFA in different studies above, that the factor structure of the original HSOPSC questionnaire (Sorra and Nieva, 2004) might not work in a satisfactory way in these countries. It appears that the original HSOPSC questionnaire measures different dimensions of patient safety culture in different countries. Table 7-1 shows the comparative CFA data of these countries (six countries).

Table 7-1 Comparative CFA data from six countries					
Study	Fit indices criteria of CFA				
Original HSOSPC (Sorra and Nieva, 2004)	Chi-square=2064, DF=746, CFI=0.94, RMSEA=0.040				
The current study	Chi-square=4995, DF=805, CFI=0.64, RMSEA=0.678, RMR=0.178, SRMR=0.193 and TLI=0.617				
The UK study (Waterson et al., 2009)	Chi-square=1906, DF=674, CFI=0.91, RMSEA=0.045, SRMR=0.046				
German study (Pfeiffer and Manser, 2010)	RMSEA=0.047, PCLOSE=0.91, CMIN/df=2.271, GFI=0.878, NFI=0.859, and TLI=0.901				
Dutch study (Smits et al., 2008)	NA				
Turkish study (Bodur and Filiz, 2010)	NA				
NA: not available.					

Table 7-1 shows that three studies reported the values of the fit indices of CFA when they tested the fitting of the original HSOPSC questionnaire (e.g. Sorra and Nieva, 2004; Waterson et al., 209; Pfeiffer and Manser, 2010) while two studies did not report this (e.g. Smits et al., 2008; Bodur and Filiz, 2010). This kind of incompleteness might be because of the difficulty of reporting fit indices. It might be also because some studies depend on reporting reliability analysis which is an important indicator but not enough to evaluate the fitting of the original model in terms of a comprehensive psychometric assessment of patient safety climate questionnaires in a different context. According to Hoyle (1995) CFA is the best procedure analysis for examining the factor structure of models.

Secondly, with regard to reliability analysis in the current study, the internal consistency of the 12 factors of the original USA questionnaire (Sorra and Nieva, 2004) was calculated with Cronbach's alpha ( $\alpha$ ) (Field, 2000). The internal consistency of the Saudi data for each dimension was greater than 0.6 except for three dimensions that were poor or unacceptable: for overall perceptions of safety ( $\alpha$ =0.31), staffing ( $\alpha$ =0.57) and hospital handoffs and transitions ( $\alpha$ =0.59). Overall, the internal consistency of the Saudi data was lower for each dimension than the American data (Sorra and Nieva, 2004).

The first dimension with low internal consistency was the overall perception of patient safety which consists of four items. This dimension relates to the extent to which procedures and systems in hospitals are good at preventing errors and there is a lack of patient safety problem (Sorra and Nieva, 2004). There is a multi cultural environment in Saudi hospitals because the staff there are from different countries. Therefore, their perceptions may be affected by their cultural backgrounds and experiences in their countries. Accordingly, they might answer these items of overall perceptions of patient safety dimension based in comparison with the procedures and systems in their countries. Moreover, the majority of staff have experiences and communications training outside Saudi Arabia such as the UK, USA and Canada. All these cultural issues might affect the overall perceptions of the staff.

On the other hand, this dimension works with the UK data that was gathered from similar cultural backgrounds in an English context (Waterson et al., 2009). Similarly, the dimension works with the Dutch data (Dutch translated version of the HSOPSC questionnaire) that was gathered from the same culture (Smits et al., 2008). It seems that this kind of dimension is affected by cultural background.

Therefore, it did not work in a multi cultural environment such as the Saudi context but it works in data that gathered from the same culture such the UK and Dutch contexts. However, it is seemed that there is a cultural variation between staff and also between contexts. Therefore, it may be the perceptions of staff toward patient safety culture dimensions in the questionnaire may be affected by their cultural backgrounds and experiences.

The second dimension that had low internal consistency was the staffing dimension ( $\alpha$ =0.57). It consists of four items and it is about the extent to which there is enough staff to handle the workload and whether work hours are appropriate to provide the best care for patients (Sorra and Nieva, 2004). This dimension had low internal consistency in a number of studies. For example, Smits et al. (2008) found the staffing dimension had low internal consistency ( $\alpha$ =0.49) when they tested the fit of the original USA HSOPSC questionnaire (Sorra and Nieva, 2004) in Dutch data. Similarly, the staffing dimension had low internal consistency ( $\alpha$ =0.58) in Waterson et al. (2009) study when they tested the fit of the original USA HSOPSC questionnaire (Sorra and Nieva, 2004) in the UK data.

The staffing dimension had very low internal consistency ( $\alpha = 0.19$ ) in a study by Bodur and Filiz (2010) when they tested the fit of the original USA HSOPSC questionnaire (Sorra and Nieva, 2004) in the Turkish data. It seemed that this dimension did not work well in different contexts. It might be that the composition of this dimension was not simple and clear as it has conflicting items (a mixture of one positive and three negative items), therefore it is difficult to interpret this kind of composition (Schmitt and Stults, 1985).

The third dimension was hospital handoffs and transitions. It consists of four items and it is about how important patient care information is transferred between hospital departments during shift changes (Sorra and Nieva, 2004). The current study indicated that the internal consistency of hospital handoffs and transitions might be acceptable because it was close to 0.6. This dimension might work with a different structure of items in the Saudi data. However, this point may be answered by EFA in the next section. In summary, the current study indicated that the internal consistency was lower for each dimension than the American data (Sorra and Nieva, 2004). Three factors were poor or unacceptable: overall perceptions of safety, staffing and hospital handoffs and transitions.

A similar finding was reported by a number of studies. For example, in the study by Smits et al. (2008) the internal consistency of the Dutch items was lower than that of the original items in the AHRQ study (Sorra and Nieva, 2004) except for communication openness, which was the same. The internal consistency of three factors was poor or even unacceptable: organisational learning-continuous improvement ( $\alpha = 0.57$ ), staffing ( $\alpha = 0.49$ ) and teamwork across hospital units ( $\alpha = 0.59$ ). The internal consistency of the UK items was lower for each dimension than that of the original items of the HSOPSC questionnaire. The internal consistency of one dimension was unacceptable, which was staffing ( $\alpha=0.58$ ) (Waterson et al., 2009).

Pfeiffer and Manser (2010) reported that the internal consistency of the German items was lower for seven dimensions than the original items of the HSOPSC questionnaire (Sorra and Nieva, 2004). Similarly, Bodur and Filiz (2010) reported that the internal consistency of the Turkish items was lower for each dimension than the original items in the HSOPSC questionnaire (Sorra and Nieva, 2004), except for frequency of events reported ( $\alpha = 0.86$ ) and teamwork within units ( $\alpha=0.83$ ). The internal consistency of two factors was poor: Staffing ( $\alpha=0.19$ ) and non-punitive response to error ( $\alpha=0.31$ ). Table 7-2 provides comparative data of internal consistency of a number of studies (six countries).

HSOPSC almensions	Internal consistency [Crondach's alpha (a)]						
	USA	The Current study	UK	Germany	Netherlands	Turkey	
Overall perceptions of safety	0.74	0.31	0.67	0.75	0.74	NA	
Frequency of error reporting	0.84	0.83	0.83	0.88	0.79	0.86	
Supervisor/manger expectations and actions promoting patient safety	0.75	0.75	0.68	0.78	0.70	NA	
Organisational learning- continuous improvement	0.76	0.63	0.66	0.68	0.57	NA	
Teamwork within units	0.83	0.75	0.73	0.73	0.66	0.83	
Communication openness	0.72	0.67	0.67	0.64	0.72	NA	
Feedback and communication about error	0.78	0.74	0.80	0.79	0.75	NA	
No punitive response to error	0.79	0.72	0.65	0.71	0.69	0.31	
Staffing	0.63	0.57	0.58	0.61	0.49	0.19	
Hospital management support for patient safety	0.83	0.65	0.69	0.71	0.68	NA	
Teamwork across hospital units	0.80	0.69	0.70	0.76	0.59	NA	
Hospital handoffs and transitions	0.80	0.59	0.77	0.71	0.68	NA	

## Table 7- 2 Comparative internal consistency data from six countries USOPSC J:

NA: not available but the internal consistency of each dimension was lower than USA dimensions in the Turkish study (Bodur and Filiz, 2010).

In summary, it seemed that the internal consistency of the current study using Saudi data, UK, German, Dutch, Turkish data was lower for each dimension than the American data (Sorra and Nieva, 2004). The lowest internal consistencies of dimensions across these different studies are overall perception of patient safety, organisational learning, non punitive response to error, staffing, teamwork across hospital units and hospital handoffs and transitions. The results of the reliability analysis of the studies provide evidence that the original model does not work well in different countries.

On basis of the results of testing the original model in the current study (CFA and reliability analysis), the question of investigating the psychometric properties of the HSOPSC questionnaire (Sorra and Nieva, 2004) in the Saudi hospital settings could be answered. It seems that the reliability and validity of the original factor structure of the HSOPSC questionnaire (Sorra and Nieva, 2004) is not replicated in Saudi hospital settings. Thus the original model of the HSOPSC should not be used in Saudi hospitals. This finding is in contrast with other recent studies in Saudi hospitals that were produced after the current study began. For example, Alahmadi's (2009) study assessed patient safety culture in Saudi hospitals by using the original USA HSOPSC questionnaire (Sorra and Nieva, 2004) before examining the reliability and validity of the questionnaire in a different context. Similarly, in a study by another author (Alahmadi. Hanan, 2010) used the original HSOPSC questionnaire (Sorra and Nieva, 2004) to assess patient safety culture in a number of Saudi hospitals without performing a proper validation of the questionnaire before its application in the Saudi context.

It seems that these two Saudi studies (Alahmadi. Talal, 2009; Alahmadi. Hanan, 2010) did not assess the psychometric properties of the HSOPSC questionnaire whereas the current study performed rigorous validation to assess the psychometric properties of the HSOPSC questionnaire. However, the HSOPSC questionnaire (Sorra and Nieva, 2004) needs to be used with caution in Saudi hospitals because of the variation of cultures and language in the Saudi context as discussed earlier in section (7.2) of this chapter. Therefore, it is important to check the validity and reliability of a patient safety climate questionnaire before applying it in a new context (Smits et al., 2008).

In terms of testing the original HSOPSC questionnaire at an international level (different countries), evidence from other studies supports the findings from testing the original HSOPSC questionnaire in the current study. For example, the factor structure of the original USA HSOPSC questionnaire (Sorra and Nieva, 2004) was tested in different countries. Smits et al. (2008) examined the applicability of the factor structure of the original HSOPSC questionnaire to the Dutch data. They indicated that the Dutch version was not able to replicate the factor structure proposed by Sorra and Nieva (2004), but found some factors corresponding to those proposed for the original instrument. Similarly, in a UK study (Waterson et al., 2009) investigated whether the factor structure of the original USA HSOPSC questionnaire (Sorra and Nieva, 2004) could be used with the UK data study. Waterson et al. (2009) found that the UK version was not able to replicate the factor structure of the HSOPSC questionnaire. Similarly, Bodur and Filiz (2010) examined the applicability of the factor structure of the original HSOPSC questionnaire (Sorra and Nieva, 2004) to the Turkish data. Bodur and Filiz (2010) found that the factor structure of the original HSOPSC questionnaire (Sorra and Nieva, 2004) cannot be used with Turkish data. In a study by Pfeiffer and Manser (2010) they examined the applicability of the original HSOPSC questionnaire to assess patient safety climate to a German speaking hospital setting in Switzerland. They also found the German version was not able to replicate the proposed factor structure of the HSOPSC by Sorra and Nieva (2004).

#### Summary

In summary, the findings from testing the original HSOPSC questionnaire (Sorra and Nieva, 2004) highlight a number of important issues. For example, the testing of the original HSOPSC questionnaire might be useful when it is based on reporting the results of CFA and reliability analysis in terms of evaluating the psychometric properties of the HSOPSC questionnaire. The results from CFA and the internal consistency of the current study using Saudi data indicated that the twelve dimensions of the original HSOPSC questionnaire (Sorra and Nieva, 2004) did not match the findings in Saudi hospitals. As a result, the original HSOPSC questionnaire cannot be used with the English language Saudi data. Therefore, it appeared that the factor structure of the original HSOPSC questionnaire may be measuring different dimensions of patient safety culture in the Saudi context as compared to the USA (Sorra and Nieva, 2004). This might be because of cultural and contextual differences between the United States and the Saudi hospital settings.
The original factor structure of the HSOPSC questionnaire (Sorra and Nieva, 2004) has recently been applied and empirically evaluated in different countries (e.g. Netherlands: Smits et al., 2008; the UK: Waterson et al., 2009; Turkey: Bodur and Filiz 2010; Germany: Pfeiffer and Manser, 2010). None of these studies were able to replicate the factor structure proposed by Sorra and Nieva (2004). Based on this, it is important to investigate whether or not the dimensional structure of patient safety climate questionnaires can be replicated in different contexts to ensure that patient safety climate questionnaire is valid and reliable, therefore, a questionnaire developed in one context should ideally always be validated before wider use in a different context (Pronovost and Sexton, 2005). Patient safety climate questionnaires need to be used with caution because of cultural and language variations between different contexts (Nieva and Sorra, 2003; Hutchinson et al., 2006; Waterson et al., 2009). These issues above show the importance of testing the original HSOPSC questionnaire (Sorra and Nieva, 2004) carefully before full application in different healthcare environments.

However, a number of recent studies in patient safety culture in Saudi Arabia (e.g. Alahmadi Talal, 2009; Alahmadi Hanan, 2010) did not investigate whether the factor structure of the original USA HSOPSC questionnaire (12 factors and 42 items) (Sorra and Nieva, 2004) can be used with the Saudi data by using psychometric analysis such as CFA and reliability analysis. The current study has addressed the need to identify an optimal model for using to assess patient safety culture in Saudi Arabia. The findings of EFA, CFA and reliability analysis in terms of developing the optimal model are discussed in the next section.

### 7.4 Developing the optimal model

As mentioned earlier, the original HSOPSC questionnaire (Sorra and Nieva, 2004) is not valid and reliable in the Saudi context using English language data. Therefore, there was a need to identify an optimal model for the Saudi data. An optimal measurement model was constructed via EFA and CFA with the Saudi data split into halves. In the current study the results of psychometric analysis (EFA, CFA and Reliability analysis) show that the optimal model for the Saudi data consists of eight factors (see table 6-39 in page 216). The optimal model could be used in Saudi hospitals to assess patient safety culture. The findings of EFA, CFA and Reliability analysis are discussed below. A comparison of optimal models across different countries is provided.

With regard to EFA, it was performed in order to investigate if there is a factor structure that better fits the Saudi data. In the current study, the Saudi data set was split randomly into two halves. The first half of the Saudi data set was used for model construction by exploring a number of factors (patient safety culture dimensions). The second half of the Saudi data set was used for model validation to test the fit of the results of the optimal factor structure. A split half validation strategy (EFA followed by CFA) was recommended by a number of studies for providing realistic results of the psychometric properties of patient safety climate questionnaires. For example, Waterson et al. (2009) split their data when they identified an optimal model in the UK context. On the other hand, a number of recent studies did not split their data. For example, in Sorra and Nieva's (2004) study, the split half validation was not undertaken and they tested the fit of their model by using the same data (Waterson et al., 2009). Smits et al. (2008) did not split Dutch data into halves when they identified their optimal model in Netherlands hospitals. Similarly, Bodur and Filiz (2010) did not split the Turkish data into halves. However, the complexity of psychometric analysis (cross validation) might be the reason for not doing the split half validation in these studies.

In terms of method of extraction, rotation and other criteria for EFA, in the current study EFA was performed to identify an optimal model, using Principle Axis Factoring and Varimax rotation, as well as both Kaiser's criterion and the Scree plot. Changing the rotations was performed (from Varimax to Oblique rotation) to enable interpretation of the factors.

According to Costello and Osborne (2005), Oblique (Orthogonal) rotation is used to produce more easily interpretable results especially in terms of cross loading items. All these criteria were taken into consideration for identifying the optimal model of the current study with no low loading, no low communalities, no cross-loaded items and including the theoretical related structure of the items of the optimal model. According to Flin (2007) identification of a patient safety climate questionnaire with a high standard of measurement means its results can be relied on. Therefore, using the criteria above was helpful in terms of ensuring a high standard measurement of the original HSOPSC questionnaire dimensions (Sorra and Nieva, 2004) in the Saudi context and identifying the optimal model.

Studies vary in their use of EFA for developing a measurement model. For example, Pfeiffer and Manser (2010) applied EFA using principle component analysis, Varimax rotation to interpret the factor loadings independently and Kaiser's criterion (Eigenvalues>1). Although, this strategy seems comprehensive, there is a lack in terms of changing rotation to produce an easily interpretable measurement model. As a result of this, their optimal model contains a number of items which cross-loaded in two factors and also some items with low loading. It might also be due to not having examined a series of possible models and they did not gradually remove low loading and cross-loaded items (Pfeiffer and Manser, 2010).

Smits et al. (2008) applied EFA using principle component analysis, Varimax rotation, Kaiser's criterion (Eigenvalues>1) and the Scree plot. Their model (11 factors) consists of highly loading items (>0.36). On the other hand, Waterson et al. (2009) applied EFA using principle axis factoring, Varimax rotation, Kaiser's criterion (Eigenvalues>1) and the Scree plot. Their model also consists of highly loading items (>0.36). An oblique rotation was used to aid interpretation of their resulting factors (9 factors). However, it seems that the methods of extraction and rotation are different in these studies because selecting extraction and rotation methods is based on the purpose of study (Conway and Allen, 2003).

In the current study, a number of exploratory factor analyses were performed and a number of solutions were identified. Each possible solution was examined in order to identify an optimal model including 11, 10 and 8 factor solutions. At the end of this stage the eight factor solution was identified as an optimal model for the current study. The total variance explained by the optimal model (8 factors) accounts for 51.5% of total variance.

The optimal model consists of 23 items while 19 items were removed gradually that were either cross-loaded or had low loading (<0.3) or no loading across different solutions. Factor loadings were between 0.43 and 0.97. All dimensions of the optimal model in the current study consist of two to four safety climate items.

The optimal model of the current study is in line with other studies. For example, Waterson et al. (2009) identified 9 factors as the optimal model of their data. Their model accounts for 51.5% of total variance. The optimal model consists of 27 items and 13 items were removed gradually that were either cross-loaded or had low loading (<0.3). Factor loadings were between 0.38 and 0.98. All dimensions consist of two to five items. Smits et al. (2008) identified 11 factors as the optimal model of their data. Their model accounts for 57.1% of total variance and the optimal model consists of 40 items (two items were removed). No items cross-loaded or had loading less than 0.3. Factor loadings were between 0.36 and 0.88. All dimensions consist of two to five items that the current study followed recommended practice. However, the studies vary in terms of the composition of dimensions.

In terms of the composition of the items and factors of the optimal model of the current study, the optimal model consists of eight dimensions. Two dimensions were formed as before, these were frequency of events reported and non punitive response to error. Three dimensions lost one item (Teamwork within units, handoffs and transitions, and communication openness). Two dimensions lost two items (staffing, supervisor expectations and actions promoting patient safety). One dimension had a new item "organisational learning-continuous improvement". There were only small shifts among items across factors. For instance, F6 moved to "Handoffs and transitions", and F8 moved to "organisational learning-continuous improvement".

On the other hand, in the optimal model of the current study, four dimensions are absent: "overall perception of patient safety", "feedback and communication about error", "teamwork across hospital units" and "hospital management support for patient safety". The composition of the optimal model of the current study consists of 23 items which represent 8 dimensions of patient safety culture, while 19 items were excluded because their loading either they did not load upon any factor or they had low loading. The composition of the optimal model is discussed below.

In the optimal model of the current study, four dimensions are absent. For example, "overall perception of patient safety", this might be absent because the respondents are from 23 countries and therefore they have different perceptions of patient safety because they have different cultural backgrounds. This dimension combined with the staffing dimension in the optimal model derived from the UK data (Waterson et al., 2009) and German data (Pfeiffer and Manser, 2010). As mentioned earlier in section 7.3, the "overall perception of patient safety" works better in other countries such as the Dutch and the British (e.g. Smits et al., 2008; Waterson et al., 2009) either because it seemed that they have shared similar perceptions of safety within their countries or because their understanding of English was similar. On the other hand, this dimension did not work in multi cultural environment such as the Saudi context. Moreover, it seems that the "overall perception of patient safety" and staffing could be formed as one dimension. This may be because of the "overall perception of patient safety" regarding the issues that could prevent problems of patient safety and staffing could be one of the issues that contribute to provide the best care for patients (Sorra and Nieva, 2004).

Another dimension that was lost in the optimal model of the current study is "feedback and communication about error". It might be because staff do not trust the incident report system because almost half (46.7%) of respondents agreed that their mistakes are held against them. This dimension works in a number of studies (Netherlands: Smits et al., 2008; UK: Waterson et al., 2009). Additionally, this dimension combined with the "communication openness" dimension in the optimal model derived from both the Turkish data and German data (Turkey: Bodur and Filiz, 2010; Germany: Pfeiffer and Manser, 2010). This might indicate that there is no a blame culture and they support reporting mistakes that happen in hospitals to learn from their mistakes and to improve patient safety. Moreover, it seems that "feedback and communication about error" dimension and "communication openness" could be formed as one dimension because both of them are concerned with how the communication between staff should be about how errors happen (Sorra and Nieva, 2004).

The teamwork across hospital units dimension was lost in the current study (except item F6). This dimension combined with the handoffs and transitions dimension in the optimal model derived from the German data (Pfeiffer and Manser, 2010). This dimension combined with hospital management support for patient safety dimension in the optimal model derived from the Turkish data (Bodur and Filiz, 2010).

It seems that the items of this dimension could be correlated with different items of dimensions. The hospital management support for patient safety dimension consists of three items. Two items were dropped whereas item number F8 moved to organisational learning-continuous improvement. This dimension also did not work in the optimal model derived from the UK data (Waterson et al., 2009). However, the absence of this dimension might be considered as an indicator of a lack of hospital management support for patient safety or may not be aware.

"Supervisor/manager expectations and actions promoting patient safety" dropped two negative items (B3-B4). These two items are about the negative role of managers towards patient safety (i.e. ignoring patient safety problem). Similarly, these two items were dropped in the optimal model derived from the UK data (Waterson et al., 2009). It is expected because if the respondents agreed with B1 and B2 as positive items of this dimension they disagreed to B3 and B4, therefore, the negative items were discarded from this dimension (Spector, 1997). Therefore, the majority of respondents in the current study (89.8%) disagreed that supervisors ignore patient safety problems and they agreed (76.1%) that supervisors promote patient safety.

"Staffing" dropped two items (A2-A14). These items are about having enough numbers of staff and how they handle the workload. It seems that almost half of respondents (51.8%) agreed that there is a shortage of staff and there is a high workload. These two items work in a number of optimal models as an indicator of having enough staff to handle the workload properly (e.g. Waterson et al., 2009; Bodur and Filiz, 2010).

A single item was dropped from "Hospital handoffs and transitions" (F3: things lost when transferring patients from one unit to another), from "teamwork within units" (A11: when one area in this unit gets busy, others help out) and "communication openness" (C6: staff are afraid to ask question when something does not seem right). This means these items are not working in any dimension of the optimal model. This dropping of some items is expected. For example, in Waterson et al's (2009) study, a single item dropped from both "teamwork within units" and "teamwork across units". Similarly, in Smits et al's (2008) study, a single item dropped from "hospital handoffs and transitions".

In the optimal model of the current study, there were only small shifts among items across factors. For instance, F6 (it is often unpleasant to work with staff from other hospital units) moved from "teamwork across units" to "handoffs and transitions", and F8 (the actions of hospital management show that patient safety is a top priority) moved from "management support for patient safety" to "organisational learning-continuous improvement". It seemed that this item movement (shifting) is acceptable because F6 is related to staff that work in same shift and shift changes. Also F8 is related because there is a link between patient safety as a top priority and learning from mistakes to improve patient safety. Similarly, shifting of items happened in different optimal models of studies. For example, F6 moved to handoffs and transitions dimension in optimal model derived from the Turkish data (Bodur and Filiz, 2010). F6 moved to communication openness in optimal model derived from the Dutch data (Smits et al., 2008). It seems that shifting of items depends on how staff understand the content of the items and the relationships between items.

In summary, the eight factor structures of the Saudi and the American HSOPSC versions are reasonably consistent. The overall structure of eight factors is the same and the composition of dimensions is approximately similar because the main part of the factor structures is the same. Table 7-3 shows the comparison between the structures of the American version and the Saudi version of eight factors of the HSOPSC questionnaire.

Eight factors (dimensions)	American version	Saudi version	Note	
No punitive response to error	A8®-A12®-A16®	A8®-A12®-A16®	Same	
Teamwork within units	A1-A3-A4-A11	A1-A3-A4	Lost one item (A11)	
Handoffs and transition	F5®-F7®-F11®-F3®	F5®-F7®-F11®-F6®	Lost one item (F3) and add one item (F6)	
Organisational learning- continues improvement	A6-A9-A13	A6-A9-A13-F8	Add one item (F8)	
Communication openness	C2-C4-C6	C2-C4	Lost one item (C6)	
Staffing	A5®-A7®-A2-A14®	A5®-A7®	Lost two items (A2-A14)	
Supervisor expectation and action promoting patient safety	B1-B2-B3®-B4®	B1-B2	Lost two items (B3-B4)	
Total	28	23	7 items lost and 2 items add	

# Table 7- 3 Comparison between the structures of the American version and the English language Saudi version of eight factors of the HSOPSC

® indicates negative items.

Table 7-3 shows that seven items were lost (A11-F3-C6-A2-A14-B3-B4) and two items (F6-F8) were added to the Saudi version across the eight factors. In total, the differences between two versions are five items. The items of each factor are theoretically structure related and the optimal model of the current study (Saudi version) demonstrates generally good psychometric properties. The construction of the optimal model of the current study was confirmed by using CFA. The following section discusses the CFA of the optimal model of the current study.

Following EFA, CFA was performed, using the validation half of the data, to test the fit of the resulting models that emerged from the EFA that was performed on the construction half of the data. These models were 11, 10, and 8 factor solutions. However, the values of the fit indices from CFA of 11 and 10 factors were outside the general agreed parameters as shown on page 174 (see appendix 47). Therefore, the model fit of both 11 and 10 factors were not satisfactory. Consequently these models were rejected. The fit indices from CFA of eight factors indicated that this fit model was satisfactory because the values of the fit indices of the Saudi data (8 factors) meet the general agreed parameters (Chi-square=407, DF=202, CFI >0.90, SRMR <0.08, RMSEA <0.06 and TLI>0.90). Consequently this model (eight factors) was acceptable as the final optimal model.

Finally, testing the fit of the optimal model for the whole data set was performed by using CFA. The result of this step also confirmed that the fit of the optimal model was appropriate to the whole Saudi data. Through this rigorous process the optimal model (eight factors) of the current study was tested carefully and confirmed via two confirmatory analyses. In terms of using CFA to confirm the optimal model, it seemed that an application of this procedure was not conducted in a number of studies (e.g. Smits et al., 2008; Alahmadi, 2009; Alahmadi, 2010; Bouder and Filiz, 2010; Pfeiffer and Manser, 2010). This situation might be referring to the complexity of CFA because most of studies used reliability analysis and correlations instead of CFA. However, some studies performed CFA to confirm their optimal model such as Waterson et al. (2009). According to Hoyle (1995) CFA is the best analysis procedure in terms of examining the factor structure of models.

With regard to reliability analysis, in the current study, the optimal model's internal consistency (reliability) was assessed using Cronbach's alpha ( $\alpha$ ). When different items are supposed to measure the same dimension, it is recommended that internal consistency (reliability) should be greater than or equal to 0.6 (Field, 2000). Since the original HSOPSC questionnaire (Sorra and Nieva, 2004) contains both positively worded items (n=24) and negatively worded items (n=18), the negatively formulated items were first recoded to make sure that a higher score always means a more positive response (Pallant, 2007).

The internal consistency was calculated for every factor according to the optimal model dimensions and items structure (8 factors) by using Cronbach's alpha values ( $\alpha$ ). The reliability analysis indicated that Cronbach's alpha values ( $\alpha$ ) of the factors are equal or above

0.68 for all factors. Factor 7 is low (<0.68). This factor consists of two items. Although three items and above upon each factor is the best, two items are acceptable especially if they are very high loadings items (above 0.5) and theoretically related. This factor is acceptable because the two items are high loading and the theoretical structure of this factor is related to staffing dimension. Furthermore, various fit indices in CFA of the optimal model met with the agreed parameters. It is important to mention that it is possible that a set of items will be below 0.7 on Cronbach's alpha ( $\alpha$ ), yet various fit indices in confirmatory factor analysis will be above the cut off (usually 0.9) levels. Alpha may be low when there are fewer items in the dimension (Kline, 2005). The value of Cronbach's alpha ( $\alpha$ ) depends on the number of items on the scale. If a number of items on the scale are increased thus Cronbach's alpha ( $\alpha$ ) will increase. Therefore, it is possible to get a large value of Cronbach's alpha ( $\alpha$ ) because a lot of items are present, and not because a scale is reliable (Field, 2009).

A number of studies found two items with low Cronbach's alpha ( $\alpha$ ). For example, the staffing dimension was 0.50 in Handler et al. (2006). Singer et al's (2007) study found Cronbach's alpha ( $\alpha$ ) ranged from 0.50 to 0.89 across the factors. However, it seemed that the application of reliability analysis was conducted in a number of studies (e.g. Smits et al., 2008; Waterson et al., 2009; Bodur and Filiz, 2010; Pfeiffer and Manser, 2010). According to Field (2009) it is useful to check the reliability of the scale in validation studies. Moreover, the application of reliability analysis may also refer to the simplicity of the reliability procedure compared with the CFA procedure.

In summary, the results of the EFA, CFA and reliability analysis indicate that eight factors solution is the optimal model of the current study. The composition of the optimal model was considered appropriate. The composition of the eight factors of the optimal model was similar to that of the American HSOPSC questionnaire (Sorra and Nieva, 2004) because the main part of the eight factors structure was unchanged. It seems that the psychometric analysis strategy of the current study contributes to the knowledge of patient safety climate measurement in terms of using rigorous assessment of psychometric properties of the original HSOPSC questionnaire in new settings. Moreover, it seems that there are common dimensions of patient safety culture across different countries (at international level). The next section examines a number of validation approaches and optimal models in different contexts in order to identify if there is a common patient safety culture dimensions.

#### Validation studies and optimal models in different contexts

The original HSOPSC questionnaire (Sorra and Nieva, 2004) has recently been validated in a number of different countries (e.g. the UK, Netherlands, Germany, and Turkey). Table 7-4 presents approaches of validation of the HSOPSC questionnaire of the current study compared with these countries.

Study	Current study	UK (Waterson et al., 2009)	Netherlands (Smits et al., 2008)	Germany (Pfeiffer and Manser, 2010)	Turkey (Bodur and Filiz, 2010)
Language	English	English	Dutch	German	Turkish
Face validity testing	Performed	Performed	Translation into Dutch	Translation into German	Translation into Turkish
Testing original model	CFA was conducted (a weak fit).	CFA was conducted (a weak fit).	CFA was conducted (not satisfactory fit).	CFA was conducted (a weak fit.	CFA was conducted (a weak fit.
	The internal consistency was calculated (Cronbach's alpha (α) was not satisfactory).	The internal consistency was calculated (Cronbach's alpha (α) was not satisfactory	The internal consistency was calculated (Cronbach's alpha (α) was not satisfactory).	The internal consistency was calculated (Cronbach's alpha (α) was Satisfactory scores).	The internal consistency was calculated (Cronbach's alpha (α) was not satisfactory).
Split data	Performed	Performed	Not performed	Not performed	Not performed
Optimal model	Eight factors	Nine factors	10 factors	Eight factors	11 factors
Confirmed optimal model	CFA was conducted. It was satisfactory fit.	CFA was conducted. It was satisfactory fit.	CFA was not conducted.	CFA was not conducted.	CFA was not conducted.
Reliability analysis for optimal model	The internal consistency was calculated with Cronbach's alpha (α).	The internal consistency was calculated with Cronbach's alpha (α).	The internal consistency was calculated with Cronbach's alpha (α).	The internal consistency was calculated with Cronbach's alpha (α).	The internal consistency was calculated with Cronbach's alpha (α).

### Table 7-4 Approaches of validation of the HSOPSC

.

Table 7-4 shows that studies are varied in terms of the extent validation approaches (psychometric assessment) conducted when using the HSOPSC questionnaire at an international level and assessing patient safety culture in different countries. However, there is adherence to recommended practice across these studies in terms of validation patient safety climate questionnaire including face validity testing, testing original model, split data, EFA, CFA and reliability analysis.

It seems that each country has developed a somewhat different optimal model. Table 7-5 shows the factor structure of the optimal model of the current study compared with optimal models that developed in these countries in addition to the original USA HSOPSC questionnaire (Sorra and Nieva, 2004). This comparison is aimed to identify the common dimension of patient safety culture across different countries.

Dimension	Original study (Sorra and Nieva, 2004)	The current study	UK study (Waterson et al., 2009)	Germany study (Pfeiffer and Manser, 2010)	Dutch study (Smits et al., 2008),	Turkey study (Bodur and Filiz, 2010)
Organisational learning – continuous improvement	$\overline{\checkmark}$	$\checkmark$		√    (Teamwork within unit)		
Frequency of event reported	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
Non punitive response to error	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
Handoffs and transitions	$\checkmark$	$\checkmark$	$\checkmark$	$\sqrt{  }$ (Teamwork across units)	$\checkmark$	$\checkmark$
Teamwork within units	$\checkmark$	$\checkmark$	$\checkmark$		$\checkmark$	$\checkmark$
Communication openness	$\checkmark$	$\checkmark$	$\checkmark$		$\checkmark$	
Supervisor expectation and action promoting patient safety	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
Staffing	$\checkmark$	$\checkmark$			$\checkmark$	$\checkmark$
Feedback and communication about error	√		$\checkmark$	√    (Communication openness)	$\checkmark$	$\sqrt{\parallel}$ (Communication openness)
Management support for patient safety	$\checkmark$			$\checkmark$	$\checkmark$	√    (Teamwork across units)
Overall perception of patient safety			√∥ (Staffing)	$\sqrt{\parallel}$ (Staffing)	$\checkmark$	$\checkmark$
Teamwork across units	$\checkmark$		$\checkmark$		$\checkmark$	
Total number of dimension	12	8	9	8	11	10

Table 7- 5 Common patient safety culture dimensions across different countries

 $\sqrt{Indicates dimension is included}$ 

||Indicates two dimensions combined in one dimension.

Two dimensions were combined into one dimension (factor) in three studies. For example, "staffing" and "overall perception of safety" combined as one factor in the UK study (Waterson et al., 2009). In the German study (Pfeiffer and Manser, 2010) "communication openness" and "feedback and communication about error" combined as one factor. "Organisational learning-continuous improvement" and "teamwork within units" also combined as one factor. "Staffing" and "overall perceptions of safety" combined as one factor factor. "Teamwork across units" and "handoffs and transitions" combined as one factor (Pfeiffer and Manser, 2010). In the Turkish study, "hospital management support for patient safety" and "teamwork across units" combined as one factor. "Communication openness" and "feedback and communication" combined as one factor. "Communication openness" and "feedback and communication" combined as one factor, while the other 8 factors work as separate factors (Bodur and Filiz, 2010). The current study has 8 factors and each factor works separately. Similarly the Dutch study has 11 factors and each factor works separately (Smits et al., 2008).

In three studies of the six studies in table 7-5, two dimensions combined in one dimension (items from two dimensions grouped together to form one dimension), still each dimension counts separately. Therefore, it appears that, there are a number of common dimension across these studies. The common patient safety culture dimensions at an international level are:

1 Frequency of event reported.

2 Non punitive response to error.

3 Communication openness.

4 Supervisor expectation and action promoting patient safety.

5 Handoffs and transitions.

6 Teamwork within units.

7 Staffing.

These dimensions are present in different countries, maybe because they are related to patient safety elements at an organisational level and not to patient safety practices at the individual level. Therefore, staff are varied in terms of their safe practice for patient care but they agreed about understanding the common dimensions of patient safety as organisational factors because safety climate is constituted by organisational factors (Neal et al., 2000).

The system approach that is based on investigation of organisational factors which lead to errors is widely used in organisations and is becoming increasingly used in healthcare to improve patient safety (Currie and Watt, 2007). Thus, errors in organisations usually happen because of poorly designed systems rather than weakly performing workers (IOM, 2000).

It appears that "teamwork across units", "overall perceptions of safety", "feedback and communication about error" and "management support for patient safety" were absent in the current study only, while, "organisational learning" was absent in two countries at the European level (UK and Netherlands). "Hospital management support for patient safety" was absent in the UK study and the current study. However, the common patient safety culture dimensions at the European level are:

1 Frequency of event reported.

2 Non punitive response to error.

3 Handoffs and transitions.

4 Communication openness.

5 Supervisor expectation and action promoting patient safety.

6 Teamwork within units.

7 Staffing.

8 Overall perceptions of safety.

9 Feedback and communication about error.

10 Teamwork across units.

However, it was difficult to identify the structure of these common dimensions across these countries because there was a lack of reporting of the structures of dimensions of some studies. Therefore, it was not possible to report and compare the structure of items and dimensions between these models in different countries. However, a number of studies mentioned that the composition of the dimensions of their optimal model was similar to that of the original HSOPSC questionnaire (Smits et al., 2008; Bodur and Filiz, 2010). This means, maybe, that the original model of the HSOPSC questionnaire (Sorra and Nieva, 2004) was not succeeding in a number of countries but found some dimensions corresponding to those proposed for the original HSOPSC questionnaire. It seems that the absence of number of patient safety climate dimensions such as organisational learning-continuous improvement and hospital management support for patient safety disagree with a number of studies that reviewed a number of patient safety climate questionnaires. For example, Fleming (2005)

mentioned that organisational learning-continuous improvement and hospital management support for patient safety are reported as common dimensions in a number of patient safety climate questionnaires. Flin et al. (2006b) stated that management commitment to safety is the most common dimension in nine patient safety climate questionnaires. Singla et al. (2006) mentioned that most of the questionnaires included four dimensions. These were: management commitment to safety, communication openness, beliefs about causes of errors and teamwork. However, this disagreement across reviews may be expected because in these review papers they compared patient safety culture dimensions across a number of different patient safety climate questionnaires. Meanwhile, in the current study the patient safety culture dimensions of the HSOPSC questionnaire (Sorra and Nieva, 2004) were compared across different countries to identify the common dimensions in different contexts. The difference arises because there are two different approaches to compare dimensions; one is comparing dimensions across different tools while the second is comparing dimensions of one tool across different countries. However, it is argued that there is a great value of comparing dimensions of one tool in different countries.

#### Summary of developing an optimal model

The current study is the first extensively validated study of a patient safety climate questionnaire in Saudi hospitals and it identifies the optimal model (eight factors) for assessing patient safety culture in Saudi Arabia. The current study indicated that the psychometric properties of the optimal model are good. The composition of the eight factors of the optimal model of the study was similar to that of the USA HSOPSC questionnaire (Sorra and Nieva, 2004) because the main part of the eight factor structure was unchanged. In general, the structure of the optimal model of the current study was considered appropriate. At the international level, it seemed that the approach to identifying an optimal model in different countries (e.g. the UK, Netherlands, Germany, and Turkey) is varied. However, these countries did in general follow recommended practice in terms of psychometric analysis strategy (EFA, CFA and reliability applications). The composition of the factors of the optimal models varies from country to country. There are some dimensions and items of the original HSOPSC questionnaire (Sorra and Nieva, 2004) that are absent in each country. In addition, some dimensions combined into one dimension and some items shifted to other dimensions. However, it does seem that there are common dimensions of patient safety culture at an international level. The next section will address some methodological issues from the current study.

### 7.5 Methodological issues

#### Negatively worded items

According to Marsh (1996) when a questionnaire includes both negative and positive items, then the proportion of negatively and positively worded items should be relatively constant for each factor. This creates balance in a scale and this balance helps to establish distinction between different factors. Most patient safety climate questionnaires include both negative and positive items (Hutchinson et al., 2006). For example, the original HSOPSC questionnaire consists of 12 factors (four positive factors, one negative factor, and seven factors are a mixture of both negative items and positive items). In total, it consists of 18 negative items and 24 positive items. The Teamwork and Safety Climate survey consists of 4 negative items and 23 positive items.

However, there is a debate in the literature regarding negatively and positively worded items that construct the factor structure of questionnaire. A number of studies argued that negative items correlate with positive items to form a separate factor (e.g. Spector et al., 1997). On the other hand, other studies argued that usually a minority of the items in most questionnaires are negative items and they are grouped together as a separate factor (e.g. Schmitt and Coyle, 1976; Schmitt and Stults, 1985). A large body of literature indicates that separate factors are associated with negatively and positively worded items. In other words, negative items are grouped together (correlated with each other) and positive items are grouped together to form a separate factor (Marsh, 1996).

The optimal model of the current study consists of both negative and positive items (nine negative items and 14 positive items). The optimal model consists of eight factors (three negative factors and five positive factors). Moreover, the negative items were grouped together in a number of dimensions such as "no punitive response to error", and "handoffs and transitions". The positive items were grouped together in a number of dimensions such as "organisational learning-continuous improvement", "frequency of event reported" and "supervisors" (see table 6-39 page 216). This result is consistent with the views of Schmitt and Stults (1985); Marsh (1996); and Spector et al. (1997).

The results of the current study differ from the UK study conducted by Waterson et al. (2009), who found four factors in their optimal model (consisting of both negative and positive items) out of nine factors overall. In some studies negative items were not used in data analysis based on responses of respondents. For example, Hutchinson et al. (2006) found that a number of respondents in their study failed to recognise the negative items of the Teamwork and Safety Climate survey. Therefore, four negatively worded items were removed from the final stage of factor analysis.

However, in the current study, it seemed that the respondents were able to distinguish between positive and negative items as mentioned in the face validity stage and the respondents did not experience difficult in interpreting negatively worded items. Therefore, it was decided that there was no need to direct participants' attention towards negative items by using a mark such as underlining. This finding was also reflected later in the psychometric analysis of negative items because the optimal model of the current study contains both negatively and positively worded items as separate factors and they did not cause problems in terms of interpreting the results. In general, the respondents in the current study distinguished between positively worded and negatively worded items and they succeeded to notice negatively worded items without any difficulty. Therefore, the negatively worded items were included in the factor analysis of the study.

#### Reporting psychometric properties of patient safety climate questionnaires

In terms of reporting psychometric data, it seemed that there is a lack of reporting the psychometric properties of patient safety climate questionnaires in the published literature, even though the need for analysing and reporting psychometric properties of patient safety climate questionnaires has been stressed by several authors (Nieva and Sorra, 2003; Colla et al., 2005; Hutchinson et al., 2006; Flin et al., 2006b). According to Nieva and Sorra (2003) more evidence is needed in relation to reporting the psychometric analysis and data of patient safety climate questionnaires. In Singla et al's (2006) review no psychometric testing had been reported for eight questionnaires from their review of 13 questionnaires.

Similarly, Flin, et al. (2006b) mentioned that the main limitation of their review was the lack of provision of psychometric data in publication of results of patient safety questionnaire use. A possible explanation of this limitation might be due to complexity of factor analysis, especially confirmatory factor analysis.

Chapter seven

Alternatively, it might be due to a misconception about assessing the psychometric properties of patient safety climate questionnaires in some studies, in particular, what psychometric analysis should be performed and how to report the psychometric data of questionnaires. It appears there is a need for better practice regarding psychometric properties of patient safety climate questionnaires in terms of reporting them properly. According to Flin et al. (2006b) it is becoming increasingly important to obtain information about the psychometric properties of patient safety climate questionnaires. Therefore, the psychometric analysis strategy of the current study was performed according to a recommended standard, to ensure a rigorous scientific approach. Moreover, the current study reports the details about the assessment of the psychometric properties of the HSOPSC questionnaire with the Saudi data. The next section will discuss the strengths and limitations of the current study.

### 7.6 Study Strengths and limitations

This study has some strengths and limitations that are discussed in this section. A mixed method approach was used. Initial assessment of the questionnaires was conducted. Three hospitals were included in the study for data collection, and 862 completed questionnaires were returned so the data was tested in a large sample. An intensive advanced psychometric assessment including EFA and CFA was performed. The rigorous psychometric analysis strategy was performed according to recommended practice to validate the psychometric properties of the HSOPSC questionnaire for use in Saudi hospitals.

This study had a number of challenges such as time available for data collection; especially as the data collection was divided in two stages. Firstly, hospital agreements were obtained and 12 interviews were conducted in Saudi Arabia. Then, the researcher returned to the place of the study (University of Sheffield in the UK) to discuss and to confirm the decisions and the results of the face validity stage with the supervisors before going back to Saudi Arabia to collect the data from Saudi hospitals. Secondly, the HSOPSC questionnaires (Sorra and Nieva, 2004) were distributed and collected in three hospitals in Saudi Arabia. All these stages lasted around six months. However, the most practical challenge of this study was achieving a high HSOPSC questionnaires return rate among hospitals staff.

The study encountered some challenges during the data collection stage. For example, summer holidays of staff during the data collection of the current study, shortage of staff in certain departments and a delay in ethics review in the third hospital due to the establishment of new regulations there.

A theoretical challenge encountered in the current study was a lack of reporting of explicit psychometric data in some relevant studies used in the comparative analyses. The measurement of patient safety climate is a dynamic growing field and includes some limitations such as not reporting currently patient safety climate questionnaires under development but not published (Colla et al., 2005). Some aspects of the questionnaires such as psychometric properties may have been performed but not yet published. Some versions of questionnaires are not available (Flin et al., 2006b).

The current study may not have included all relevant existing patient safety climate questionnaires in the literature review; however, it included the most widely used questionnaires which are representative of the patient safety climate questionnaires currently available in the published literature. Moreover, the current study discussed key studies that report data about the psychometric properties of patient safety climate questionnaires.

This research was carried out as a PhD project; therefore, all these stages were undertaken by the researcher. However, all the stages of the current study was discussed and supervised by research supervisors to ensure the quality of the study including developing the research question, aim and objectives, the methodology, the data collection and data analysis.

The researcher received statistical analysis training and support course to aid with their psychometric analysis of the HSOPSC questionnaire. Written materials provided information on correlation, validity, reliability, data splitting, EFA and CFA. Workshops were also conducted on factor analysis (EFA and CFA) and the use of AMOS software was conducted. The course instructor explained how to undertake both EFA and CFA. The research was undertaken independently. Follow up meetings were held to ensure that the psychometric analysis (CFA and EFA) were performed correctly and satisfactorily and to discuss the results.

### 7.7 Summary of the main findings and impact of the study

The current chapter offers a general overview of the most significant findings of the study and provides an interpretation of these findings in the context of the patient safety culture measurement. The originality and the research contributions of the study findings are discussed in this chapter.

In general, the current study followed recommended practice in terms of assessing the psychometric properties of patient safety climate questionnaires from key studies (e.g. Hutchinson et al., 2006; Smits et al., 2008; Waterson et al., 2009; Bouder and Filiz, 2010; Pfeiffer and Manser, 2010). For example, the current study established the face validity of the two patient safety climate questionnaires for comparison (Hutchinson et al., 2006). It also used CFA and reliability analysis when testing the original HSOPSC questionnaire in Saudi hospitals. A split half validation strategy (EFA followed by CFA) was performed as it is recommended by a number of studies for providing realistic results of the psychometric properties of patient safety climate questionnaires (e.g. Waterson et al., 2009). Finally, an extensive EFA, CFA and reliability analysis was performed in order to identify the optimal model for Saudi hospitals.

The current study provides four main findings. Firstly, in terms of the importance of face validity, the current study demonstrated that establishing face validity of a patient safety climate questionnaire is very important and an essential element before wider application in a new context. Secondly, the current study demonstrated that the factor structure of the original USA HSOPSC questionnaire (12 factors and 42 items) is not valid and reliable when used with the English language Saudi data collected in this study because the psychometric properties of the original HSOPSC questionnaire may be effectively measuring different dimensions of patient safety culture within Saudi Arabia, as compared to the USA (Sorra and Nieva, 2004). Therefore, the factor structure of the original USA HSOPSC model was not suitable for use in Saudi hospitals.

Thirdly, the results of psychometric analysis (EFA, CFA and Reliability analysis) showed that eight factors was the optimal model for the Saudi data. The optimal model could be used in Saudi hospitals to assess patient safety culture. Finally, the current study compared number of optimal models in different countries. The comparison shows the seven common patient safety culture dimensions in different contexts. The next chapter will provide the conclusion of the current study and recommendation for the future use of the HSOPSC questionnaire in Saudi Arabia.

## **Chapter eight: Conclusion**

#### 8.1 Introduction

This chapter concludes this thesis by summarising the key findings in relation to the research question, aim and objectives of the study. In section 8.2 the implications of the study are presented, including theoretical implications, practical implications and recommendations for future research of patient safety culture measurement, in particular, patient safety culture measurement in Saudi Arabia. Finally, in section 8.3 the conclusion of this chapter is provided.

The current study investigated whether there was an existing patient safety climate questionnaire that would be suitable for assessing patient safety culture in hospitals in Saudi Arabia. The main aim of the study was to identify a suitable measure for assessing patient safety culture for use in hospitals in Saudi Arabia. In this investigation four main objectives needed to be met in order to achieve the aim of the study. The main objectives of the study were:

- 1 To select an appropriate questionnaire to assess hospital patient safety culture.
- 2 To evaluate the face validity of the selected patient safety climate questionnaire.
- 3 To assess the psychometric properties of the selected patient safety climate questionnaire in hospitals in Saudi Arabia.
- 4 To develop the most appropriate measure for assessing patient safety culture for use in hospitals in Saudi Arabia.

In terms of achievement of the objectives, a mixed methods study design has been employed to identify a suitable patient safety climate questionnaire for use in hospitals in Saudi Arabia. The original HSOPSC questionnaire (Sorra and Nieva, 2004) and Teamwork and Safety Climate Survey (Sexton et al., 2004) were identified as appropriate questionnaires in the first stage of the research. The second objective was testing the face validity of the two questionnaires above for comparison.

The original HSOPSC questionnaire (Sorra and Nieva, 2004) was the most appropriate questionnaire for use in the current study although some changes were made to the wording. Thirdly, this questionnaire was used to collect the data from three Saudi Arabian hospitals using 862 completed questionnaires, and finally, the evaluation of the psychometric properties of the HSOPSC using Saudi data.

In terms of findings, the current study provides three main findings. Firstly, the current study demonstrated that establishing face validity of a patient safety climate questionnaire is very important and an essential element before the full application of a questionnaire. Secondly, the current study demonstrated that the factor structure of the original USA HSOPSC questionnaire (12 factors and 42 items) (Sorra and Nieva, 2004) was not valid and reliable in Saudi hospitals. Thirdly, the current study developed an English language version of the HSOPSC questionnaire, consisting of eight patient safety culture dimensions, as an optimal model which can be used in Saudi hospitals for assessing patient safety culture.

### 8.2 Implications of the study

This research contributes to the measurement of patient safety culture in three ways: contribution to the current knowledge (theoretical implications), implications for practice and implications for future research (empirical implications).

#### **Contribution to the current knowledge**

In terms of theoretical implications, measuring safety climate in healthcare is still a developing science. Therefore, there is a need to measure patient safety climate in different healthcare environments carefully to avoid the risk of inappropriately applying patient safety climate questionnaires in a different environment (Waterson et al., 2009). The current study highlights the necessity of assessing the face validity of patient safety climate questionnaire before full application in different settings. It indicates that there is a need for caution in using the original English language version of the HSOPSC questionnaire (Sorra and Nieva, 2004) in Saudi Arabia and underlines the importance of appropriate validation of patient safety climate questionnaires before extending their usage in healthcare contexts different from those in which they were developed.

The current study shows that the HSOPSC questionnaire may be measuring different dimensions of patient safety culture within Saudi Arabia, as compared to the USA (Sorra and Nieva, 2004). In addition, the HSOPSC questionnaire may also be measuring different dimensions of patient safety culture within different countries such as the UK, Netherlands, Germany and Turkey as compared to the USA (Sorra and Nieva, 2004). Moreover, the current study provides international comparative information regarding the use of the HSOPSC questionnaire in different countries. This comparison has identified the seven common patient safety culture dimensions of this questionnaire across these countries.

#### **Implications for practice**

In terms of practical implications, based on this study, the original English language version of the HSOPSC questionnaire (Sorra and Nieva, 2004) is not appropriate for use in Saudi hospitals, while the validated Saudi English language version of the HSOPSC questionnaire can be used to assess patient safety culture in the Saudi hospitals. The optimal model of the current study (Saudi English language version of the HSOPSC) demonstrates generally good psychometric properties. The eight factor structures of the Saudi and the American HSOPSC versions are reasonably consistent because the overall structure of eight factors is the same and the composition of dimensions is approximately similar. This study demonstrates that the Saudi English language version of the HSOPSC questionnaire is an appropriate patient safety climate questionnaire to assess patient safety culture in Saudi hospitals.

#### Implications for future research

Several areas for further research have emerged from the findings of the current study. In terms of empirical implications, the current study highlights that there is a weakness in reporting psychometric properties and psychometric analysis in the relevant literature. More safety climate questionnaire validation studies are needed that include the reporting of all the assessment data regarding the psychometric properties of patient safety climate questionnaires. Additional studies on patient safety culture assessment in Saudi healthcare using the validated Saudi English language version of the questionnaire developed in the current study will help to provide better understanding of patient safety in Saudi Arabia.

### 8.3 Conclusion

The current study is one of few studies to provide an evaluation of an American patient safety climate questionnaire using data from Saudi Arabia and applying a rigorous psychometric analysis strategy. It also appears to be the first study to identify the Saudi English language version of the HSOPSC questionnaire as the optimal model for assessing patient safety culture in Saudi hospitals. In conclusion, the findings of the current study contribute to the measurement of patient safety climate in healthcare organisations. Moreover, the current study established a basis for measuring the patient safety culture in Saudi hospitals, thereby potentially contributing to creating a safer environment for hospital patients in Saudi Arabia.

# References

Abdullah, N, Spickett, J, Rumchev, K, & Dhaliwal, S. (2009). "Validity and reliability of the safety climate measurement in Malaysia". *International Review of Business Research Papers*, 5, 111-141.

Alahmadi, H. (2010). "Assessment of patient safety culture in Saudi Arabia hospitals". *Quality and Safety in Health Care*, 10, 1-5.

Alahmadi, T. (2009). "Measuring patient safety culture in Riyadh hospitals: A comparison between public and private hospitals". *Egypt Public Health Association*, 84 (5-6) 479-500.

Alanzay, S. (2006). "Factors associated with implementation of electronic health records in Saudi Arabia". PhD, University of Medicine and Dentistry of New Jersey.

Amanda, L, Michael, S, & Meurer, S. (2007). "Development of a patient safety culture measurement tool for ambulatory health care settings: analysis of content validity". *Health Care Management Science*, 10, 139-149.

Arah, O. & Klazinga, N. 2004, "How safe is the safety paradigm?". Quality & Safety in Health Care, 13, 226-232.

Arbuckle, J. (2005). Amos 6.0 users guide IL: SPSS. Chicago: Inc.

Arezes, P. & Miguel, A. (2003). "The role of safety culture in safety performance measurement". *Measuring Business Excellence*, 7 (4) 20-28.

Baily, C. & Petersen, D. (1989). "Using perception surveys to assess safety system effectiveness". *Professional Safety*, 10 (2) 277-295.

Bartlett, M. 1954, "A note on multiplying factors for various chi-squared approximations", Journal of the Royal Statistics Society, 16, 296-298.

Beck, M. & Woolfson, C. (1999). "Safety culture: a concept too many". The Safety & Health Practitioner, 14-16.

Beuzekom, M, Akerboom, S, & Boer, F. (2007). "Assessing system failures in operating rooms and intensive care units", *Quality and Safety in Health Care*, 16, 45-50.

Bodur, S. & Filiz, E. (2010). "Validity and reliability of Turkish version of Hospital Survey on Patient Safety Culture and perception of patient safety in public hospitals in Turkey", *BMC Health Services Research*, 10 (8), 1-9.

Bone, P, Sharma, S, & Shimp, T. (1989) "A bootstrap procedure for evaluating goodness-offit indices of structural equation and confirmatory factor models", *Journal of Marketing Research*, 26 (1), 105-111. Bourque, L & Fielder, E. (1995). How to conduct self-administered and mail survey. USA: Sage.

Bowling, A. (2005). "Techniques of questionnaire design," in Handbook of health research methods investigation, measurement and analysis, Open University Press, England, pp. 394-427.

Brennan, T, Leape, L, Laird, N, Hebert, L, Localio, A, Lawther, A, Newhouse, J, Weiler, P, & Hiatt, H. (1991). "Incidence of adverse events and negligence in hospitalized patients. Results of the Haravrd Medical Practice Study I", *The new England journal of medicine*, 324 (6), 370-376.

Brown, R. & Holmes, H. (1986). "The use of a factor-analytic procedure for assessing the validity of an employee safety climate model", *Accident anlysis and prevention*, 18 (6), 455-470.

Burr, M, Sorra, J, & Nieva, V. (2002). Analysis of the Vetrans Administration(VA) National Centre for Patient Safety(NCPS) FY 2000 Patient Safety Questionnaier, Westat.

Campbell, M, Machin, D, & Walters, S. (2007). *Medical Statistics: A textbook for health science (medical statistics)*. 4<sup>th</sup> ed. England: John Wiley& Sons Ltd.

Carayon, P, Alvarado, C, Hudt, A, Springman, S, Borgsdorf, A, & Hoonakker, P. (2005). "An employee questionnaire for assessing patient safety in outpatient safety surgery", *Advances in Patient Safety*, 4, 461-473.

Carmines, E. & Zeller, R. (1979). Reliability and validity assessment. London: Sage publications.

Cattell, R. (1966). "The scree test for the number of factors", Multivariate Behavioral Research, 1 (2), 245-276.

Cheyne, A, Cox, S, Oliver, A, & Tomas, J. (1998). "Modelling safety climate in the prediction of levels of safety activity", *Work& Stress*, 12 (3), 255-271.

Cheyne, A, Oliver, A, Tomas, J & Cox, S. (2002). "The architecture of employee attitudes to safety in the manufacturing sector", *Personnel Review*, 31 (5-6), 649-670.

Choudhry, R, Fang, D, & Mohamed, S. (2007). "The nature of safety culture: A survey of the state-of-the-art", *Safety Science*, 45, 993-1012.

Clarke, S. (1998). "Safety culture on the U k railway network", Work& Stress, 12 (3) 285-292.

Clarke, S. (1999). "Perceptions of organizational safety: implications for the development of safety culture", *journal of Organizational Behaviour*, 20, 185-198.

Clarke, S. (2003). "The contemporary workforce implications for organisational safety culture.", *Personnel Review*, 32 (1), 40-57.

Coakes, S. & Steed, L. (2003). SPSS Analysis without anguish version 11 for windows Wiley, Australia.

Colla, J, Bracken, A, Kinney, M, & Weeks, W. (2005). "Measuring patient safety climate: a review of surveys.", *Quality & Safety in Health Care*, 14 (5) 364-366.

Collins, D. (2003). "Pretesting survey instruments: An overview of cognitive methods", *Quality of Life Research*, 12, 229-238.

Comrey, A. & Lee, H. (1992). A first course in factor analysis. Lawrence Erlbaum, Hillsdale, NJ.

Conway, J. & Allen, H. (2003). "A review and evaluation of exploratory factor analysis practices in organizational research", *Organizational Research Methods*, 6, 147-167.

Cooke, A. & Rousseau, D. (1998). "Behavioral Norms and Expectations a quantitative approach to the assessment of organisational culture", *Group & Organization Studies*, 13 (3) 254-273.

Cooper, M. & Phillips, R. (2004). "Expolratory analysis of the safety climate and safety behavior relationship", *Journal of Safety Research*, 35, 497-512.

Cooper, M. & Philips, R. (1994). "Validation of a safety climate measure", in *The BPS Occupational Psychology Conference*.

Cooper, M. (2000). "Towards a model of safety culture", Safety Science, 36, 111-136.

Costello, A. & Osborne, J. (2005). "Best practices in exploratory factor analysis: Four recommendations for getting the most from your analysis", *Practical Assessment, Research & Evaluation*, 10 (7) 1-9.

Cox, S. & Cheyne, A. (2000). "Assessing safety culture in offshore environments", Safety Science, 34, 111-129.

Cox, S. & Cox, T. (1991). "The structure of employee attitudes to safety: A European example.", *Work & Stress*, 5, 93-104.

Cox, S. & Flin, R. (1998). "Safety culture: philosopher's stone or man of straw?", Work and Stress, 13 (12) 189-201.

Cronbach, L. (1951). "Coefficient Alpha and the internal structure of tests", *Psychometrika*, 16 (3) 297-333.

Currie, L. & Watt, S. (2007). "An Introduction to Patient Safety," in Understanding Patient Safety, L. Currie, ed., London: MA Healthcare Limited.

Dalling, I. (1997). "Understanding and assessing safety culture", *Journal Radiol prot*, 17 (4) 261-274.

Day, S, Fayers, P, & Harvey, D. (1998). "Double data entry: What value, what price?", Controlled Clinical Trials, 19 (1) 15-24.

Dean, B, Schachter, M, Vincent, C, & Barber, N. (2002). "Causes of prescribing errors in hospital inpatients: a prospective study", *THE lANCENT*, 359, 2002.

Dedobbeleer, N. (1991). "A safety climate measure for construction sites", Journal of Safety Research, 22, 97-103.

Deilkas, E. & Hofoss, D. (2008). "Psychometric properties of the Norwegian version of the Safety Attitudes Questionnaire (SAQ), generic version (short form 2006)", *BMC Health Services Research*, 8 (191) 1-10.

Denison, D. (1996). "What is the difference between Organizational Culture and Organizational Climate? A Natives Point of View on a Decade of Paradigm Wars", *The Academy of Management Review*, 21 (3) 619-654.

Drennan, J. (2003). "Cognitive interviewing: verbal data in the design and pretesting of questionnaires", *Advanced Nursing*, 42 (1), 57-63.

Ehsani, J, Jackson, T, & Duckett, S. (2006). "The incidence and cost of adverse events in Victorian hospitals 2003-04", MJA, 184 (11) 551-555.

Etchegaray, J. & Fischer, W. (2006). "Survey research: be careful where you step", *Quality & Safety in Health Care*, 15, 154-155.

Everitt, S. (1975). "Multivariate analysis: The need for data, and other problems", British Journal of Psychiatry, 126, 237-240.

Fabrigar, L, Wegener, D, MacCallum, R, & Strahan, E. (1999). "Evaluating the use of exploratory factor analysis in psychological research", *Psychological Methods*, 4, 272-299.

Fernandez-Muniz, B, Montes-Peon, J, & Vazquez-Ordas, C. (2007). "Safety culture: Analysis of the causal relationships between its key dimensions", *Journal of Safety Research*, 38, 627-641.

Field, A. (2000). Discovering statistics using SPSS for windows: advanced techniques for the beginner. London: Sage publications.

Field, A. (2009). Discovering statistics using SPSS. London: Sage publications.

Fink, A. (1995). The survey handbook. 2nd ed. London: Sage publications.

Fink, A. (2003). How to Sample in Surveys. 2<sup>nd</sup> ed. London: Sage publications.

Firth-Cozens, J. (2001). "Cultures for improving patient safety through learning: the role of teamwork", *Quality and Safety in Health Care*, 10, 26-31.

Fleming, M. (2005). "Patient safety culture measurement and improvement: a "how to" guide", *Healthcare Quarterly*, 8 Spec No, 14-19.

Fleming, M. & Wentzell, N. (2008). "Patient safety culture improvement tool: Development and guidelines for use", *Healthcare Quartely*, 11, 10-15.

Flin, R, Mearns, K, Connor, P, & Bryden, R. (2000). "Measuring safety climate: identifying the common features", *Safety Science*, 34, 177-192.

Flin, R. (2007). "Measuring safety culture in healthcare: A case for accurate diagnosis", *Safety Science*, 45 (6) 653-667.

Flin, R, Burns, C, Mearns, K, Yule, S, & Robertson, M. (2006). "Measuring safety climate in health care", *Quality and Safety in Health Care*, 15, 109-115.

Flin, R, Mearns, C, Gordon, R, & Fleming, M. (1996). "Risk perception by offshore workers on UK oil and gas platforms", *Safety Science*, 22 (1-3) 131-145.

Flin, R, Fletcher, G, McGeorge, P, Sutherland, A, & Patey, R. (2003). "Anaesthetists' attitudes to teamwork and safety", *Anaesthesia*, 58 (3) 233-242.

Flin, R. & Yule, S. (2003). "Advances in patient safety: non-technical skills in surgery", *Surgeon News*, 4 (3) 83-86.

Flin, R, Yule, S, McKenzie, L, Paterson-Brown, S, & Maran, N. (2006a). "Attitudes to teamwork and safety in the operating theatre", *Surgeon Journal of the Royal Colleges of Surgeons of Edinburgh & Ireland*, 4, (3) 145-151.

Flin, R, Burns, C, Mearns, K., Yule, S, & Robertson, E. M. (2006b). "Measuring safety climate in health care. ", *Quality & Safety in Health Care*, 15 (2) 109-115.

Floyd, F. & Widaman, K. (1995). "Factor analysis in the development and refinement of clinical assessment instruments", *Psychological Assessment*, 7, 286-299.

Flynn, E. (2004). "Summary of crossing the quality chasm: a new system for the 21st century," in *The patient safety handbook*, B. Youngberg & M. Hatlie. eds, London: Jones and Bartlett Publisher International.

Ford, J, MacCallum, R, & Tait, M. (1986). "The application of exploratory factor analysis in applied psychology: A critical review and analysis", *Personnel Psychology*, 39, 291-314.

Gadd, S. & Collins, A. (2002). Safety Culture: A review of the literature, Health & Safety Laboratory. Sheffield, HSL/2002/25.

Glendon, A. & Stanton, N. (2000). "Percepectives on safety culture", *Safety Science*, 34, 193-214.

Gonzales-Roma, V., Peiro, J., Lioret, S., & Zornoza, A. (1999). "The validity of collective climates", *Journal of Occupational and Organizational Psychology*, 72, 25-40.

Gorsuch, R. (1983). Factor analysis. 2<sup>nd</sup> ed. NJ: Lawrence Erlbaum, Hillsdale.

Gorsuch, R. (1997). "Exploratory factor analysis: its role in item analysis", Journal of Personality Assessment, 68, 532-560.

Guadagnoli, E. & Velicer, W. (1988). "Relation of Sample Size to the Stability of Component Patterns", *Psychological Bulletin*, 103 (2) 265-275.

Guldenmund, F. (2000). "The nature of safety culture: a review of theory and research", *Safety Science*, 34, 215-257.

Guldenmund, F. (2007). "The use of questionnaires in safety culture research-an evaluation", *Safety Science*, 45, 723-743.

Hair, J, Anderson, R, Tatham, R, & Black, W. (1998). Multivariate data analysis. 5th ed. New Jersy: Prentice Hall.

Hale, A. (2000). "Editorial:culture confusions", Safety Science, 34, 193-214.

Handler, S, Castle, N, Studenski, S, Perera, S., Fridsma, D, Nace, D, & Hanlon, J. (2006). "Patient safety culture assessment in the nursing home", *Quality and Safety in Health Care*, 15, 400-404.

Harvey, J, Erdos, G, Jackson, H, & Dennison, S. (2004). "Is safety culture in differing organizations the same thing? a comparison of safety culture measures in three organizations.", *Risk Decision and Policy*, 9, 391-400.

Havold, J. & Nesset, E. (2009). "From safety culture to safety orientation: validation and simplification of a safety orientation scale using a sample of seafarers working for Norwegian ship owners", *Safety Science*, 47 (3) 305-326.

Health and Safety Commission (HSC) (1993). Organizing for safety: Third report of the human factors study group of ACSNI., Sudbury: HSE Books.

Health and Safety Excutive (HSE) (2005). A review of safety and safety climate literature for the development of the safety culture inspection toolkit. Research report 367.

Health and Safety Excutive (HSE) (2007). Health and safety statistics 2007/2008.

Helen, M. (2010). The effect of disease adaptation information on general population values: A case study using rheumatoid arthritis states, PhD, University of Sheffield.

Hellings, J, Schrooten, W, Klazinga, N, & Vleugels, A. (2007). "Challenging patient safety culture: survey results", *International Journal of Health Care Quality Assurance*, 20 (7) 620-632.

Helmreich, R, Sexton, J. B, & Merritt, A. (1997). The operating room managemnt attitudes questionnaire (ORMAQ), The university of Texas, Austin, Texas, 97-6.

Helmreich, R. & Merrrit, A. (1998). Culture at work in aviation and medicine: national, organizational, and professional influences. Aldershot, UK: Ashgate.

Helmreich, R. (2000). "On error management: lessons from aviation", British Medical Journal, 320, 781-785.

Hofmann, D, Jacobs, R, & Landy, F. (1995). "High Reliability Process Industries: Individual, Micro, and Macro Organizational Influences on Safety Performance", *Journal of Safety Research*, 26 (3) 131-149.

Hofmann, D. & Stezer, A. (1996). "A cross-level investigation of factors influencing unsafe behaviors and accidents", *Personnel Psychology*, 49, 307-339.

References

Hofmann, D. & Stetzer, A. (1998). "The role of safety climate and communication in accident implications for learning from negative events", *Academy of Management Journal*, 41 (6) 644-657.

Hsu, M, Ju, T, Yen, C, & Chang, C. (2007). "Knowledge sharing behaviour in virtual communities: The relationship between trust, self-efficacy, and outcome expectations", Human-Computer studies, 65, 153-169.

Hoyle, R. (1995). Structural Equation Modelling: Concepts, Issues, and Applications. London: Sage publications.

Hu, L. & Bentler, P. (1998). "Fit indices in convariance structure modeling: sensitivity to underparameterized model misspecification", *Psychological Methods*, 4, 424-453.

Hu, L. & Bentler, P. (1999). "Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives", *Structural Equation Modeling: A Multidisciplinary Journal*, 6, (1) 1-55.

Hudson, P. (2003). "Applying the lessons of high risk industries to health care", *Quality Safe Health Care*, 12 7-12.

Hutchinson, A, Cooper, KL, Dean, JE, McIntosh, A, Patterson, M., Stride, CB, Laurence, BE., & Smith, CM. (2006). "Use of a safety climate questionnaire in UK health care: factor structure, reliability and usability", *Quality & Safety in Health Care*, 15 (5) 347-353.

Institute for Safety Medication Practices. (2000). Medication Safety Self-Assessment for Community/Ambulatory Pharmacy. [Online]. Available from: www.ismp.org/survey [Accessed 15th December 2007].

Institute of Medicine (2000). To err is human: building a safer health system. Washington DC: National Academy Press.

Institute of Medicine (2004). *Patient safety: Achieving a new standard for care*. Washington DC: National Academy Press.

James, H. & Thrall, M. (2004). "Quality and safety revolution in health care", *Radiology*, 233, 3-6.

James, L, Hater, J, Gent, M, & Bruni, J. (1978). "Psychological climate: implications from cognitive social learning theory and interactional psychology", *Personnel Psychology*, 31, 783-811.

Johnson, C. (2002). "The causes of human error in medicine", *Cognition, Technology & work*, 4, 65-70.

Johnson, K. (2004). "Keeping patients safe: an analysis of organisational culture and caregiver training", *Journal of healthcare management/American College*, 49 (3) 171-178.

Joreskog, K. & Sorbom, D. (1986). LISREL VI: analysis of linear structural relationships by maximum likelihood, instrumental variables, and least square methods. 4<sup>th</sup> ed. Sweden: University of Uppsula Department of Statistics.

Joreskog, K. & Sorbom, D. (1993). Lisrel 8: the simplis command language Scientific Software International, Lincolnwood.

Kaissi, A, Johnson, T, & Kirschbaum, M. (2003). "Measuring teamwork and patient safety attitudes of high-risk areas", *Nurse Economics*, 21 (5) 211-218.

Kho, M. E, Carbone, J. M, Lucas, J, & Cook, D. J. (2005). "Safety Climate Survey: reliability of results from a multicenter ICU survey", *Quality & Safety in Health Care*, 14 (4) 273-278.

Kim, J. & Mueller, C. (1978). Factor analysis statistical Methods and Practical Issues, 5<sup>th</sup> ed, London: Sage publications.

Kirk, S, Parker, D, Claridge, T, Esmail, A, & Marshall, M. (2007). "Patient sfety culture in primary care: developing a theoretical framework for practical use", *Quality & Safety in Health Care*, 16, 313-320.

Kirk, S, Marshall, M, Claridge, T, Esmail, A., & Parker, D. (2006). "Evaluating safety culture," in *Patient safety research into practice*, K. Walshe & R. Boaden, eds., Open University Press, Berkshire.

Kline, P. (1994). An easy guide to factor analysis, London: Routledge.

Kline, R. (2005). *Principles and practices of structural equation modelling*. 2<sup>nd</sup> ed. New York: Guilford Press.

Leape, L, Brennan, T, Laird, N, Lawther, A, Localio, A, Hebert, L, Newhouse, J, Weiler, P, & Hiatt, H. (1991). "The nature of adverse events in hospitalized patients. Results of the Harvard Medical Practice Study II", *The New England journal of medicine*, 324 (6) 377-384.

Leape, L. (1993). "Preventing medical injury", Qual Rev Bull, 19 (5) 144-149.

Leape, L. & Berwick, D. (2000). "Safe health care: are we up to it?", BMJ. 725-726.

Leape, L. (1994). "Error in medicine", JAMA, 272, 1851-1857.

Lee, T. (1996). "Perceptions, attitudes and behaviour: the vital elements of a safety culture", *Health and Safety* Bulletin, October, 1-15.

Lee, T. & Harrison, K. (2000). "Assessing safety culture in nuclear power stations", Safety Science, 34, 61-97.

Lievens, F. & Anseel, F. (2004). "Confirmatory factor analysis and invariance of an organisational citizenship behaviour measure across samples in a Dutch-speaking context", *Journal of Occupational and Organizational Psychology*, 77, 299-306.

MacCallum, R, Widman, K, Zhang, S, & Hong, S. (1999). "Sample size in factor analysis", *Psychological Methods*, 103, 391-410.

March, H. (1996). "Positive and negative global self-esteem: A substantively meaningful distinction or artifactors?", *Journal of Personality and social psychology*, 70, 810-819.

Marsh, H. & Balla, J. (1994). "Goodness of fit in confirmatory factor analysis: the effect of sample size and model parsimony", *Quality and Quantity*, 28, 185-217.

Mary, B, Ginette, P, & Joseph, R. (2005). "Safety climate on hospital units: A new measure", *Advances in Patient Safety: From Research to Implementation*, 4, 415-427.

Mckeon, L, Oswaks, J, & Cunningham, P. (2006,)."Safeguarding Patients: Complexity Science, High Reliability Organizations, and Implications for Team Training in Healthcare", *Ajournal for Advanced Nursing Practice*, 20 (6) 298-304.

Mearns, K, Flin, R, Gordan, R, & Fleming, M. (1998). "Measuring safety climate on offshore installations", *Work& Stress*, 12 (3) 238-254.

Mearns, K. & Flin, R. 1999, "Assessing the state of organizational safety culture or climate?", *Current Psychology*, 18 (1) 5-17.

Mearns, K, Flin, R, & OConnor, P. (2001). "Sharing worlds of risk; improving communication with crew resource management", *Journal of Risk Research*, 4 (4) 377-392.

Mearns, K, Whitaker, S, & Flin, R. 2003, "Safety climate, safety management practice and safety performance in offshore environments", *Safety Science*, 41, 641-680.

Mohar, J, Batalden, P, & Barach, P. (2004). "Integrating patient safety into the clinical microsystem", *Quality & Safety in Health Care*, 13 34-38.

Moran, T. & Volkwein, F. (1992). "The Cultural Approach to the Formation of Organizational Climate", *Human Relations*, 45 (1) 19-47.

National Patient Safety Agency. (2003). what is a patient safety incident? 1-8.

Neal, A. & Griffin, M. (2002). "Safety climate and safety behavior.", Australian Journal of Managemnt, 27, 67-75.

Neal, A, Griffin, M, & Hart, P. (2000). "The impact of organizational climate on safety climate and individual behavior", *Safety Science*, 34, 99-109.

Nieva, V. & Sorra, J. (2003). "Safety culture assessment: a tool for improving patient safety in healthcare organisations", *Quality & Safety in Health Care*, 12, 17-23.

Norden-Hagg, A, Sexton, J, kalvemark-Sporrong, S, Ring, L, & Kettis-Lindblad, A. (2010). "Assessing safety culture in pharmacies: the psychometric validation of the Safety Attitudes Questionnaire(SAQ) in a national sample of community pharmacies in Sweden.", *BMC Clinical Pharmacology*, 10 (8) 1-12.

Nusair, K. & Hua, N. (2010). "Comparative assessment of structural equation modeling and multiple regression research methodologies: E-commerce context", *Tourism Management*, 31, 314-324.

Oliver, A, Cheyne, A, & Cox, S. (2002). "The effects of organizational and individual factors on occupational accidents", *Journal of Occupational and Organizational Psychology*, 75, 473-488.
Ostrom, L, Wilhemsen, C, & Kaplan, B. (1993). "Assessing safety culture", Nuclear Safety, 34 (2) 163-171.

Pallant, J. (2007). SPSS Survival manual. Berkshire: Open University Press.

Parker, D, Matthew, L, & Hudson, P. (2006). "A framework for understanding the development of organisational safety culture", *Safety Science*, 44, 551-562.

Pett, M., Lackey, N., & Sullivan, J. (2003). Making sense of factor analysis: The use of factor analysis for instrument development in health care research. London: Sage publications.

Pfeiffer, Y. & Manser, T. (2010). "Development of the German version of the Hospital Survey on Patient Safety Culture: Dimensionality and psychometric properties", *Safety Science*, 48, 1452-1462.

Pidgeon, N. (1991). "Safety culture and risk management in organizations.", Journal of Cross-Cultural Psychology, 22, 129-141.

Pidgeon, N. & Oleary, M. (2000). "Man-made disasters: why technology and organisations (sometimes) fail", *Safety Science*, 34, 15-30.

Pidgeon, N. (1998). "Safety culture: key theoretical issues", Work& Stress, 12, (3) 202-216.

Pittet, D. & Donaldson, L. (2006). "Challenging the world: Patient safety and healthcareassociated infection", *International Journal for Quality in Healthcare*, 18 (1) 4-8.

Pronovost, P. & Sexton, S. (2005). "Assessing safety culture: guidelines and recommendations", *Quality & Safety in Health Care*, 14, 231-233.

Pronovost, P., Weast, B., Holzmueller, C., Rosentein, B., Kidwell, R., Haller, K., Feroli, E., Sexton, J., & Rubin, H. (2003). "Evaluation of the culture of safety:survey of clinicians and managers in an academic medical centre", *Quality and Safety in Health Care*, 12, 405-410.

Pronovost, P, Holzmueller, C, Needham, D, Sexton, J, Miller, M, Berenholtz, S, Perl, T, Davis, R, Baker, D, Winner, L, Morlock, L, & Albert, W. (2006). "How will we know patients are safer? An organization-wide approach to measuring and improving safety.", *Critical Care Medicine*, 34 (7) 1988-1995.

Raykov, T. & Marcoulides, G. (2000). "A method for comparing completely standardized solutions in muliple groups", *Structural Equation Modelling*, 7, 292-308.

Reason, J. (1998). "Achieving a safe culture: Theory and practice", Work and Stress, 12 (3) 293-306.

Reason, J. (1990). Human error. Cambridge: University Press.

Reason, J. (1997). Managing the risks of organisational accidents Aldershot, Ashgate.

Reason, J. (2000). "Human error: models and management", BMJ, 320, 768-770.

Regenstein, M. (2004). "Understanding the first Institute of Medicine report and it's on patient safety," in *The patient safety handbook*, B. Youngberg & M. Hatlie, eds., Jones and Bartlett, London.

Robyn, G, Christine, K., James, G, Lawrence, M, Antonio, E., Patricia, F, Edward, B, Christine, K, & Linda, M. (2000). "Hospital safety climate and its relationship with safe work practices and workplace exposure incidents.", *American Journal of Infection Control*, (28) (3) 211-221.

Roemer, M. (1991). National health system of the world Oxford University Press, New York.

Runciman, W. (2002). "Lesson from the Australian Patient Safety Foundation: setting up a national patient safety surveillance system-is this the right model?", *Quality & Safety in Health Care*, 11 (3) 246-251.

Salminen, S. & Seppala, A. (2005). "Safety climate in Finnish-Swedish Speaking companies", *International Journal of Occupational Safety and Ergonomics*, 11 (4) 389-397.

Sammer, C, Lykens, K, Singh, K, Mains, D, & Lackan, N. (2010). "What is patient safety culture? A review of the literature", *Journal of Nursing Scholarship*, 42 (2) 156-165.

Sanislow, C, Grilo, C, Morey, L, Bender, D, Skodol, A, Gunderson, J, Shea, M, Stout, R, Zanarini, M, & Glashan, T. (2002). "Confirmatory factor analysis of DSM-IV creteria for bordline personality disorder: findings from the collaborative longitudinal personality disorders study", *AM J Psychiatry*, 159 (2) 284-290.

Schmitt, N. & Coyle, B. (1976). "Applicant decisions in the employment interview", *Journal of Applied Psychology*, 61, 184-192.

Schmitt, N. & Stults, D. (1985). "Factors defined by negatively keyed items: The result of careless respondents?", *Applied Psychological Measurement*, 9, 367-373.

Schreiber, J, Stage, F, & King, J. (2006). "Reporting structural equation modeling and confirmatory factor analysis results: a review", *The Journal of Education Research*, 99 (6) 323-337.

Schumacker, R. & Lomax, R. (2004). *A beginner's guide to structural equation modeling*, 2<sup>nd</sup> ed, Lawrence Erlbaum Associates.

Sexton, J, Helmreich, R, Neilands, T, Rowan, K, Vella, K, Boyden, J, & Roberts, P. (2004). *Frontline Assessment of Healthcare Culture: Safety Attitudes Questionnaire Norms and Psychometric Properties*, The University of Texas Center of Excellence for Patient Safety Research and Practice, 04-01.

Sexton, J, Helmreich, R, Pronovost, P, & Thomas, E. (2003), Safety Climate Survey The Centre of Excellence for Patient Safety Research and Practice University of Texas.[online] available from: www.uth.tmc.edu/schools/med/imd/patient-safety/survey&tools.htm [Accessed 15<sup>th</sup> December 2008].

Sexton, J, Thomas, E, & Helmreich, R. (2000), "Error, stress, and teamwork in medicine and aviation: cross sectional surveys", *BMJ*, 320 (7237) 745-749.

Sexton, J, Helmreich, R., Neilands, T, Rowan, K, Vella, K., Boyden, J, Roberts, P, & Thomas, E. (2006a). "The Safety Attitudes Questionnaire: psychometric properties, benchmarking data, and emerging research", *BMC Health* Services Research, 6, 44.

Sexton, J, Makary, M, Tersigni, A, Pryor, D, Hendrich, A, Thomas, E, Holzmueller, C, Knight, A., Wu, Y., & Pronovost, P. (2006b). "Teamwork in the operating room: frontline perspectives among hospitals and operating room personnel.", *Anaesthesiology*, 105 (5) 877-884.

Singer, S, Meterko, M, Baker, L, Gaba, D, Falwell, A, & Rosen, A. (2007), "Workforce perceptions of hospital safety culture: development and validation of the patient safety climate in healthcare organizations survey", *Health Services Research*, 42 (5) 1999-2021.

Singer, S, Gaba, D, Geppert, J, Sinaiko, A, Howard, S, & Park, K. (2003). "The culture of safety: results of an organization-wide survey in 15 California hospitals.", *Quality & Safety in Health Care*, 12 (2) 112-118.

Singla, A, Barrett, K, Weissman, J, & Campbell, E. (2006). "Assessing patient safety culture: A review and synthesis of the measurement tools", *Journal of Patient Safety*, 2 (3) 105-115.

Smits, M, Christiaans-Dingelhoff, I, Wanger, C, Wal, G, & Groenewegen, P. (2008). "The psychometric proerties of the Hospital Survey on Patient Safety Culture in Dutch hospitals", *BMC Health Services Research*, 8 (230) 1-9.

Snook, S. & Gorsuch, R. (1989). "Component analysis versus common factor analysis: A Monte Carlo study ", *Psychological Bulletin*, 106 (1), 148-154.

Sorensen, J. (2002). "Safety culture: a survey of the state-of-the-art", *Reliability Engineering* and System Safety, 76, 189-204.

Sorra, J. & Nieva, V. (2002). "Safety culture and event reporting in hospital transfusion services.", *Transfusion*, 42, 1-38.

Sorra, J. & Nieva, V. (2004). *Hospital Survey on Patient Safety Culture*, AHRQ publication No 04-0041. Rockville, Md: Agency for Healthcare Research and Quality.

Sorra, S. & Dyer, N. (2010). "Multilevel psychometric properties of the AHRQ hospital survey on patient safety culture", *BMC Health Services Research*, 10 (199) 1-13.

Spector, P, Van Katwyk, P, Brannick, M, & Chen, P. (1997). "When two factors do not reflect two constructs: How item characteristics can produce artifactual factors", *Journal of Management*, 23, 659-677.

Sperber, A. (2004). "Translation and validation of study instruments for cross-cultural research", *Gastroenterology*, 126, 124-128.

Statistics Directorate. (2009). *Health statistical book for the year of 2009*. Riyadh: Ministry of Health.

Streiner, D. & Norman, G. (1995). Health measurment scales: A practical guide to their development and use, 2<sup>nd</sup> ed. Oxford: Oxford University Press.

Tabachnick, B. & Fidell, L. (2007). Using multivariate statistics, 5<sup>th</sup> ed. Boston: Pearson Education.

Tinsley, H. & Tinsley, D. (1987). "Uses of factor analysis in counseling psychology research", *Journal of Counseling Psychology*, 34, 414-424.

Tomas, J, Melia, J, & Oliver, A. (1999). "A cross-validation of a structural equation model of accidents: organizational and psychological variables as predictors of work safety", *Work &Stress*, 13 (1) 49-58.

University of Texas Centre of Excellence for Patient Safety Research and Practice. [Online]. Available from: www.uth.tmc.edu/schools/med/imed/patient-safety/survey&tools.htm [Accessed 15th December 2007].

VanMaanen, J. & Schein, E. (1979). "Toward a theory of organisation socialization.", *Research in Organizational Behavior*, 1, 209-264.

Vecchio-Sadus, A. & Griffiths, S. (2004). "Marketing strategies for enhancing safety culture", *Safety Science*, 42, 601-619.

Vincent, C, Neale, G, & Woloshynowych, M. (2001). "Adverse events in British hospitals: preliminary retrospective record review", *BMJ*, 322, 517-519.

Vincent, C, Taylor-Adams, S, & Stanhope, N. (1998). "Framework for analysing risk and safety in clinical medicine", *BMJ*, 316, 1154-1157.

Vincent, C, Stanhope, N, & Taylor-Adams, S. (2000). "Developing a systematic method of analyzing serious accidents in mental health", *Journal of Mental Health*, 9 (1) 89-103.

Voulntray Hospital of America/America Hospital Association. Strategies for leadership: An organisational approach to patient safety (2000). [Online]. Available from: www.aha.org/aha. [Accessed 5th January 2008].

Walshe, K. & Boaden, R. (2006). Patient safety research in to practice. Open University Press, New York.

Walston, S, Alomar, B, & Almutari, F. (2010). "Factors affecting the climate of hospital patient safety; Astudy of hospitals in Saudi Arabia", *International Journal of Health Care Quality Assurance*, 23 (1) 35-50.

Waterson, P, Griffiths, P, Stride, C, & et al (2009). "Psychometric properties of the Hospital Survey on Patient Safety Culture: findings from the UK", *Quality and Safety in Health Care*, 10, 1-5.

Weick, K, Sutcliffe, K, & Obstfeld, D. (1999). "Organizing for reliability: process of collective mindfulness.", *Research in Organizational Behavior*, 21, 81-123.

Weingart, S, Farbstein, K., Davis, R., & Philips, R. (2004). "Using a multi hospital survey to examine the safety culture", *Joint Commission Journal on Quality and Safety*, 30 (3) 125-132.

Wet, C, Spence, W, Mash, R., Johnson, P, & Bowie, P. (2010). "The development and psychometric evaluation of a safety climate measure for primary care", *Quality and Safety in Health Care*, 10 1-7.

Wiegmann, D, Zhang, H, Thaden, T, Sharma, G, & Mitchell, A. (2002). A Synthesis of Safety Culture and Safety Climate Research, Aviation Research Lab Institute of Aviation, University of Illinois, ARL-02-3/FAA-02-2.

Wilson, R, Runciman, W, Gibberd, R, Harrison, B, Newby, L, & Hamilton, J. (1995). "The quailty in Australian Health Care Study", The Medical Journal of Australia, 163 (6) 458-471.

World Health Organisation Report (2000), Why do health systems matter. [Online]. Available from: www.who.int/en. [Accessed 20 January 2008].

World Health Organisation 2004, World Alliance for Patient Safety. [Online]. Available from: www.who.int/patientsafety. [Accessed 20 January 2008].

World Health Organisation 2006, World Alliance for Patient Safety. [Online]. Available from: www.who.int/patientsafety. [Accessed 20 January 2008].

World Health Organisation 2008, World Alliance for Patient Safety. [Online]. Available from: www.who.int/patientsafety. [Accessed 10 March 2008].

Yassi, A. & Hancock, T. (2005). "Patient safety-worker safety: Building a culture of safety to improve healthcare worker and patient well-being", *Healthcare Quarterly*, 8, 32-38.

Yule, S, Flin, R, Paterson-Browne, S, & Maran, N. (2006). "Non-technical skills for surgeons in the operating room: a review of the literature", *Surgery*, 139 (2) 140-149.

Zohar, D. (1980). "Safety climate in industrial organisations: Theoritical and applied implications", *Journal of Applied Psychology*, 65 (1) 96-102.

Zohar, D. (2002). "The effects of leadership dimensions, safety climate, and assigned priorities onminor injuries in work groups", *Journal of Organizational Behavior*, 23, 75-92.

Zohar, D. (2003). "Safety climate: conceptual and measurement issues," in *Handbook of occupational health psychology*, J. Quick & L. Tetrick, eds., American Psychological Association, Washington, DC, pp. 123-142.

Zwick, W & Velicer, W. (1986), "Comparison of five rules for determining the number of components to retain". *Psychological Bulletin*, 99, 432-442.

# Constant and

The University Of Sheffield.

# School of Health and Related Research

Appendix 1 Consent form

Title of Project: An evaluation of a	patient safety culture tool in	Saudi Arabia								
Name of Researcher: Mamdooh Shrier Alonazi										
Participant Identification Number for this project:										
		Please initial box								
1. I confirm that I have read and understand the information sheet dated for the above project and have had the opportunity to ask questions.										
2. I understand that my participation is voluntary and that I am free to withdraw at any time without giving any reason.										
<ol> <li>I understand that my responses I give permission for members of to my responses.</li> </ol>	will not be anonymised befor f the research team to have a	re analysis.								
4. I agree to take part in the above	research project.	Las Bodi activitas <u>10 lik</u> (i								
Name of Participant	Date	Signature								
Lead Researcher	Date	Signature								
To be signed and dated in presence of the participant										
<b>Copies:</b> Once this has been signed by all parties the participant should receive a copy of the signed and dated participant consent form, the information sheet and any other written information provided to the participants. A copy for the signed and dated consent form should be placed in the project's main record (e.g. a site file), which must be kept in a secure location.										

The University of Sheffield Western Bank Sheffield S10 2TN UK



The University Of Sheffield.

# School of Health and Related Research

#### **Appendix 2 Participant Information Sheet**

Research project title: An evaluation of a patient safety culture tool in Saudi Arabia

# Invitation to participate:

You are being invited to take part in this PhD research project. It is important for you to understand why this research is being done and what it will involve before you consider your participation. Please read the following information, which explains the purpose of this research and what participation will involve and you may discuss this with others if you wish. If you would like further information or you have questions, please contact the researcher. You will be given a period of 5-7 days to decide whether or not you wish to participate.

Thank you for reading this information. If you decide to participate, you will be given a copy of this information sheet and your signed consent form.

#### Purpose of the research:

The purpose of this research is to identify a suitable measure for assessing patient safety culture for use in hospitals in Saudi Arabia. The methods that will be used to collect the information include interviews and questionnaires related to the development of the patient safety culture tool. It is expected that data collection will be undertaken over a period of five months.

## Why have I been chosen?

The participants in this study are frontline healthcare providers (doctors and nurses). Therefore, you have been selected because you are one of the doctors or the nurses in the hospital. You would be one of a total of twelve doctors and nurses helping with this part of the study.

# Do I have to participate?

No, it is up to you to decide whether or not you want to take part. If you decide to take part you will be given this information sheet to keep and you will be asked to sign a consent form. Even after deciding to take part in this research, you are still free to withdraw from the research at any point in time and without giving a reason. If you decide to participate in the interviews we will ask you not to take part in completing the questionnaires.

# What will happen to me if I participate?

If you decide to participate then you will be interviewed in an informal conversational style. The researcher will ask you to sign a consent form. Then the researcher will organise an interview schedule. The interview will last about 50 to 60 minutes, you will fill in the questionnaire (Hospital Survey on Patient Safety Culture Questionnaire) and notes will be taken by the researcher during the interview. Three questions will be asked for each item of the questionnaire:

- Q1: Is the wording of the item clear for you?
- Q2: What does the item mean to you?
- Q3: Do you find the item easy or difficult to understand?

The interview will be conducted in a private office and everything said and written during the interview will remain in confidence. There is no potential for physical and/or psychological harm / distress to participants during the research.

# What are the possible benefits of taking part?

This study seeks to contribute to the knowledge base by investigating the suitability of an appropriate patient safety culture measure for use in hospitals in Saudi Arabia. It is hoped that the knowledge generated will contribute to the literature on the measurement of the patient safety culture, in particular in Saudi Arabia. There are no direct benefits to you.

# What will happen to the results of the research study?

The results will be published in the PhD thesis. No named information about you will be mentioned in the thesis.

# Will my taking part in this research be kept confidential?

All of the data that will be collected will be securely stored by the researcher. The interviews notes will be disposed of 24 months after the researcher receives the PhD degree. The computer used for analysing the data will be password protected. Research notes will be stored in a locked filing cabinet in an office that is locked. Your anonymity and confidentiality will be preserved at all times. Access to the research data is limited to the research team, which include the supervisors of this research and the researcher (Professor Allen Hutchinson, Dr. Jenny Freeman, Dr. Rachel O'Hara and Mr Mamdooh Shereir Alonazi).

## Who is organising and funding the research?

The research is part of the PhD studies and no organisation is involved in funding.

# Who has ethically reviewed the project?

The research has been scientifically reviewed by the University of Sheffield, School of Health and Related Research in the United Kingdom. It has also been reviewed and approved by the School of Health and Related Research Ethics Committee. In addition, it has been reviewed by the Medical and Health Sciences Research Centre at King Abdul-Aziz Medical City and Research department at King Faisal Specialist Hospital and Research Centre Riyadh, Saudi Arabia.

# **Contact for further information:**

Should you need more information you can contact the supervisors of the research and the researcher:

1. Professor Allen Hutchinson

Professor in Public Health

School of Health and Related Research (ScHARR)

Section of Public Health

Regent Court, 30 Regent Street

Sheffield, S1 4DA, UK

Tel: +44 114 222 0813

Email: Allen.hutchinson@sheffield.ac.uk

2. Dr. Jenny Freeman School of Health and Related Research (ScHARR) Section of Health Services Research Telephone: (+44) (0) 114 222 0695 Fax: (+44) (0) 114 222 0749 Email: j.v.freeman@sheffield.ac.uk 3. Dr. Rachel O'Hara School of Health and Related Research (ScHARR) Section of Public Health Regent Court, 30 Regent Street Sheffield, S1 4DA Tel: +44 114 222 0680 Email: r.ohara@sheffield.ac.uk 4. Mamdooh Sherier Alonazi PhD student in Section of Public Health Tel: [in Saudi Arabia+966505386492] [in the U.K+447735627677] Email: M.Alonazi@sheffield.ac.uk

Thank you for taking the time to participate in the research.

Appendix 3 Original Hospital Survey on Patient Safety Culture Survey

# **HOSPITAL SURVEY ON PATIENT SAFETY CULTURE**

# INSTRUCTIONS

This survey asks for your opinions about patient safety issues, medical error, and event reporting in your hospital and will take about 10 to 15 minutes to complete.

- An "event" is defined as any type of error, mistake, incident, accident, or deviation, regardless of whether or not it results in patient harm.
- "Patient safety" is defined as the avoidance and prevention of patient injuries or adverse events resulting from the processes of healthcare delivery.

# SECTION A: Your Work Area/Unit

In this survey, think of your "unit" as the work area, department, or clinical area of the hospital where you spend <u>most of your work time or provide most of your clinical services</u>

What is your primary work area or unit in this hospital? Mark ONE answer by filling in the circle.

Pa. Many different hospital units/No specific unit

b. Medicine (non-surgical)	g. Intensive care unit (any type)	I. Radiology
? c. Surgery	Ph. Psychiatry/mental health	? m. Anaesthesiology
? d. Obstetrics	<b>?</b> i. Rehabilitation	n. Other, please specify
? e. Pediatrics	? j. Pharmacy	
? f. Emergency department	R. Laboratory	

Please indicate your agreement or disagreement with the following statements about your work area/unit. Mark your answer by filling in the brackets.

Th	ink about your hospital work area/unit	Strongly Disagree	Disagree	Neither	Agree	Strongly Agree
1.	People support one another in this unit	[1]	[2]	[3]	[4]	[5]
2.	We have enough staff to handle the workload	[1]	[2]	[3]	[4]	[5]
3.	When a lot of work needs to be done quickly, we work together as a team to get the work done	[1]	[2]	[3]	[4]	[5]
4.	In this unit, people treat each other with respect.	[1]	[2]	[3]	[4]	[5]
5.	Staff in this unit work longer hours than is best for patient care	[1]	[2]	[3]	[4]	[5]
6.	We are actively doing things to improve patient safety	[1]	[2]	[3]	[4]	[5]
7.	We use more agency/temporary staff than is best for patient care	[1]	[2]	[3]	[4]	[5]
8.	Staff feel like their mistakes are held against them	[1]	[2]	[3]	[4]	[5]
9.	Mistakes have led to positive changes here	[1]	[2]	[3]	[4]	[5]
10.	It is just by chance that more serious mistakes don't happen around here	[1]	[2]	[3]	[4]	[5]
11.	When one area in this unit gets really busy, others help out	[1]	[2]	[3]	[4]	[5]
12.	When an event is reported, it feels like the person is being written up, not the problem	[1]	[2]	[3]	[4]	[5]

# SECTION A: Your Work Area/Unit (continued)

Think about your hospital work area/unit	Strongly Disagree	Disagree	Neither	Agree	Strongly Agree
13. After we make changes to improve patient safety, we evaluate their effectiveness	[1]	[2]	[3]	[4]	[5]
14. We work in "crisis mode" trying to do too much, too quickly	[1]	[2]	[3]	[4]	[5]
15. Patient safety is never sacrificed to get more work done	[1]	[2]	[3]	[4]	[5]
16. Staff worry that mistakes they make are kept in their personnel file	[1]	[2]	[3]	[4]	[5]
17. We have patient safety problems in this unit	[1]	[2]	[3]	[4]	[5]
18. Our procedures and systems are good at preventing errors from happening	[1]	[2]	[3]	[4]	[5] <sub>.</sub>

#### SECTION B: Your Supervisor/Manager

Please indicate your agreement or disagreement with the following statements about your immediate supervisor/manager or person to whom you directly report. Mark your answer by filling in the brackets.

		Strongly Disagree	Disagree	Neither	Agree	Strongly Agree
1.	My supervisor/manager says a good word when he/she sees a job done according to established patient safety procedures	[1]	[2]	[3]	[4]	[5]
2.	My supervisor/manager seriously considers staff suggestions for improving patient safety	[1]	[2]	[3]	[4]	[5]
3.	Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it means taking shortcuts	[1]	[2]	[3]	[4]	[5]
4.	My supervisor/manager overlooks patient safety problems that happen over and over	[1]	[2]	[3]	[4]	[5]

# **SECTION C: Communications**

How often do the following things happen in your work area/unit? Mark your answer by filling in the brackets.

Th	ink about your hospital work area/unit	Never	Rarely	Some times	Most of the time	Always
1.	We are given feedback about changes put into place based on event reports	[1]	[2]	[3]	[4]	[5]
2.	Staff will freely speak up if they see something that may negatively affect patient care	[1]	[2] ·	[3]	[4]	[5]
3.	We are informed about errors that happen in this unit	[1]	[2]	[3]	[4]	[5]
4.	Staff feel free to question the decisions or actions of those with more authority	[1]	[2]	[3]	[4]	[5]
5.	In this unit, we discuss ways to prevent errors from happening again	[1]	[2]	[3]	[4]	[5]
6.	Staff are afraid to ask questions when something does not seem right	[1]	[2]	[3]	[4]	[5]

# SECTION D: Frequency of Events Reported

In your hospital work area/unit, when the following mistakes happen, how often are they reported? Mark your answer by filling in the brackets.

	a so some constant i calificatione	Never	Rarely	Some times	Most of the time	Always
1.	When a mistake is made, but is caught and corrected before affecting the patient, how often is this reported	[1]	[2]	[3]	[4]	[5]
2.	When a mistake is made, but has no potential to harm the patient, how often is this reported	[1]	[2]	[3] .	[4]	[5]
3.	When a mistake is made that could harm the patient, but does not, how often is this reported?	[1]	[2]	[3]	[4]	[5]

## SECTION E: Patient Safety Grade

Please give your work area/unit in this hospital an overall grade on patient safety. Mark ONE answer.

Excellent	Very Good	Acceptable	Poor	Failing
A	В	C	D	
?	?	?	?	?

## SECTION F: Your Hospital

Please indicate your agreement or disagreement with the following statements about your hospital. Mark your answer by filling in the brackets.

Th	ink about your hospital	Strongly Disagree	Disagree	Neither	Agree	Strongly Agree
1.	Hospital management provides a work climate that promotes patient safety	[1]	[2]	[3]	[4]	[5]
2.	Hospital units do not coordinate well with each other	[1]	[2]	[3]	[4]	[5]
3.	Things "fall between the cracks" when transferring patients from one unit to another	[1]	[2]	[3]	[4]	[5]
4.	There is good cooperation among hospital units that need to work together	[1]	[2]	[3]	. [4]	[5]
5.	Important patient care information is often lost during shift changes	[1]	[2]	[3]	[4]	[5]
6.	It is often unpleasant to work with staff from other hospital units	[1]	[2]	[3]	[4]	[5]
7.	Problems often occur in the exchange of information across hospital units	[1]	[2]	[3]	[4]	[5]

8.	The actions of hospital management show that patient safety is a top priority	[1]	[2]	[3]	[4]	[5]
9.	Hospital management seems interested in patient safety only after an adverse event happens	[1]	[2]	[3]	[4]	[5]
10.	Hospital units work well together to provide the best care for patients	[1]	[2]	[3]	[4]	[5]
11.	Shift changes are problematic for patients in this hospital	[1]	[2]	[3]	[4]	[5]

# SECTION G: Number of Events Reported

In the past 12 months, how many event reports have you filled out and submitted? Mark ONE answer .

? a. No event reports

? d. 6 to 10 event reports

D. 1 to 2 event reports

? e. 11 to 20 event reports

? c. 3 to 5 event reports

? f. 21 event reports or more

## SECTION H: Background Information

This information will help in the analysis of the survey results. Mark ONE answer by filling in the circle .

1. What is your staff position in this hospital? Mark ONE answer that best describes your staff position.

? a. Physician in training

? g. Head nurse

? h. Registered nurse

D. Physician (general physician)

? c. Resident physician

? i. Other, please specify:.....

? d. Specialist

? e. Consultant

? f. Professor

2. What is your gender?? a. Male

Pb. Female

3. What is your nationality?

? a. Saudi

b. Non-Saudi, please specify:.....

4. What is your educational level?? a. PhD

? b. Master degree

? c. Bachelor degree

? d. Diploma

e. other, Please specify.....

5. How long have you worked in this hospital ?
? a. Less than 1 year
? d. 11 to 15 years

? b. 1 to 5 years

? e. 16 to 20 years

? c. 6 to 10 years

? f. 21 years or more

6. How long have you worked in your current hospital work area/unit ?
? a. Less than 1 year
? d. 11 to 15 years

? b. 1 to 5 years

? e. 16 to 20 years

? c. 6 to 10 years

? f. 21 years or more

7. Typically, how many hours per week do you work in this hospital ?
a. Less than 20 hours per week
d. 60 to 79 hours per week

D. 20 to 39 hours per week

? e. 80 to 99 hours per week

? c. 40 to 59 hours per week

? f. 100 hours per week or more

8. In your staff position, do you typically have direct interaction or contact with patients?

? a. YES, I typically have direct interaction or contact with patients

b. NO, I typically do NOT have direct interaction or contact with patients

9. How long have you worked in your current specialty or profession ?? a. Less than 1 year? d. 11 to 15 years

? b. 1 to 5 years

? e. 16 to 20 years

? c. 6 to 10 years

? f. 21 years or more

#### SECTION I: Your Comments

Please feel free to write any comments about patient safety, error, or event reporting in your hospital.

#### THANK YOU FOR COMPLETING THIS QUESTIONNAIRE

# Appendix 4 Teamwork and Safety Climate

Please answer the following items with respect to your specific unit or clinical area. Choose your responses using the scale below:

No	Item	Disagree	Disagree	neutral	Agree	Agree	Not
		strongly	slightly	at suches	slightly	strongly	Applicable
1	Nurse input is well received in this clinical area.						ert yum
2	In this clinical area, it is difficult to speak up if I perceive a problem with patient care.			estrines. Conchevere	ar-brazi Skribi ski s nin teri	ines (c.p. ht protect	olug to maard
3	Decision-making in this clinical area utilizes input from relevant personnel.						8.d2 308
4	The physicians and nurses here work together as a well –coordinated team.				a interes	n most i	R PROV
5	Disagreements in this clinical area are resolved appropriately (i.e., not who is right, but what is best for the patient).				about		Medica
6	I am frequently unable to express disagreement with the attending/staff physicians here.			to direct stately jr	charoneli ig philen	Ne prope Straggtos	werut i chizace
7	It is easy for personnel here to ask questions when there is something that they do not understand.			01 THUS		inical pro	anti ak Ruseik
8	I have the support I need from other personnel to care for patients.			10.419365	ord onich	ly compr	mworrs -
9	I know the first and last names of all the personnel I worked with during my last shift.			ne for the one	ofic period 8 insectiv	ki nostuti ulety nou	This its patient
10	Important issues are well communicated at shift changes.			vistern o			Leader
11	Briefing personnel before the start of a shift (i.e., to plan for possible contingencies) is important for patient safety.			od bisow od dia	1916-100 1 092241	estion an on V est mens	My Sug ected u manage
12	Briefings are common in this clinical area.						
13	I am satisfied with the quality of collaboration that I experience with staff physicians in this clinical area.						
14	I am satisfied with the quality of collaboration that I experience with nurses in this clinical area.	Баненка					
	10 mart 0 5.21 m	a of loors					

Safe	ty climate			nati ti çê M Tên tinîna	pe has in	a termina Single const	olamoz – A Na sosti – A
15	The levels of staffing in this clinical area are sufficient to handle the number of patients.	d anna	in the second		a i Publici Secolori - C	bit gen out	o knowines
16	I would feel safe being treated here as a patient.	eralend Aufmeis	Di signa				anst!
17	I am encouraged by my colleagues to report any patient safety concern I may have.		(Rinord)) - Landson - Stranger	eirth chis	aylaget i	174 ST 255	11 (1231) 21
18	Personnel frequently disregard rules or guidelines (e.g., hand-washing, treatment protocols/clinical pathways, sterile field, etc.) that are established for this clinical area.			o mia itiw nefi jisw isa	5 615 214 - 2 2012 42 3 4 1 11 11 2 21 11 2 21 11 2 21 11 2 21 11 2 21 11 2 21 11 2 21 12 - 2 2 2 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2	nicut are nicut are are are are	In this c spiration ontiont Decision
19	The culture in this clinical area makes it easy to learn from the errors.			Markey (1994) Markey (1994)	1210-4 (1121) 1210-4 (1121)	MI SUMPLY ADD TOM	्रत्याच्याच्य हर्ताच्च इत्य
20	I receive appropriate feedback about my performance.			narat signa na signa la	inia cini	ni shingin	autestor
21	Medical errors are handled appropriately here.			1993 (M) (M)	41.00 (0.00) 38-92 (0.6 (	e periodes e territorio	nd anno an
22	I know the proper channels to direct questions regarding patient safety in this clinical area.			geress Jeines Sile	क को को को कि जोदित की जिसके जनस्थ की जिसके की	1111年1月 1912年1月 1915年 1915年	evernid Aliðerip Rusinu i
23	In this clinical area, it is difficult to discuss errors.			her in giaithe	nariyin Dê Losêr	i ned 10) I neder in	dieses a l
24	Hospital management does not knowingly compromise the safety of patients.				al basin t a basin t shirt u tur		1 (12 (12 (12 (12 (12 (12 (12 (12 (12 (1
25	This institution is doing more for patient safety now, than it did one year ago.	•		ne in can guingen	na inin b thu biblin tractic	e ten av et i brinde , itel	the period
26	Leadership is driving us to be a safety- centred institution.			1935	ाख्यः स्व स्वतः प्रीत्तीव	tausti K Isilahsin	sintegint vintingo
27	My suggestion about safety would be acted upon if I expressed them to management.		ti ti	in i en en en del setting set	। जनवन्त्रत्व द्राव्यकृत्वन्तुः अर्थकृत्युः	anticerae and as us al centan	gansna 1 fint a galinus

## **Background information:**

- 1. What is your staff position in this hospital? Mark ONE answer that best describes your staff position.
- O. Physician in training
- O. Physician (general physician)
- O. Resident physician

O. Registered nurse

O. Head nurse

O. other, please specify:

- O. Specialist
- O Consultant
- O. Professor
  - 2 What is your gender?O. MaleO. Female
  - 3 What is your nationality?O. SaudiO. non-Saudi, please specify.....

4 Experience in organisation

- O. Less than 6 months
- O. 6 to 11 months
- O. 1 to 2 years
- O. 3 to 7 years
- O. 8 to 12 years
- O. 13 to 20 years
- O. 21 or more
  - 5 How long have you worked in your current hospital work area/unit?

O. a. Less than 1 year	O. d. 11 to 15 years
O. b. 1 to 5 years	O. e. 16 to 20 years
O. c. 6 to 10 years	O. f. 21 years or more

A-17

- 6 Unit clinical area: please write in your unit title/location:.....
- 7 What is your educational level? O. PhD degree

O. Master degree

O. Bachelor degree

O. Diploma degree

O. e other, please specify

8 In your staff position, do you typically have direct interaction or contact with patients?

O. YES, I typically have direct interaction or contact with patients

O. NO, I typically do NOT have direct interaction or contact with patients

9 Typically, how many hours per week do you work in this hospital?

O. Less than 20 hours per week O. 60 to 79 hours per week

O. 20 to 39 hours per week

O. 80 to 99 hours per week

O. 40 to 59 hours per week

O. 100 hours per week or more

# THANK YOU FOR COMPLETING THIS QUESTIONNAIRE

#### Appendix 5 University ethics approval 1



# The University Of Sheffield.

Cheryl Oliver Ethics Committee Administrator

Regent Court 30 Regent Street Sheffield SI 4DA Telephone: +44 (0) 114 2220871 Fax: +44 (0) 114 272 4095 (non confidential) Email: c.a.oliver@sheffield.ac.uk

Our ref: /CAO

DATE

Mamdooh Alonazi ScHARR

Dear Mamdooh,

# An evaluation of a patient culture tool in Saudi Arabia

Thank you for submitting the above research project for approval by the ScHARR Research Ethics Committee. On behalf of the University ethics reviewers who reviewed your project, I am pleased to inform you that the project was approved.

If during the course of the project you need to deviate significantly from the documents you submitted for review, please inform me since written approval will be required.

Yours sincerely

Cheryl Oliver Ethics Committee Administrator

# Appendix 6 University ethics approval 2



The University Of Sheffield.

> Cheryl Oliver Ethics Committee Administrator

Regent Court 30 Regent Street Sheffield SI 4DA **Telephone:** +44 (0) 114 2220871 **Fax:** +44 (0) 114 272 4095 (non confidential) **Email:** c.a.oliver@sheffield.ac.uk

Our ref: /CAO

16 June 09

Mamdooh Alonazi ScHARR

Dear Mamdooh,

# An evaluation of a patient culture tool in Saudi Arabia

Thank you for the additional papers you have submitted to Jennifer Burr.

ScHARR ethics has looked through the changes and is happy to approve the amendment. Please note this does not replace the ethical approval that you may require from the King Fahad Hospital in Saudi Arabia.

Yours sincerely

Cheryl Oliver Ethics Committee Administrator

Research ethics review was required from the School of Health and Related Research at the University of Sheffield. Thereafter, ethics review and agreements were sought from the three hospitals in Saudi Arabia.

# Ethics in School of Health and Related Research

The first stage involved ethical approval from the University of Sheffield. The present study followed the university ethics review procedure ('UER Procedure') from the School of Health and Related Research at the University of Sheffield as follows:

# Scientific Review

The study was reviewed not only in terms of ethical aspects, but also from scientific quality such as the methodology and strategy of data analysis. Therefore, this stage involved the scientific review and the ethics approval in March-April 2009. As a starting point, the scientific review document was sent to the two scientific reviewers in the School of Health and Related Research on 4th March 2009. Once the feedbacks received from the scientific peer review, the following suggestions have been taken into account and acted upon.

#### Reviewer 1:

The application should be approved with the suggested, optional amendments as follows:

- 1. Clarifying as to whether the patient safety tool is to be used only in English with English speaking staff or whether it will be used in translation.
- 2. Providing copies of consent forms and information sheets to demonstrate how the confidentiality of interviews will be assured.
- 3. Clarifying the differences between the two hospitals particularly in terms of the justification of conducting the interviews in only one hospital.

# Reviewer 2:

The study is well thought out and scientifically sounds. The application should be approved with the suggested optional amendments as follows:

- 1. Will the questionnaire be in English or will it need to be translated? Suggested that it will need to be clarified in the protocol.
- 2. Suggestion that the cover letter is printed on the same document as the questionnaire, or that information is included directly on the questionnaire about how to return it.
- 3. Will the student undertake the factor analysis and if so will he have access to a statistician to advise him on this?
- 4. There are two potential scenarios described for the factor analysis. The suggestion is that there is a lack of clarity about the first stage and the second stage of the proposed analytical strategy (which stage comes first?), but it is also strongly recommended to check with a statistician with expertise in factor analysis.
- 5. Re-time schedule between the interviews and questionnaires, ensuring that enough time is allowed for making and agreeing changes to the questionnaires and then getting them printed.
- 6. Ensure identification in advance of somewhere in Saudi Arabia to get the questionnaires printed.

#### **Ethics Review**

After all the comments of the scientific reviewers were considered to account and acted upon. The protocol includes the University Research Ethics Application Form, the Interview Consent Form and Participant Information Sheet (see appendix 4). The protocol then was submitted to the ethics administrator at School of Health and Related Research on 12<sup>th</sup> March 2009 to obtain ethics approval for the study. The ethics reviewer's comments indicated the need for some work before approval can be given, these comments were as follows:

- 1. Application form [We have made the changes you resolved, please see the responses below and changes in documentation].
- 2. Start date of project is stated as in 2007 (presumably the start of the PhD). We cannot give retrospective ethical permission to proceed, so this should be changed [The date has been changed]

A-22

- 3. In stage 1 there should be a clear statement about the balance between nurses and doctors in the interview sample? (It may mean 12 doctors and 12 nurses or 6 of each, etc.). There should be a statement if more than 12 or 24 volunteer after reading the information how selection and rejection will be handled [It has been clarified]
- 4. For stage 1 in the application form it needs to be clear what the doctor/nurse does once they decide whether to participate. It is also not clear how the 12 participants are selected or what happens if they refuse [I have clarified]
- 5. Needs to be clear that those who have taken part in Stage 1 will be asked not to take part in Stage 2 (Information sheet) [We have mentioned it in information sheet]
- 6. In A9.1 it should state there is an opportunity for the participant to ask questions?! (see covering letter)[we have done this]
- 7. In A10 it states data 'will be saved securely on the managed desktop at the university and be password protected during the study period'; There must be a statement that explains what happens to the data after the study.[We have mentioned what will happen to the data and the period in application form]
- 8. Not clear from the application whether the student would, in his normal capacity, have access to the hospital administration. If not then he should not expect to access the personal details for staff through this system and should request that this information be obtained, and letters circulated to staff, on his behalf. Please confirm.[The local research ethics and scientific committee approval in each hospital will enable the student to approach member of staff]
- 9. Covering letter Change "I would be grateful if you would take a few minutes ......" to "I would be grateful if you would volunteer to take a few minutes ......" there is nothing to clearly indicate participation is voluntary. There should be some indication that as many 300 participants are sought.[We have done this]
- 10. Information sheet. Somewhere the potential participants should be told that they would be 1 of 12 or 24 people helping with this part of the study. Also should mention right to withdraw at any time. [It has been done]
- 11. Those who have taken part in Stage 1 should be asked not to take part in Stage 2 (see comment above)[ It has been done]
- 12. In the information sheet specify what the doctor/nurse should do when they decide to participate in the interview. Some indication of the type of interview questions expected is necessary[We mentioned it]

- 13. Under 'What will happen if I take part', there is reference to a questionnaire. Is this a different questionnaire to the one being circulated later in the study? If so we would need a copy. If it's the same one please could this be clarified [The questionnaire mentioned is the same questionnaire]
- 14. Questionnaire. Reformat questionnaire, as currently several questions go across 2 pages and this means that they are difficult to compete e.g. section D, H6 etc.[We have made some formatting changes but further work required after the interview stage]
- 15. Consent form. The following should be resolved and amendments made as necessary. Although item 3 evidently implies that at least the researcher will collect participant identifiable data they are not given permission to have access to it by the participant ticking the box! The questionnaire as it stands has no name on it so at the point of collection it could be anonymised. If the latter is the case and also in recording what happens during the interview the researcher does not record the name of the participant (other than on the consent form) then item 3 could stand as worded. However this does raise the issue that if the data is anonymous at the point of collection then it cannot subsequently be withdrawn from the study (if a participant decides to withdraw). It could be argued that participants should be told that once anonymised their data can no longer be removed from the study. [Item three has been changed on the interview consent form].

The above suggestions have been taken into account and acted upon and the documentation has resubmitted on 7/4/2009 to get the ethical approval from the ethics committee in the School of Heath and Related Research at the University of Sheffield. Finally, the study was approved (See copy of ethics approval in appendix 2).

# Ethics and agreements in Saudi hospitals

The second stage was the hospitals' approval to conduct the study. Communication has been established with the research centres in the three hospitals in Riyadh for ethics application forms and regulations (instructions) for research requirements to obtain their agreement to undertake the survey as follows:

1. Institutional Review Board Committee of the Research Centre, King Fahad Hospital (submitted on April 2009)

- 2. Research department in King Faisal Specialist Hospital and Research Centre (submitted on April 2009)
- 3. Institutional Review Board of the King Fahad Medical City (submitted on June 2009)

The proposal was submitted to the three research centres above in hospitals in Riyadh in Saudi Arabia. These hospitals are the largest tertiary hospitals in terms of bed capacity and workforce in Saudi Arabia.

This task started on 11 April 2009 by preparing research proposal and the application form of King Fahad Hospital. The researcher completed the application form according to the general instructions of the research centre of the hospital. After that the proposal and all relevant forms were completed and submitted to the Medical & Health Sciences Research Centre (MHSRC) in King Fahad Hospital to obtain a reference number for the study. Thereafter it was forwarded to the Research Committee meeting and Institutional Review Board (IRB) for peer review and discussion. Finally, the decision approving the proposal was made by the Chairman of the Research Committee (See appendices for the copy of the approval letter from Institutional Review Board of King Abdullah International Medical & Health Sciences Research Centre).

King Faisal Specialist Hospital and Research Centre (KFSH&RC) was the second hospital on 14/4/2009. According to the King Faisal Specialist Hospital and Research Centre's guidelines and instructions for submission of research proposals, the researcher must have successfully completed the National Institute of Health (NIH) Web based training course "Protecting Human Research Participants" which was done on 22/3/2009 (see appendix for the copy of the certificate). Then the principal investigator was identified. Subsequently, the research proposal format and the application form were completed. The principal investigator submitted the proposal with all relevant forms to the Office of Research Affairs (ORA). This office screened the proposal for compliance with submission guidelines, then forwarded it for peer review and sent it to the appropriate research committee for evaluation. There was a delay in obtaining the ethical approval because of the establishment of a new research committee there, once the new committee members were designated the proposal has submitted. However below is a time history of the research. "An Evaluation of Patient Safety Culture Tool in Saudi Arabia".

20 April 2009: Proposal submitted to the Research Advisory Council, King Faisal Specialist Hospital and Research Centre for evaluation. 02 May 2009: Proposal forwarded to the Research Ethics Committee (REC). 04 May 2009: approved by the REC through the review process. 18 May 2009: Scheduled for discussion by the Clinical Research Committee (CRC) on next meeting. 26 May 2009: Proposal discussed by CRC - some clarifications were requested from the investigator. The clarification was provided and the proposal was scheduled for discussion again by the CRC. However, this may take some time due to the change of the membership of the Research Advisory Council Supporting Committees, Research Ethics Committee (REC) and Clinical Research Committee (CRC). Finally, the decision approving the proposal was made by the Research Advisory Council (See appendices for the copy of the approval letter from the Research Advisory Council, King Faisal Specialist Hospital and Research Centre).

The third hospital was King Fahad Medical City Riyadh, Saudi Arabia approached on 6/6/2009. The proposal and relevant forms were submitted to the External Research Review Committee, a subcommittee of the Institutional Review Board in King Fahad Medical City. The proposal was reviewed by the External Research Review Committee after that the principal investigator was informed the proposal has approved by the Institutional Review Board (See appendices for the copy of the approval letter from Institutional Review Board of the King Fahad Medical City).

It is important to mention that time is an important factor in the fieldwork of the study. Therefore, the face validity interviews were held in the hospital which issued its agreement the first. King Fahad Hospital was the first one where the interviews were conducted.

#### Appendix 8 King Fahad National Guard Hospital agreement

and the set of the set	in firms the new second of the second second second	a particular and the second second second second		
Health Affairs Ex 4/19/2009gdom005	Saudiana Sa	File # 4513/2009	لمملكة العربية السعودية	J
National Guard	I-Health Affairs	MEREN	س الوطئي – الشؤون الصحية	الحرب
King Abdulazi	a Madical City	12 ST	a ubilitio ellus dittatu	
ing Abuulazi	2 Medical City	Constraint of the second	يعه (عنه عبد)محرين العبيه	
	-	~ ~	-	
Institutional R	eview Board	16586/16669 1515	CLNResearch 1@ngha.r	ned.sa
		MEMODANDI		
		MEMUNANDO	) m B	
		Ref. #: IRBC/021/09		
	10540k33420		Heesson and the second s	
Date:	(G) 14 April 200	9		
	(H) 18 Rabi'll 143	30		
To:	Mr. Mamdooh Sh	ierier Alonazi		
	PhD Student			
	University of She	ffield U.K.		
Subject:	Protocol RC09/0	015 - "An Evaluation of a	Patient Safety Culture Tool in Sau	ibu
	Arabia			

This is in reference to your subject proposal, which has been expedited reviewed by the IRB on 14<sup>th</sup> of April 2009. Upon recommendation of the Research Committee, and following the review of the IRB on the ethical aspects of the proposal, you are granted permission to conduct your study.

Your research proposal is approved for one year commencing from the above date with the following conditions:

#### TERMS OF APPROVAL:

- 1. Annual Reports: Continued approval of this project is dependent on the submission of an Annual Report. Please provide KAIMRC with an Annual Report determined by the date of your letter of approval.
- 2. Amendments to the approved project: Changes to any aspect of the project require the submission of a Request for Amendment to KAIMRC and must not begin without an approval from KAIMRC. Substantial variations may require a new application.
- 3. Future correspondence: Please quote the project number and project title above in any further correspondence.
- 4. Monitoring: Projects may be subject to an audit or any other form of monitoring by KAIMRC at any time.
- 5. Retention and storage of data: The PI is responsible for the storage and retention of original data pertaining to a project for a minimum period of five years.

Prof. Amin Kashmeery Head, Biomedical Ethics Section National Guard Health Affairs

P. O. BoxAld miatiyadh 11426 Tel. 2520088 Telex: 403450 NGRMED SJ

KFII-MATERIALS 14574 (05/96) (ORACLE 29795)

Dr. Mohammed Al Jumah Executive Director, KAIMRC National Guard Health Affairs

Dr. Bandar Al Knawy ()Chief Executive Officer

National Guard Health Affairs

ص. ب ۲۲۶۹۰ الرياض ۲۲۶۹۰ تلفوت : ۸۸ . . ۲۵۲ تلکس NGRMED : ۳٤٥٠ تلکس PALT .A

#### Appendix 9 Support letter from Associate Executive Director of Nursing department

Kingdom of Saudi Arabia National Guard-Health Affairs King Abdulaziz Medical City



المملكة العربية السعودية الحرس الوطني - الشؤون الصحية مدينة الملك عبدالعزيز الطبية

NURSING SERVICES Reference: NSG.JM152.09

DATE: 19 May 2009

#### Memorandum

TO:	Mr. Mandooh Sherier Alonazi
	PHD Student, University of Sheffield, UNK
FROM:	Rob Hemmelder
	(A) Associate Executive Director, Nursing Services, Central Region
SUBJECT:	Research Survey –
	"An Evaluation of a Patient Safety Culture Tool in Saudi Arabia" Ref: IPRC/021/09

Attached is the above referenced memorandum confirming the approval of the Research Committee for you to conduct the above research survey in KAMC-Riyadh.

We support your request to conduct this research survey; and in view thereof, you are authorized to coordinate with the nursing leadership and staff to "face validate" your tool and distribute your research questionnaire in the clinical units / wards.

Best regards.

JM/mj

File

Cc: H,E, Dr. Bandar Al Knawy, Chief Executive Officer, NGHA Mr. Abdullah Al Aamer, Executive Director, Operations Dr. Mohammed Al Jumah, Executive Director, KAIMRC Directors, Clinical Nursing

Att: IPRC/021/09

Research Chrono

⊠1522

2 +9661-2520088 ext. 11584 / 12818 / 11882 / 11780

P. O. Box 22490, Riyadh 11426 Tel. 2520088 Telex : 403450 NGRMED SJ

KFH-MATERIALS 14574 (05/96) (ORACLE 29795)

ص. ب ۲۲۴۹۰ الریاض ۱۱۴۲۹ تلفرن : ۲۰۲۰۰۸۸ تلکس ، NGRMED ٤۰۳٤٥ ۲۵/۱۰۸

A nurserv1@ngha.med.sa

₫ +9661-2520088 ext. 12778

Appendix 10 Support letter from Chairman of Surgery department



**Department of Surgery** 



KINGDOM OF SAUDI ARABIA

	46 1	966-1-252-0088	x14118/14117/14	119/14137/1413	5/14122	966-1-252-0	051 (direct - inc	oming) or 955-1-	252-0088x1413		emall: surgery Ha	ngha med sa
DR AZZAM Chairman Pgr 9988	DR YOUSEF Deputy Chairman Pgr 8271	DR MAZROU Head ENT Pgr 8333	DR BEDAH Head Gen Surgery Pgr 7060	DR RUSSELL Head Neurosurgery Pgr 8241	DR JOSAIR Head Onntha Pgr 8888	DR TUWAIRGI Head O&MFS Pgr 8395	DR KHELAIFI Head Ortho Pgr 7236	DR NAMSHAN Head Paeds Pgr 8265	DR THUNYAN Head Plastic Surg Pgr 8300	DR GANNASS Director Pod Surg Pgr 6868	DR ABOULATION Head Thoracic Pgr 8283	DR SHAMMAR Head Urology Pgr 7226
Offic	e of	the C	hairm	nan								
REF: DATE:	036-05- 18 May 23 Juma	2009rvs 2009 da I 1430										
				M	EN	OR	AN	DU	M			
то		•	ALL DIV Departm National	ISION H ent of Su Guard He	HEADS ( irgery ealth Aff	as per di	stributio	n list)				
FROM	I	:	<b>DR. SAL</b> Chairmar Departm National	EH AL A n ent of Su Guard He	<b>ZZAM</b> rgery ealth Aff	airs	5	/	1			
SUBJ	ест	:	RESEAR	CH PRO	ARABI	N EVALUA	TION O	F A PATI	ENT SA	FETY CUI	TURE	

Please be advised that Mr. Mandoch Alonazi has the approval of the CEO to conduct research on the above mentioned subject, which consists of two parts:

1) Interview

2) Filling up questionnaire

The first part which is the interview is conducted by Mr. Mandooh Alonazi over 50-60 minutes, this interview is confidential and if you are willing to participate it will be carried out at your convenient time and location and the information obtained will help the principle's investigation in his thesis for PhD please see attached form if you do agree to participate, please sign below as follows:

AGICE LI NO II TES ILVES DEASE CONTINETE AL VOUL CONVENIEL	Aaree	D No		Yes if	ves please	complete	at your	convenier	ice
--	-------	------	--	--------	------------	----------	---------	-----------	-----

Time:

Date:

Location:

I look forward to your support and I thank you for taking time to read and participate in either the interview, the questionnaire or both, for any further information please contact Mr. Mandooh Alonazi, please see attached form.

Notice: Please drop your response in the box assigned for this project in the Department of Surgeries office.

Kind regards.

**Distribution List:** 

- 1.
- 3. 4.
- Dr. Khalid Al Mazrou, Head, Division of ENT Surgery Dr. Khalid Al Bedah, Head, Division of General Surgery Dr. Neville Russeli, Head, Division of Neurosurgery Dr. Khalid Al Jobair, Head, Division of Ophthalmology Surgery Dr. Othman Al Tuwairgi, Head, Division of Oral and Maxillofacial Surgery Dr. Ahmad Al Khelaifi, Head, Division of Orthopedic Surgery Dr. Ahmad Al Khelaifi, Head, Division of Pediatric Surgery Dr. Abnammed Al Namshan, Head, Division of Pediatric Surgery 5.
- 6.
- 8.
- Dr. Abdullah Al Thunyan, Head, Division of Plastic Surgery Dr. Mohammed Al Abdulatief, Head, Division of Thoracic Surgery 9.

Administrative Assistant 1 Ronel ext: 14118

#### Appendix 11 King Fahad Medical City agreement

Kingdom of Saudi Arabia Ministry of Health King Fahad Medical City



الملكة العربية السعودية وزارة الصحة مدينة اللك ههد الطبية

Dear Mr Mamdooh AlOnazi,

June 16, 2009

ERRC Number: 09-013

It is my pleasure to inform you that the External Research Review Committee, a subcommittee of the Institutional Review Board, has approved your study titled: "Evaluation of a Patient Safety Culture Tool in Saudi Arabia".

Please be informed that in conducting this study, you as the Principal Investigator is required to abide by the rules and regulations of the Government of Saudi Arabia and KFMC/ERRC. The approval of this proposal will automatically be suspended on June 15, 2010 pending the reapplication to renew the approval.

Please observe the following:

- 1. Personal identifying data should only be collected when necessary for research;
- 2. The data collected should only be used for this proposal;
- 3. Secondary disclosure of personal identifiable data is not allowed.
- 4. No medical tests for study purposes are permitted to be conducted on patients.

We wish you every success in your research endeavor.

Sincerely, Mohamad Al Tannir, DMD, MPH

Head of External Research Review Committee Institutional Review Board King Fahad Medical City, Riyadh, Saudi Arabia Tel: 966 1 288 999 Ext.8391/1299 Email: maltannir@kfunc.med.sa

المرفقات ا

09-0

الرقىم : التاريخ :

# Appendix 12 Support letter from Executive director of Medical Affairs

Kingd Mi King I Executive Di	om of Saudi Arabia inistry of Health Fahad Medical City rector of Medical Affairs Office	يدية بية رن الطبية	المملكة العربية السعو وزارة الصحة مدينة الملك فهد الطر مكتب المدير التنفيذي للشوو
	Executive Director of Medi	cal Affairs 020	Office 7933/4336
То:	Directors, Hospitals/Centers Directors, Medical Administrations Department Heads	Date:	20 Sha'ban 1430 11 August 2009
From:	Dr. Abdulrahman M. Al Mazrou Executive Director, Medical Affairs Chairman, Medical Consultant Council	Ref. No.: Attachment:	30001/ /1430 Questionnaire (8p)
Subject:	Completion of Research Questionnai	re	
In ord colleag researc Your c toward	liversity of Sheffield, UK. ler to make this study significant and s uses to participate in completing the attach cher. cooperation will bring more value to this is creating a safer environment for our patier	uccessful, I am ed questionnair research as this nts.	encouraging my reprepared by the swould be a step
Thank	you and best regards.		
# Appendix 13 King Faisal Specialist Hospital & Research Centre agreement

	A STATE OF STATE	and a second	
	A A A A A A A A A A A A A A A A A A A		
		ndetaa s	No. In the second second second
	صل التخصصي و مركز الإبحاث 8 King Faisal Specialist Hospital سة عامة	نشفی الملك فيد Research Co عز	entre
	OFFICE OF RESEAR 232937 & 27894	CH AFFA	AIRS
	INTERNAL ME	емо	
то:	Nahar Alanezi, MD Associate Consultant Vascular Surgeon Department of Surgery	DATE:	02 Ramadan 1430 23 August 2009
	~. A. ~~	>	
FROM:	Mohamed M. Al-Turki, CCRP Co-Director Office of Besearch Affairs	REF:	ORA/1140/30
SUBJECT:	PROPOSAL # 2091 025	ol in Saudi Ar	abia

Further to the Office of Research Affairs (ORA) memo, ORA/0992/30, your Emailed reply to the Clinical Research Committee (CRC), (received at 05 July 2009), was reviewed by the CRC at 19 August 2009.

We are pleased to inform you that the CRC has accepted the reply and recommended the above-referenced proposal for approval. We take this opportunity to congratulate you on behalf of the Research Advisory Council (RAC).

In conducting this study, the Investigators are required to abide by the rules and regulations of the Government of Saudi Arabia, KFSH&RC, and the RAC. Further, you are required to submit a Final report by 19 July 2010, so it can be reviewed by the Committees without a lapse in approval. The approval of this proposal will automatically be suspended on 10 August 2010, pending the acceptance of the Final Report. You also need to notify the ORA as soon as possible in the case of:

- 1 Any amendments to the proposal
- 2 Termination of the study
- 3 Any event or new information that may affect the benefit/risk ratio of the proposal

Please observe the following:

- 1 Personally identifying data should only be collected when necessary for research
- 2 The data collected should only be used for this proposal
- 3 Data should be stored securely so that only a few authorized users are permitted access to the database
- 4 Secondary disclosures of personally identifiable data are not allowed
- 5 Should there be a need to contact the research subjects for follow-up information, you will need to seek the authorization of the RAC prior to such contact.

We wish you success in your research endeavors.

ORA cc: Chairman, Clinical Research Committee A Chairman of PI(s) Formf11100-01 (Rev. 09-27) I.C. 202028 E -Mail: pra@kfshrc.edu.sa

Appendix 14 King Faisal Specialist Hospital & Research Centre agreement



مستشفى الملك فيصل التخصصى ومركز الأبحاث King Faisal Specialist Hospital & Research Centre مؤسسة عامة . (ien. Org

#### **OFFICE OF RESEARCH AFFAIRS** MBC: 03, TEL: 32937, FAX: 27894

#### INTERNAL MEMO

TO:	Nahar Alanezi, MD Associate Consultant Vascular Surgeon	DATE: 11 Rajab 1430 04 July 2009
	Department of Surgery	0,00,000
THROUGH:	Mohamed Al Turki, CCRP Co- Director	<b>REF:</b> ORA/0992/30
	Office of Research Affairs	
FROM:	Sasha M. Abu Rass, MPharm Proposal Processing Section Office of Research Affairs	<u>199</u>
SUBJECT:	Proposal # 2091 025 An Evaluation of Patients Safety Culture Too	ol in Saudi Arabia – Research Study
The above-refe	renced proposal was reviewed by the Clinica	Research Committee (CRC) and the

Th Research Ethics Committee (REC) on 26 May and 04 May 2009, respectively.

The REC has recommended approving the proposal as submitted.

The CRC was not able to accept the proposal in the current format and request the following:

- 1. Abstract/Summary of the proposal
- 2. Copy of the Questionnaire and Data Collection Sheet
- 3. Detailed work plan for each investigator.
- 4. The investigators were planning to start the project early in year 2009, during the active period of the year for 5 months? What will be the situation if it will start now in summer? Will the investigators be able to conclude the study in the specified period of time?
- 5. There is no indication in the recruitment plan for which hospitals will be included in the study and who are the participants?
- 6. How the participants will be selected (i.e. randomized or invited)? How the selection bias be minimized?

Please forward your reply/ revised proposal to the Office of Research Affairs (ORA) at your convenience, but not later than two months from the date of this memo

ORA m 11100-01 (Rev. 09-27) I.C. 202028 C

SI

E-Mail: ora@kfshrc.edu.sa

#### Appendix 15 Support letter from Deputy Executive Director Nursing Affairs



مستشفى الملك فيصل التخصصي ومركز الإبحاث King Faisal Specialist Hospital & Research Centre مؤسسة عامة . Gen. Org

Ext: 23442	NURSING AFFAIRS - MBC 73	Fax: 23021		
TO:	Nursing Leadership Team DATE: Nursing Affairs	11 Ramadan 1430 01 September 2009		
FROM:	Rita Anderson, RN, BSN, MSN Deputy Executive Director Nursing Affairs	DEDNA: 090 30		
SUBJECT:	PROPOSAL # 2091 025 - An Evaluation of Patients Safety Cultur Ref: ORA/1140/30	re Tools in Saudi Arabic		

As per the attached memo from RAC, Mr. Mamdooh Sherier Alonazi's research proposal is approved for application in King Faisal Specialist Hospital & Research Center.

Please extend your support in this important study for Patient Safety.

Thank you.

RA/ert

11100-01 (Rev. 09-27)

I.C. 202028

A-34

Printed by Reprographics KFSH&RC

	Uncompleted questionnaires																																																									
Q.N	A	A1	A2	A3	A4	A5	A6	A7	A8	A9	A	10 A	11 /	12	A 13	Al	A	A	16 A	17 A	18 H	31 1	B2	B3	<b>B</b> 4	C1	C2	C3	C	4 C:	5 C	6 D	1 D	2 1	D3 H	E	F1	F2	F3	F4	F5	F6	F7	F8	F9	F 10	F 11	G	H1	H2	H3	H4	H5	H6	H7	H8	H9	H 10
1	line :	100	300		105	880	202	169	1999	0.0						103									100	1989		16												2.001	1000	100		1000			1999		100			100					803	
2		1000	55.	100	100	1000	100	1888	1233							194	100			10		1919		100	100	10.15	1923	1993		120	1.00							100	200	100		1433	100	1000	162		1000	100		100	100	1000					201	No.
3	40	100	1000		1949	100	100	100	1000	-					863	100	100									20	120	100	100	1.00							2011			182	(0)	1900		1998	and the			1000		198	100	196					-	
4	1000	100	30241		100	200	100	(inst	1.00	1.000					0.21	1950	100						100	100		200	1 and	100	100	1.10		0. 78	11					198	100	100	1938	1000	1000	100		1202	100	1000	1000	1000	1000	10.00			100			1000
5		1000	2021	2201	2.0		100	182	1000						1000	1986									1000													100	120	105	1002	1	100			0.00	100	1000	100		200		1.44		100	100		
6	100		0.00	1000	100	1.200			1.55	1.5%						0.04		1	1							100	1865	100		1		1						100	1	100	1000	100	200		2.6	100	100	1000	1.00	100	10.00	1000	100		CO.			
7	1005	100		1223	-	1000	1.00								100	127									1962	2007	1.5%	1000					1					122	1999	120	1000		12.0			1.5.1			1000	1986	100	1.167				100		
8	1000	1000	100	1000	1997	1000	100	1000		1			+		Sec.	100																	+	+	-					100	100	1000	3.5		1000							1100						
9	1000	100	100		10	1000	1251		100	1	+			+		1993			+			-	+		100			100	1	1	-	+	1	+	-							1000	-	1.50			100		100		100						Test.	
10				1		1940	1010		1	1	+	+		1		200		1	+						10.5	1.25		100		-		+	1		-	+					CONT.	-						-	1	1					-			and a
11	1000	-					1000		1000		+	+	+			1000			+	-		+			12302	1230			+	-	-		-	+	-	+				and the second	-	Sint	1000	3012	and the	1000	1000		0.00		1000	100	-		-			
12	1000	-		-	1000	and a	2000		-		+	+	+	+		1010	-	+	+	+		-			-	1000	1.000				-	-	+	+		+					1000	1.000			-	-	-	-	-		-	-	-		-			-
13	-		1000		100	-			-		+			-		100	-	-	+		-	-	-	-		1000	-	-	+	-	-	+	+	+	-	-		-	-	-			-		-	-	-	-	-	-	-	-	-		-	-		-
14	-40823	-	1000		10.0				-		+	+	+	-		1.0	-	-	+	+	-	+	+	-		-			+	-	+	+	+	+	-	+		-		A STATE	-	-	1.000	-			-			1.00	-		1000			-		
15		1000	10000		7.08			-	-		+	+	+	+		-	+	+	+	+	+	-	-	-		-		-	-	+	+	+	+	+	-	+	-	-			1000		-	-			-		-	-	-	-	-		-	-		-
16								-	-	-	+	+	+	+	1000		-	+	+	+	+	-	+	-	100	-	-		+	+	+	+	+	+	-	+	-	-	-	Training of	1000					1000	100.00	-	-	-	-	-	-		-+	-	1016	_
17			100		-		-	-	-	-	+	+	+	-		-	-	-	+	+	+	-	+	-	1000		-	-	-	+	+	+	+	+	-	+	-		-	a har	-	-	-	-			-	-	-	-	-	-	-		-+	-	1	
18	-	ALC: NO	1000	-	-			-	+	+	+	+	+	-	Same I	-	-	-	+	+	+	-			-	-	-	-	-	-	+	+	+	+	-	-		-		-	-	-	-				-	-	-	-	-	-	-		-	-		
10	1.5	-			(Section)		-	-	-	-	+	+	-	-	-	-	-	-	+	-	+	-	-	-	1000	-		-	-	-	+	+	+	+	-	+	-		-				-	-			-	-	-	-	-	-	-		-	-		
20	1.000	1.1.1.1						-	+	-	+	+	+	-	-	-	+		+	+	-	-	-			-	-	-	+	+	+	+	+	+	-	-	-	-	-	-	-	-					-	-	-	-	-	-	-		-	-		-
2		1000				-	10000		-	100	+	+	-	-		-	+	+	+	+	-	-	-	-	-	-	100	-	+	-	+	+	+	+	-	-	-	-	-	-			-	-		-		-	-	-	-	-	-	$\vdash$		-		
22	-	100-04-		-	-		-	-	-	-	+	+	-	-		-	-	+	+	-	-	-	-	_			-	-	+	+	-	+	+	+	-	-	-	-	-	100	-		-	-	-	1000		-			-	-	-	$\square$	-	-		_
23		1.0			-	-		-	-	-	+	+	-	-	24		+	+	+	+	-	-	-				-	-	+	+	+	+	+	+	-	-	-	_			-	-	-	-	-	-	-	-	-	-	-	-	-	$\vdash$	-	-		-
20	1992	1995.0	1000	200			-	100	-	-	+	+	-	-		-	-	+	+	+	-	-	-		-		-	-	+	+	-	-	+	-	-	-		-			-	-	-			-	-	-	-	-			-	$\square$	-	-		_
24			-	_		-	-	-	-	-	+	+	-	-		-	-	+	+	-	-	-	-		-	-	-	-	+-	+	+	+	+	+	-	-	-		_		-	-	-	-	-	-	-	-	-	-	-	-	-	$\square$	-	-	-	_
20					-	100		-	-	-	+	+	-	-	5.950	-	-	+	+	+	-	-	-		-	-	-	-	+	+	+	+	+	+	-	-	1000				-			-			-	-	-	-	-		-	$\square$				_
20					120	-		-	-	+	+	+	+	-	100	-	+	+	+	+	-	-	-	-	-		-	+	+	+	+	+	+	+	-	-	-		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	$\vdash$		-	-	-
21	1000	140			100	1000	-	-	-	-	+	-	-	_		-	+	+	-	-	-	-	-	100	-	-	-	-	+	+	+	+	+	+	-	-	1992		100			-	-	1	-		-	-	-	-	-	-	-	+			-	
20	1	1000		-	1	-	-	-	-	-	+	+	-	_		-	+	+	-	-	-	-	-	100	-	-	-	+	+	+	+	-	+	-	-	-		100	-		-	-	-		-	-	-	-	-	-	-	-	-	+		-	100	-
23	1		39.95	100			-	-	-	-	+	-	-	_		-	+	+	+	-	-	-	-	100	-		-	-	-	-	-	-	-	-	_	_			1989	1	-	-	-	-	-	-	-	-	-	-	-	-	-					_
30		1000		1000		100	100	-	1000	-	+	+	-	_	101	-	+	+	+	-	-	-	-		-		-	-	+	-	-	-	-	+	_	_					-	-	-	-			-	-	-		-							_
10		100	_			-			-	-	-	-	-			-	-	-	+	-	-	-					-	-	+	-	-	-	-	-	-	_	17.8		1000	100	-	-	-	-	-		-	-	-	-	-	-	-			_		
34		19.88	1996		1				-	-	-	-			1.19		-	-	-	-	-	-					-	-	+	-	-	-	-	-		_				-	-	-	-	-			-	-		-	-		-					
30	1963	122	100	1996	100	199	1	-	-	-	-	-	2	2	23		-	-	-	-	-						-	-	-	-	-	-		-	_	_			1.00	1.00	-					100	123			-	-	100	-					
34			1	- 61				-		-	-	-			100	-	-	-	-	-		_	_				-	-	-	-	-	-	-	-	-					100	-	-	-			202		-	-	-		-				100		
3		100	1	200	30		-	-		-	-	-		100	1	-	1	-		-							-	-	-	-	-	-	-	-			832		- 87			-	-				1	-	-	-	-	-	-	1				1
1 30				1		-	-		-	-				1200		-	-	-											1	-	-	-	-	-		-	-		1			-	-		-		-	-	-	-	-		-	1		100		
31		1976	100		100					-				200				-									-	-	-	-						1000	1		1		-	-	-					-					-			100	122	
38	200	100/	1007		12	100	100							2292					10			192			100											0.8	1111		100					1		100							1					
39	1	1	12.5	2.0		100	100						100	100									128	100		12									120	100	12		200	199								100	1							0.0	-39	
40		1922	100	1949	193	1013	181	199							10.0			2.3				100	100	152	100			1.53			2 2						100.		1005	1200	20	100	199		100	100			1		12		1			1000	10	
4	100	1000	30	100		198		100				25	38		100								1001							1							-		1280	199	18	100			10													
42	199	1	100	198	363		1997								100		Т		Т				100	110		110									100		100		100	1107		100			100											100	100	
43	100	(in):		1000	1953	100	100						600	00	185								101	1000		11									1999				1982	12302	125		120								100					100	100	
44	100	1	100	100	185			100				T		100	100		T	T		T				190										T		001	1			1		1.8								1						1212	107	
4	1000	33	23	101										in S									100	1200	100												1233		1.28	1800	180	100		13												100	100	
40			1000			100		1															Sec.	200													101		1								1			1		T	1				100	
4			100			1	1	1		1				100		1	1	1	1				100	199		1	1	1	1				1				1000			1992			1		1	1	1	1	1		1	1	-			123		
4			1					1		1	1			liney.		1	1	-	1				1997	1	1	1	1	-	1	-	-	1									1	T	1	1	1		1	1	-	1		1		1				
4			1		1		1	-	1	1	+	1		10mm		-	+	+	+				100		1	1		1	+		+		1			10	-			-	1	1	1	1	1	1	1	1	1	-	-	1	-	-				
5		100	1	1966	1		1	-		+	+					1	+	+	+	1	1		1000	-	1	1	1	+	+	-	+		+	1					-	1	1	1	1	1	1	1	1	-	-	-	-	1	-	-		1000		
5	1					100	1	1	-		1			352	1	100	+	+	+	-	-		1		1	1	1	+	+	+	+	-	+	+		102					1	1	1	1	1	1	1	1	-	1	-	1		1		-		
-	1			COL			0.00	601	10	110		200	10.4		140	-	_	-	-	-	_	-	-	-	1	-	-	-	-	1	-	_	-	_	_		-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-		1.00	1	

Number	Variable label	Туре	Measurement
of			
variable			<b>MATHOME</b>
Α	Work area of respondent	Numeric	Nominal
H1	Current position	Numeric	Nominal
H2	Gender	Numeric	Nominal
H3	Nationality	Numeric	Nominal
H4	Highest educational qualification	Numeric	Ordinal
H5	Training outside S.A	Numeric	Nominal
H6	Years of working in this hospital	Numeric	Ordinal
H7	Years of working in current hospital work area	Numeric	Ordinal
H8	Working hours in the hospital	Numeric	Ordinal
H9	Direct contact with patients	Numeric	Nominal
H10	Years of experience in current specialty	Numeric	Ordinal

### Appendix 17 Data management table (background items)

# Appendix 18 Data management (safety climate items)

Number of	Variable	Туре	Measurement
variable	a substitution of the state of the second		to the fit of the
A1	Staff support one another in this unit	Numeric	Ordinal
A2	We have enough staff to handle the workload	Numeric	Ordinal
A3	When a lot of work needs to be done quickly, we work together as a team to get the work done	Numeric	Ordinal
A4	In this unit, people treat each other with respect	Numeric	Ordinal
A5	Staff in this unit work longer hours than they should, which is not good for patient care	Numeric	Ordinal
A6	We are actively doing things to improve patient care	Numeric	Ordinal
A7	We use more temporary staff than we should, which is not good for patient care	Numeric	Ordinal
A8	Staff feel like their mistakes are held against them	Numeric	Ordinal
A9	Mistakes have led to positive changes here	Numeric	Ordinal
A10	It is just by chance that more serious mistakes do not happen around here	Numeric	Ordinal
A11	When one area in this unit gets really busy, others help out	Numeric	Ordinal
A12	When an event is reported, it feels like the person is being written up, not the problem	Numeric	Ordinal
A13	After we make changes to improve patient safety, we evaluate their effectiveness	Numeric	Ordinal
A14	We work in a hurry, trying to do too much, too quickly	Numeric	Ordinal
A15	Patient safety never takes second place to get more work done	Numeric	Ordinal
A16	Staff worry that mistakes they make are kept in their personnel file	Numeric	Ordinal
A17	We have patient safety problems in this unit	Numeric	Ordinal

A18	Our procedures and systems are good at preventing errors from happening	Numeric	Ordinal
B1	My supervisor/manager says a good word when he/she sees a job done according to established patient safety procedures	Numeric	Ordinal
B2	My supervisor/manager seriously considers staff suggestions for improving patient safety	Numeric	Ordinal
B3	Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it means taking shortcuts	Numeric	Ordinal
B4	My supervisor/manager ignores patient safety problems that happen over and over	Numeric	Ordinal
C1	We are given feedback about changes put into place based on event reports	Numeric	Ordinal
C2	Staff will freely speak up if they see something that may negatively affect patient care	Numeric	Ordinal
C3	We are informed about errors that happen in this unit	Numeric	Ordinal
C4	Staff feel free to question the decisions or actions of those with more authority	Numeric	Ordinal
C5	In this unit, we discuss ways to prevent errors from happening again	Numeric	Ordinal
C6	Staff are afraid to ask questions when something does not seem right	Numeric	Ordinal
D1	When a mistake is made, but is caught and corrected before affecting the patient, how often is this reported?	Numeric	Ordinal
D2	When a mistake is made, but has no potential to harm the patient, how often is this reported?	Numeric	Ordinal
D3	When a mistake is made that could harm the patient, but does not, how often is this reported?	Numeric	Ordinal

Е	An overall grade on patient safety	Numeric	Ordinal
F1	Hospital management provides a work climate that promotes patient safety	Numeric	Ordinal
F2	Hospital units do not coordinate well with each other	Numeric	Ordinal
F3	Some things do not happen or get missed when transferring patients from one unit to another	Numeric	Ordinal
F4	There is good cooperation among hospital units that need to work together	Numeric	Ordinal
F5	Important patient care information is often lost during shift changes	Numeric	Ordinal
F6	It is often unpleasant to work with staff from other hospital units	Numeric	Ordinal
F7	Problems often occur in the exchange of information across hospital units	Numeric	Ordinal
F8	The actions of hospital management show that patient safety is a top priority	Numeric	Ordinal
F9	Hospital management seems interested in patient safety only after an adverse event happens	Numeric	Ordinal
F10	Hospital units work well together to provide the best care for patients	Numeric	Ordinal
F11	Shift changes cause problems for patients in this hospital	Numeric	Ordinal
G	Number of event reports in the past 12 months	Numeric	Ordinal

# Appendix 19 Duple data entry (15%)

NO	Number of questionnaire	Number of item	Value in DF1A1 (original file)	Value in DF1A2 (re- entering)	Correct Value in questionnaire	Site of data entry errors
1	728	A4	5	3	3	DF1A1
2	728	A6	3	4	4	DF1A1
3	728	A8	4	2	2	DF1A1
4	728	A9	2	4	4	DF1A1
5	520	A5	2	3	3	DF1A1
6	687	A5	3	2	2	DF1A1
7	687	H10	6	3	6	DF1A2
8	486	A9	4	2	4	DF1A2
9	486	A11	2	5	2	DF1A2
10	486	A12	5	3	5	DF1A2
11	486	A14	3	4	3	DF1A2
12	486	A17	4	3	4	DF1A2
13	486	A18	3	2	3	DF1A2
14	246	A11	4	2	4	DF1A2
15	232	A13	2	4	4	DF1A1
16	339	A15	2	3	3	DF1A1
17	720	B3	2	1.	1	DF1A1
18	720	B4	2	1	1	DF1A1
19	577	C1	3	2	3	DF1A2
20	707	D1	3	2	2	DF1A1
21	707	D2	3	2	2	DF1A1
22	583	D2	3	2	3	DF1A2
23	653	D2	5	3	3	DF1A1
24	633	F1	4	2	4	DF1A2
25	633	F2	2 -	3	2	DF1A2
26	633	F3	3	4	3	DF1A2
27	633	F4	4	1	4	DF1A2
28	633	F6	1	2	1	DF1A2
29	633	F7	2	5	2	DF1A2

30	653	F8	5	2	5	DF1A2
31	633	F9	2	4	2	DF1A2
32	633	F11	4	2	4	DF1A2
33	633	G	2	6	2	DF1A2
34	633	H1	6	1	6	DF1A2
35	633	H5	1	4	1	DF1A2
36	633	H6	4	2	4	DF1A2
37	633	H7	2	3	2	DF1A2
38	633	H8	3	1	3	DF1A2
39	633	Н9	1	3	1	DF1A2
40	390	G	2	1	2	DF1A2
41	268	H1	9	11	9	DF1A2
42	268	H4	5	3	5	DF1A2
43	570	H1	12	11	12	DF1A2
44	570	H10	1	2	1	DF1A2
45	7	H3	2	1	2	DF1A2
46	196	H4	4	5	5	DF1A1
47	231	H4	4	3	3	DF1A1
48	462	H4	5	4	5	DF1A2
49	722	H5	1	2	2	DF1A1
50	822	H5	1	2	2	DF1A1
51	835	H5	1	2	2	DF1A1
Total	23	51	51	51	51	51

,

# Appendix 20 Duple data entry for the rest of the questionnaires

NO	Number of	Number	Value in First	Value in	Correct Value	Site of
	questionnaire	of item	entry	Double entry	in	data entry
					questionnaire	errors
1	10	A	3	4	3	Double
2	10	A2	4	5	4	Double
3	10	A3	5	4	5	Double
4	10	A5	4	5	4	Double
5	10	A6	5	1	5	Double
6	10	A8	1	5	1	Double
7	10	A10	5	4	5	Double
8	10	A11	4	5	4	Double
9	10	A12	5	4	5	Double
10	10	A14	4 .	5	4	Double
11	10	A16	5	4	5	Double
12	14	H5	1	2	1	Double
13	14	H6	3	1	3	Double
14	14	H10	3	1	3	Double
15	38	H5	2	1	2	Double
16	41	F9	2	4	2	Double
17	41	F10	4	2	4	Double
18	64	H10	3	2	2	First
19	67	H4	5	4	5	Double
20	74	H2	1	2	1	Double
21	74	H3	1	2	1	Double
22	74	H8	3	5	5	First
23	77	A12	3	4	4	First
24	79	A15	4	5	4	Double
25	79	A16	5	4	5	Double
26	79	A17	2	4	2	Double
27	79	A18	4	3	4	Double
28	79	B2	3	2	3	Double
29	79	C3	2	1	2	Double
30	79	C4	1	3	1	Double
31	79	C5	3	4	3	Double

32	79	C6	4	1	4	Double
33	79	D1	1	2	1	Double
34	79	D2	2	3	2	Double
35	134	H2	2	1	2	Double
36	136	H1	12	11	12	Double
37	136	H8	3	4	4	First
38	137	C1	2	4	4	First
39	137	C2	2	4	4	First
40	137	C3	2	4	4	First
41	137	C5	2	4	4	First
42	137	C4	2	4	4	First
43	144	A1	4	3	3	First
44	144	A2	5	4	4	First
45	144	A3	3	5	5	First
46	144	A4	1	3	3	First
47	144	A5	5	1	1 .	First
48	144	A6	1	5	5	First
49	144	A7	5	1	1	First
50	145	H8	4	6	6	First
51	162	H5	1	2	1	Double
52	164	A	4	11	4	Double
53	168	H5 .	1	2	1	Double
54	170	H8	3	1	3	Double
55	170	H7	2	3	2	Double
56	171	H5	1	2	1	Double
57	179	H5	1	2	1	Double
58	252	F10	2	3	3	First
59	252	F9	2	3	3	First
60	252	Н5	1	2	2	First
61	253	H4	5	4	5	Double
62	255	C5	4	3	4	Double
63	261	H5	2	1	1	First
64	262	H8	4	3	3	First
65	263	H5	2	1	1	First
66	264	H5	2	1	1	First
COPPER STOL				and the second s		

67	266	H5	2	1	1	First
68	276	H9	2	1	1	First
69	284	F8	4	3	4	Double
70	288	B4	2	4	2	Double
71	288	C4	4	3	4	Double
72	288	H1	12	11	11	First
73	314	H10	4	5	5	First
74	318	H1	9	11	11	First
75	319	H10 ·	1	6	1	Double
76	322	A	11	2	11)	Double
77	324	C4	4	5	4	Double
78	324	C5	5	4	5	Double
79	324	C6	2	5	2	Double
80	324	D1	4	2	4	Double
81	324	D2	3	4	3	Double
82	324	D3	5	3	5	Double
83	325	A5	5	2	2	First
84	325	A4	2	4	4	First
85	325	A6	4	5	5	First
86	326	H1	9	11	9	Double
87	335	A17	3	5	3	Double
88	337	A17	2	4	2	Double
89	339	A13	and the second second	4	3	Double
90	341	A8	2	4	2	Double
91	341	A9	4	2	4	Double
92	341	A10	2	4	2	Double
93	341	A11	4	2	4	Double
94	341	A12	2	4	2	Double
95	345	A9	5	2	5	Double
96	346	A	4	7	7	First
97	367	C6	1	4	1	Double
98	367	D1	4	5	4	Double
99	391	D3	3	4	4	First
100	393	A	5	7	5	Double
101	396	A1	4	2	4	Double

102	396	A2	2	4	2	Double
103	396	A5	4	5	4	Double
104	396	A6	5	2	5	Double
105	396	A7	2	4	2	Double
106	396	A9	4	2	4	Double
107	396	A10	2	4	2	Double
108	396	A12	4	5	4	Double
109	396	A13	5	4	5	Double
110	396	A14	4	5	4	Double
111	396	A15	5	4	5	Double
112	396	A16	4	3	4	Double
113	396	A17	3	4	3	Double
114	396	B2	4	2	4	Double
115	396	B4	2	5	2	Double
116	400	B4	4	2	4	Double
117	400	B2	2	4	4	First
118	407	F1	4	5	5	First
119	424	F7	4	3	4	Double
120	430	H3	2	1	2	Double
121	455	F1	4	2 .	4	Double
122	455	F2	2	4	2	Double
123	455	F4	4	3	4	Double
124	455	F6	3	4	3	Double
125	457	Е	3	4	3	Double
126	463	F1	5	4	5	Double
127	471	A18	2	4	4	First
128	476	B2	2	4	4	First
129	476	B3	1	2	2	First
130	476	B4	4	1	1 6	First
131	478	F8	2	4	4	First
132	478	F9	4	2	2	First
133	478	F10	2	4	4	First
134	478	F11	4	2	2	First
135	494	A8	4	2	2	First
136	494	F11	2	3	3	First
	the second s	and the second s			and the second se	and the second

138 $513$ A8455139 $521$ $F10$ 434140 $522$ $F9$ 233141 $522$ $F8$ 244142 $526$ $H3$ 211143 $531$ $H5$ 121144 $532$ $A12$ 322145 $532$ $A13$ 535146 $532$ $A14$ 252147 $532$ $A15$ 323	
139 $521$ $F10$ 434140 $522$ $F9$ 233141 $522$ $F8$ 244142 $526$ $H3$ 211143 $531$ $H5$ 121144 $532$ $A12$ 322145 $532$ $A13$ 535146 $532$ $A14$ 252147 $532$ $A15$ 323	First
140       522       F9       2       3       3         141       522       F8       2       4       4         142       526       H3       2       1       1         143       531       H5       1       2       1         144       532       A12       3       2       2         145       532       A13       5       3       5         146       532       A14       2       5       2         147       532       A15       3       2       3	Double
141       522       F8       2       4       4         142       526       H3       2       1       1         143       531       H5       1       2       1         144       532       A12       3       2       2         145       532       A13       5       3       5         146       532       A14       2       5       2         147       532       A15       3       2       3	First
142       526       H3       2       1       1         143       531       H5       1       2       1         143       531       H5       1       2       1         144       532       A12       3       2       2         145       532       A13       5       3       5         146       532       A14       2       5       2         147       532       A15       3       2       3	First
143531H5121144532A12322145532A13535146532A14252147532A15323	First
144       532       A12       3       2       2         145       532       A13       5       3       5         146       532       A14       2       5       2         147       532       A15       3       2       3	Double
145     532     A13     5     3     5       146     532     A14     2     5     2       147     532     A15     3     2     3	First
146         532         A14         2         5         2           147         532         A15         3         2         3	Double
147 532 A15 3 2 3	Double
	Double
148 532 A17 2 3 2	Double
149 532 A18 4 2 4	Double
150 534 A14 3 2 3	Double
151 534 A15 3 2 3	Double
152 536 H9 1 2 1	Double
153 536 H10 3 2 3	Double
154 537 A5 2 3 2	Double
155 537 A12 2 3 2	Double
156 540 F9 4 5 4	Double
157 543 C6 1 5 1	Double
158 543 D1 2 1 2	Double
159 543 E 5 2 5	Double
160 543 F2 2 5 2	Double
161 543 F3 5 2 5	Double
162 543 F4 4 5 4	Double
163 543 F5 1 4 1	Double
164 543 F6 2 1 2	Double
165 543 F8 5 2 5	Double
166 543 F9 1 5 1	Double
167 543 F10 5 1 5	Double
168 543 F11 3 5 3	Double
169 543 G 1 3 1	Double
170 547 C5 2 3 3	First
171 547 C6 2 3 3	First

172	548	H3	2	1	1	First
173	550	H1	6	11	11	First
174	554	А	2	5	5	First
175	554	A1	3	4	4	First
176	554	A2	3	4	4	First
177	554	A3	3	4	4	First
178	554	A4	3	4	4	First
179	554	A5	3	2	2	First
180	554	A6	4	2	2	First
181	554	A7	3	4	4	First
182	554	A9	4	2	2	First
183	554	A8	3	4	4	First
184	554	A11	4	2	2	First
185	554	A13	3	2	2	First
186	554	A18	4	2	2	First
187	554	B1	4	1	1	First
188	554	B2	4	1	1	First
189	554	B3	4	2	2	First
190	554	B4	4	2	2	First
191	554	C1	3	2	2	First
192	554	C2	3	2	2	First
193	554	C3	3	2	2	First
194	554	C4	3	2	2	First
195	554	C5	3	2	2	First
196	554	D1	3	2	2	First
197	554	D2	3	4	4	First
198	554	D3	3	5	5	First
199	554	F1	4	2	2	First
200	554	F3	4	5	5	First
201	554	F4	4	2	2	First
202	554	F6	4	5	5	First
203	554	F7	4	5	5	First
204	554	F8	3	2	2	First
205	554	F9	3	4	4	First
206	554	F10	3	2	2	First
		the second s		and the second		

207	554	H1	1	6	6	First
208	554	H2	1	2	2	First
209	554	H4	3	1	1	First
210	554	H5	2	1	1	First
211	554	H6	2	6	6	First
212	554	H7	2	6	6	First
213	554	H10	2	6	6	First
214	556	A9	2	4	2	Double
215	565	E	3	4	3	Double
216	548	H2	2	1	2	Double
217	567	A17	2	3	2	Double
218	567	C5	5	4	5	Double
219	567	D1	2	3	2	Double
220	578	B1	3	5	3	Double
221	578	B2	3	2	3	Double
222	575	G	3	2	3	Double
223	590	F7	2	3	3	First
224	590	F9	3	4	4	First
225	590	F10	4	3	3	First
226	590	F11	3	1	1	First
227	602	F11	2	4	2	Double
228	602	G	1	2		Double
229	615	A	11	3	3	First
230	619	A	11	3	3	First
231	641	A10	2	4	2	Double
232	635	H9	2	1	1	First
233	638	D1	2	3	3	First
234	641	A13	4	2	2	First
235	641	F8	5	4	4	First
236	641	H6	4	5	5	First
237	642	H9	2	1 1	1	First
238	646	H9	2	1	1	First
239	652	G	4	5	5	First
240	689	C3	3	4	4	First
241	692	A12	2	3	3	First

243       703       F3       2       3       3       First         244       711       A5       3       2       2       First         245       712       D1       3       2       2       First         246       715       A7       4       5       4       Double         247       715       D2       5       4       4       First         248       719       A9       4       5       4       Double         249       719       A10       3       2       3       Double         250       719       A11       4       3       4       Double         251       719       A12       3       2       3       Double         253       719       A13       5       4       3       Double         253       719       A16       4       3       4       Double         254       719       A16       4       3       4       Double         255       719       B1       4       5       4       Double         255       719       B2       4       5 <td< th=""><th>242</th><th>702</th><th>F1</th><th>5</th><th>4</th><th>5</th><th>Double</th></td<>	242	702	F1	5	4	5	Double
244711A5322First245712D1322First246715A7454Double247715D2544First248719A9454Double249719A10323Double250719A11434Double251719A12323Double252719A13545Double253719A14343Double254719A15232Double255719A16434Double256719A18434Double257719B1454Double258719B2454Double260719B3232Double261719C1434Double262719C2233Double263719C3545Double264719C4444Double275719D1545Double264719C5545Double265719C5545D	243	703	F3	2	3	3	First
245 $712$ D1322First246 $715$ $A7$ 454Double247 $715$ D2544First248 $719$ $A9$ 454Double249 $719$ $A10$ 323Double250 $719$ $A11$ 434Double251 $719$ $A12$ 323Double252 $719$ $A13$ 545Double253 $719$ $A14$ 343Double254 $719$ $A16$ 434Double255 $719$ $A16$ 434Double256 $719$ $A16$ 434Double257 $719$ $B1$ $4$ 54Double258 $719$ $B2$ $4$ 54Double260 $719$ $B3$ 232Double261 $719$ $C1$ $4$ 34Double262 $719$ $C2$ $2$ $3$ $3$ Double263 $719$ $C3$ $5$ $3$ $5$ Double264 $719$ $C1$ $4$ $4$ 4Double265 $719$ $C2$ $2$ $3$ $3$ Double264 $719$ $C2$ $2$ $3$ $5$ Double265 $719$ $C5$	244	711	A5	3	2	2	First
246715A7454Double247715D2544First248719A9454Double249719A10323Double250719A11434Double251719A12323Double252719A13545Double253719A14343Double254719A15232Double255719A16434Double256719A18434Double257719B1454Double258719B2454Double259719B2454Double260719B4232Double261719C1434Double262719C2233Double263719C3545Double264719C444Double265719C5545Double264719D1545Double265719D2545Double266719D2545Double <td>245</td> <td>712</td> <td>D1</td> <td>3</td> <td>2</td> <td>2</td> <td>First</td>	245	712	D1	3	2	2	First
247         715         D2         5         4         4         First           248         719         A9         4         5         4         Double           249         719         A10         3         2         3         Double           250         719         A11         4         3         4         Double           251         719         A12         3         2         3         Double           252         719         A13         5         4         5         Double           253         719         A14         3         4         3         Double           254         719         A16         4         3         4         Double           255         719         A16         4         3         4         Double           255         719         A18         4         3         4         Double           256         719         B1         4         5         4         Double           257         719         B2         4         5         4         Double           257         719         B2         4	246	715	A7	4	5	4	Double
248         719         A9         4         5         4         Double           249         719         A10         3         2         3         Double           250         719         A11         4         3         4         Double           251         719         A12         3         2         3         Double           251         719         A13         5         4         5         Double           253         719         A14         3         4         3         Double           254         719         A15         2         3         2         Double           255         719         A16         4         3         4         Double           255         719         A18         4         3         4         Double           256         719         B1         4         5         4         Double           257         719         B1         2         3         2         Double           258         719         B2         4         5         Double         2           260         719         B3         2	247	715	D2	5	4	4	First
249         719         A10         3         2         3         Double           250         719         A11         4         3         4         Double           251         719         A12         3         2         3         Double           252         719         A13         5         4         5         Double           253         719         A14         3         4         3         Double           254         719         A16         4         3         4         Double           255         719         A16         4         3         4         Double           255         719         A18         4         3         4         Double           256         719         A18         4         3         4         Double           257         719         B1         4         5         4         Double           258         719         B2         4         5         4         Double           260         719         B4         2         3         2         Double           261         719         C1         4	248	719	A9	4	5	4	Double
250         719         A11         4         3         4         Double           251         719         A12         3         2         3         Double           252         719         A13         5         4         5         Double           253         719         A14         3         4         3         Double           254         719         A15         2         3         2         Double           255         719         A16         4         3         4         Double           255         719         A16         4         3         4         Double           256         719         A18         4         3         4         Double           257         719         B1         4         5         4         Double           258         719         B2         4         5         4         Double           259         719         B4         2         3         2         Double           261         719         C1         4         3         4         Double           262         719         C2         2	249	719	A10	3	2	3	Double
251       719       A12       3       2       3       Double         252       719       A13       5       4       5       Double         253       719       A14       3       4       3       Double         254       719       A15       2       3       2       Double         255       719       A16       4       3       4       Double         255       719       A16       4       3       4       Double         256       719       A18       4       3       4       Double         257       719       B1       4       5       4       Double         258       719       B2       4       5       4       Double         259       719       B3       2       3       2       Double         260       719       B4       2       3       2       Double         261       719       C1       4       3       4       Double         262       719       C2       2       3       5       Double         264       719       C4       4       4	250	719	A11	4	3	4	Double
252       719       A13       5       4       5       Double         253       719       A14       3       4       3       Double         254       719       A15       2       3       2       Double         255       719       A16       4       3       4       Double         255       719       A16       4       3       4       Double         256       719       A18       4       3       4       Double         257       719       B1       4       5       4       Double         258       719       B2       4       5       4       Double         259       719       B3       2       3       2       Double         260       719       B4       2       3       2       Double         261       719       C1       4       3       4       Double         262       719       C2       2       3       5       Double         263       719       C3       5       3       5       Double         264       719       C6       1       2	251	719	A12	3	2	3	Double
253       719       A14       3       4       3       Double         254       719       A15       2       3       2       Double         255       719       A16       4       3       4       Double         256       719       A18       4       3       4       Double         256       719       A18       4       3       4       Double         257       719       B1       4       5       4       Double         258       719       B2       4       5       4       Double         259       719       B3       2       3       2       Double         260       719       B4       2       3       2       Double         261       719       C1       4       3       4       Double         262       719       C2       2       3       5       Double         263       719       C3       5       3       5       Double         264       719       C6       1       2       1       Double         265       719       D3       1       2	252	719	A13	5	4	5	Double
254 $719$ $A15$ $2$ $3$ $2$ Double $255$ $719$ $A16$ $4$ $3$ $4$ Double $256$ $719$ $A18$ $4$ $3$ $4$ Double $257$ $719$ $B1$ $4$ $5$ $4$ Double $257$ $719$ $B1$ $4$ $5$ $4$ Double $258$ $719$ $B2$ $4$ $5$ $4$ Double $259$ $719$ $B3$ $2$ $3$ $2$ Double $260$ $719$ $B4$ $2$ $3$ $2$ Double $261$ $719$ $C1$ $4$ $3$ $4$ Double $262$ $719$ $C2$ $2$ $3$ $3$ Double $263$ $719$ $C2$ $2$ $3$ $5$ Double $264$ $719$ $C4$ $4$ $4$ $4$ Double $265$ $719$ $C5$ $5$ $4$ $5$ Double $266$ $719$ $C6$ $1$ $2$ $1$ Double $266$ $719$ $D1$ $5$ $4$ $5$ Double $268$ $719$ $D3$ $1$ $2$ $3$ $2$ $3$ $270$ $719$ $E$ $3$ $2$ $3$ $2$ Double $271$ $719$ $F1$ $4$ $3$ $4$ Double $272$ $719$ $F2$ $2$ $3$ $2$ Double $273$ $719$ $F3$ $4$ $3$ $4$ Double <td>253</td> <td>719</td> <td>A14</td> <td>3</td> <td>4</td> <td>3</td> <td>Double</td>	253	719	A14	3	4	3	Double
255       719       A16       4       3       4       Double         256       719       A18       4       3       4       Double         257       719       B1       4       5       4       Double         258       719       B2       4       5       4       Double         259       719       B3       2       3       2       Double         260       719       B4       2       3       2       Double         260       719       B4       2       3       2       Double         261       719       C1       4       3       4       Double         262       719       C2       2       3       3       Double         263       719       C3       5       3       5       Double         264       719       C4       4       4       Double       2         265       719       C5       5       4       5       Double         266       719       D1       5       4       5       Double         266       719       D3       1       2 <td< td=""><td>254</td><td>719</td><td>A15</td><td>2</td><td>3</td><td>2</td><td>Double</td></td<>	254	719	A15	2	3	2	Double
256       719       A18       4       3       4       Double         257       719       B1       4       5       4       Double         258       719       B2       4       5       4       Double         259       719       B3       2       3       2       Double         260       719       B4       2       3       2       Double         261       719       C1       4       3       4       Double         261       719       C1       4       3       4       Double         262       719       C2       2       3       S       Double         263       719       C3       5       3       S       Double         264       719       C4       4       4       Double       Double         265       719       C5       5       4       S       Double         266       719       D1       5       4       S       Double         267       719       D2       5       4       S       Double         268       719       D3       1       2	255	719	A16	4	3	4	Double
257       719       B1       4       5       4       Double         258       719       B2       4       5       4       Double         259       719       B3       2       3       2       Double         260       719       B4       2       3       2       Double         261       719       B4       2       3       4       Double         262       719       C1       4       3       4       Double         262       719       C2       2       3       3       Double         263       719       C3       5       3       5       Double         264       719       C4       4       4       Double       265         264       719       C5       5       4       5       Double         266       719       C6       1       2       1       Double         267       719       D1       5       4       5       Double         268       719       D3       1       2       1       Double         270       719       E       3       2	256	719	A18	4	3	4	Double
258 $719$ $B2$ $4$ $5$ $4$ Double $259$ $719$ $B3$ $2$ $3$ $2$ Double $260$ $719$ $B4$ $2$ $3$ $2$ Double $261$ $719$ $C1$ $4$ $3$ $4$ Double $262$ $719$ $C2$ $2$ $3$ $3$ Double $263$ $719$ $C2$ $2$ $3$ $5$ Double $263$ $719$ $C3$ $5$ $3$ $5$ Double $264$ $719$ $C4$ $4$ $4$ $4$ Double $265$ $719$ $C5$ $5$ $4$ $5$ Double $266$ $719$ $C6$ $1$ $2$ $1$ Double $266$ $719$ $D1$ $5$ $4$ $5$ Double $266$ $719$ $D1$ $5$ $4$ $5$ Double $268$ $719$ $D2$ $5$ $4$ $5$ Double $269$ $719$ $D3$ $1$ $2$ $1$ Double $270$ $719$ $E$ $3$ $2$ $3$ Double $271$ $719$ $F1$ $4$ $3$ $4$ Double $272$ $719$ $F2$ $2$ $3$ $4$ Double $273$ $719$ $F3$ $4$ $3$ $4$ Double $274$ $719$ $F4$ $4$ $3$ $4$ Double $274$ $719$ $F4$ $4$ $3$ $4$ Double	257	719 .	B1	4	5	4	Double
259       719       B3       2       3       2       Double         260       719       B4       2       3       2       Double         261       719       C1       4       3       4       Double         262       719       C2       2       3       3       Double         263       719       C2       2       3       3       Double         264       719       C3       5       3       5       Double         265       719       C5       5       4       4       Double         265       719       C6       1       2       1       Double         266       719       D1       5       4       5       Double         267       719       D1       5       4       5       Double         268       719       D2       5       4       5       Double         269       719       D3       1       2       1       Double         270       719       F1       4       3       4       Double         271       719       F2       2       3       2	258	719	B2	4	5	4	Double
260 $719$ $B4$ $2$ $3$ $2$ $Double$ $261$ $719$ $C1$ $4$ $3$ $4$ $Double$ $262$ $719$ $C2$ $2$ $3$ $3$ $Double$ $263$ $719$ $C2$ $2$ $3$ $3$ $Double$ $263$ $719$ $C3$ $5$ $3$ $5$ $Double$ $264$ $719$ $C4$ $4$ $4$ $4$ $Double$ $265$ $719$ $C5$ $5$ $4$ $5$ $Double$ $266$ $719$ $C6$ $1$ $2$ $1$ $Double$ $267$ $719$ $D1$ $5$ $4$ $5$ $Double$ $268$ $719$ $D2$ $5$ $4$ $5$ $Double$ $269$ $719$ $D3$ $1$ $2$ $1$ $Double$ $270$ $719$ $E$ $3$ $2$ $3$ $Double$ $271$ $719$ $F1$ $4$ $3$ $4$ $Double$ $272$ $719$ $F2$ $2$ $3$ $2$ $Double$ $273$ $719$ $F3$ $4$ $3$ $4$ $Double$ $274$ $719$ $F4$ $4$ $3$ $4$ $Double$ $275$ $719$ $F4$ $4$ $3$ $4$ $Double$	259	719	B3	2	3	2	Double
261 $719$ $C1$ $4$ $3$ $4$ Double $262$ $719$ $C2$ $2$ $3$ $3$ Double $263$ $719$ $C3$ $5$ $3$ $5$ Double $264$ $719$ $C4$ $4$ $4$ $4$ Double $265$ $719$ $C5$ $5$ $4$ $5$ Double $266$ $719$ $C6$ $1$ $2$ $1$ Double $266$ $719$ $C6$ $1$ $2$ $1$ Double $267$ $719$ $D1$ $5$ $4$ $5$ Double $268$ $719$ $D2$ $5$ $4$ $5$ Double $269$ $719$ $D3$ $1$ $2$ $1$ Double $270$ $719$ $E$ $3$ $2$ $3$ Double $271$ $719$ $F1$ $4$ $3$ $4$ Double $272$ $719$ $F2$ $2$ $3$ $2$ Double $273$ $719$ $F3$ $4$ $3$ $4$ Double $274$ $719$ $F4$ $4$ $3$ $4$ Double $275$ $719$ $F5$ $1$ $2$ $1$ Double	260	719	B4	2	3	2	Double
262 $719$ $C2$ $2$ $3$ $3$ Double $263$ $719$ $C3$ $5$ $3$ $5$ Double $264$ $719$ $C4$ $4$ $4$ $4$ $4$ Double $265$ $719$ $C5$ $5$ $4$ $5$ Double $266$ $719$ $C6$ $1$ $2$ $1$ Double $266$ $719$ $C6$ $1$ $2$ $1$ Double $267$ $719$ $D1$ $5$ $4$ $5$ Double $268$ $719$ $D2$ $5$ $4$ $5$ Double $269$ $719$ $D3$ $1$ $2$ $1$ Double $270$ $719$ $E$ $3$ $2$ $3$ Double $271$ $719$ $F1$ $4$ $3$ $4$ Double $272$ $719$ $F2$ $2$ $3$ $4$ Double $273$ $719$ $F3$ $4$ $3$ $4$ Double $274$ $719$ $F4$ $4$ $3$ $4$ Double $275$ $719$ $F5$ $1$ $2$ $1$ Double	261	719	C1	4	3	4	Double
263 $719$ $C3$ $5$ $3$ $5$ Double $264$ $719$ $C4$ $4$ $4$ $4$ $4$ Double $265$ $719$ $C5$ $5$ $4$ $5$ Double $266$ $719$ $C6$ $1$ $2$ $1$ Double $267$ $719$ $D1$ $5$ $4$ $5$ Double $268$ $719$ $D2$ $5$ $4$ $5$ Double $269$ $719$ $D2$ $5$ $4$ $5$ Double $269$ $719$ $D3$ $1$ $2$ $1$ Double $270$ $719$ $E$ $3$ $2$ $3$ Double $271$ $719$ $F1$ $4$ $3$ $4$ Double $272$ $719$ $F2$ $2$ $3$ $2$ Double $273$ $719$ $F3$ $4$ $3$ $4$ Double $274$ $719$ $F4$ $4$ $3$ $4$ Double $275$ $719$ $F5$ $1$ $2$ $1$ Double	262	719	C2	2	3	3	Double
264 $719$ $C4$ $4$ $4$ $4$ $4$ $Double$ $265$ $719$ $C5$ $5$ $4$ $5$ $Double$ $266$ $719$ $C6$ $1$ $2$ $1$ $Double$ $267$ $719$ $D1$ $5$ $4$ $5$ $Double$ $268$ $719$ $D2$ $5$ $4$ $5$ $Double$ $268$ $719$ $D2$ $5$ $4$ $5$ $Double$ $269$ $719$ $D3$ $1$ $2$ $1$ $Double$ $270$ $719$ $E$ $3$ $2$ $3$ $Double$ $271$ $719$ $F1$ $4$ $3$ $4$ $Double$ $272$ $719$ $F2$ $2$ $3$ $2$ $Double$ $273$ $719$ $F3$ $4$ $3$ $4$ $Double$ $274$ $719$ $F4$ $4$ $3$ $4$ $Double$ $275$ $719$ $F5$ $1$ $2$ $1$ $Double$	263	719	C3	5	3	5	Double
265 $719$ $C5$ $5$ $4$ $5$ Double $266$ $719$ $C6$ $1$ $2$ $1$ Double $267$ $719$ $D1$ $5$ $4$ $5$ Double $268$ $719$ $D2$ $5$ $4$ $5$ Double $268$ $719$ $D2$ $5$ $4$ $5$ Double $269$ $719$ $D3$ $1$ $2$ $1$ Double $270$ $719$ $E$ $3$ $2$ $3$ Double $271$ $719$ $F1$ $4$ $3$ $4$ Double $272$ $719$ $F2$ $2$ $3$ $2$ Double $273$ $719$ $F3$ $4$ $3$ $4$ Double $274$ $719$ $F4$ $4$ $3$ $4$ Double $275$ $719$ $F5$ $1$ $2$ $1$ Double	264	719	C4	4	4	4	Double
266 $719$ $C6$ $1$ $2$ $1$ Double $267$ $719$ $D1$ $5$ $4$ $5$ $Double$ $268$ $719$ $D2$ $5$ $4$ $5$ $Double$ $269$ $719$ $D3$ $1$ $2$ $1$ $Double$ $270$ $719$ $E$ $3$ $2$ $3$ $Double$ $271$ $719$ $F1$ $4$ $3$ $4$ $Double$ $272$ $719$ $F2$ $2$ $3$ $2$ $Double$ $273$ $719$ $F3$ $4$ $3$ $4$ $Double$ $274$ $719$ $F4$ $4$ $3$ $4$ $Double$ $275$ $719$ $F5$ $1$ $2$ $1$ $Double$	265	719	C5	5	4	5	Double
267 $719$ D1 $5$ $4$ $5$ Double $268$ $719$ D2 $5$ $4$ $5$ Double $269$ $719$ D3 $1$ $2$ $1$ Double $270$ $719$ E $3$ $2$ $3$ Double $270$ $719$ F1 $4$ $3$ $4$ Double $271$ $719$ F1 $4$ $3$ $4$ Double $272$ $719$ F2 $2$ $3$ $2$ Double $273$ $719$ F3 $4$ $3$ $4$ Double $274$ $719$ F4 $4$ $3$ $4$ Double $275$ $719$ F5 $1$ $2$ $1$ Double	266	719	C6	1	2	1	Double
268 $719$ $D2$ $5$ $4$ $5$ $Double$ $269$ $719$ $D3$ $1$ $2$ $1$ $Double$ $270$ $719$ $E$ $3$ $2$ $3$ $Double$ $271$ $719$ $F1$ $4$ $3$ $4$ $Double$ $272$ $719$ $F2$ $2$ $3$ $2$ $Double$ $273$ $719$ $F3$ $4$ $3$ $4$ $Double$ $274$ $719$ $F4$ $4$ $3$ $4$ $Double$ $275$ $719$ $F5$ $1$ $2$ $1$ $Double$	267	719	D1	5	4	5	Double
269 $719$ D3121Double $270$ $719$ E323Double $271$ $719$ F1434Double $272$ $719$ F2232Double $273$ $719$ F3434Double $274$ $719$ F4434Double $275$ $719$ F5121Double	268	719	D2	5	4	5	Double
270 $719$ E323Double $271$ $719$ F1434Double $272$ $719$ F2232Double $273$ $719$ F3434Double $274$ $719$ F4434Double $275$ $719$ F5121Double	269	719	D3	1 .	2	1	Double
271       719       F1       4       3       4       Double         272       719       F2       2       3       2       Double         273       719       F3       4       3       4       Double         274       719       F4       4       3       4       Double         275       719       F5       1       2       1       Double	270	719	Е	3	2	3	Double
272       719       F2       2       3       2       Double         273       719       F3       4       3       4       Double         274       719       F4       4       3       4       Double         275       719       F5       1       2       1       Double	271	719	F1	4	3	4	Double
273       719       F3       4       3       4       Double         274       719       F4       4       3       4       Double         275       719       F5       1       2       1       Double	272	719	F2	2	3	2	Double
274         719         F4         4         3         4         Double           275         719         F5         1         2         1         Double	273	719	F3	4	3	4	Double
275 719 F5 1 2 1 Double	274	719	F4	4	3	4	Double
	275	719	F5	1	2	1	Double
276 719 F6 2 1 2 Double	276	719	F6	2	1	2	Double

211	/19	F/	2	1	2	Double
278	719	F8	4	2	4	Double
279	719	F9	2	1	2	Double
280	719	F10	4	3	4	Double
281	719	F11	2	4	2	Double
282	719	A17	5	4	5	Double
283	732	A11	1	4	4	First
284	732	A12	5	1	1	First
285	732	A13	1	5	5	First
286	732	A14	5	1	1	First
287	732	A15	2	5	5	First
288	732	A16	5	2	2	First
289	732	A17	4	5	5	First
290	732	A18	5	4	4	First
291	736	B2	4	2	2	First
292	743	A3	3	2	3.	Double
293	743	A4	3	2	3	Double
294	743	B1	3	4	3	Double
295	743	B2	3	4	3	Double
296	743	H5	1	2	1	Double
297	754	A15	5	4	5	Double
298	754	A16	1	5	1	Double
299	756	B4	2	1	1	First
300	763	B3	2	3	3	First
301	766	C4	4	3	4	Double
302	766	C5	3	4	4	First
303	769	A5	2	3	3	First
304	770	A2	3	2	2	First
305	782	C6	2	5	2	Double
306	792	A14	3	2	2	First
307	800	C1	4	2	4	Double
308	800	C2	2	2	4	Double
309	803	F4	4	3	3	First
310	818	C1	4	5	5	First
311	818	C2	4	5	5	First

312	818	C3	4	5	5	First
313	826	A6	4	5	5	First
314	829	C2	4	5	5	First
315	832	A10	4	2	2	First
316	836	F9	3	2	2	First
317	836	H1	9	8	8	First
318	847	H1	11	8	8	First
319	854	F9	2	3	3	First
320	860	А	5	7	5	Double
Total	118	320	320	320	320	de anni tantinin.



The University Of Sheffield. School of Health and Related

Research

#### **Appendix 21 Cover letter**

Dear respondents,

I am a PhD student at the University of Sheffield in the U.K and I am currently in the process of collecting data for my study entitled "An evaluation of patient safety culture tool in Saudi Arabia". This study aims to identify a suitable measure for assessing the patient safety culture for use in hospitals in Saudi Arabia. The knowledge generated will contribute to the measurement of the patient safety culture in hospitals in Saudi Arabia, and ultimately the creation of a safer environment for patients in hospital. As a healthcare professional, your assistance is very important and useful to this study to get at least 300 completed questionnaires. Your Participation is voluntary in the study.

I would be grateful if you would volunteer to take a few minutes of your valuable time to fill in this questionnaire. Your privacy and confidentiality will be fully respected as this questionnaire was designed in such a way as to be completely anonymous. If you have any question or queries or concern about the questionnaire you are welcome to contact me through my contact numbers or be sending an email to the address below.

Please post the completed questionnaire to secretary of your department. Alternatively, the questionnaire can be returned to head nurse in your clinical area. I am very grateful for your participation in this study and your commitment and dedication to the healthcare services in Saudi Arabia.

Thank you very much for your patience and cooperation,

Yours sincerely,

Mamdooh Sherier Alonazi The University of Sheffield Faculty of Medicine School of Health and Related Research Phone: [Saudi Arabia: 00966505386492]-[U.K: 00447735627677] Email:M.Alonazi@sheffield.ac.uk Appendix 22 Hospital Survey on Patient Safety Culture with changes

# HOSPITAL SURVEY ON PATIENT SAFETY CULTURE

#### INSTRUCTIONS

This survey asks for your opinions about patient safety issues, medical error, and event reporting in your hospital and will take about 10 to 15 minutes to complete.

- An "event" is defined as any type of error, mistake, incident, accident, or deviation, regardless of whether or not it results in patient harm.
- "Patient safety" is defined as the avoidance and prevention of patient injuries or adverse events resulting from the processes of health care delivery.

#### SECTION A: Your Work Area/Unit

In this survey, think of your "unit" as the work area, department, or clinical area of the hospital where you spend <u>most of your work time or provide most of your clinical</u> <u>services</u>

What is your primary work area or unit in this hospital? Mark ONE answer by filling in the brackets  $\sqrt{1}$ 



Please indicate your agreement or disagreement with the following statements about your work area/unit. Mark your answer by filling in the brackets.

Think about your hospital work area/unit	Strongly Disagree	Disagree	Neither	Agree	Strongly Agree
1. Staff support one another in this unit	[1]	[2]	[3]	[4]	[5]
2. We have enough staff to handle the workload	[1]	[2]	[3]	[4]	[5]
3. When a lot of work needs to be done quickly, we work together as a team to get the work done	[1]	[2]	[3]	[4]	[5]
4. In this unit, staff treat each other with respect	[1]	[2]	[3]	[4]	[5]
5. Staff in this unit work longer hours than they should, which is not good for patient care	[1]	[2]	[3]	[4]	[5]
<ol> <li>We are actively doing things to improve patient safety</li> </ol>	[1]	[2]	[3]	[4]	[5]
7. We use more temporary staff than we should, which is not good for patient care	[1]	[2]	[3]	[4]	[5]
8. Staff feel like their mistakes are held against them	[1]	[2]	[3]	[4]	[5]
9. Mistakes have led to positive changes here	[1]	[2]	[3]	[4]	[5]
10.It is just by chance that more serious mistakes do not happen around here	[1]	[2]	[3]	[4]	[5]
11.When one area in this unit gets really busy, others help out	[1]	[2]	[3]	[4]	[5]
12. When an event is reported, it feels like the person is being written up, not the problem	[1]	[2]	[3]	[4]	[5]

# SECTION A: Your Work Area/Unit (continued)

Think about your hospital work area/unit	Strongly Disagree	Disagree	Neither	Agree	Strongly Agree
13.After we make changes to improve patient safety, we evaluate their effectiveness	[1]	[2]	[3]	[4]	[5]
14.We work in a hurry, trying to do too much, too quickly	[1]	[2]	[3]	[4]	[5]
15.Patient safety never takes second place to get more work done	[1]	[2]	[3]	[4]	[5]
16.Staff worry that mistakes they make are kept in their personnel file	[1]	[2]	[3]	[4]	[5]
17.We have patient safety problems in this unit	[1]	[2]	[3]	[4]	[5]
18.Our procedures and systems are good at preventing errors from happening	[1]	[2]	[3]	[4]	[5]

## SECTION B: Your Supervisor/Manager

Please indicate your agreement or disagreement with the following statements about your immediate supervisor/manager or person to whom you directly report. Mark your answer by filling in the brackets.

	Strongly Disagree	Disagree	Neither	Agree	Strongly Agree
1. My supervisor/manager says a good word when he/she sees a job done according to established patient safety procedures	[1]	[2]	[3]	[4]	[5]
2. My supervisor/manager seriously considers staff suggestions for improving patient safety	[1]	[2]	[3]	[4]	[5]
3. Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it means taking shortcuts	[1]	[2]	[3]	[4]	[5]
4. My supervisor/manager ignores patient safety problems that happen over and over	[1]	[2]	[3]	[4]	[5]

#### **SECTION C: Communications**

How often do the following things happen in your work area/unit? Mark your answer by filling in the brackets.

TI	nink about your hospital work area/unit	Never	Rarely	Some times	Most of the time	Always
1.	We are given feedback about changes put into place based on event reports	[1]	[2]	[3]	[4]	[5]
2.	Staff will freely speak up if they see something that may negatively affect patient care	[1]	[2]	[3]	[4]	[5]
3.	We are informed about errors that happen in this unit	[1]	[2]	[3]	[4]	[5]
4.	Staff feel free to question the decisions or actions of those with more authority	[1]	[2]	[3]	[4]	[5]
5.	In this unit, we discuss ways to prevent errors from happening again	[1]	[2]	[3]	[4]	[5]
6.	Staff are afraid to ask questions when something does not seem right	[1]	[2]	[3]	[4]	[5]

# SECTION D: Frequency of Events Reported

In your hospital work area/unit, when the following mistakes happen, how often are they reported? Mark your answer by filling in the brackets.

		Never	Rarely	Some times	Most of the time	Always
1	When a mistake is made, <u>but is caught</u> <u>and corrected before affecting the patient</u> , how often is this reported	[1]	[2]	[3]	[4]	[5]
2	When a mistake is made, but has <u>no</u> <u>potential to harm the patient</u> , how often is this reported	[1]	[2]	[3]	[4]	[5]
3	When a mistake is made that <u>could harm</u> <u>the patient</u> , but does not, how often is this reported?	[1]	[2]	[3]	[4]	[5]

# SECTION E: Patient Safety Grade

Please give your work area/unit in this hospital an overall grade on patient safety. Mark ONE answer.

[]		[]	[]	[]
Α	В	С	D	E
Excellent	Very Good	Acceptable	Poor	Failing
SECTION F	: Your Hospital			A State Laboration

Please indicate your agreement or disagreement with the following statements about your hospital. Mark your answer by filling in the brackets.

Think	about your hospital	Strongly Disagree	Disagree	Neither	Agree	Strongly Agree
1	Hospital management provides a work climate that promotes patient safety	[1]	[2]	[3]	[4]	[5]
2	Hospital units do not coordinate well with each other	[1]	[2]	[3]	[4]	[5]
3	Some things do not happen or get missed when transferring patients from one unit to another	[1]	[2]	[3]	[4]	[5]
4	There is good cooperation among hospital units that need to work together	[1]	[2]	[3]	[4]	[5]
5	Important patient care information is often lost during shift changes	[1]	[2]	[3]	[4]	[5]
6	It is often unpleasant to work with staff from other hospital units	[1]	[2]	[3]	[4]	[5]
7	Problems often occur in the exchange of information across hospital units	[1]	[2]	[3]	[4]	[5]
8	The actions of hospital management show that patient safety is a top priority	[1]	[2]	[3]	[4]	[5]
9	Hospital management seems interested in patient safety only after an adverse event happens	[1]	[2]	[3]	[4]	[5]
10	Hospital units work well together to provide the best care for patients	[1]	[2]	[3]	[4]	[5]
11	Shift changes cause problems for patients in this hospital	[1]	[2]	[3]	[4]	[5]

# SECTION G: Number of Events Reported

In the past 12 months, how many event reports have you filled out and submitted? Mark ONE answer.

a	No event reports [ ]
b	1 to 2 event reports [ ]
с	3 to 5 event reports []

d	6 to 10 event reports [ ]
e	11 to 20 event reports [ ]
f	21 event reports or more [ ]

### **SECTION H: Background Information**

This information will help in the analysis of the survey results. Mark ONE answer by filling in the brackets.

1. What is your current position in this hospital? Mark ONE answer that best describes your current position.

a	Physician in training [ ]	ernies	g	Head nurse/Nurse manager [ ]
b	Physician (general physician) [ ]		h	Nurse supervisor [ ]
c	Resident physician [ ]		i	Registered nurse [ ]
d	Assistant consultant [ ]		j	Clinical resource nurse [ ]
e	Associate consultant [ ]	ynurdi	k	Staff nurse [ ]
f	Consultant [ ]	tione	1	Other [ ]
		and the		Please specify
	2. What is your gender?	-larae-th		
a	Male [ ]	No. 19	b	Female [ ]
	3. What is your nationality?			
a	Saudi [ ]	Solotos	b	Non-Saudi [ ]
-				Please specify
	4. What is your highest educational qualification	?		
а	PhD Doctorate / Board Certified/Fellowship [ ]	(Cilitar	с	Bachelor degree [ ]
b	Master degree [ ]		d	Diploma [ ]
		mikery	e	Other [ ]
		ist- m	12	Please specify

<ul> <li>5. <u>Do you have any training outside Saudi Arabia?</u></li> <li>a Yes [] If yes, please specify</li> </ul>	b No [ ]
<ul> <li>6. How long have you worked in this hospital ?</li> <li>a Less than 1 years []</li> <li>b 1 to 5 years []</li> <li>c 6 to 10 years []</li> </ul>	d       11 to 15 years []         e       16 to 20 years []         f       21 years or more []
<ul> <li>7. How long have you worked in your current hospit.</li> <li>a Less than 1 years []</li> <li>b 1 to 5 years []</li> <li>c 6 to 10 years []</li> </ul>	al work area/unit? d 11 to 15 years [] e 16 to 20 years [] f 21 years or more []
<ul> <li>8. Typically, how many hours per week do you work</li> <li>a Less than 20 hours per week []</li> <li>b 20 to 39 hours per week []</li> <li>c 40 to 59 hours per week []</li> </ul>	<pre>in this hospital ?     d 60 to 79 hours per week [ ]     e 80 to 99 hours per week [ ]     f 100 hours per week or more [ ]</pre>
<ul> <li>9. In your current position, do you typically have direct</li> <li>a YES, I typically have direct interaction or contact with</li> <li>b NO, I typically do NOT have direct interaction or contact</li> </ul>	ect interaction or contact with patients? h patients [ ] tact with patients [ ]
<ul> <li>10. How long have you worked in your current special</li> <li>a Less than 1 year []</li> <li>b 1 to 5 years []</li> <li>c 6 to 10 years []</li> </ul>	lty or profession ? d 11 to 15 years [] e 16 to 20 years [] f 21 years or more []

### **SECTION I: Your Comments**

Please feel free to write any comments about patient safety, error, or event reporting in your hospital.



Appendix 23 Comparison between an overall grade of patient safety and number of event reports

Item		G: Number of eve	ent report in the pas	t 12 months	
A A DECEMBER OF A A A A A A A A A A A A A A A A A A		No event reports and 1 to 2 event reports	3 to 5event reports and 6 to 10 event reports	11 to 20 event reports and 21 event reports or more	Total
E: An overall grade of patient safety	Very good and Excellent	433	88	15	536
	Acceptable	242	58	8	308
DA AL	Poor and Failing	12	5	1 a a a a a a a a a a a a a a a a a a a	18
	Total	687	151	24	862

The table above presents the comparison between an overall grade of patient safety and number of event reports in the past 12 months. It shows that only one respondent reported poor and failing whom reported lots of event while 12 respondents also reported poor and failing they reported no event reports and 1 to 2 event reports. This means those reporting poor/failing were not the same individuals who reported lots of events.

## Appendix 24 Communalities of the initial solution (11 factors)

Safety climate items	Initial	Extractio
A1 Staff support one another in this unit	.51	.64
A2 We have enough staff to handle the workload	.24	.24
A3 When a lot of work needs to be done quikly, we work together as a team to get the work	.48	.56
A4 In this unit, people treat each other with respect	.48	.62
A5 Staff in this unit work longer hours than they should, which is not good for patient care	.24	.31
A6 We are actively doing things to improve patient care	.40	.50
A7 We use more temporary staff than we should, which is not good for patient care	.23	.31
A8 Staff feel like their mistakes are held against them	.45	.64
A9 Mistakes have led to positive changes here	.33	.38
A11 When one area in this unit gets really busy, others help out	.36	.40
A12 When an event is reported, it feels like the person is being written up, not the problem	.43	.55
A13 After we make changes to improve patient safety, we evaluate their effectiveness	.46	.55
A14 We work in a hurry, trying to do too much, too quickly	.35	.38
A16 Staff worry that mistakes they make are kept in their personnel file	.30	.35
A18 Our procedures and systems are good at preventing errors from happening	.41	.42
B1 My supervisor/manager says a good word when he/she sees a job done according to	.59	.81
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety	.62	.69
B3 Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it	.38	.46
B4 My supervisor/manager ignores patient safety problems that happen over and over	.39	.60
C1 We are given feedback about changes put into place based on event reports	.41	.45
C2 Staff will freely speak up if they see something that may negatively affect patient care	.45	.52
C3 We are informed about errors that happen in this unit	.42	.48
C4 Staff feel free to question the decisions or actions of those with more authority	.40	.57
C5 In this unit, we discuss ways to prevent errors from happening again	.54	.56
C6 Staff are afraid to ask questions when something does not seem right	.27	.33
D1 When a mistake is made, but is caught and corrected before affecting the patient, how	.53	.58
D2 When a mistake is made, but has no potential to harm the patient, how often is this	.66	.88
D3 When a mistake is made that could harm the patient, but does not, how often is this	.53	.55
F2 Hospital units do not coordinate well with each other	.44	.46
F4 There is good cooperation among hospital units that need to work together	.45	.56

Appendix 24 (continued)	niol de l	sphodely.
F5 Important patient care information is often lost during shift changes	.38	.46
F6 It is often unpleasant to work with staff from other hospital units	.33	.36
F7 Problems often occur in the exchange of information across hospital units	.42	.54
F8 The actions of hospital management show that patient safety is a top priority	.43	.44
F10 Hospital units work well together to provide the best care for patients	.52	.63
F11 Shift changes cause problems for patients in this hospital	.33	.39

Extraction Method: Principal Axis Factoring.

0

	1.1.1	Initial Eigenvalues			Extraction Sums of Squared Loadings			
Factor	Total	% of Variance	Cumulative %	Total	% of Variance	Cumulative %	Total	
1	9.168	25.4	25.4	8.7	24.1	24.1	2.9	
2	2.223	6.1	31.6	1.7	4.9	29.1	3.7	
3	2.175	6.0	37.6	1.6	4.7	33.8	2.8	
4	1.764	4.8	42.5	1.2	3.5	37.4	3.3	
5	1.598	4.4	47.0	1.1	3.1	40.6	3.7	
6	1.452	4.0	51.0	.9	2.7	43.3	2.3	
7	1.214	3.3	54.4	.7	2.1	45.4	4.3	
8	1.149	3.1	57.6	.6	1.8	47.2	3.1	
9	1.008	2.8	60.4	.5	1.5	48.8	4.8	
10	.897	2.4	62.9	.4	1.1	49.9	.6	
11	.867	2.4	65.3	.3	.9	50.8	3.3	
12	.857	2.3	67.6	100			15 2 18	
13	.748	2.0	69.7	Na (ringia na			30 15	
14	.722	2.0	71.7	ang blander i	and the country		11 42	
15	.713	1.9	73.7	a Fire your	the parts stated	and to be	10 1 11	
16	.679	1.8	75.6	deren wiele	in a transmitter	Stearages	6	
17	.663	1.8	77.4	ile a secolo	and the second second	ever if it	36 46	
18	.630	1.7	79.2	Mikos Sui	ality and state	estre 1	35 60	
19	.607	1.6	80.9	Danice	a design of the second second		1400 45	
20	.589	1.6	82.5	in the second	and she is	and care	11 1 12	
21	.571	1.5	84.1				State of the	
22	.509	1.4	85.5	a the second second	harren en der	L.	40.	
23	.492	1.3	86.9	z finentaria	- Artural		34 3 36 1	
24	.486	1.3	88.2	ela se se	lo suidit		27 1 1 13	
25	.451	1.2	89.5		Review the rate	na, have	151 J. 140	
26	.441	1.2	90.7	i-ductor ha	international	tars -	26 1 183	
27	.422	1.1	91.9	in an ana		a they	52 35	
28	.412	· 1.1	93.0				an a	
29	.379	1.0	94.1	langaist in s	ani organijevi i		14 A	

# Appendix 25 Total Variance Explained of the initial solution (11 factors)

Append	Appendix 25 (continued)								
30	.375	1.0	95.1						
31	.353	.9	96.1						
32	.331	.9	97.0			and the content of	e la feite ha ces barres		
33	.311	.8	97.9		gradere el base	anges per long phen	a text to start and		
34	.292	.8	98.7	and the second	Second participation	abilitation and	ani di mandri sikas		
35	.253	.7	99.4	iger (1074) but	national set of	let of kelten sig on a	e ha start a tom		
36	.198	.5	100	print and	test dependence turne	encing and sound (12	e sole shere is a sur		

Safety climate items	Factor											
	1	2	3	4	5	6	7	8	9	10	11	
C3 We are informed about errors that happen in this unit	.47					1	-					
C1 We are given feedback about changes put into place based on event reports	.45	10				12						
A13 After we make changes to improve patient safety, we evaluate their effectiveness	.39	18				2.5		2			3	
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?		.96			13		and a state	1				
D3 When a mistake is made that could harm the patient, but does not, how often is this reported?		.73			14	1.						
D1 When a mistake is made, but is caught and corrected before affecting the patient, how often is	iniger)	.72	eleveration (	bina Parat	and	al Sage par	-	2.				
A8 Staff feel like their mistakes are held against them			.78		15			1				
A12 When an event is reported, it feels like the person is being written up, not the problem			.66		47.			1				
A16 Staff worry that mistakes they make are kept in their personnel file			.51		48.			1.4				
A14 We work in a hurry, trying to do too much, too quickly			.33		10,							
F7 Problems often occur in the exchange of information across hospital units				70	198			1.3				
F5 Important patient care information is often lost during shift changes				55					and the second			
F6 It is often unpleasant to work with staff from other hospital units				51					1. A.			
F11 Shift changes cause problems for patients in this hospital				50								
A1 Staff support one another in this unit			5.0		.76							
A4 In this unit, people treat each other with respect					.73		1					
A3 When a lot of work needs to be done quikly, we work together as a team to get the work done					.56							
A2 We have enough staff to handle the workload									ke hurub			
A7 We use more temporary staff than we should, which is not good for patient care						.52						
A5 Staff in this unit work longer hours than they should, which is not good for patient care						.45						
B3 Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it means						.41			a china ch			
B1 My supervisor/manager says a good word when he/she sees a job done according to established							90					
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety							67					
C4 Staff feel free to question the decisions or actions of those with more authority								67				
C2 Staff will freely speak up if they see something that may negatively affect patient care		- 364					1	51				
C6 Staff are afraid to ask questions when something does not seem right								.41				
C5 In this unit, we discuss ways to prevent errors from happening again							ſ	31				
F10 Hospital units work well together to provide the best care for patients								250	.70			
F4 There is good cooperation among hospital units that need to work together						1.			.64			

# Appendix 26 Pattern Matrix of the initial solution (11 factors)

Appendix 26 (continued)	icitulos h	instal.				0.0	210 h				
F2 Hospital units do not coordinate well with each other							110	(Taile	46		
F8 The actions of hospital management show that patient safety is a top priority		11 811	103.7	fisce.	Carlo	non	2.11		.45		
B4 My supervisor/manager ignores patient safety problems that happen over and over	tilo p si	6.60	ot a		ind.	0.97	o ann	12.6		.52	
A11 When one area in this unit gets really busy, others help out	paratras	- day	1010	in an	digor	n he					
A6 We are actively doing things to improve patient care	Silver-Gio	1.015	1.00	A. stre	82 (L.B.)	a profi					56
A9 Mistakes have led to positive changes here	na george			1 2 30	the vi	WIRE.		14 a.			48
A18 Our procedures and systems are good at preventing errors from happening	should be	and	(here	(28130)		mon		14.5			30

Extraction Method: Principl Axis Factoring. Rotation Method: Oblimin with Kaiser Normalization a. Rotation converged in 21 iterations.
# Appendix 27 Communalities of initial solution (11 factors) with safety climate items (n=31)

Safety climate items	Initial	Extraction
A1 Staff support one another in this unit	.49	.65
A3 When a lot of work needs to be done quickly, we work together as a team to get the	.43	.52
A4 In this unit, people treat each other with respect	.47	.61
A5 Staff in this unit work longer hours than they should, which is not good for patient care	.19	.34
A6 We are actively doing things to improve patient care	.39	.61
A7 We use more temporary staff than we should, which is not good for patient care	.21	.28
A8 Staff feel like their mistakes are held against them	.43	.62
A9 Mistakes have led to positive changes here	.31	.35
A12 When an event is reported, it feels like the person is being written up, not the problem	.42	.55
A13 After we make changes to improve patient safety, we evaluate their effectiveness	.45	.56
A16 Staff worry that mistakes they make are kept in their personnel file	.28	.35
B1 My supervisor/manager says a good word when he/she sees a job done according to	.59	.84
B2 My supervisor/manager seriously considers staff suggestions for improving patient	.62	.68
B3 Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it	.35	.64
B4 My supervisor/manager ignores patient safety problems that happen over and over	.36	.40
C1 We are given feedback about changes put into place based on event reports	.41	.46
C2 Staff will freely speak up if they see something that may negatively affect patient care	.44	.55
C3 We are informed about errors that happen in this unit	.40	.47
C4 Staff feel free to question the decisions or actions of those with more authority	.36	.56
C6 Staff are afraid to ask questions when something does not seem right	.26	.34
D1 When a mistake is made, but is caught and corrected before affecting the patient, how	.53	.58
D2 When a mistake is made, but has no potential to harm the patient, how often is this	.65	.89
D3 When a mistake is made that could harm the patient, but does not, how often is this	.52	.55
F2 Hospital units do not coordinate well with each other	.43	.47
F4 There is good cooperation among hospital units that need to work together	.43	.52
F5 Important patient care information is often lost during shift changes	.37	.445
F6 It is often unpleasant to work with staff from other hospital units	.328	.37
F7 Problems often occur in the exchange of information across hospital units	.39	.56
F8 The actions of hospital management show that patient safety is a top priority	.39	.42
F10 Hospital units work well together to provide the best care for patients	.51	.68

Appendix	x 27 (continued)			nationadi le	Doublest Store	and into T 61 21	A pprod
F11 Shift	t changes cause pr	oblems for patie	ents in this hos	spital		.39	.37
Extractio	n Method: Princip	al Axis Factori	ng.	1 Montale	Bigravalues Variades Coun	Town Lour	notae 1
			1.0.1				
					. h.F.		
						200	

	and the second	Initial Eigenval	ues	Extractio	Extraction Sums of Squared Loadings						
Factor	Total	% of Variance	Cumulative %	Total	% of Variance	Cumulative %	Total				
1	7.897	25.4	25.4	7.4	24.0	24.0	2.1				
2	2.078	6.7	32.1	1.7	5.5	29.5	3.4				
3	2.053	6.6	38.7	1.5	5.0	34.6	3.0				
4	1.689	5.4	44.2	1.2	4.0	38.6	2.2				
5	1.441	4.6	48.8	1.0	3.3	41.9	3.3				
6	1.362	4.3	53.2	.8	2.8	44.7	2.5				
7	1.169	3.7	57.0	.7	2.3	47.1	3.2				
8	1.112	3.5	60.6	.6	2.0	49.1	1.4				
9	.983	3.1	63.8	.5	1.7	50.8	4.0				
10	.855	2.7	66.5	.3	1.1	51.9	3.1				
11	.800	2.5	69.1	.2	.9	52.9	2.5				
12	.763	2.4	71.6	and motor	dusadre dete	Addies for all	35				
13	.707	2.2	73.8	$\frac{1}{2}(x,x) = \frac{1}{2}(x,y)$	at post in column	ove	36 30				
14	.682	2.1	76.0	dan biyand na	event months		din				
15	.623	2.0	78.1	an ahig nen	en Streiten på	initi atre	-4				
16	.594	1.9	80.0	ilara -							
17	.565	1.8	81.8	and designed of	hotora miligian		124				
18	.533	1.7	83.5	rian lancar	a nga		26				
19	.519	1.6	85.2	$\log \log (d_{\rm elev})$	there is the part	all low	3.5 ··· · · · 25%				
20	.498	1.6	86.8	halisi burine	en providen	ethan .	.65				
21	.487	1.5	88.4	ndjasitase	nut, sivi atm	5.904					
22	.460	1.4	89.8	dine -			6 . 5 6				
23	.437	1.4	91.3	ing supply as	cea a gradios		10 10 50				
24	.430	1.3	92.6	ing and a	waxee		sel de				
25	.393	1.2	93.9	v fiotopicas e			3254				
26	.387	1.2	95.2	see constant	spractigads - c		.14				
27	.363	1.1	96.3	satat taristy	e es a policip		14 16				
28	.335	1.0	97.4	Station of the	participa -						

# Appendix 28 Total Variance Explained of the initial solution (11 factors) with safety climate (n=31)

Г

Appendix 28 (continued)								
29	.327	1.0	98.5			10-m mary		
30	.255	.8	99.3					
31	.204	.6	100		a kitt al ainpait in	and should found		

A-71

# Appendix 29 Pattern Matrix of the initial solution (11 factors) with safety climate items

Safety climate items (n=31)	Factor											
	1	2	3	4	5	6	7	8	9	10	11	
C3 We are informed about errors that happen in this unit	.45	i i i i i i i i i i i i i i i i i i i	1923 1923		1	The second		Sec.	12			
A13 After we make changes to improve patient safety, we evaluate their effectiveness	.44	100	24 mb	a la cher	ENERGY I		a farmining a	120 14955				
C1 We are given feedback about changes put into place based on event reports	.41					1.1			2.4			
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?		.97				1.2			3.5			
D1 When a mistake is made, but is caught and corrected before affecting the patient, how often is		.73										
D3 When a mistake is made that could harm the patient, but does not, how often is this reported?		.72				100			3.4			
F7 Problems often occur in the exchange of information across hospital units			.73						25			
F6 It is often unpleasant to work with staff from other hospital units			.54			1.1			3.2			
F5 Important patient care information is often lost during shift changes			.52			0.2			3.0			
F11 Shift changes cause problems for patients in this hospital	58		.46			(1.8) (1.8)						
A8 Staff feel like their mistakes are held against them				.74		·			1.1			
A12 When an event is reported, it feels like the person is being written up, not the problem	15			.66		1			5.0			
A16 Staff worry that mistakes they make are kept in their personnel file		14.0		.50								
A1 Staff support one another in this unit			1		.77							
A4 In this unit, people treat each other with respect					.72				~			
A3 When a lot of work needs to be done quikly, we work together as a team to get the work done					.57					1		
C4 Staff feel free to question the decisions or actions of those with more authority						.65						
C2 Staff will freely speak up if they see something that may negatively affect patient care			- Land			.51						
C6 Staff are afraid to ask questions when something does not seem right						41						
B1 My supervisor/manager says a good word when he/she sees a job done according to							91					
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety							64					
A5 Staff in this unit work longer hours than they should, which is not good for patient care								.55				
A7 We use more temporary staff than we should, which is not good for patient care								.37				
F10 Hospital units work well together to provide the best care for patients		1							.74			
F4 There is good cooperation among hospital units that need to work together	1								.59			
F8 The actions of hospital management show that patient safety is a top priority									.44			
F2 Hospital units do not coordinate well with each other									41			
A6 We are actively doing things to improve patient care		1	1							70		
A9 Mistakes have led to positive changes here							No.			43		

B3 When	ever pressure buil	ds up, my su	pervisor/manager wants us to work faster, even if it	.73
B4 My su	pervisor/manager	ignores pat	ient safety problems that happen over and over	.39
	Extraction Rotation I Rotation o	Method: Method: Converged	Principl Axis Factoring. Oblimin with Kaiser Normalization in 21 iterations	
			1. [D1 When a metales is made, but is usual and conscied before allocang the	

### Appendix 30 Communalities of the 12 factors solution

Communalities									
	Initial	Extraction							
A1 Staff support one another in this unit	.52	.59							
A2 We have enough staff to handle the workload	.25	.29							
A3 When a lot of work needs to be done quickly, we work together as a team to get	.50	.58							
A4 In this unit, people treat each other with respect	.50	.68							
A5 Staff in this unit work longer hours than they should, which is not good for	.25	.28							
A6 We are actively doing things to improve patient care	.41	.47							
A7 We use more temporary staff than we should, which is not good for patient care	.25	.36							
A8 Staff feel like their mistakes are held against them	.46	.66							
A9 Mistakes have led to positive changes here	.35	.39							
A10 It is just by chance that more serious mistakes do not happen around here	.13	.17							
A11 When one area in this unit gets really busy, others help out	.38	.47							
A12 When an event is reported, it feels like the person is being written up, not the	.44	.53							
A13 After we make changes to improve patient safety, we evaluate their	.47	.53							
A14 We work in a hurry, trying to do too much, too quickly	.36	.37							
A15 Patient safety never takes second place to get more work done	.20	.25							
A16 Staff worry that mistakes they make are kept in their personnel file	.31	.36							
A17 We have patient safety problems in this unit	.18	.19							
A18 Our procedures and systems are good at preventing errors from happening	.42	.43							
B1 My supervisor/manager says a good word when he/she sees a job done	.60	.76							
B2 My supervisor/manager seriously considers staff suggestions for improving	.63	.72							
B3 Whenever pressure builds up, my supervisor/manager wants us to work faster,	.38	.48							
B4 My supervisor/manager ignores patient safety problems that happen over and	.42	.58							
C1 We are given feedback about changes put into place based on event reports	.42	.44							
C2 Staff will freely speak up if they see something that may negatively affect	.46	.54							
C3 We are informed about errors that happen in this unit	.44	.49							
C4 Staff feel free to question the decisions or actions of those with more authority	.41	.56							
C5 In this unit, we discuss ways to prevent errors from happening again	.56	.57							
C6 Staff are afraid to ask questions when something does not seem right	.28	.35							
D1 When a mistake is made, but is caught and corrected before affecting the	.54	.59							
D2 When a mistake is made, but has no potential to harm the patient, how often is	.66	.88							
D3 When a mistake is made that could harm the patient, but does not, how often is	.54	.56							
F1 Hospital management provides a work climate that promotes patient safety	.41	.41							
F2 Hospital units do not coordinate well with each other	.45	.47							
F3 Some things do not happen or get missed when transferring patients from one	.15	.18							

Appendix 30 Continued	Server 1	o Mining and
F4 There is good cooperation among hospital units that need to work together	.47	.56
F5 Important patient care information is often lost during shift changes	.39	.46
F6 It is often unpleasant to work with staff from other hospital units	.37	.39
F7 Problems often occur in the exchange of information across hospital units	.42	.48
F8 The actions of hospital management show that patient safety is a top priority	.45	.49
F9 Hospital managemnet seems interested in patient safety only after an adverse	.36	.40
F10 Hospital units work well together to provide the best care for patients	.53	.63
F11 Shift changes cause problems for patients in this hospital	.34	.42

<b>F</b> t-		Initial Eig	envalues	Extr	action Sun Load	ns of Squared ings	Rotation Sums of Squared Loadings					
Factor	Total	% of Variance	Cumulative %	Total	% of Variance	Cumulative %	Total	% of Variance	Cumulative %			
1	9.930	23.6	23.6	9.434	22.5	22.5	2.387	5.6	5.6			
2	2.413	5.7	29.3	1.867	4.4	26.9	2.205	5.2	10.9			
3	2.227	5.3	34.6	1.827	4.3	31.3	2.166	5.1	16.0			
4	1.869	4.4	39.1	1.380	3.2	34.5	2.152	5.1	21.2			
5	1.626	3.8	43.0	1.180	2.8	37.4	1.943	4.6	25.8			
6	1.516	3.6	46.6	1.033	2.4	39.8	1.928	4.5	30.4			
7	1.288	3.0	49.6	.809	1.9	41.7	1.786	4.2	34.6			
8	1.188	2.8	52.5	.669	1.5	43.3	1.649	3.9	38.6			
9	1.095	2.6	55.1	.629	1.4	44.8	1.519	3.6	42.2			
10	1.086	2.5	57.7	.482	1.1	46.0	1.311	3.1	45.3			
11	1.057	2.5	60.2	.437	1.0	47.0	.623	1.4	46.8			
12	.961	2.2	62.5	.399	.9	48.0	.501	1.1	48.0			
13	.913	2.1	64.6	in line	size in the	a competence fil						
14	.847	2.0	66.6	a See				·				
15	.786	1.8	68.5	1	www.engle.	service it and the	11111		Star Inc.			
16	.761	1.8	70.3	Same	nesus are	a share when the	1.12					
17	.737	1.7	72.1	(mid)	S. F. all Sans	and the second second	minute					
18	.731	1.7	73.8	diama a	anthone that	r takihi na toʻy	Sec. 21		3 15 AR			
19	.698	1.6	75.5	ioni a	lerviorabi	nà tha taom	and and	and in	1 Carlos			
20	.661	1.5	77.1	les para	tran (	nakes on ovalit	in the second					
21	.634	1.5	78.6	ROEM	tine fine	env iterativehi						
22	.619	1.4	80.0	america	in this ten				1			
23	.612	1.4	81.5	day e	allento	Some har has	Frid	and the	1			
24	.598	1.4	82.9		ord finites	a second second						
25	.579	1.3	84.3	- and the second	At an a day							
26	.561	1.3	85.6		Summer la	. Same a Carrie	athe					
27	.519	1.2	86.9		T NI DO E DO	the article k	NS OR	the last				
28	.491	1.1	88.0		or had not	ten does abit	17 F 201					
29	.472	1.1	89.2		Constants	annoise antini						
30	.460	1.0	90.3									
31	.436	1.0	013						ALC: NO			

# Appendix 31 Total variance explained of the 12 factors solution

Appe	endix 31 (co	ontinued)						1		
32	.431	1.0	92.3	20	Bularia	and \$1.5	10.24	taked before	a pojesti de 22 k	
33	.412	.8	93.3	E						
34	.396	.4	94.2							
35	.370	.8	95.1							
36	.351	.8	96.0				r nie (ook	oppoint longs	and all the times of	To amplifier?
37	.342	.8	96.8	a second			1	in less patrices	Antipolat som tek	of sealoons
38	.325	.7	97.6	in an			and the second	and anything	a main manage	gradu tatili
39	.284	.6	98.2	Se a pice in	in low		-	ise local Turn	ner Henry of micesel	ini, centra se M
40	.278	.6	98.9	1 Cash				il.	and property inclusion	
41	.251	.5	99.5							
42	.194	.4	100							

Safety climate items (n=42)		Factor											
	1	2	3	4	5	6	7	8	9	10	11	12	
F7 Problems often occur in the exchange of information across hospital units	.64	AP.	1			2							
F5 Important patient care information is often lost during shift changes	.59	38		\$24	8.	6	103		5.5				
F11 Shift changes cause problems for patients in this hospital	.54			25	t si	2	125		0.01				
F6 It is often unpleasant to work with staff from other hospital units	.54	200	12.5		<i>à</i> .		184		103				
F9 Hospital management seems interested in patient safety only after an adverse event happens	.34	10			ð. ***		34	-	0				
A17 We have patient safety problems in this unit		他		1.51		1.1	12		24				
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?	and the	.90		the second	la arcoarciae interaction	100	JAR .	underst wittens	Contraction of				
D3 When a mistake is made that could harm the patient, but does not, how often is this reported?		.70		191		X		an the second	3.0				
D1 When a mistake is made, but is caught and corrected before affecting the patient, how often is		.69	41,4	19		E.			2.3				
C3 We are informed about errors that happen in this unit		15	.57			Arts.			6.3				
C1 We are given feedback about changes put into place based on event reports		.63	.54	13		4			16.3				
C5 In this unit, we discuss ways to prevent errors from happening again		18.6	.48			and internet			15.19				
A13 After we make changes to improve patient safety, we evaluate their effectiveness			.48			-	.41						
A4 In this unit, people treat each other with respect				.72									
A1 Staff support one another in this unit			100	.76						1		5.73	
A3 When a lot of work needs to be done quikly, we work together as a team to get the work done				.61									
F10 Hospital units work well together to provide the best care for patients					.65			18					
F4 There is good cooperation among hospital units that need to work together					.58								
F8 The actions of hospital management show that patient safety is a top priority					.45		.36				11 P. D		
F2 Hospital units do not coordinate well with each other	.39				44								
A18 Our procedures and systems are good at preventing errors from happening			4 6		.35		.33						
F1 Hospital management provides a work climate that promotes patient safety		3			.33		.32						
A8 Staff feel like their mistakes are held against them						.77							
A12 When an event is reported, it feels like the person is being written up, not the problem					륗뵑	.65							
A16 Staff worry that mistakes they make are kept in their personnel file			12			.52							
A14 We work in a hurry, trying to do too much, too quickly						.39			.37			1900 and	
A6 We are actively doing things to improve patient care							.51		2.2				
A9 Mistakes have led to positive changes here							.50						

#### Appendix 32 Rotated Matrix of the 12 factors solution

Appendix 32 (continued)	1, 61, <u>963 10 2988600</u> 0	0000.010	. cilites	996			
A15 Patient safety never takes second place to get more work done	100 hos 80 0 h	.3	0				
B1 My supervisor/manager says a good word when he/she sees a job done according to	enti katel schipto en	e sit im	.76	IA			
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety	evilond at the pla	A MALE W	.68				
B4 My supervisor/manager ignores patient safety problems that happen over and over	ta annen an	a ne als	36	E.		.34	.31
A5 Staff in this unit work longer hours than they should, which is not good for patient care	朝朝鮮和	197.00	100	.48			
A7 We use more temporary staff than we should, which is not good for patient care	ing space is the refer	all of the particular	Tinti	.48			
B3 Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it means	is the state	1201	14.93	.44			
A11 When one area in this unit gets really busy, others help out	.33	2010223-2	191	35		.31	
A2 We have enough staff to handle the workload	.31		in the	31			
C4 Staff feel free to question the decisions or actions of those with more authority	REAL PROPERTY			64	.61		
C2 Staff will freely speak up if they see something that may negatively affect patient care	.38	Port M		13/1	.48		
C6 Staff are afraid to ask questions when something does not seem right	ang fun ein me	ally a sole		14	45		
A10 It is just by chance that more serious mistakes do not happen around here	A log top at a	1.19	a 197	14		.38	
F3 Some things do not happen or get missed when transferring patients from one unit to another	a ch ages to here	a carrie	1 A.A.	1A			

Extraction Method: Principal Axis Factoring. Rotation Method: Varimax with Kaiser Normalization a. Rotation converged in 19 iterations.

A-79

# Appendix 33 Communalities of the 10 factors solution

Safety climate items (N=42)	Initial	Extraction
A1 Staff support one another in this unit	.52	.57
A2 We have enough staff to handle the workload	.25	.26
A3 When a lot of work needs to be done quickly, we work together as a team to	.50	.58
A4 In this unit, people treat each other with respect	.50	.70
A5 Staff in this unit work longer hours than they should, which is not good for	.25	.28
A6 We are actively doing things to improve patient care	.41	.42
A7 We use more temporary staff than we should, which is not good for patient	.25	.29
A8 Staff feel like their mistakes are held against them	.46	.61
A9 Mistakes have led to positive changes here	.35	.33
A10 It is just by chance that more serious mistakes do not happen around here	.13	.10
A11 When one area in this unit gets really busy, others help out	.38	.38
A12 When an event is reported, it feels like the person is being written up, not	.44	.55
A13 After we make changes to improve patient safety, we evaluate their	.47	.54
A14 We work in a hurry, trying to do too much, too quickly	.36	.38
A15 Patient safety never takes second place to get more work done	.20	.19
A16 Staff worry that mistakes they make are kept in their personnel file	.31	.36
A17 We have patient safety problems in this unit	.18	.17
A18 Our procedures and systems are good at preventing errors from happening	.42	.42
B1 My supervisor/manager says a good word when he/she sees a job done	.60	.71
B2 My supervisor/manager seriously considers staff suggestions for improving	.63	.74
B3 Whenever pressure builds up, my supervisor/manager wants us to work	.38	.44
B4 My supervisor/manager ignores patient safety problems that happen over and	.42	.48
C1 We are given feedback about changes put into place based on event reports	.42	.44
C2 Staff will freely speak up if they see something that may negatively affect	.46	.50
C3 We are informed about errors that happen in this unit	.44	.48
C4 Staff feel free to question the decisions or actions of those with more	.41	.57
C5 In this unit, we discuss ways to prevent errors from happening again	.56	.57
C6 Staff are afraid to ask questions when something does not seem right	.28	.34
D1 When a mistake is made, but is caught and corrected before affecting the	.54	.56
D2 When a mistake is made, but has no potential to harm the patient, how often	.66	.88

Appendix 33(continued)		
D3 When a mistake is made that could harm the patient, but does not, how often	.54	.55
F1 Hospital management provides a work climate that promotes patient safety	.41	.40
F2 Hospital units do not coordinate well with each other	.45	.45
F3 Some things do not happen or get missed when transferring patients from one	.15	.12
F4 There is good cooperation among hospital units that need to work together	.47	.52
F5 Important patient care information is often lost during shift changes	.39	.46
F6 It is often unpleasant to work with staff from other hospital units	.37	.39
F7 Problems often occur in the exchange of information across hospital units	.42	.49
F8 The actions of hospital management show that patient safety is a top priority	.45	.47
F9 Hospital management seems interested in patient safety only after an adverse	.36	.36
F10 Hospital units work well together to provide the best care for patients	.53	.53
F11 Shift changes cause problems for patients in this hospital	.34	.36

		Initial Eig	envalues	Extra	ction Sun	ns of Squared	Rotation Sums of Squared					
Factor	Total	% of Variance	Cumulative %	Total	% of Variance	Cumulative %	Total	% of Variance	Cumulative %			
1	9.930	23.6	23.6	9.432	22.4	22.4	3.626	8.6	8.6			
2	2.413	5.7	29.3	1.852	4.4	26.8	2.509	5.9	14.5			
3	2.227	5.3	34.6	1.816	4.3	31.1	2.183	5.1	19.7			
4	1.869	4.4	39.1	1.356	3.2	34.4	2.029	4.8	24.6			
5	1.626	3.8	43.0	1.166	2.7	37.1	1.853	4.4	29.0			
6	1.516	3.6	46.6	1.009	2.4	39.5	1.649	3.9	32.9			
7	1.288	3.0	49.6	.791	1.8	41.4	1.482	3.5	36.4			
8	1.188	2.8	52.5	.651	1.5	43.0	1.467	3.4	39.9			
9	1.095	2.6	55.1	.602	1.4	44.4	1.395	3.3	43.2			
10	1.086	2.5	57.7	.444	1.0	45.5	.934	2.2	45.5			
11	1.057	2.5	60.2									
12	.961	2.2	62.5									
13	.913	2.1	64.6									
14	.847	2.0	. 66.6			a prosessione						
15	.786	1.8	68.5									
16	.761	1.8	70.3				2 2 C 24-4					
17	.737	1.7	72.1									
18	.731	1.7	73.8									
19	.698	1.6	75.5									
20	.661	1.5	77.1									
21	.634	1.5	78.6									
22	.619	1.4	80.0		Source and		211167					
23	.612	1.4	81.5	12 Starter								
24	.598	1.4	82.9		6463.02.04	Innero wraz Gao						
25	.579	1.3	84.3		e co la como	an an an a' su						
26	.561	1.3	85.6		and all sto	a con a cin (12)						
27	.519	1.2	86.9		COS. SELO	A PERSON ACCOUNTS	39.1116					
28	.491	1.1	88.0						and the second			

### Appendix 34 Total variance explained of the 10 factors solution

Appe	ndix 34 (co	ontinued)			o hail 91 act to	RUDER IV	en lett i Neve	off of vitrosida
29	.472	1.1	89.2	ing this out				- Kinchenering
30	.460	1.0	90.3					
31	.436	1.0	91.3	sale sales y	n traditioners	e de sing her:	-	num langest beinges
32	.431	1.0	92.3		and and and the	ene ren de	a da sata	That have not be
33	.412	.9	93.3	1			repáraje a	ent plot dispose
34	.396	.9	94.2		n in the second s	neg linester	a fairing and a	interest the second second
35	.370	.8	95.1	A comparison	a la set visit e		-	
36	.351	.8	96.0	a apple	je nave som och		the fill gran	and understanding of
37	.342	.8	96.8		in the second second	true site long		
38	.325	.7	97.6	a la la			a frank i fo	al quit televise in
39	.284	.6	98.2	a marine		had bed ato	Income in	
40	.278	.6	98.9	and the second second	in the second second	ensi e sencila	i binoraat	and the second second
41	.251	.5	99.5		and it		- Arterio	and a set of a section in
42	.194	.4	100			a e deined	the sequences	distances in the

### Appendix 35 Rotated Factor Matrix of the 10 factors solution

Safety climate items (n=42)			Factor											
	1	2	3	4	5	6	7	8	9	10				
F8 The actions of hospital management show that patient safety is a top priority	.61	Sai		1.845		15 A.S.	1	N.						
F10 Hospital units work well together to provide the best care for patients	.55	35		RE.		T.	1	αđ.						
A6 We are actively doing things to improve patient care	.54	100				165.		2.24						
A18 Our procedures and systems are good at preventing errors from happening	.52					Q.E.,		4						
A13 After we make changes to improve patient safety, we evaluate their effectiveness	.51	-				305		.40						
F4 There is good cooperation among hospital units that need to work together	.46					35.				Niz.				
F1 Hospital management provides a work climate that promotes patient safety	.46	1				34		1						
A9 Mistakes have led to positive changes here	.42					TE.								
C5 In this unit, we discuss ways to prevent errors from happening again	.41					22		.41						
F9 Hospital managemnet seems interested in patient safety only after an adverse event happens	35							Day.						
A15 Patient safety never takes second place to get more work done	1982					25								
F7 Problems often occur in the exchange of information across hospital units	41	.66				or:		1. 64						
F5 Important patient care information is often lost during shift changes	is my have	.59	ingen i	Print		orași Mat	Aurony Net							
F6 It is often unpleasant to work with staff from other hospital units		.54				4	arised.	1						
F11 Shift changes cause problems for patients in this hospital	1	.49		3										
F2 Hospital units do not coordinate well with each other	36	.46												
F3 Some things do not happen or get missed when transferring patients from one unit to another														
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?		12	.90							0.71				
D3 When a mistake is made that could harm the patient, but does not, how often is this reported?	1		.70						1.8					
D1 When a mistake is made, but is caught and corrected before affecting the patient, how often is	1.010		.68											
A4 In this unit, people treat each other with respect				.75										
A1 Staff support one another in this unit				.68					1					
A3 When a lot of work needs to be done quickly, we work together as a team to get the work done	.33			.59			51	U.						
A8 Staff feel like their mistakes are held against them					.73									
A12 When an event is reported, it feels like the person is being written up, not the problem	1				.66									
A16 Staff worry that mistakes they make are kept in their personnel file					.51									
B1 My supervisor/manager says a good word when he/she sees a job done according to established						.73								
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety						.71		1. Cal						
A5 Staff in this unit work longer hours than they should, which is not good for patient care							.48							

Appendix 35 (continued)	1.4.607.1	o ki di k		10.1	105				
A7 We use more temporary staff than we should, which is not good for patient care	1.5	1.500			init .	.42			
A14 We work in a hurry, trying to do too much, too quickly	widter	is lips	10	.37	1.00	.39			
B3 Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it means	r officient	le qu	i úg	202		.39	4	1	.34
A2 We have enough staff to handle the workload	4.05.05	100	10/47	0.10		37			
All When one area in this unit gets really busy, others help out	(prog	6.49	.30		(and	35			
C3 We are informed about errors that happen in this unit	.32	us slar	a da	Lizie	1013	12	.51		
C1 We are given feedback about changes put into place based on event reports	1913-190	ing an	in ve	in the	in the	1	.48		
C4 Staff feel free to question the decisions or actions of those with more authority	with the	a ano	ap)	noc		14	1	.62	
C2 Staff will freely speak up if they see something that may negatively affect patient care	1810, 19	(Astron			-	-	.33	.46	
C6 Staff are afraid to ask questions when something does not seem right	de e di	in pres	0617	944		10		-,45	
B4 My supervisor/manager ignores patient safety problems that happen over and over	28 G 128	() d) =	a.s.d			1950			.47
A10 It is just by chance that more serious mistakes do not happen around here	Ang in	numb.	1.00						
A17 We have patient safety problems in this unit	istai he	A gran							

Extraction Method: Principal Axis Factoring. Rotation Method: Varimax with Kaiser Normalization Rotation converged in 15 iterations

A-85

#### Appendix 36 Communalities of the 9 factors solution

Safety climate items (n=42)	Initial	Extraction
A1 Staff support one another in this unit	.51	.57
A2 We have enough staff to handle the workload	.24	.24
A3 When a lot of work needs to be done quikly, we work together as a team to	.50	.58
A4 In this unit, people treat each other with respect	.50	.71
A5 Staff in this unit work longer hours than they should, which is not good for	.25	.28
A6 We are actively doing things to improve patient care	.41	.41
A7 We use more temporary staff than we should, which is not good for patient	.25	.25
A8 Staff feel like their mistakes are held against them	.46	.62
A9 Mistakes have led to positive changes here	.35	.33
A10 It is just by chance that more serious mistakes do not happen around here	.13	.07
A11 When one area in this unit gets really busy, others help out	.38	.36
A12 When an event is reported, it feels like the person is being written up, not	.44	.53
A13 After we make changes to improve patient safety, we evaluate their	.47	.52
A14 We work in a hurry, trying to do too much, too quickly	.36	.37
A15 Patient safety never takes second place to get more work done	.20	.16
A16 Staff worry that mistakes they make are kept in their personnel file	.31	.36
A17 We have patient safety problems in this unit	.18	.15
A18 Our procedures and systems are good at preventing errors from happening	.42	.41
B1 My supervisor/manager says a good word when he/she sees a job done	.60	.69
B2 My supervisor/manager seriously considers staff suggestions for improving	.63	.76
B3 Whenever pressure builds up, my supervisor/manager wants us to work	.38	.45
B4 My supervisor/manager ignores patient safety problems that happen over	.42	.39
C1 We are given feedback about changes put into place based on event reports	.42	.41
C2 Staff will freely speak up if they see something that may negatively affect	.46	.50
C3 We are informed about errors that happen in this unit	.44	.44
C4 Staff feel free to question the decisions or actions of those with more	.41	.58
C5 In this unit, we discuss ways to prevent errors from happening again	.56	.57
C6 Staff are afraid to ask questions when something does not seem right	.28	.28
D1 When a mistake is made, but is caught and corrected before affecting the	.54	.55
D2 When a mistake is made, but has no potential to harm the patient, how often	.66	.89

Appendix 36 (Continued)	KORT INTO I	C KIORNAGE/
D3 When a mistake is made that could harm the patient, but does not, how	.54	.55
F1 Hospital management provides a work climate that promotes patient safety	.41	.40
F2 Hospital units do not coordinate well with each other	.45	.45
F3 Some things do not happen or get missed when transferring patients from	.15	.11
F4 There is good cooperation among hospital units that need to work together	.47	.55
F5 Important patient care information is often lost during shift changes	.39	.42
F6 It is often unpleasant to work with staff from other hospital units	.37	.38
F7 Problems often occur in the exchange of information across hospital units	.42	.46
F8 The actions of hospital management show that patient safety is a top priority	.45	.46
F9 Hospital managemnet seems interested in patient safety only after an	.36	.34
F10 Hospital units work well together to provide the best care for patients	.53	.53
F11 Shift changes cause problems for patients in this hospital	.34	.36

	Ι	nitial Eig	envalues	Extrac	ction Sum	ns of Squared	Rota	tion Sum	s of Squared
Factor	Total	% of	Cumulative %	Total	% of	Cumulative %	Total	% of	Cumulative %
1	9.930	23.6	23.6	9.421	22.4	22.4	4.235	10.0	10.0
2	2.413	5.7	29.3	1.841	4.3	26.8	2.957	7.0	17.1
3	2.227	5.3	34.6	1.812	4.3	31.1	2.167	5.1	22.2
4	1.869	4.4	39.1	1.348	3.2	34.3	1.993	4.7	27.0
5	1.626	3.8	43.0	1.161	2.7	37.1	1.835	4.3	31.3
6	1.516	3.6	46.6	.998	2.3	39.4	1.590	3.7	35.1
7	1.288	3.0	49.6	.782	1.8	41.3	1.566	3.7	38.9
8	1.188	2.8	52.5	.642	1.5	42.8	1.443	3.4	42.3
9	1.095	2.6	55.1	.588	1.3	44.2	.808	1.9	44.2
10	1.086	2.5	57.7	11465		Shidhlig to ca	atiline		anto Bide 1 1
11	1.057	2.5	60.2	15.5		jation 19	2013) 2013		a M ministerio
12	.961	2.2	62.5	vigeste		st gestide th			
13	.913	2.1	64.6	tid up is					
14	.847	2.0	66.6	Sec. 2	elen sennyi	storti telep	1		7 . 16
15	.786	1.8	68.5	Stender.		r parasition (j)			
16	.761	1.8	70.3	ALE NO					1 1.13
17	.737	1.7	72.1	tana ar		ergystiona la	<i>iptedes</i>		1 10 41
18	.731	1.7	73.8	t weet		a sere s arb il	ris (		-
19	.698	1.6	75.5	in the second	and Trees	positivo i rije	No.		76
20	.661	1.5	77.1	agina io		r yigan is ta'r	cvk.		1.45
21	.634	1.5	78.6	lend (2)		ter the paper	(and		1 19
22	.619	1.4	80.0	the first		anga na dudi	isbour.		40.1
23	.612	1.4	81.5	(pro-)		the aspending	affecta		417
24	.598	1.4	82.9	aiped					1 44
25	.579	1.3	84.3	Sta . 2		Line wild bid	6.4		100
26	.561	1.3	85.6	inst de		napetrong sign			1.0
27	.519	1.2	86.9	and inte		i tas sooming			
28	.491	1.1	88.0	for an		Second affects	g thất		1 19
29	.472	1.1	89.2	: 00.03	lot to bar	a the patient. I	ne vrite	1	80

Appendix 37 Total variance explained of the nine factors solution

Appe	ndix 37 (co	ontinued)		and the second	0.0002 0.001 10.21	Inversion Independent	t brankford s	N NEWSCHART
30	.460	1.0	90.3	a sun				
31	.436	1.0	91.3	all marters	i secureta			and the
32	.431	1.0	92.3	in a sure line	e distriction of the	see or the see	any service	a and a state of the
33	.412	.9	93.3		prompto ese qui	ana an	and a spectrum	an program
34	.396	.9	94.2			1000 20.000	(Translater)	per den fer
35	.370	.8	95.1		. And price		a increase a c	an spectra pre-
36	.351	.8	96.0	1		Arrest of	-	the most spore
37	.342	.8	96.8		gamena and a	to pulle	a Beerland	sine his said
38	.325	.7	97.6		o stra teaco de o	et such part		and plant is
39	.284	.6	98.2	In the particular				renperta
40	.278	.6	98.9		and the part of			a fare being pro
41	.251	.5	99.5		chies the logal		i hara	- Ten spectra in ser
42	.194	.4	100		a consequencial	include a presi		(b) reference bear

Extraction Method: Principal Axis

#### Appendix 38 Rotated Factor Matrix of the 9 factors solution

fety climate items (n=42)		Factor									
ety climate items (n=42)  3 After we make changes to improve patient safety, we evaluate their effectiveness The actions of hospital management show that patient safety is a top priority We are actively doing things to improve patient care In this unit, we discuss ways to prevent errors from happening again We are informed about errors that happen in this unit 8 Our procedures and systems are good at preventing errors from happening We are given feedback about changes put into place based on event reports Mistakes have led to positive changes here ) Hospital units work well together to provide the best care for patients Hospital management provides a work climate that promotes patient safety My supervisor/manager ignores patient safety problems that happen over and over 5 Patient safety never takes second place to get more work done Problems often occur in the exchange of information across hospital units Inportant patient care information is often lost during shift changes It is often unpleasant to work with staff from other hospital units Hospital management seems interested in patient safety only after an adverse event happens 7 We have patient safety problems in this hospital Hospital management seems interested in patient safety only after an adverse event happens 7 We have patient safety problems in this unit Some things do not happen or get missed when transferring patients from one unit to another E When a mistake is made, but has no potential to harm the patient, how often is this reported? When a mistake is made, but is caught and corrected before affecting the patient, how often is this reported?	ic.m	1	2	3	4	5	6	7	8	9	
A13 After we make changes to improve patient safety, we evaluate their effectiveness	22	.67	1	r ite	S. San	á.	1	sil.			
F8 The actions of hospital management show that patient safety is a top priority	1281	.57		6.3	4	15.	N.L.	133			
A6 We are actively doing things to improve patient care	154	.56		1.2		NE.	2				
C5 In this unit, we discuss ways to prevent errors from happening again	13:21	.55		. a		NE 8.		.37			
C3 We are informed about errors that happen in this unit	10.44	.59				al.		136			
A18 Our procedures and systems are good at preventing errors from happening	13136	.48				15	13				
C1 We are given feedback about changes put into place based on event reports	and a	.47	6			32				7	
A9 Mistakes have led to positive changes here	1 Sec.	.46	1	3		88.		100			
F10 Hospital units work well together to provide the best care for patients		.44	42	. 1	4	12	4	mi.			
F1 Hospital management provides a work climate that promotes patient safety	18.95	.38	35		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	25					
B4 My supervisor/manager ignores patient safety problems that happen over and over	001	34				.01					
A15 Patient safety never takes second place to get more work done	diaran 1	LE. IE	in the second	energian Al de	1469	99999 1 110	loween Kogit				
F7 Problems often occur in the exchange of information across hospital units			.65								
F5 Important patient care information is often lost during shift changes			.56								
F6 It is often unpleasant to work with staff from other hospital units		100	.55								
F2 Hospital units do not coordinate well with each other			.50								
F11 Shift changes cause problems for patients in this hospital			.50								
F9 Hospital management seems interested in patient safety only after an adverse event happens		35	.36					1		4	
A17 We have patient safety problems in this unit											
F3 Some things do not happen or get missed when transferring patients from one unit to another											
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?				.91							
D3 When a mistake is made that could harm the patient, but does not, how often is this reported?				.70							
D1 When a mistake is made, but is caught and corrected before affecting the patient, how often is this				.68				1.000			
A4 In this unit, people treat each other with respect		and an			.76						
A1 Staff support one another in this unit					.67						
A3 When a lot of work needs to be done quikly, we work together as a team to get the work done		.39			.57						
A8 Staff feel like their mistakes are held against them				1997		.74					
A12 When an event is reported, it feels like the person is being written up, not the problem					-New	.65					
A16 Staff worry that mistakes they make are kept in their personnel file						.51					

Appendix 38 (continued)	MII Ja a							
A5 Staff in this unit work longer hours than they should, which is not good for patient care				1.21	.48			
B3 Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it means taking	i iii	i en in	a serve	han	.44	ing in	ien 15	.33
A7 We use more temporary staff than we should, which is not good for patient care	1000	ti si si		12.12	.41	3/4		57
A14 We work in a hurry, trying to do too much, too quickly	149/10	0.00 20	-	.36	.40	1.00	10%	8.0
A2 We have enough staff to handle the workload	(B) (A)	dia ba	1.7	ale (	36	115	11	24
A11 When one area in this unit gets really busy, others help out	.30	herd big			35			
C4 Staff feel free to question the decisions or actions of those with more authority	wind			10		.68	2Vi	
C2 Staff will freely speak up if they see something that may negatively affect patient care	.36	fails and	e oner		- ann	.52	010	ς <sub>Λ</sub>
C6 Staff are afraid to ask questions when something does not seem right	SALE!	t sin he			-	37	10.2	
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety	.30				anger.	0	.71	
B1 My supervisor/manager says a good word when he/she sees a job done according to established patient	.36	ar i ar				No.	.69	
F4 There is good cooperation among hospital units that need to work together	1993	34	i int					.40
A10 It is just by chance that more serious mistakes do not happen around here	a la s			1			N.	1914

Extraction Method: Principal Axis Factoring. Rotation Method: Varimax with Kaiser Normalization Rotation converged in 10 iterations

### Appendix 39 Communalities of the 8 factors solution

Safety climate items (n=42)	Initial	Extraction
A1 Staff support one another in this unit	.52	.60
A2 We have enough staff to handle the workload	.25	.24
A3 When a lot of work needs to be done quickly, we work together as a team to get the work done	.50	.54
A4 In this unit, people treat each other with respect	.50	.63
A5 Staff in this unit work longer hours than they should, which is not good for patient care	.25	.28
A6 We are actively doing things to improve patient care	.41	.41
A7 We use more temporary staff than we should, which is not good for patient care	.25	.25
A8 Staff feel like their mistakes are held against them	.46	.62
A9 Mistakes have led to positive changes here	.35	.33
A10 It is just by chance that more serious mistakes do not happen around here	.13	.06
A11 When one area in this unit gets really busy, others help out	.38	.36
A12 When an event is reported, it feels like the person is being written up, not the problem	.44	.53
A13 After we make changes to improve patient safety, we evaluate their effectiveness	.47	.51
A14 We work in a hurry, trying to do too much, too quickly	.36	.37
A15 Patient safety never takes second place to get more work done	.20	.15
A16 Staff worry that mistakes they make are kept in their personnel file	.31	.35
A17 We have patient safety problems in this unit	.18	.15
A18 Our procedures and systems are good at preventing errors from happening	.42	.41
B1 My supervisor/manager says a good word when he/she sees a job done according to established	.60	.50
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety	.63	.61
B3 Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it means	.38	.43
B4 My supervisor/manager ignores patient safety problems that happen over and over	.42	.39
C1 We are given feedback about changes put into place based on event reports	.42	.37
C2 Staff will freely speak up if they see something that may negatively affect patient care	.46	.49
C3 We are informed about errors that happen in this unit	.44	.41
C4 Staff feel free to question the decisions or actions of those with more authority	.41	.60
C5 In this unit, we discuss ways to prevent errors from happening again	.56	.56
C6 Staff are afraid to ask questions when something does not seem right	.28	.27
D1 When a mistake is made, but is caught and corrected before affecting the patient, how often is	.54	.55
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?	.66	.88

Appendix 39 (continued)	an all a chail	
D3 When a mistake is made that could harm the patient, but does not, how often is this reported?	.54	.53
F1 Hospital management provides a work climate that promotes patient safety	.41	.37
F2 Hospital units do not coordinate well with each other	.45	.45
F3 Some things do not happen or get missed when transferring patients from one unit to another	.15	.10
F4 There is good cooperation among hospital units that need to work together	.47	.45
F5 Important patient care information is often lost during shift changes	.39	.41
F6 It is often unpleasant to work with staff from other hospital units	.37	.38
F7 Problems often occur in the exchange of information across hospital units	.42	.46
F8 The actions of hospital management show that patient safety is a top priority	.45	.46
F9 Hospital management seems interested in patient safety only after an adverse event happens	.36	.33
F10 Hospital units work well together to provide the best care for patients	.53	.53
F11 Shift changes cause problems for patients in this hospital	.34	.36
		and the second second second second

		Initial Eig	envalues	Extra	ction Sum	ns of Squared	Rotation Sums of Squared					
Factor	Total	% of Variance	Cumulative %	Total	% of Variance	Cumulative %	Total	% of Variance	Cumulative %			
1	9.930	23.6	23.6	9.3	22.3	22.3	4.402	10.4	10.4			
2	2.413	5.7	29.3	1.834	4.3	26.7	2.837	6.7	17.2			
3	2.227	5.3	34.6	1.796	4.2	31.0	2.173	5.1	22.4			
4	1.869	4.4	39.1	1.336	3.1	34.1	2.1	5.0	27.4			
5	1.626	3.8	43.0	1.123	2.6	36.8	1.820	4.3	31.8			
6	1.516	3.6	46.6	.981	2.3	39.2	1.641	3.9	35.7			
7	1.288	3.0	49.6	.745	1.7	40.9	1.452	3.4	39.1			
8	1.188	2.8	52.5	.632	1.5	42.4	1.393	3.3	42.4			
9	1.095	2.6	55.1	100				dareter and a	a new manda			
10	1.086	2.5	57.7	S. B. S.	in large	Bloghet gi						
11	1.057	2.5	60.2	Si wa					1.10			
12	.961	2.2	62.5	142-12	and states							
13	.913	2.1	64.6	hora	a de la composition de la comp	en e						
14	.847	2.0	66.6									
15	.786	1.8	68.5	a kata	LOS DIA	Secondia -						
16	.761	1.8	70.3	Anch		e para tra da		an bille basi	1. 1. 10 1. 1			
17	.737	1.7	72.1	6-13)	10.7419	Distance of		iny -	425			
18	.731	1.7	73.8	94.8.324	South State	e adh thank		Lannais				
19	.698	1.6	75.5	44.24	a nakalas	here entry and						
20	.661	1.5	77.1		1.1.1.1.1.1.1	ad spaces			1.1.1			
21	.634	1.5	78.6	Sec. 1	1 And	als share as l	R Sand		a la la			
22	.619	1.4	80.0	S. Sale								
23	.612	1.4	81.5		Sec. Par	tridente la de						
24	.598	1.4	82.9	Karah J	harden	1994						
25	.579	1.3	84.3									
26	.561	1.3	85.6	144.83			Hereita	Merchan	Se par la			
27	.519	1.2	86.9	Million	NR SPACE	"Deal which	hain	49.253	1			

# Appendix 40 Total variance explained of the 8 factors solution

•

Appen	Appendix 40 (continued)											
28	.491	1.1	88.0									
29	.472	1.1	89.2									
30	.460	1.0	90.3	the selected enders in the set of select the set of the								
31	.436	1.0	91.3	a la la compañía de l								
32	.431	1.0	92.3	and the second state in the second state of th								
33	.412	.9	93.3	La constantine for an and the second								
34	.396	.9	94.2	and the series has and points an good operations and series of the								
35	.370	.8	95.1	The provide the second second second second to be								
36	.351	.8	96.0	The second sets and the period structure and second structures								
37	.342	.8	96.8	and the second problem of the second s								
38	.325	.7	97.6	a compare programmer don's doubt but pour bloch if								
39	.284	.6	98.2	a wante shirt been been been shirt and shirt and shirt a								
40	.278	.6	98.9	and appropriate printing and allow a product of the second state of the								
41	.251	.5	99.5									
42	.194	.4	100	en a la company de								

### Appendix 41 Rotated Factor Matrix of the 8 factors solution

Safety climate items (n=42)		Factor									
	1	2	3	4	5	6	7	8			
A13 After we make changes to improve patient safety, we evaluate their effectiveness	.65					tawe Care					
F8 The actions of hospital management show that patient safety is a top priority	.59		Automation	h							
A6 We are actively doing things to improve patient care	.50	h.				105					
C5 In this unit, we discuss ways to prevent errors from happening again	.53			1	4	.37					
A18 Our procedures and systems are good at preventing errors from happening	.51	6		1 jai				-			
A9 Mistakes have led to positive changes here	.48		1	1	6.1	33					
F10 Hospital units work well together to provide the best care for patients	.48	46			0	10					
C3 We are informed about errors that happen in this unit	.48			0	5	35					
F1 Hospital management provides a work climate that promotes patient safety	.44	33	20			139					
C1 We are given feedback about changes put into place based on event reports	.43			1.00		120					
F4 There is good cooperation among hospital units that need to work together	.36	32		.33							
A15 Patient safety never takes second place to get more work done		10									
F7 Problems often occur in the exchange of information across hospital units		.65		- Just	1	1					
F5 Important patient care information is often lost during shift changes	1.0. S	.56	and the second	strain -	and and a	anglasty a					
F6 It is often unpleasant to work with staff from other hospital units		.55			in a	102.5					
F11 Shift changes cause problems for patients in this hospital		.50									
F2 Hospital units do not coordinate well with each other	31	.50	5								
F9 Hospital managemnet seems interested in patient safety only after an adverse event happens	34	.36									
A17 We have patient safety problems in this unit											
F3 Some things do not happen or get missed when transferring patients from one unit to another			2.2.4			142					
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?			.90								
D3 When a mistake is made that could harm the patient, but does not, how often is this reported?			.69								
D1 When a mistake is made, but is caught and corrected before affecting the patient, how often is			.68								
A4 In this unit, people treat each other with respect				.71							
A1 Staff support one another in this unit				.70							
A3 When a lot of work needs to be done quickly, we work together as a team to get the work done	.37			.57							
A8 Staff feel like their mistakes are held against them					.74						
A12 When an event is reported, it feels like the person is being written up, not the problem					.65						
A16 Staff worry that mistakes they make are kept in their personnel file					.51						

Appendix 41 (continued)	o su sustella.	anten a sibre a	44
C4 Staff feel free to question the decisions or actions of those with more authority			70
C2 Staff will freely speak up if they see something that may negatively affect patient care	.34		51
C6 Staff are afraid to ask questions when something does not seem right	12 24	-	37
A5 Staff in this unit work longer hours than they should, which is not good for patient care			.49
A7 We use more temporary staff than we should, which is not good for patient care			.41
B3 Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it means	5		.4140
A14 We work in a hurry, trying to do too much, too quickly		.36	.40
A2 We have enough staff to handle the workload			34
A11 When one area in this unit gets really busy, others help out	.32	.33	33
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety	.37		.54
B1 My supervisor/manager says a good word when he/she sees a job done according to	.43		.47
B4 My supervisor/manager ignores patient safety problems that happen over and over	31		42
A10 It is just by chance that more serious mistakes do not happen around here			

Extraction Method: Principal Axis Factoring. Rotation Method: Varimax with Kaiser Normalization Rotation converged in 14 iterations

### Appendix 42 Pattern Matrix of the 8 factors solution

climate items (n=42)					Factor								
	1	2	3	4	5	6	7	8					
A13 After we make changes to improve patient safety, we evaluate their effectiveness	.60	1	ales.	1		and the set	1.1						
F8 The actions of hospital management show that patient safety is a top priority	.50		1										
A6 We are actively doing things to improve patient care	.49		2.4.4										
A9 Mistakes have led to positive changes here	.44												
A18 Our procedures and systems are good at preventing errors from happening	.39												
F1 Hospital management provides a work climate that promotes patient safety	.30		Same										
A8 Staff feel like their mistakes are held against them	1	.76											
A12 When an event is reported, it feels like the person is being written up, not the problem		.66											
A16 Staff worry that mistakes they make are kept in their personnel file	and the second	.49											
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?			98			alles (s							
D3 When a mistake is made that could harm the patient, but does not, how often is this reported?			74										
D1 When a mistake is made, but is caught and corrected before affecting the patient, how often is this reported?	19-19-19-19-19-19-19-19-19-19-19-19-19-1	and has	74	tri t		an pan Trideoec							
F7 Problems often occur in the exchange of information across hospital units	nose		ita na mat	.67		al-al-							
F6 It is often unpleasant to work with staff from other hospital units	199			.56									
F5 Important patient care information is often lost during shift changes				.54									
F11 Shift changes cause problems for patients in this hospital				.47									
F2 Hospital units do not coordinate well with each other				.45			144						
F10 Hospital units work well together to provide the best care for patients	.31		14	35									
F9 Hospital management seems interested in patient safety only after an adverse event happens				.30									
F3 Some things do not happen or get missed when transferring patients from one unit to another	in the					SP.							
A17 We have patient safety problems in this unit	1 975-42	1											
A4 In this unit, people treat each other with respect	to service	n la			76								
A1 Staff support one another in this unit	ing all			1 m	74								
A3 When a lot of work needs to be done quickly, we work together as a team to get the work done					54								
F4 There is good cooperation among hospital units that need to work together													
C4 Staff feel free to question the decisions or actions of those with more authority	miline					.82	2						
C2 Staff will freely speak up if they see something that may negatively affect patient care						.59							
C5 In this unit, we discuss ways to prevent errors from happening again	.31					.44							
C 6 Staff are afraid to ask questions when something does not seem right						39							

appendix 42 (continued)	an gala sin a	1900	in other	6312			
C3 We are informed about errors that happen in this unit	.32		l'ester		.33		
C1 We are given feedback about changes put into place based on event reports	Dage and the		0.0	12 PC	.31		
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety	nen eter	ining 1	10,810	sei a	.33	49	
B1 My supervisor/manager says a good word when he/she sees a job done according to established patient safety	arver 275		ane a	-	david 1	43	
B4 My supervisor/manager ignores patient safety problems that happen over and over	0 100 1000	1.2%	10 2019	100	forda	.38	
A10 It is just by chance that more serious mistakes do not happen around here	reprint de lega	10 90	1.28	1000	1.1381		
A5 Staff in this unit work longer hours than they should, which is not good for patient care	an on the star	gine	tint y	1000	198		49
A7 We use more temporary staff than we should, which is not good for patient care	alat mang	-	inde e	il inc	(jant)		-,40
B3 Whenever pressure builds up, my supervisor/manager wants us to work faster, even if it means taking	agent a gene	ion of	441 74		anabri.	.36	37
A2 We have enough staff to handle the workload	Fare I.A.	ni-grad	i ma	1.000	12.046		.36
A14 We work in a hurry, trying to do too much, too quickly	.31	Contrast	6-24 <u>1</u>		defe to		35
A11 When one area in this unit gets really busy, others help out	eshen garti e	125-12	Senia.				.32
A15 Patient safety never takes second place to get more work done	bear a sys		men				

Extraction Method: Principal Axis Factoring. Rotation Method: Oblimin with Kaiser Normalization Rotation converged in 46 iterations

### Appendix 43 Communalities of the step eight

Safety climate items (n=25)	Initial	Extraction
A1 Staff support one another in this unit	.48	.64
A3 When a lot of work needs to be done quickly, we work together as a team	.42	.52
A4 In this unit, people treat each other with respect	.46	.63
A5 Staff in this unit work longer hours than they should, which is not good	.17	.31
A6 We are actively doing things to improve patient care	.38	.45
A7 We use more temporary staff than we should, which is not good for	.17	.23
A8 Staff feel like their mistakes are held against them	.42	.61
A9 Mistakes have led to positive changes here	.31	.34
A12 When an event is reported, it feels like the person is being written up,	.41	.54
A13 After we make changes to improve patient safety, we evaluate their	.39	.49
A16 Staff worry that mistakes they make are kept in their personnel file	.27	.34
B1 My supervisor/manager says a good word when he/she sees a job done	.57	.66
B2 My supervisor/manager seriously considers staff suggestions for	.61	.82
C2 Staff will freely speak up if they see something that may negatively affect	.42	.50
C4 Staff feel free to question the decisions or actions of those with more	.36	.64
C5 In this unit, we discuss ways to prevent errors from happening again	.50	.55
D1 When a mistake is made, but is caught and corrected before affecting the	.52	.56
D2 When a mistake is made, but has no potential to harm the patient, how	.64	.89
D3 When a mistake is made that could harm the patient, but does not, how	.50	.52
F2 Hospital units do not coordinate well with each other	.35	.37
F5 Important patient care information is often lost during shift changes	.35	.43
F6 It is often unpleasant to work with staff from other hospital units	.31	.38
F7 Problems often occur in the exchange of information across hospital units	.37	.54
F8 The actions of hospital management show that patient safety is a top	.32	.36
F11 Shift changes cause problems for patients in this hospital	.31	.39

	]	Initial Eigen	values	Extraction Sums of Squared				
Factor	Total	% of Variance	Cumulative %	Total	% of Variance	Cumulative %	Total	
1	6.404	25.6	25.6	5.948	23.7	23.7	3.3	
2	2.053	8.2	33.8	1.681	6.7	30.5	3.0	
3	1.963	7.8	41.6	1.462	5.8	36.3	2.7	
4	1.521	6.0	47.7	1.040	4.1	40.5	2.0	
5	1.331	5.3	53.0	.927	3.7	44.2	2.7	
6	1.203	4.8	57.8	.679	2.7	46.9	2.9	
7	1.033	4.1	62.0	.603	2.4	49.3	3.1	
8	.986	3.9	65.9	.481	1.9	51.2	1.5	
9	.801	3.2	69.1			anning strike is a		
10	.746	2.9	72.1		1	god nil d. Calman		
11	.717	2.8	75.0					
12	.663	2.6	77.6			and particular		
13	.597	2.3	80.0	and many set			A BORNAN TANK	
14	.566	2.2	82.3			nan in an	least construction both	
15	.545	2.1	84.5		124 (1992)	· · · · · · · · · · · · · · · · · · ·	in a subserve serve	
16	.520	2.0	86.5			and a subscription target		
17	.472	1.8	88.4	And the Party	The second second second	and of Phenoper		
18	.470	1.8	90.3	n was juit	Analysis and the	(Company)		
19	.445	1.7	92.1	1 Service States	alls then bit i	a ha shi ka		
20	.417	1.6	93.8		Surga grootes	al tradi parti ini ar		
21	.395	1.5	95.3		Sarangeuspell medie	ng min Turke e drinner av		
22	.348	1.3	96.7		a section growth	and each lane by	and transmission	
23	.331	1.3	98.1	SUD RISING	Phone You proved	e bloger vertrauit en		
24	.255	1.0	99.1	The second s				
25	.217	.8	100	nose	ALCONOM SURVEY	the number of	Continues ?	

### Appendix 44 Total variance explained of the step eight

Safety climate items (n=25)	zilici	Factor									
	1	2	3	4	5	6	7	8			
A13 After we make changes to improve patient safety, we evaluate their effectiveness	.60	1994	ight.v	1.4	T &	stop					
A6 We are actively doing things to improve patient care	.56	a na an		-	and the	territory (	2				
F8 The actions of hospital management show that patient safety is a top priority	.45										
A9 Mistakes have led to positive changes here	.434			I.N.		1. 1					
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?	ALC: N	.98		il i ci							
D1 When a mistake is made, but is caught and corrected before affecting the patient, how often is this		.73									
D3 When a mistake is made that could harm the patient, but does not, how often is this reported?		.72		1. A	- Autor						
F7 Problems often occur in the exchange of information across hospital units			.74								
F6 It is often unpleasant to work with staff from other hospital units		al at a	.56		1						
F5 Important patient care information is often lost during shift changes			.54	Luna		2.5					
F11 Shift changes cause problems for patients in this hospital	14-14		.47								
F2 Hospital units do not coordinate well with each other	here's	Arrent .	.38								
A8 Staff feel like their mistakes are held against them	57.00	and the second		73							
A12 When an event is reported, it feels like the person is being written up, not the problem	Kessag			65							
A16 Staff worry that mistakes they make are kept in their personnel file	1.00			48			1				
A1 Staff support one another in this unit	$\beta_{2} = i \eta$	No.			76						
A4 In this unit, people treat each other with respect	ind, th			1944 1	73						
A3 When a lot of work needs to be done quikly, we work together as a team to get the work done	are for				57	01					
C4 Staff feel free to question the decisions or actions of those with more authority		12.13				.80					
C2 Staff will freely speak up if they see something that may negatively affect patient care	100					.54					
C5 In this unit, we discuss ways to prevent errors from happening again	.30			Citta		.41					
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety	apata	no.1 Natis				103	84				
B1 My supervisor/manager says a good word when he/she sees a job done according to established	1.11.201						- 74				
A5 Staff in this unit work longer hours than they should, which is not good for patient care		the state		SAA							
A7 We use more temporary staff than we should, which is not good for patient care			S. Contractor	and defe		and the second		с.			
Extraction Method: Principal Axis Factoring.		WS Yes	1.18	1527		1.20		.4			

#### Appendix 45 Pattern Matrix of the 8 factors solution

Extraction Method: Principal Axis Factoring. Rotation Method: Oblimin with Kaiser Normalization Rotation converged in 9 iterations

Factor one	Item loadings	
A6:	.699	and the second second
A9:	.563	
A13:	.672	
F8:	.583	
Factor two	Item loadings	
D1:	.730	i mulant
D2:	.953	
D3:	.725	
Factor three	Item loadings	In maker
A8:	.805	
A12:	.707	A. Andama
A16:	.541	
Factor four	Item loadings	
F5:	.639	an then
F6:	.548	
F7:	.706	
F11:	.581	
Factor five	Item loadings	
A1:	.818	
A3:	.662	
A4:	.709	
Factor six	Item loadings	dive are
C2:	.694	
C4:	.994	
Factor seven	Item loadings	N. Carlos
B1	.846	
B2	.846	
Factor eight	Item loadings	
A5:	.510	
A7:	.510	

Appendix 46 EFA of each scale of the optimal model (8 factors)

The table above shows that each set of items was loaded in one factor as they were structured in the optimal model. The overall of all items (23 items) and 8 factors satisfactorily explained by a single factor.
#### Appendix 47 Extra factor analysis investigations

In order to explore potential improvements and because exploratory factor analysis is based on exploration approach another investigation was performed by exploring the conclusion of the comparison between 11-12-10-9-8 factors solutions that produced the common factor structure which consists 10 factors. The summary of the conclusion indicated that the common factor structure consisted of 10 factors. In addition 10 factor solution was suggested by scree plot (see page 12) and also it was suggested by 11 and 12 factors solution. Items number A10-A15-A17-F3 did not load upon any factor. Items number A2-A5-A10-A15-A17-F3 had low communalities. Items number F1-F3-F9-A11-A14-A18-A2-C5-B3 had low loadings. Therefore, Items numbers A10-A15-A17-F3-A2-A11-A14-A18-B3-C5-F1-F9 were removed in the next step.

#### Step 1:

Run EFA exclude A10-A15-A17-F3-A2-A11-A14-A18-B3-C5-F1-F9 and number of factors (10) with using oblique rotation to aid interpretation of the items in each factor of the solution. The following tables show the communalities, Total Variance Explained and Pattern Matrix of this step.

# Table 1 Communalities of the 10 factors solution

Safety climate items (n=30)	Initial	Extraction
A1 Staff support one another in this unit	.494	.653
A3 When a lot of work needs to be done quickly, we work together as a team to get the work done	.433	.526
A4 In this unit, people treat each other with respect	.468	.622
A5 Staff in this unit work longer hours than they should, which is not good for patient care	.187	.290
A6 We are actively doing things to improve patient care	.392	.595
A7 We use more temporary staff than we should, which is not good for patient care	.196	.318
A8 Staff feel like their mistakes are held against them	.432	.620
A9 Mistakes have led to positive changes here	.316	.365
A12 When an event is reported, it feels like the person is being written up, not the problem	.424	.551
A13 After we make changes to improve patient safety, we evaluate their effectiveness	.453	.552
A16 Staff worry that mistakes they make are kept in their personnel file	.278	.345
B1 My supervisor/manager says a good word when he/she sees a job done according to established	.591	.827
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety	.619	.698
B4 My supervisor/manager ignores patient safety problems that happen over and over	.313	.327
C1 We are given feedback about changes put into place based on event reports	.399	.450
C2 Staff will freely speak up if they see something that may negatively affect patient care	.436	.529
C3 We are informed about errors that happen in this unit	.400	.489
C4 Staff feel free to question the decisions or actions of those with more authority	.364	.591
C6 Staff are afraid to ask questions when something does not seem right	.262	.345
D1 When a mistake is made, but is caught and corrected before affecting the patient, how often is this reported?	.535	.586
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?	.656	.887
D3 When a mistake is made that could harm the patient, but does not, how often is this reported?	.524	.552
F2 Hospital units do not coordinate well with each other	.436	.472
F4 There is good cooperation among hospital units that need to work together	.424	.531
F5 Important patient care information is often lost during shift changes	.366	.446
F6 It is often unpleasant to work with staff from other hospital units	.323	.377
F7 Problems often occur in the exchange of information across hospital units	.397	.575
F8 The actions of hospital management show that patient safety is a top priority	.392	.422

Table 1 (continued)	11	
F10 Hospital units work well together to provide the best care for patients where the second of the second second	.514	.649
F11 Shift changes cause problems for patients in this hospital	.323	.384

Extraction Method: Principal Axis Factoring.

### Table 2 Total Variance Explained of the 10 factors solution

2.14		Initial Eigenval	lues	Extraction	n Sums of Squar	red Loadings	Rotation Sums
Factor	Total	% of Variance	Cumulative %	Total	% of Variance	Cumulative %	Total
1	7.707	25.692	25.692	7.253	24.176	24.176	2.329
2	2.069	6.897	32.589	1.709	5.695	29.871	3.387
3	2.044	6.813	39.402	1.546	5.155	35.026	2.955
4	1.623	5.409	44.810	1.171	3.905	38.931	2.285
5	1.388	4.627	49.438	.995	3.317	42.248	3.288
6	1.282	4.272	53.710	.778	2.592	44.840	2.483
7	1.161	3.871	57.581	.714	2.380	47.220	3.938
8	1.112	3.705	61.287	.555	1.850	49.070	1.552
9	.975	3.250	64.537	.513	1.709	50.779	4.175
10	.847	2.825	67.362	.340	1.132	51.911	2.903
11	.781	2.603	69.965	ALL A DESCRIPTION OF	aa nan Juransi	na dire Karli je di	Sunde Visari, (i)
12	.755	2.517	72.482		<ul> <li>New state of the</li> </ul>	(ind tails around	unde bottypai.
13	.695	2.318	74.799	era sunversa	es to anothic fo	Hite decartons	taking takan bu
14	.631	2.102	76.901	Gen soleten die	energi gaurinanan	Cristiae (Hesterio)	e alta at hittle e
15	.600	1.999	78.900	er solf i e solar.	d persone has	Higgs is and a	nan aradahara i
16	.591	1.971	80.871				
17	.565	1.883	82.754	a Hadakiya da	dened of inbang	e, bui ites po ju	nan HESSINDER I
18	.533	1.776	84.530	1911 2010 1	d Monzon als a	and blumin david m	nta al olleistica a
19	.500	1.666	86.196		raio i dans iltr	e then observings	o tos do ener h
20	.491	1.638	87.834	n for which in	io inte manufal	ihoq Maham av	s good coopera
21	.463	1.543	89.377	nganin fin	Sump ting the	o zi dolterražn	tras tending ap
22	.444	1.478	90.855	stum into	eou neire noch	The distriction	t instasligiu is
23	.430	1.432	92.288	nacional ann i	a orateurada) 1	agrafian da i	1 SUGOD ANSKO NO

Table 2	(continued)	s of the 19 lac	torn endlander		•					
24	.428	1.426	93.714	ot go in oco	single day.	nan si	ed and a bay	a trop a	-	1999
25	.392	1.306	95.020		topinad.		athainna			
26	.367	1.224	96.244		19.099.09	-	tem abien a	ender pro		
27	.340	1.133	97.377				alla Janes	in sur		
28	.328	1.092	98.469		buo in a	enter:	they shock	a con th		
29	.255	.851	99.320		alana a		el sel stares	0.10		
30	.204	.680	100.000		langet dysal	in line	with this is being			

Extraction Method: Principal Axis Factoring.

#### Table 3 Pattern Matrix of the 10 factors solution

Safety climate items (n=30)			Factor								
and the second second and the second seco	1	2	3	4	5	6	7	8	9	10	
C3 We are informed about errors that happen in this unit	.488	2 Ket	-	100	Bino				500		
C1 We are given feedback about changes put into place based on event reports	.444			304	0.13		ad.				
A13 After we make changes to improve patient safety, we evaluate their effectiveness	.418				(in)	Tian .	101				
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?		.968									
D3 When a mistake is made that could harm the patient, but does not, how often is this reported?		.734									
D1 When a mistake is made, but is caught and corrected before affecting the patient, how often is this		.732					12				
F7 Problems often occur in the exchange of information across hospital units		in.	.741	1	1100	A.S					
F6 It is often unpleasant to work with staff from other hospital units	COND.S.	1007	.534	1	in the second						
F5 Important patient care information is often lost during shift changes	1	ant.	.513	2212			1115	1			
F11 Shift changes cause problems for patients in this hospital		Pred a	.462								
A8 Staff feel like their mistakes are held against them				755							
A12 When an event is reported, it feels like the person is being written up, not the problem				664					1		
A16 Staff worry that mistakes they make are kept in their personnel file				506							
A1 Staff support one another in this unit					.773						
A4 In this unit, people treat each other with respect					.721						
A3 When a lot of work needs to be done quikly, we work together as a team to get the work done					.569		12:35				
C4 Staff feel free to question the decisions or extings of those with more outhority.						679					
C2 Staff will freely speels on if the spees speed is a day to be specific to a first state of the speed spee						490					
or the first set up if they see something that may negatively affect patient care						.470					
to stall are alraid to ask questions when something does not seem right						410	0.8				

Table 3 (continued)					n ai t	en de	ń.		
B1 My supervisor/manager says a good word when he/she sees a job done according to established patient	1 10 10 10			5.1		926			
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety	BRIL			2.3	and the second	704			
B4 My supervisor/manager ignores patient safety problems that happen over and over	ia deterio 19 I.C. I	an a	in a second	parinden Man	Success .	.311		and the	
A7 We use more temporary staff than we should, which is not good for patient care	had			- En	0.0		.521		
A5 Staff in this unit work longer hours than they should, which is not good for patient care	15 10.1		- 19	35	-		.478		
F10 Hospital units work well together to provide the best care for patients	C B IGen	receipter-	1	ficer		and the second		.714	
F4 There is good cooperation among hospital units that need to work together		136		n Bern	12108	Salar La la		.637	
F2 Hospital units do not coordinate well with each other	3 36		and the second		Lagi	1-1-1-5 26-1-5		458	
F8 The actions of hospital management show that patient safety is a top priority			4.87			2.3%		.440	
A6 We are actively doing things to improve patient care			0.87			3.32			67
A9 Mistakes have led to positive changes here	1 1.91 1 1.91	HO KI	No. of Street,	19 2023 	RUNE A	142462			44

Rotation Method: Oblimin with Kaiser Normalization

a. Rotation converged in 46 iterations.

The pattern matrix above showed that B4 had low loading 0.311. Therefore this item was excluded in the next step.

#### Step 2:

Run EFA exclude A10-A15-A17-F3-A2-A11-A14-A18-B3-C5-F1-F9-B4 with using oblique rotation and number of factors (10). The following tables show the communalities, Total Variance Explained and Pattern Matrix of this step.

# Table 4 Communalities of the 10 factors solution

Safety climate items (n=29)	Initial	Extraction
A1 Staff support one another in this unit	.490	.645
A3 When a lot of work needs to be done quickly, we work together as a team to get the work done	.433	.532
A4 In this unit, people treat each other with respect	.465	.621
A5 Staff in this unit work longer hours than they should, which is not good for patient care	.186	.315
A6 We are actively doing things to improve patient care	.391	.591
A7 We use more temporary staff than we should, which is not good for patient care	.183	.283
A8 Staff feel like their mistakes are held against them	.430	.628
A9 Mistakes have led to positive changes here	.316	.370
A12 When an event is reported, it feels like the person is being written up, not the problem	.423	.549
A13 After we make changes to improve patient safety, we evaluate their effectiveness	.453	.553
A16 Staff worry that mistakes they make are kept in their personnel file	.277	.343
B1 My supervisor/manager says a good word when he/she sees a job done according to established patient safety	.578	.808
B2 My supervisor/manager seriously considers staff suggestions for improving patient safety	.619	.706
C1 We are given feedback about changes put into place based on event reports	.393	.444
C2 Staff will freely speak up if they see something that may negatively affect patient care	.436	.531
C3 We are informed about errors that happen in this unit	.400	.492
C4 Staff feel free to question the decisions or actions of those with more authority	.362	.585
C6 Staff are afraid to ask questions when something does not seem right	.262	.347
D1 When a mistake is made, but is caught and corrected before affecting the patient, how often is this reported?	.535	.591
D2 When a mistake is made, but has no potential to harm the patient, how often is this reported?	.654	.882
D3 When a mistake is made that could harm the patient, but does not, how often is this reported?	.523	.553
F2 Hospital units do not coordinate well with each other	.436	.472
F4 There is good cooperation among hospital units that need to work together	.424	.529
F5 Important patient care information is often lost during shift changes	.365	.449
F6 It is often unpleasant to work with staff from other hospital units	.322	.375
F7 Problems often occur in the exchange of information across hospital units	.396	.577
F8 The actions of hospital management show that patient safety is a top priority	.391	.422
F10 Hospital units work well together to provide the best care for patients	.513	.654
F11 Shift changes cause problems for patients in this hospital	.322	.390

Extraction Method: Principal Axis Factoring.

		Initial Eigenval	lues	Extractio	n Sums of Squar	red Loadings	Rotation Sums
Factor	Total	% of Variance	Cumulative %	Total	% of Variance	Cumulative %	Total
1	7.442	25.662	25.662	6.995	24.120	24.120	2.232
2	2.068	7.131	32.793	1.704	5.876	29.996	3.325
3	2.034	7.013	39.806	1.538	5.304	35.300	2.861
4	1.619	5.583	45.389	1.170	4.036	39.336	2.270
5	1.386	4.779	50.168	.985	3.398	42.734	3.132
6	1.239	4.274	54.441	.736	2.537	45.271	2.454
7	1.138	3.926	58.367	.709	2.445	47.716	3.686
8	1.100	3.793	62.160	.552	1.903	49.619	1.451
9	.973	3.354	65.515	.509	1.756	51.375	4.048
10	.846	2.918	68.432	.335	1.156	52.531	2.843
11	.755	2.605	71.037			网络哈尔 图书图记书	n der Herbauk
12	.731	2.521	73.558		Estropola divis	en fillen er en staar de seelen en staar de seelen en staar de seelen en staar de seelen de seelen de seelen d Seelen de seelen de s	triandigial and
13	.690	2.378	75.936		astra principality	perf spectrum h	u Ange (land)
14	.623	2.148	78.084		<ul> <li>160 (000)</li> </ul>	enter 0.0 atra	n molde konstole.
15	.598	2.063	80.147	rong Laro- c	adi to zacijan K	i annististraigh	(Reo la appoint)
16	.567	1.955	82.102		a good is think	or of the section	in dis schliebs
17	.536	1.847	83.949	anter an		anti-alati in 1631	indical checkle
18	.500	1.724	85.673		do traval maleria Esta	ing and the	ningtal officials
19	.491	1.694	87.367	it posteril	ang manod my	engel Marstrage.	ilitat di selapan
20	.464	1.600	88.968		antin, dunari	enges het er sokundige	tool test of a to su
21	.444	1.530	90.498	en Herria. Alteria	iner-toir chab (	tispe strike opplyg	utangga basy
22	.434	1.496	91.994	Hauto A		nto Producenti Litto Producenti	ni franciski politik S
23	.428	1.476	93.470		lige i nautri poli	究心:	en mand any
24	.395	1.363	94.833	indig of sa	na na harant	la constance al	0) teoro azilio (
25	.368	1.270	96.103			esto Ultra angella.	t litingend he per
26	.340	1.172	97.275	1.2019			received stated
27	.328	1.130	98.405	and an and a second			Pri devilo angge

Table 5 Total Variance Explained of the 10 factors solution

Table 5	(continued)						
28	.255	.880	99.285	malia di no cu	on flore struck	ido aerizan	icu bigraoti Si
29	.207	.715	100.000	ne da ryminen		an tang orth	indication at

### Table 6 pattern matrix of the final solution (10 factors)

Safety climate items (n=29)	Factor									
	1	2	3	4	5	6	7	8	9	10
C3 We are informed about errors that happen in this unit	.490				Tinte					
C1 We are given feedback about changes put into place based on	.439									
A13 After we make changes to improve patient safety, we evaluate	.416		2016) 	10 M			1.03	1003		
D2 When a mistake is made, but has no potential to harm the patient,		.963								
D1 When a mistake is made, but is caught and corrected before		.736			1.					
D3 When a mistake is made that could harm the patient, but does not,		.734								
F7 Problems often occur in the exchange of information across			.738							
F6 It is often unpleasant to work with staff from other hospital units			.529							
F5 Important patient care information is often lost during shift			.518							
F11 Shift changes cause problems for patients in this hospital			.468							
A8 Staff feel like their mistakes are held against them			0	.762						
A12 When an event is reported, it feels like the person is being				.657			L'ANA	1.00		
A16 Staff worry that mistakes they make are kept in their personnel			7010	.502	1.			2.30		
A1 Staff support one another in this unit				SIN	-			in the		
A4 In this unit, people treat each other with respect		i alter			-		2.53	1.1		
A3 When a lot of work needs to be done quikly, we work together as		. Ast	21				13	New Y		
C4 Staff feel free to question the decisions or actions of those with					1	.675		1		
C2 Staff will freely speak up if they see something that may		123		Nelv.		.488	beer			
C6 Staff are afraid to ask questions when something does not seem						-				
B1 My supervisor/manager says a good word when he/she sees a job							-			
B2 My supervisor/manager seriously considers staff suggestions for							-			
A5 Staff in this unit work longer hours than they should, which is not								.511		
A7 We use more temporary staff than we should, which is not good	11172	201	VICE	a alt		it we	i de la	.477		
F10 Hospital units work well together to provide the best care for									.719	
F4 There is good cooperation among hospital units that need to work									.628	



Rotation Method: Oblimin with Kaiser Normalization

a. Rotation converged in 11 iterations.

Figure 1 Scree Plot of the final solution (10 factors)



The following table shows the structure of the 10 factor solution and its loadings.

Factor number	Factor	Items (loading)	Number of items		
1	Feedback and communication about error	C1-C3-A13	3		
2	Frequency of event reported	D1-D2-D3	3		
3	No punitive response to error	A8-A12-A16	3		
4	Handoffs and transitions	F5-F6-F7-F11	4		
5	Teamwork within units	A1-A3-A4	3		
6	Communication openness	C2-C4-6	3		
7	Supervisor expectation and action promoting patient safety	B1-B2	2		
8	Staffing	A5-A7	2		
9	Teamwork across units	F2-F4-F10-F8	4		
10	Organisational learning- continuous improvement	A6-A9-	2		

Table 7 Structure of the final solution (10 factors)

The 10 factors solution included 29 safety climate items in 13 factors while 11 items were excluded (A2-A10-A11-A14-A15-A17-A18-B3-B4-C5-F1-F3-F9). Factor loadings were between 0.41 and 0.97. There were only small shifts among items across factors for instance, A13 moved to Feedback and communication about error. F6 moved to Handoffs and transitions. F8 moved to Teamwork across units. All dimensions consisted of two to four safety climate items. The composition of the factors of this solution was very similar to that of the 11 factors solution except B3 and B4 were lost in the 10 factors solution which they grouped as a factor in 11 factors solution. In conclusion three factors solutions (11-10-8) were identified by explorative factor analysis. The following table shows the structure of each solution produced in EFA.

Table 8 Comparison of the three final solutions

Solution	Number of factors	Structure	Items	Total Variance
	(number of items)		excluded	Explained (%)
One	11 factors	F1: D1-D2-D3	A10-A15-	52.9
	(31 items)	F2: A8-A12-A16	A17-F3-F1-	
		F3: A1-A3-A4	F9-C5-A2-	
		F4: F5-F6-F7-F11	A11-A14-	
		F5: A6-A9	A18	
		F6: C2-C4-C6		
		F7:A5-A7		
		F8:B1-B2		
		F9: C1-C3-A13		
		F10:F2-F4-F8-F10		
		F11: B3-B4		
Two	10 factors	F1: D1-D2-D3	A10-A15-	52.5
	(29 items)	F2: A8-A12-A16	A17-F3-F1-	
		F3: A1-A3-A4	F9-C5-A2-	
		F4: F5-F6-F7-F11	A11-A14-	
		F5: A6-A9	A18-B3-B4	
		F6: C2-C4-C6		
		F7:A5-A7		
		F8:B1-B2		
		F9: C1-C3-A13		
		F10:F2-F4-F8-F10		
Three	8 factors	F1: D1-D2-D3	A10-A15-	51.5
	(23 items)	F2: A8-A12-A16	A17-F3-F1-	
		F3: A1-A3-A4	F9-C5-A2-	
		F4: F5-F6-F7-F11	A11-A14-	
		F5: A6-A9-A13-F8	A18-B3-B4-	
		F6: C2-C4	F4-F10-F2-	
		F7: A5-A7	C6-C3	
		F8: B1-B2		

### Step 3: Testing the fit of optimal models

This step aimed to test, using the validation half of the data, the fit of the resulting models that emerged from the EFA that was performed on the construction half of the data. These models included 11, 10, and 8 factor solutions.

### **11 factors solution**

The process of CFA of the 11 factor solution was performed (as practical steps that described in the process of CFA of the original 12 factors in page 2)

- 1 AMOS Graphics was used to open up a blank diagram window
- 2 Opening the data (The validation half of the data, 431 questionnaires)
- 3 Building the model (Drawing the11 factors by using model drawing area in step one)
- 4 Adding the variable names to the model
- 5 Checking the diagram and running the model and using the output window

Table 4 shows the output of CFA of 11 factors solution.

### Table 9 Output of CFA of 11 factors solution

Model	Chi- square	DF	CMIN/DF	CFI	RMSEA	RMR	SRMR	11.1
11 Factors	922	381	2.4	0.881	0.678	0.057	0.093	0.855

The output above shows the value of CMIN/DF was greater than 2 (good model  $\leq 2$ ). The value of CFI < 0.90 (good model > 0.90). The value of SRMR > 0.08 (good model <0.05). The value of RMSEA > 0.06 (good model  $\leq 0.06$ ). Thus, the values of the fit indices from CFA of 11 factors are outside the general agreed parameters in table one and this model fit was deemed not satisfactory. Consequently this model was rejected.

#### **10 factors solution**

The process of CFA of the 10 factors solution was as follows:

- 1 AMOS Graphics was used to open up a blank diagram window
- 2 Opening the data (The validation half of the data, 431 questionnaires)
- 3 Building the model (Drawing the10 factors by using model drawing area in step one)
- 4 Adding the variable names to the model
- 5 Checking the diagram and running the model and using the output window

Table 5 shows the output of CFA of 10 factors solution.

#### Table 10 Output of CFA of 10 factors solution

Model	Chi- square	DF	CMIN/DF	CFI	RMSEA	RMR	SRMR	TLI
10 Factors	716	332	2.1	0.90	0.52	0.046	0.052	0.88

The output above shows the value of CMIN/DF was greater than 2 (good model  $\leq$ 2). The value of CFI=0.90. The value of SRMR <0.08. The value of RMSEA <0.06. Overall, the values of the fit indices from CFA of 10 factors indicated that this model fit was not satisfactory.

### **Eight factors solution**

The process of CFA analysis of the eight factors solution was followed:

- 1 AMOS Graphics was used to open up a blank diagram window
- 2 Opening the data (The validation half of the data, 431 questionnaires)
- 3 Building the model (Drawing the eight factors by using model drawing area in step one)
- 4 Adding the variable names to the model
- 5 Running the model and using the output window

Table 6 shows the output of CFA of 8 factors solution.

Table 11 Output of CFA of 8 factors solution

Model	Chi- square	DF	CMIN/DF	CFI	RMSEA	RMR	SRMR	TLI
8	407	202	2	0.93	0.049	0.044	0.047	0.916
Factors	(good)	(good)	(good)	(good)	(good)	(good)	(good)	(good)

The output above shows the value of CMIN/DF was 2 (good model  $\leq$ 2). The value of CFI >0.90. The value of SRMR <0.08. The value of RMSEA <0.06. Overall, the fit indices from CFA of eight factors indicated that this model fit was good.

The values of the fit indices of the Saudi data (8 factors) meet the general agreed parameters in table one. Therefore, this model fit was a good fit. Consequently this model was acceptable as the optimal model. Table 7 shows the comparison of CFA between USA and UK models and the three solutions of the current study (11, 10 and 8 factor solutions).

Model	Chi-	DF	CMIN/DF	CFI	RMSEA	RMR	SRMR	TLI
USA data (Sorra and Nieva, 2004)	square 2064	746	-	0.94	0.040	-	-	-
UK data (original model) (Waterson et al, 2009)	1906	674	-	0.91	0.045	-	0.046	-
UK data 9 factors, 27 items(Waterson et al, 2009)	587	288	-	0.94	0.043	-	0.043	0.93
			Curre	ent study				
12 Factors (original model)	4995	805	6.2	0.642	0.678	0.178	0.1931	0.617
11 Factors	922	381	2.4	0.881	0.678	0.057	0.093	0.855
10 Factors	716	332	2.1	0.90	0.52	0.046	0.052	0.88
8 Factors	407	202	2	0.93	0.049	0.044	0.047	0.916

Table 12 Comparison of CFA of USA and UK models and the current study

It is clear that the optimal model that fitting the Saudi data is the eight factor solutions because of the value of the fit indices indicate that this model meets the general agreed parameters.

N	ID	Job	Actual text
1	6	Nurse	Shortcuts still been practised in the hospital. Difficult to ask some medical staff. (-) 2
2	10	Nurse	Polices and producers are excellent. Feedbacks on reports are almost available. Reporting system is effective (+) 3
3	14	Doctor	Many patients left in pediatric emergency. Many days for transfer to wards. 1-5 days waiting in very crowded atmosphere (a) 1
4	15	Doctor	The only thing affect the patient safety in emergency is crowded and no bed available in the wards. Patients stay for two to five days in the emergency
5	21	Nurse	No cooperation between staff. The staff are not beloing in problems (-) 1
6	22	Nurse	I believe when you take care of patients you have to be oriented to your work place and with people you work with them (+) 1
7	23	Nurse	We always talk about patient safety, patient comfort and patient rights. Management never helped the staff and their safety and welfare. Shift staff without giving them enough time to orient themselves. Nurses having basic knowledge on nursing nurses fear to speak. Management is not helpful and supportive the staff.
8	24	Nurse	If we will maintain patient safety we should in the first place maintain staff safety, so when they come to work, their mind will be concentrated on safe patient care e.g. if the staff living outside the hospital compound with family and no medical and education coverage all these so how the staff is going to concentrate and provide safe care to the patients (-) one
9	26	Nurse	We are not familiar with staff from other area. (-) one
10	27	Doctor	The questionnaire concentrated on patient safety in patient not outpatient. We see more problems with patient safety due to short time spent with patient. Lack of cooperation between different specialities (-) 3
11	34	Nurse	Our hospital following international nations safety goals as we are ICIA hospital $(+)$ 1
12	35	Doctor	This hospital is probably is the best hospital in the KSA regarding this issue. Good research $(+)$ 2
13	36	Nurse	Sometimes we are concentrating more with paper works than patient care. Honestly they should decrease paper work and we should concentrate on patient care and safety. Thank you, (-) 2.
14	45	Doctor	I feel that patient safety in our hospitals depend on the physician himself; I mean his own rules and beliefs not on the hospital regulations.
15	53	Nurse	Patient safety is the first and ultimate goal in this unit and we as nurse are really practicing it in an excellent way. (+) 2
16	58	Nurse	We have <u>electronic incident reporting system</u> . In the past it was a paper based. Patient safety has been <u>made number one priority</u> by senior management. <u>Quality management</u> is effectively striving to enhance patient safety through its leadership. (+) 3 themes
17	62	Doctor	Encourage staff to disclose his/her medical errors to others for learning purpose. Holding regular session to present morbidity/mortality cases to all staff in each units for learning. Thanks (+) 2
18	63	Doctor	They are working on good patient safety maintaining. (+) 1
19	65	Doctor	Easy and clear to complete questionnaire. (+) 1
20	67	Doctor	They have excellent reporting system. (+) 1
21	71	Doctor	Good survey we would be interested in seeing the results. (+) 1
22	72	Doctor	My hospital is a tertiary hospital in Saudi Arabia in which patient all over Saudi Arabia come plus international patients. I believe in such hospital are efficient number of medical staff should work especially in critical area but it is not the case in my hospital. I believe there is no respect to junior staff opinion although sometimes it is important and true. I think sometimes patient care is sometimes missed when discharged from hospital. I mean the outpatient follow up is not optimal. Many patients are lost between clinics. Many patients do not know why to visit the clinics. Last point is patient education is still need more to do. I wish your study will beln us. God bless you (c) 6
23	77	Doctor	With environment of working when everybody concentrates on patient safety these action affect patient treatment, because everybody wants to protect himself than patients (-) 1

# Appendix 48 Actual texts of answers of open question in the HSOPSC questionnaire

24	96	Doctor	We still need further improvement of patient safety system although it is better than other hospital. Reporting the physician who made the mistakes need more confidential. Thank you for your effort. (-) 3
25	97	Doctor	There is no justice in the working hours, load of the work, and position of physicians. Even there is no balance in the salary. Emergency department is not large. Some departments are supported by their head; even if they did mistake nobody care. (-) 4
26	100	Doctor	I wish you all luck. This questionnaire is too long and some questions are reported in other with same idea (-) 2
27	106	Doctor	This is a good hospital but the problem here, it is from the patient contact, and knowledge and contact with their relative it is difficult. Thank you. (-) 3
28	111	Doctor	Wish you best research good luck (+) 1
29	112	Doctor	We hope that we can share patient safety so we can learn from the people mistakes. (+) 2
30	116	Doctor	I hope we will receive feedback
31	117	Doctor	Clear to patient safety research no comments thank you so much. (+)
32	138	Nurse	Science our department personnel involved a multi disciplinary team from different culture patient safety is always our priority. Our scope of practice must be reported mistakes even near miss. Reporting system. In behalf of our department we wish you good luck, doctor + our congratulations. (+) 4.
33	144	Nurse	Blame on what happened rather than what caused the problem. (-) 1
34	148	Nurse	My observation is those critical incidences are not freely reported. Professionals continue to cover up mistakes, I believe will continue until individuals are no longer blamed, punished for incidents in which they are involved.
35	146	Nurse	Patient safety is number issue in the hospital. Event reporting is good tool to improve patient safety (+) 2
36	140	Nurse	The hospital setting is good. The structure is good. Programs some are good but most very unnecessary and stressful for the staff on the floor. Every unit is doing its own thing that may be good or bad but at least there is competition on trying to get better. Staff on the floor is overworked, overloaded with patients. Nobody is taking care of the staff on the floor. (-) 6
37	166	Nurse	Patient safety precautions program have been presented by means of posters, conference and workshop that provides full information to staff. (+) 1
38	170	Nurse	Theory practice gap still exist. It is difficult at times to practice what is being taught in clinical areas. Too much paper work more than patient care. Staffing problems (in balance of staffing) (-) 3
39	171	Nurse	The hospital focuses on patient safety. JCIA put policy and regulations and guidelines related what should be implemented and too many checklists about patient safety.(+) 2
40	181	Nurse	We are very much focus on patient safety (+) 1
41	192	Nurse	Patient safety should be our priority. We should be careful about all things related to patient care (+) 2
42	191	Nurse	I grateful to inform you that the hospital I am working with has a high regard when it comes to patient safety. (+) 1
43	190	Nurse	Our hospital is currently implementing measures to ensure quality patient care for patients. The responsibilities of all healthcare worker to work together to ensure safety to all patients. Polices are being put in place to facilitate patient safety. Errors or events affect patient safety reported and reviewed by the quality assurance management team.
44	197	Nurse	Communication problems sometimes lead to patient safety problem because the patient difficult to understand physicians instructions. (-) 1
45	202	Nurse	I personally think the issues regarding patient safety are of utmost importance in this organisation. (+) 1
46	205	Nurse	More on paper works which delay patient care. (-) 1
47	204	Nurse	Too many paper works to be done, less direct patient care. Patient safety here is our priority. (-) 2
48	206	Nurse	An event reported becomes essential and important. (-) 1
49	207	Nurse	Reporting events is common helpful factor in patient care and safety. (+) 1
50	210	Nurse	For patient safety staff play an vital part for that the staff welfare must be taken care. Nurse mangers and supervisors must attend consulting course regarding how to communicate to staff. Stress for staff will cause more damage in patient safety. Staff satisfaction is important and giving a better safety environment for patients. Patient safety depends on staff satisfaction. Staff with highly stress will causing more harm for

	-	1	
	-		patient and affected patient safety.
51	212	Nurse	Patient safety becomes a focus at accreditation times.
52	239	Nurse	More information on how to enhance patient safety in this hospital.
53	247	Nurse	Patient safety is the number one priority in this hospital in relation to this all staff should adhere with the international patient safety goals. Identify patient correctly using two unique identities. Improve staff communication using SBAR. Correct site of surgery a time out. Reduce risk of infection by hand hygiene. Do necessary action to correct mistakes
54	248	Nurse	Patients' awareness is important. Communication between staff and between patients as well is very important to decrease errors.
55	249	Nurse	Honestly, this hospital is very particular in patient safety focus in improvement towards patient care. Mistakes happened yes but there is always room for improvement. Patient safety is the top priority can be achieved through teamwork of excellent staff.
56	274	nurse	Our goal is patient and their family rights first which include them to be safe.
57	280	Nurse	The head nurse is not held accountable for her behaviour and very poor treatment of the staff. Despite numerous reports to the management through the chain of command nothing has been done. She does not lesson to the staff. Lack of communication between and preventing discussion all these lead to unsafe practice.
58	283	Nurse	Visitors could easily come and go towards there is on security guard at the doors.
59	292	Nurse	More time is spent on documentation than bedside care.
60	304	Nurse	Babies should not bring to mothers during visiting hours for safety reason.
61	303	Nurse	Patient safety measures only be taken after event has happened. I am very happy to be given this opportunity to fill up this questionnaire. Thank you.
62	306	Nurse	Patient safety is definitely a priority but due to workload sometimes it becomes practically impossible to ensure safety. The high turnover of patients makes it difficult/impossible to ensure safety of patients because you discharge one patient when you have not even finished the discharge procedure another patient is being transferred to your care. As much as the safety of patient s is important sometimes the practicality becomes impossible.
63	310	Nurse	Top management speaks about patient safety but physically not too much is done. Patient safety should not only be on papers, it should be practiced.
64	312	Nurse	Feedback of this survey would be strongly appreciated. We work without any Arabic speaking. We do not have unit assistance.
65	313	Nurse	I feel nurse want to report near misses but they think it's not worth because of the head of department. People take reporting as an insult instead of something positive for patient outcomes. I do not view reporting as a make against anyone because I came from my country that encouraging reporting so that we can put better systems in place to prevent it from happening again
66	314	Nurse	A feedback on this survey will beneficial to the staff. Patient safety is a priority in this hospital but is compromised due to the workload that they have. Too much time spent on electronic and written documentation and limited time given to direct patient care. Directives are given to nurses to do things but the problems encountered by the nursing staff are not heard
67	322	Nurse	Management is active in trying to implement patient safety as one of focus. There seems to be resistance from some staff regarding change and patient safety. It seems that whatever is fastest is the method they will use. They cover each other regarding mistakes that are made. I find alot of problems with compliance to hospital policy's. There is little respect shown to each other and I find manner of staff to be unprofessional at time.
68	334	Nurse	Staff support depends on who one is?. Teamwork : people still find themselves leaving the unit at 9:00 or 10:00 especially if the work is not completed. People still do not feel free to ask question
69	341	Nurse	The most patient safety issue as the nurse that employed in intensive care unit not ICU bored nurses (not qualified).
70	350	Nurse	I would like to suggest for enforcement in the effort to create a non balanced culture in managing event reporting related to patient safety. The system failure should be focused on when there is event reporting such as medication errors, patient safety issue.
71	358	Nurse	Pressure time, when we work faster errors can easily happen. But now a days lots of things are treated by policies and procedures that can help us to work better to prevent

			errors and more safe environment for patients.
72	360	Nurse	As part of member of this hospital I found the hospital and management are working hard in order to keep patient safety continuously.
73	368	Nurse	I think I was just lucky enough working here in KAMC, Riyadh. All programs related and with positive effect to patient are updated. These institution prioritize their patients safety by encourage and support staff who work very well. Thanks for KF NG.
74	369	Nurse	Documentation takes a lot of time than providing patient care because the system emphasise more importance on paper works. There is no enough staff for work therefore the pressure increase because of workload.
75	372	Nurse	Nursing staff are the first person in direct contact with patient. They are potentially hazardous to patient health and life physically and mentally exhausted nurse due to staff shortage, compensated by mandatory overtime, eventually result in compromised patient safety. More errors done than prevented and not being reported. Healthy nurse can actively prevent medico legal or health hazards only if they are given time to rest and do paper work once than many times. Errors should be reported.
76	385	Nurse	No comments thank you.
77	392	Nurse	I think safety for patient is very good in this hospital. Thanks
78	395	Nurse	I did not see much positive changes or action or improvement after near miss incident being reported. This will make staff are rarely willing to report any incident any more.
79	396	Nurse	Patient safety is number one priority in this hospital.
80	399	Nurse	I find my nurse manager in supportive, open, honest and team player. Due to the leadership skills as a staff nurse I have improved my nursing care and patient safety has improved almost 95% I love coming to work now.
81	422	Nurse	Patient safety is our top priority, circumstances like patients with sitter put risk in patient safety. For example patient going to O.R, sitter instructed not to give patient by oral, they cannot resist to give water.
82	424	Nurse	Double up of patients in one cubic. Taking care of 2-3 critically ill patients. 1:3 nursing. Short cuts in doing nursing care. Not checking medication by two nurses. Un experienced nurses in the department. Lack of knowledge and skills
83	426	Nurse	I feel that staff still has misconception of reporting of errors. Fair from reporting system because they feel it kept in their file.
84	437	Nurse	What the hospital says and does are two different things. I feel a big part of good quality care is looking after staff. We do not have temporary staff in times of crisis so all staff work overtime. And the hospital issued a memo saying staff are not allowed to refuse the allocated overtime shifts. Staff therefore are overworked and patient care is at risk. Workload is real problem for patient safety and still no problem solving. Double check procedure for patients is absent i.e. there is no question of elective surgery or elective admission about patient identifications. Under these conditions staff take short cut in practice to cope with the increase workload. Basic practice principles are lost.
85	447	Nurse	Patient safety is the first top priority. Reporting immediately about errors. Provide good quality.
86	453	Nurse	No comments and thank you
87	468	Nurse	Patient safety is one of our major goals in the hospital. We have policies and procedures regarding all aspects regarding patient safety. (+) 2 themes
88	469	Nurse	We often write errors because that is not used for punching people it is for improving and teaching staff member thus improving safety. (+) one theme (report errors and learning from errors (organisational learning not punching people).
89	473	Nurse	Any type of report incident is not regarding as punitive in my hospital. Some of the staff still get worried that the reported mistakes they made go to their files for future response. Reporting incidents is only for correction and improvement purpose not carried over to any staff evaluation. (+) 4 themes.
90	477	Nurse	Patient safety should be first top priority. Reporting immediately any medication errors. Provide good quality of patient care. (+) 3 themes.
91	482	Nurse	Workload effect patient care negatively.
92	491	Nurse	Patient safety awareness has recently only become an important issue as we are being accredited
93	495	Nurse	Most of the errors have been reported but most of staff are taking reporting errors in negative way not in positive one. We should continue reporting errors

94	498	Nurse	Patient safety is number one priority. Supervisors enhance staff to protect patient safety. Thank you.
95	500	Nurse	Do not have good reporting system. No feedback given. Focus on individual rather than systems. No quick response. Communication breakdown. Blame culture exist. Thank you. (-) 7 themes.
96	502	Nurse	I believe patient safety should be maintained all the time. There is workload too many admissions of patients with limited bed capacity. More papers work than nursing your patients.
97	536	Nurse	It does not seem to help if we spoke up onset any problems, our supervisors listen to us nut there is not too much action taken, honestly they will print out on specific person and they will not see that there is too much pressure put in nurses.
98	543	Nurse	All the staff waking together with management to improve patient care. Hospital has good training for staff to upgrade their knowledge. Hospital has new program for the new staff which is very good. All the staff had very high level of confidence when they work in unit. Staff can get latest information from internet.
99	551	Nurse	The management /supervisor must dial with issue with more sensitive of staff who is involved with mistakes/errors in the unit. Majority of the staff feels threaten and fear of punishment in their workplace. Errors should be investigated rather than blame of staff.
100	553	Nurse	The most common error that we are not reporting the near miss errors. It is very important for next time to know how to avoid errors. Thank you Allah help you.
101	560	Nurse	When a mistake or error happens in the unit it is sometimes or most of the time blamed on the nurses, the person is the one being written up and talked about rather than the problem. It should be about the problem or the system and not the person involved.
102	571	Doctor	Thank you for this survey. I think in our institution the patient safety is considering top priority in comparison to other local hospitals. One of disadvantage of reporting errors is that most of staff of the hospital think that is considering personnel and will be recorded in the personnel file.
103	594	Doctor	There is a chance for analysing errors which help to improve patient safety.
104	601	Nurse	Some times staff are not working for improving patient safety. There is no proper communication and cooperative between staff
105	609	Nurse	We do more paper works than providing good and safety care for patients.
106	614	Nurse	Reporting system is important that enable us to give feedback an incident report.
107	620	Nurse	Patient safety is our aim but it is important to activate reporting system and avoid blame and fear from reporting errors. Also mangers role should be supportive to provide safe and good care for patients.
108	635	Nurse	Patient safety education is important for staff and for patients as well.
109	650	Doctor	I hope when the researcher is done the results it becomes start on real life to make it useful for patient safety improvement.
110	658	Nurse	Patient safety is one of the major objectives that every member of the staff in this hospital is aiming to achieve. There are a lots of patient safety programs about patient safety that are followed working towards ensuring patient safety. Errors and events are reported.
111	680	Nurse	Patient identification is very important issue to prevent errors.
112	710	Doctor	I remember one patient who was admitted by mistakes to isolated room for RSV+VE bronchitis patient while the patient was RSV-VE bronchitis. The patient condition was getting worse more respiratory distress, required O2 therapy review by paecl. Pulmonary service who discovered this mistakes, because patient got RSV+VE bronchitis direct contact e RSV+VE bronchitis patient in isolated room in hospital and the patient stayed long time in hospital unit improved (high risk patient). This event was reported.
113	720	Nurse	All is well.
114	721	Doctor	Every individual in medical team must work for patient safety which is goal of our profession.
115	736	Nurse	When problems occur in our area it is solving urgent away to prevent future damage. Good luck with your research.
116	756	Nurse	Enough staff to prevent any errors.
117	763	Nurse	Thank you . failing down the most common risk for the patient in my hospital, so we hope to increase the safety to all patients in the hospital.
118	769	Nurse	The utilisation of opportunity report in our unit is very essential as an excellent tool in

			maintaining patient safety a top priority. Some staff feel that errors held against them which prevent them to report errors.
119	771	Nurse	We need more staff to give the care to patient completely. Thank you
120	775	Nurse	In order to achieve the mission and vision of the hospital patient safety remains a priority in order to avoid errors. Staff feel the reports will be against them. Report should be focused on the problem not to the person who made the error.
121	797	Nurse	The hospital management promotes the patient safety by providing the staff with useful information through seminars and research. All the units and staff coordinate well with each other to provide safe patient care.
122	798	Nurse	Our hospital motto is patient first; we try our best to commit mistakes and if so happens to correct it immediately. If a mistake was committed opportunity report is done not to punish the person involved but to find a way to avoid committing the same mistakes again.
123	801	Nurse	Some important information was not completed such as consent form and patient identification.
124	806	Nurse	To be honest I think we spend too much time filling out papers and forms when we could be directly tending to our patients.
125	815	Nurse	This hospital implements more paper works which affect negatively patient acre. Compulsory overtime which excludes the working time in a week.
126	828	Nurse	I personally feel that the hospital documentation system took more time and efforts from the staff instead of giving more attention and time to patients.

This qualitative data has not analysed completely because the most important is psychometric analysis however, it will use in future.